

Future physics opportunities for high-density QCD at the LHC with heavy-ion and proton beams

Report from Working Group 5 on the Physics of the HL-LHC, and Perspectives at the HE-LHC

Editors:

Z. Citron³, A. Dainese²⁴, J.F. Grosse-Oetringhaus⁶, J.M. Jowett⁶, Y.-J. Lee⁵³, U.A. Wiedemann⁶, M. Winn^{33,43}

Chapter coordinators:

A. Andronic⁵², F. Bellini⁶, E. Bruna²⁶, E. Chapon⁶, H. Dembinski⁵¹, D. d'Enterria⁶, I. Grabowska-Bold¹, G.M. Innocenti^{6,53}, C. Loizides⁶¹, S. Mohapatra¹³, C.A. Salgado³⁸, *M.* $Verweij^{68,101}$, *M.* $Weber^{74}$

Contributors:

Contributors: J. Aichelin⁷¹, A. Angerami⁴⁸, L. Apolinario^{35,44}, F. Arleo⁴⁵, N. Armesto³⁸, R. Arnaldi²⁶, M. Arslandok¹⁹, P. Azzi²⁴, R. Bailhache³⁹, S.A. Bass¹⁶, C. Bedda⁹⁹, N.K. Behera³⁶, R. Bellwied⁸⁸, A. Beraudo²⁶, R. Bi⁵³, C. Bierlich^{50,59}, K. Blum^{6,103}, A. Borissov⁵², P. Braun-Munzinger¹⁷, R. Bruce⁶, G.E. Bruno⁶⁵, S. Bufalino⁶⁶, J. Castillo Castellanos³³, R. Chatterjee¹⁰⁰, Y. Chen⁶, Z. Chen⁶⁹, C. Cheshkov³¹, T. Chujo⁹⁷, Z. Conesa del Valle⁸, J.G. Contreras Nuno¹⁴, L. Cunqueiro Mendez⁶¹, T. Dahms¹⁸, N.P. Dang⁹², H. De la Torre⁵⁴, A.F. Dobrin⁶, B. Doenigus³⁹, L. Van Doremalen⁹⁹, X. Du⁷⁷, A. Dubla¹⁷, M. Dumancic¹⁰³, M. Dyndal¹⁵, L. Fabbietti⁷⁶, E.G. Ferreiro³⁸, F. Fionda⁸³, F. Fleuret⁴⁵, S. Elecerchinger¹⁹, C. Giageloug³², A. Giagguagana and S. P. Gassiguag⁷¹, C. Guagiaga²³, M. Grage⁸² S. Floerchinger¹⁹, G. Giacalone³², A. Giammanco⁸¹, P.B. Gossiaux⁷¹, G. Graziani²³, V. Greco⁸² A. Grelli⁹⁹, F. Grosa⁶⁶, M. Guilbaud⁶, T. Gunji¹⁰, V. Guzey^{20,64,90}, C. Hadjidakis³⁰, S. Hassani³⁴, M. He⁵⁶, I. Helenius^{80,90}, P. Huo⁷⁵, P.M. Jacobs⁴⁷, P. Janus¹, M.A. Jebramcik^{6,39}, J. Jia^{4,75}, A.P. Kalweit⁶, H. Kim¹², M. Klasen⁵², S.R. Klein⁴⁷, M. Klusek-Gawenda²¹, J. Kremer¹, *G.K. Krintiras*⁸¹, *F. Krizek*², *E. Kryshen*⁶⁴, *A. Kurkela*^{6,73}, *A. Kusina*²¹, *J.-P. Lansberg*³⁰, *R. Lea*⁹⁶, *M. van Leeuwen*^{60,99}, *W. Li*⁶⁹, *J. Margutti*⁹⁹, *A. Marin*¹⁷, *C. Marquet*⁹, *J. Martin Blanco*⁴⁵, *L. Massacrier*³⁰, *A. Mastroserio*⁸⁶, *E. Maurice*⁴⁵, *C. Mayer*²¹, *C. Mcginn*⁵³, *G. Milhano*^{6,35,44}, *A. Milm*¹⁰³, *V. M. Li*⁹⁹, *C. M.*⁵³, *A. M.*⁵³, *A. M.*⁵³, *A. M.*⁵⁵, *A. M.*⁵⁹, *M. M.*⁶⁶, *A. M.*⁶⁶, *A. M.*⁶⁶, *A. M.*⁶⁷, *A. M.*⁶⁶, *A. M.*⁶⁷, *A. M.*⁶⁶, *A. M.*⁶⁷, *A. M.*, *A. M.*⁶⁷, *A. M.*, *A. M.*⁶⁷, *A. M.*, *A.* A. Milov¹⁰³, V. Minissale²⁹, C. Mironov⁵³, A. Mischke^{*99}, N. Mohammadi⁶, M. Mulders⁶, M. Murray⁹¹, M. Narain⁵, P. Di Nezza²⁸, A. Nisati²⁵, J. Noronha-Hostler⁷⁰, A. Ohlson¹⁹, V. Okorokov⁵⁸, F. Olness⁷², M. Narain^{*}, P. Di Nezza^{-*}, A. Nisati^{-*}, J. Noronha-Hostler^{1*}, A. Ohlson^{1*}, V. Okorokov⁵⁸, F. Olness⁷², P. Paakkinen⁹⁰, L. Pappalardo⁸⁵, J. Park⁴², H. Paukkunen^{20,90}, C.C. Peng⁶⁷, H. Pereira Da Costa³³, D.V. Perepelitsa⁸⁴, D. Peresunko⁵⁷, M. Peters⁵³, N.E. Pettersson⁹³, S. Piano²⁷, T. Pierog⁴⁰, J. Pires^{7,35}, M. Płoskoń⁴⁷, S. Plumari⁸², F. Prino²⁶, M. Puccio⁹⁵, R. Rapp⁷⁷, K. Redlich^{17,98}, K. Reygers¹⁹, C.L. Ristea³⁷, P. Robbe⁴³, A. Rossi⁹⁴, A. Rustamov^{17,19,55}, M. Rybar¹³, M. Schaumann⁶, B. Schenke⁴, I. Schienbein⁴⁶, L. Schoeffel³⁴, I. Selyuzhenkov^{17,58}, A.M. Sickles⁸⁹, M. Sievert⁷⁰, P. Silva⁶, T. Song⁸⁷, M. Spousta¹¹, J. Stachel¹⁹, P. Steinberg⁴, D. Stocco⁷¹, M. Strickland⁴¹, M. Strikman⁶³, J. Sun⁷⁸, D. Tapia Takaki⁹¹, K. Tatar⁵³, C. Tarrevoli⁸⁸, A. Timmins⁸⁸, S. Tuccol-⁹⁵, P. T. (199), A. T. (21) D. Tapia Takaki⁹¹, K. Tatar⁵³, C. Terrevoli⁸⁸, A. Timmins⁸⁸, S. Trogolo⁹⁵, B. Trzeciak⁹⁹, A. Trzupek²¹, R. Ulrich⁴⁰, A. Uras³¹, R. Venugopalan⁴, I. Vitev⁴⁹, G. Vujanovic^{62,102}, J. Wang⁵³, T.W. Wang⁵³, R. Xiao⁶⁷, Y. Xu¹⁶, C. Zampolli^{6,22}, H. Zanoli⁷⁹, M. Zhou⁷⁵, Y. Zhou⁵⁹ * deceased

- ¹ AGH University of Science and Technology, Kraków, Poland, ² Academy of Sciences, Prague, Czech Republic,
- ³ Ben-Gurion University of the Negev, Beersheba, Israel, ⁴ Brookhaven National Laboratory, Upton, USA,
- ⁵ Brown University, Rhode Island, USA, ⁶ CERN, Geneva, Switzerland, ⁷ CFTP, Lisbon, Portugal,
- ⁸ CNRS/IN2P3, Université Paris-Sud, Université Paris-Saclay, Orsay, France,
- ⁹ CPHT, CNRS, École polytechnique, Université Paris-Saclay, Palaiseau, France,
- ¹⁰ Center for Nuclear Study, Graduate School of Science, The University of Tokyo, Japan,
- ¹¹ Charles University, Prague, Czech Republic, ¹² Chonnam National University, Gwangju, China,
- ¹³ Columbia University, New York City, USA, ¹⁴ Czech Technical University in Prague, Czech Republic,
- ¹⁵ DESY, Hamburg, Germany, ¹⁶ Duke University, Durham, USA,
- ¹⁷ EMMI/GSI Helmholtzzentrum für Schwerionenforschung GmbH, Darmstadt, Germany,
- ¹⁸ Excellence Cluster Universe, Technical University Munich, Germany, ¹⁹ Heidelberg University, Germany,
- ²⁰ Helsinki Institute of Physics, Finland, ²¹ IFJ PAN, PL-31342 Kraków, Poland, ²² INFN Sezione di Bologna, Italy,
- ²³ INFN Sezione di Firenze, Italy, ²⁴ INFN Sezione di Padova, Italy, ²⁵ INFN Sezione di Roma, Roma, Italy,
- ²⁶ INFN Sezione di Torino, Italy, ²⁷ INFN Sezione di Trieste, Italy, ²⁸ INFN-LNF, Frascati, Italy,
- ²⁹ INFN-LNS, Catania, Italy, ³⁰ IPN Orsay, CNRS/IN2P3, Université Paris-Sud, Université Paris-Saclay, Orsay, France,
- ³¹ IPN-Lyon, CNRS/IN2P3, Université de Lyon, Lyon, France,
- ³² IPhT, CEA Saclay, CNRS, Université Paris-Saclay, Saclay, France,
- ³³ IRFU/DPhN, CEA Saclay, Université Paris-Saclay, Saclay, France,
- ³⁴ IRFU/DPhP, CEA Saclay, Université Paris-Saclay, Saclay, France, ³⁵ IST Lisbon, Portugal,
- ³⁶ Inha University, Incheon, Korea, ³⁷ Institute of Space Science, Bucharest, Romania,
- ³⁸ Instituto Galego de Fisica de Altas Enerxias (IGFAE), Universidade de Santiago de Compostela, Spain,
- ³⁹ Johann-Wolfgang-Goethe Universität, Frankfurt, Germany, ⁴⁰ Karlsruhe Institute of Technology, Germany,
- ⁴¹ Kent State University, USA, ⁴² Korea University, Seoul, Korea,
- ⁴³ LAL, CNRS/IN2P3, Université Paris-Sud, Université Paris-Saclay, Orsay, France, ⁴⁴ LIP, Lisbon, Portugal,
- ⁴⁵ LLR, CNRS/IN2P3, École polytechnique, Université Paris-Saclay, Palaiseau, France,
- ⁴⁶ LPSC Grenoble, CNRS/IN2P3, Grenoble INP, Université Grenoble Alpes, Grenoble, France,
- ⁴⁷ Lawrence Berkeley National Laboratory, Berkeley, USA, ⁴⁸ Lawrence Livermore National Laboratory, Livermore, USA,
- ⁴⁹ Los Alamos National Laboratory, Los Alamos, USA, ⁵⁰ Lund University, Sweden,
- ⁵¹ MPI for Nuclear Physics, Heidelberg, Germany, ⁵² Münster University, Germany,
- ⁵³ Massachusetts Institute of Technology, Cambridge, USA, ⁵⁴ Michigan State University, East Lansing, USA,
- ⁵⁵ NNRC, Baku, Azerbaijan, ⁵⁶ Nanjing University of Science and Technology, China,
- ⁵⁷ National Research Centre Kurchatov Institute, Moscow, Russia,
- ⁵⁸ National Research Nuclear University MEPhI, Moscow, Russia, ⁵⁹ Niels Bohr Institute, Copenhagen, Denmark,
- ⁶⁰ Nikhef, Amsterdam, The Netherlands, ⁶¹ Oak Ridge National Laboratory, Oak Ridge, USA,
- ⁶² Ohio State University, Columbus, USA, ⁶³ Pennsylvania State University, University Park, USA,
- ⁶⁴ Petersburg Nuclear Physics Institute, Gatchina, Russia, ⁶⁵ Politecnico di Bari and INFN Sezione di Bari, Italy,
- ⁶⁶ Politecnico di Torino and INFN Sezione di Torino, Italy, ⁶⁷ Purdue University, West Lafayette, USA,
- ⁶⁸ RIKEN BNL Research Center, Upton, USA, ⁶⁹ Rice University, Houston, USA,
- ⁷⁰ Rutgers University, New Brunswick, USA, ⁷¹ SUBATECH, CNRS/IN2P3, IMT Atlantique, Université de Nantes, France,
- ⁷² Southern Methodist University, Dallas, USA, ⁷³ Stavanger University, Norway,
- ⁷⁴ Stefan Meyer Institute Vienna, Austria, Austrian Academy of Sciences, ⁷⁵ Stony Brook University, USA,
- ⁷⁶ Technical University Munich, Germany, ⁷⁷ Texas A&M University, College Station, USA,
- ⁷⁸ Tsinghua University, Beijing, China, ⁷⁹ Universidade de Sao Paulo, Brazil, ⁸⁰ Universität Tübingen, Germany,
- ⁸¹ Université catholique de Louvain, Louvain-la-Neuve, Belgium, ⁸² Università di Catania and INFN-LNS, Catania, Italy,
- ⁸³ University of Bergen, Norway, ⁸⁴ University of Colorado Boulder, USA,
- ⁸⁵ University of Ferrara and INFN Sezione di Ferrara, Italy, ⁸⁶ University of Foggia and INFN Sezione di Bari, Italy,
- ⁸⁷ University of Gießen, Germany, ⁸⁸ University of Houston, USA, ⁸⁹ University of Illinois, Urbana-Champaign, USA,
- ⁹⁰ University of Jyvaskyla, Finland, ⁹¹ University of Kansas, Lawrence, USA, ⁹² University of Louisville, USA,
- ⁹³ University of Massachusetts, Amherst, USA, ⁹⁴ University of Padova and INFN Sezione di Padova, Italy,
- ⁹⁵ University of Torino and INFN Sezione di Torino, Italy, ⁹⁶ University of Trieste and INFN Sezione di Trieste, Italy,

⁹⁷ University of Tsukuba, Japan, ⁹⁸ University of Wroclaw, Poland, ⁹⁹ Utrecht University, The Netherlands,
¹⁰⁰ VECC Calcutta, India, ¹⁰¹ Vanderbilt University, Nashville, USA, ¹⁰² Wayne State University, Detroit, USA,
¹⁰³ Weizmann Institute of Science, Rehovot, Israel

Abstract

The future opportunities for high-density QCD studies with ion and proton beams at the LHC are presented. Four major scientific goals are identified: the characterisation of the macroscopic long wavelength Quark–Gluon Plasma (QGP) properties with unprecedented precision, the investigation of the microscopic parton dynamics underlying QGP properties, the development of a unified picture of particle production and QCD dynamics from small (pp) to large (nucleus–nucleus) systems, the exploration of parton densities in nuclei in a broad (x, Q^2) kinematic range and the search for the possible onset of parton saturation. In order to address these scientific goals, high-luminosity Pb–Pb and p–Pb programmes are considered as priorities for Runs 3 and 4, complemented by high-multiplicity studies in pp collisions and a short run with oxygen ions. High-luminosity runs with intermediate-mass nuclei, for example Ar or Kr, are considered as an appealing case for extending the heavy-ion programme at the LHC beyond Run 4. The potential of the High-Energy LHC to probe QCD matter with newly-available observables, at twice larger center-of-mass energies than the LHC, is investigated.

Contents

1	Introduction	6
1.1	Macroscopic QGP properties	8
1.2	Accessing the inner workings of hot QCD matter	9
1.3	Developing a unified picture of QCD collectivity across system size	11
1.4	Nuclear parton densities and search for non-linear QCD evolution	12
1.5	Physics performance studies by ALICE, ATLAS, CMS and LHCb	13
2	Heavy-ion performance of LHC, HL-LHC and HE-LHC	14
2.1	Heavy-ion performance of LHC in Runs 1 and 2	14
2.2	Pb-Pb luminosity in Run 3 and Run 4 (HL-LHC)	14
2.3	Proton-lead operation in Run 3 and HL-LHC	17
2.4	Colliding lighter nuclei at HL-LHC	18
2.5	Short run for p–O	20
2.6	Heavy-ion performance of HE-LHC	20
3	Light flavour sector: (anti-)(hyper-)nuclei and fluctuations of conserved charges	22
3.1	Introduction	22
3.2	(Anti-)(hyper-)nuclei production	22
3.3	Fluctuations of conserved charges	31
4	Flow and Correlations	36
4.1	Introduction	36
4.2	Review of current status of theory on bulk and flow observables	36
4.3	Experimental constraints from Run 3 and 4	39
4.4	Vorticity and polarization	42
4.5	Chiral Magnetic Effect	44
4.6	Summary	46
5	Open heavy flavour	48
5.1	Perspectives for heavy-flavour observables in LHC Run 3 and 4	48
5.2	Impact of detector upgrades on heavy-flavour measurements	49
5.3	Nuclear modification factor and collective flow	50
5.4	Studies of heavy-quark hadronisation	54
5.5	Heavy-flavour correlations and jets	58
5.6	Sensitivity to early magnetic fields and vorticity phenomena	60
5.7	Heavy flavour measurements in small colliding systems	60
6	Jets and parton energy loss	64
6.1	Introduction	64
6.2	Out-of-cone radiation	65
6.3	Jet deflection	66
6.4	Jet internal structure	70
6.5	Opportunities for jet quenching studies with light-ion collisions	76
7	Quarkonia	80

7.1	Introduction
7.2	Charmonia in Pb–Pb collisions
7.3	Bottomonia in Pb–Pb collisions
7.4	Quarkonia in p-Pb and pp collisions
8	Electromagnetic radiation
8.1	Thermal radiation and in-medium spectral function
8.2	Two-photon and photonuclear interactions
8.3	Dark photons
8.4	Limitations and outlook
9	Emergence of hot and dense QCD matter in small systems
9.1	Introduction
9.2	Overview of experimental results and critical assessment
9.3	Open questions and new opportunities at HL-LHC
9.4	Proton-proton collisions at extreme multiplicities
9.5	Global-event properties
9.6	Particle correlations
9.7	Strangeness enhancement
9.8	Energy loss
9.9	Thermal Radiation
9.10	Potential of O–O Collisions
9.11	Summary
10	High energy QCD with proton-nucleus collisions and ultra-peripheral collisions 127
10.1	Introduction
10.2	The physics of ultra-peripheral collisions
10.3	The physics of inelastic p-Pb collisions
10.4	Constraints on nuclear PDFs
10.5	Perspectives with lighter ions
11	Other opportunities with ion and proton beams at the LHC
11.1	Physics motivation for collisions of light ions
11.2	Physics of $\gamma\gamma$ interactions in heavy-ion collisions $\ldots \ldots \ldots$
11.3	Proton-oxygen collisions for cosmic ray research
11.4	Fixed-target prospects with LHC beams
12	Summary of luminosity requirements and proposed run schedule
13	First considerations on a heavy-ion programme at a High Energy LHC (HE-LHC) 174
13.1	Introduction
13.2	Global characteristics of nucleus–nucleus collisions at the HE-LHC
13.3	QGP studies with hard probes
13.4	Nuclear PDF measurements and search for parton saturation
13.5	Photon–photon collisions

1 Introduction

Experiments with heavy-ion collisions at the LHC create and diagnose strongly-interacting matter under the most extreme conditions of high density and temperature accessible in the laboratory. Under these conditions, QCD calculations on the lattice predict that matter undergoes a phase transition to a Quark– Gluon Plasma (QGP) in which colour charges are deconfined and chiral symmetry is restored. Aside of its intrinsic interest, this line of research is central to our understanding of the Early Universe and the evolution of ultra-dense stars. In practice, the main focus of experimentation with nuclear beams at the LHC is on learning how collective phenomena and macroscopic properties, involving many degrees of freedom, emerge under extreme conditions from the microscopic laws of strong-interaction physics. In doing so, proton–nucleus (p–A) and nucleus–nucleus (A–A) collision experiments at the LHC aim at applying and extending the Standard Model of particle physics to matter properties that govern dynamically evolving systems of finite size.

Recent experiments with nuclear beams at collider energies have identified opportunities to further strengthen the connection between the rich phenomenology of ultra-dense matter, and its understanding in terms of the fundamental laws of strong-interaction physics. Two broad classes of phenomena may be highlighted in this context.

First, the observation of flow-like phenomena in essentially all measured soft particle spectra and particle correlations lends strong support to understanding bulk properties of heavy-ion collisions in terms of viscous fluid dynamics. The fluid-dynamic evolution is solely based on combining conservation laws with thermodynamic transport theories that are calculable from first principles in quantum field theory. Hence, this provides an experimentally accessible inroad to constraining QCD matter properties via soft flow, correlation and fluctuation measurements. As further explained in this document, this motivates future improved measurements of flow and transport phenomena, including in particular measurements of soft heavy flavour and electromagnetic radiation. It also motivates improved experimental control over the system-size dependence of flow phenomena to better constrain under which conditions and in which kinematic regime ultra-relativistic p–A and A–A collisions show fluid dynamic behaviour and where this picture fails.

Second, the observation of quantitatively-large quenching phenomena in essentially all measured hard hadronic observables in A–A collisions has established the feasibility of testing the produced OCD matter with a broad set of probes whose production rates are controlled with good precision with pp reference measurements and perturbative QCD calculations. Hard quarks and gluons are known to interact with the medium and they thus probe medium properties. As detailed in this document, important physics opportunities are related to analysing hard probes in p-A and A-A collisions with the greater precision and kinematic reach accessible in future LHC runs. For instance, the identification of (Rutherford-type) large-angle jet-medium scattering could constrain the quasi-particle nature of the fluid-like medium. This is of central importance since it critically tests the working hypothesis that the matter produced in A-A collisions is a fluid with a ratio of shear viscosity over entropy density η/s close to the theoretical minimum value. Such a fluid would be void of quasi-particles, while QCD is definitive in predicting that a microscope with sufficiently high resolution will reveal partonic (quasi-)particle structure. Identifying the scale at which inner structure (such as quasi-particles and related non-vanishing mean free paths) arises would provide a microscopic understanding of how fluid-like behaviour arises in nucleus-nucleus collisions. Therefore, probing the inner workings of the QGP by resolving its properties at shorter and shorter length scales is one of the main motivations for future experimentation with hard probes. Also at intermediate transverse momentum, this document identifies important questions that will be accessible experimentally. For instance, more differential studies of quarkonium bound-state dissociation as a function of transverse momentum, rapidity and system size are expected to yield further insights into the mechanisms of colour deconfinement and recombination. Moreover, there is the general question of how fluid-like phenomena at low momentum scales transition to quenching phenomena at intermediate and high momentum scales.

Capitalizing on previous discoveries at RHIC, the LHC experimental programme with Pb–Pb and p–Pb collisions has significantly advanced the state of the art in both the soft- and the hard-physics sector. In the soft-physics sector, differential measurements at the LHC have in particular allowed for precise $p_{\rm T}$, particle species and rapidity-dependent measurements of all higher flow harmonics $v_{\rm n}$, their mode–mode couplings and the resulting reaction-plane correlations. This flow systematics extends to charmed flavoured hadrons and possibly even to beauty ones. It is at the basis of constraining QCD transport properties today. In the hard-physics sector, the wider kinematic reach of LHC has given qualitatively novel access to quarkonium suppression and jet-quenching phenomenology, including precision measurements of bottomonium, novel observations of charmonium enhancement, and a rich phenomenology of calorimetrically defined jets and jet substructure in nuclear collisions.

LHC experiments have also led to surprises that pose significant novel challenges for the understanding of p-A and A-A collisions. Most notable in this context is the discovery that flow-like phenomena are not limited to nucleus-nucleus collisions but they persist with significant magnitude in p-A collisions and in high-multiplicity pp collisions at the LHC and some of their signatures have been observed even in minimum-bias pp collisions. However, pp collisions are typically expected to show vanishingly small re-interaction rates between produced final-state particles. In contrast, the perfect-fluid paradigm, that underlies the successful phenomenology of flow-like phenomena in nucleus-nucleus collisions implicitly, assumes that re-interaction rates are so large that even the notions of quasi-particle and mean-free path become meaningless. Does the persistence of flow-like phenomena in p-A and pp collisions indicate, in contrast to previous belief, that the perfect-fluid paradigm applies to these smaller collision systems? Or, if the prefect-fluid paradigm is not applicable to pp and p-A collisions, is it conceivable that significant corrections to a fluid-dynamic picture of vanishing mean free path persist also in the larger A–A collision systems? The LHC discoveries in pp and p–A collisions that give rise to these questions provide arguably the strongest motivation for a future programme of detailed experimentation that aims at constraining microscopic structures and length-scales in the produced QGP matter and that is expected to clarify in this way the microscopic mechanisms underlying the apparent fluid-dynamic behaviour of pp, p-A and A-A collisions.

Historically, experimental heavy-ion programmes have always addressed a very diverse set of phenomenological opportunities. Some of the proposed experimental measurements have always reached out to other areas of science and could be clearly related to fundamental open question such as the origin of mass in the Universe, QCD deconfinement, or the determination of thermodynamic transport properties (that led in the past to unforeseen connections between string theory and the thermodynamics of quantum field theories). Other parts of the experimental programme were originally not related to a working hypothesis based on an open fundamental question, but they sometimes revealed themselves a posteriori as elements of crucial insight. This can be said for instance about the LHC p-A programme, that was not part of the original LHC design, that was first conceived mainly as a set of benchmark measurements for establishing the cold nuclear matter baseline for interpretation of heavy-ion data, and that has resulted in one of the most surprising discoveries made in the LHC nuclear-beams programme. We therefore emphasize that heavy-ion physics at the LHC, in the future as well as in the past, is likely to have multiple ways of reaching out and contributing to physics at large. At the time of writing this report, questions about the origin of collectivity in small pp and p-A collision systems, and their implications for the interpretation of collective phenomena observed in A-A collisions, are arguably identified as the most pressing conceptual issue in the scientific debate. As outlined so far, they are clearly related to further experimentation with soft processes, and to research on the internal structure of QGP matter utilizing hard processes. However, future experimentation at the LHC is not limited to this set of questions. From improved constraints on nuclear parton distribution functions that may inform us about the physics reach of future electron-ion facilities, to improved measurements of anti-nuclei, to ultra-peripheral collisions of electromagnetic Weizsäcker-Williams photons at unprecedented field strength, to the search for qualitatively novel signatures of ultra-strong QED magnetic fields, the LHC nuclear beams programme

can provide new insight in a much broader range of subject areas.

As detailed in this report, the HL/HE-LHC physics working group 5 has identified future physics opportunities for high-density QCD with ions and proton beams that can be grouped broadly into the following four goals that are coming now within experimental reach:

- 1. Characterizing the macroscopic long-wavelength QGP properties with unprecedented precision.
- 2. Accessing the microscopic parton dynamics underlying QGP properties.
- 3. Developing a unified picture of particle production from small (pp) to larger (p-A and A-A) systems.
- 4. Probing parton densities in nuclei in a broad (x, Q^2) kinematic range and searching for the possible onset of parton saturation.

In the following, we summarize how the four general goals are addressed by the measurements discussed in this report. As the density and temperature attained in hadronic collisions changes mildly but distinctly with the centre-of-mass energy, the physics opportunities listed below can be further enhanced by a combined interpretation of future measurements at the LHC and at RHIC. In particular, new or upgraded experiments at RHIC aim at providing measurements of highly-improved quality and precision in the sectors of hard probes (jets and jet correlations, heavy flavour, quarkonia) and electromagnetic observables.

1.1 Macroscopic QGP properties

At sufficiently long wavelength, essentially all forms of matter can be described by fluid dynamics. The observation of flow-like behaviour in nucleus–nucleus collisions and in smaller collision systems demonstrates that this universal long-wavelength limit of hot and dense QCD matter can be accessed experimentally at the LHC. This provides an experimental inroad to fundamental questions about QCD thermodynamics and hydrodynamics since i) the QGP properties entering a fluid dynamic description are calculable from first principles in quantum field theory, and ii) hydrodynamic long wavelength properties depend on the effective physical degrees of freedom in the plasma and they are thus sensitive to the microscopic dynamics that governs their interactions. The following properties of the QCD matter produced in TeV-scale collisions are accessible via future measurements at the LHC

1. Temperature

Within the programme of determining the QCD equation of state, QCD lattice simulations at finite temperature have established since long a precise relation between the QCD energy density and pressure that determine the fluid dynamic expansion, and the temperature of QCD matter. While energy density and pressure can be constrained experimentally by many measurements, an independent determination of temperature is of great value for testing the idea of local equilibration in heavy-ion collisions or for establishing deviations thereof. Future LHC experiments will constrain the temperature and its time evolution with unprecedented precision, in particular via thermal radiation of real and virtual (dileptons) photons (Chapter 8).

2. *QCD phase transition at* $\mu_B \simeq 0$

Collisions at the LHC realize systems of close-to-zero baryo-chemical potential. QCD calculations on the lattice predict in this regime a smooth cross-over transition from a hot partonic plasma to a cold hadron gas. Fluctuation measures of conserved charges are sensitive to the characteristics of the phase transition. In future LHC experiments, they are accessible with unprecedented precision and completeness (Chapter 3). Future measurements of low-mass dileptons will also be, for the first time at the TeV-scale, sensitive to in-medium modifications of the ρ -spectral function (Chapter 8). This provides unique access to the transition between phases with restored and broken chiral symmetry that is predicted by lattice QCD.

3. Viscosity and further QCD transport coefficients

Existing flow measurements provide tight upper bounds on the value of η/s and they have been a cornerstone in supporting the perfect-fluid paradigm. In the future, measuring higher-order cross-correlations of flow coefficients will significantly extend this line of research. In addition, we note that the value of η/s can be related to the existence and size of the mean free path (isotropization length scale), which in turn results from the existence of (quasi-)particle-like excitations in the produced matter. This motivates increasing the precision on η/s with the aim of establishing the tightest lower bound on this quantity (Chapter 4), and it motivates detailed studies of flow in smaller systems with the idea of identifying the scale at which the system size becomes comparable to the mean free path (Chapter 9). Also, measurements of soft dileptons will give access to the electric conductivity of the strongly interacting medium (Chapter 8), and flow and correlation measurements may help to constrain the bulk viscous corrections (Chapter 4).

4. Heavy-quark transport coefficients

Heavy quarks provide unique tools for testing collective phenomena in nuclear collisions. As they are produced in initial hard-scattering processes and flavour is conserved throughout the collective dynamical evolution, they are the best experimental proxy to the idea of putting coloured test charges of well-defined mass into the medium and testing how they participate in the evolution. Of particular interest are precision measurements of transverse-momentum anisotropies v_n and nuclear-modification factors of open heavy-flavoured mesons that are known to constrain e.g. the heavy-quark diffusion coefficient $2\pi TD_s$ and its dependence on the temperature T, that can be compared to first-principle calculations of QCD on the lattice (Chapter 5).

5. Searching for transport phenomena related to the presence of strong electrodynamic fields.

Heavy-ion collisions produce the largest electromagnetic field of any system accessible to laboratory experiments. The field is largest in the early phase of the collision, thus the early-produced heavy quarks are expected to be the most sensitive to its strength (Chapter 5). As the maximal field strengths are estimated to be of the order of the pion mass ($eB^2 \sim m_{\pi}^2$), effects are also likely to be present for light-flavour charged hadrons (Chapter 4). Other measurements of interest include transport coefficients such as the electric conductivity with which the plasma responds to an electromagnetic field, and that are calculable within QCD. In addition, as a consequence of the chiral anomaly, QCD coupled to QED gives rise to *anomalous hydrodynamics* that displays various qualitatively novel phenomena, including for instance a component of the electromagnetic currents that flows parallel to the magnetic field. The existence of these anomalous phenomena follows from first principles in field theory and thermodynamics, but the size of potential experimental signatures is model-dependent. Beyond determining conventional QED transport phenomena, LHC allows to search with increased precision for these intriguing signatures of anomalous fluid dynamics (Chapter 4).

1.2 Accessing the inner workings of hot QCD matter

Previous experiments at the LHC and at lower centre-of-mass energy have established that the QCD matter produced in nucleus–nucleus collisions is subject to strong collective evolution. However, the nature of the effective constituents of that matter, and its characteristic inner length scales (such as screening lengths or mean scattering times, if any) are not yet understood. The scale dependence of QCD implies that one must be able to resolve partonic constituents of hot QCD matter at sufficiently high resolution scale. This motivates the use of high-momentum transfer processes (hard probes) to study the inner workings of hot QCD matter. In addition, the current status of phenomenological modelling does not exclude the existence of a sizeable mean free path of hot QCD matter (which is assumed in transport model simulations of heavy-ion collisions, but which is not assumed in almost perfect fluid dynamic models). This motivates to learn about the inner workings at hot QCD matter also from particle production at intermediate transverse momentum (such as heavy quark transport at intermediate and low $p_{\rm T}$). Here, we highlight the following opportunities for further experimentation.

1. Constraining with jet quenching the colour field strength of the medium

- In general, the fragments of jets produced in nucleus–nucleus collisions are medium-modified due to interactions with the hot QCD matter. These jet quenching effects depend on the inner structure of that matter. In particular, the average medium-induced colour field strength experienced by the escaping jet can be quantified e.g. with the quenching parameter \hat{q} , which measures the average exchanged transverse momentum squared per unit path length. Experiments at the LHC will provide improved constraints on this field strength measurement (Chapter 6). Qualitatively novel opportunities for testing the time evolution of the medium opacity to hard partons could arise if boosted tops could be studied in nuclear matter. A run at the LHC with lighter (than Pb) nuclei, like e.g. ⁴⁰Ar, would provide sufficient luminosity to this end, as well as largely enhanced kinematic and statistics reach for γ -jet and Z-jet recoil measurements (Chapter 11.1). The opportunities for boosted top measurements in Pb–Pb and lighter nuclei collisions at the HE-LHC are also discussed (Chapter 13).
- 2. Investigating the quasi-particle structure of QCD matter with jet and heavy-quark measurements Hard quarks and gluons with different energies can be used to investigate the constituents of the QCD matter at various resolution scales. On one side of the scale, high-energy quarks and gluons (leading to high-energy jets) address the smallest spatial scale. While \hat{q} characterizes the effects of jet-medium interactions in the coherent regime in which individual constituents in the medium are not resolved, Z/γ -jet correlations and modern jet substructure measurements on the highstatistics samples of future LHC Runs are expected to access a regime of Rutherford-type large angle jet-medium scattering, in which the detection of recoil or of large angle deflections gives insight into the microscopic structure of the produced matter (Chapter 6). On the other side of the scale, the scattering of low-momentum heavy quarks, characterized using for example the diffusion coefficient, addresses the nature of the QCD constituents at long-wavelength scale (Chapter 5).
- 3. Testing colour screening with bottomonium production

The family of bottomonium bound states gives access to a set of well-defined length scales that are embedded in the hot and dense QCD matter. As a consequence, the dissociation of the various bound states sensitively depends on the interplay of color screening and the coupling to the strongly interacting medium through dissociation reactions. This opens a unique window on the in-medium modifications of the fundamental QCD force that binds bottomonium. Increasingly-tight bound states are expected to melt with increasing temperature, providing a laboratory for in-medium spectroscopy. The increased precision of future measurements will allow to test for additional physics mechanisms in the production of bottomonium, such as regeneration processes that may affect the yield of Y(2S) and Y(3S). Also, the higher rates of future experiments will allow one to cross-correlate measurements of bottomonium suppression with other manifestations of collectivity, such as elliptic flow (Chapter 7).

4. Testing colour screening and regeneration dynamics with charmonium production

In close similarity to bottomonium, the medium-modification of charmonium bound states is sensitive to colour screening and it is subject to the same QCD dissociation dynamics. However, since charm quarks and anti-quarks are produced abundantly in nucleus–nucleus collisions at the TeV scale, c and \bar{c} produced in different hard processes can form bound states. Indications of this

qualitatively novel bound-state formation process are accessible at low and intermediate transverse momentum, and they motivate high-precision measurements of nuclear modification factor R_{AA} and elliptic flow v_2 . In addition, future open heavy-flavour measurements reaching down to zero p_T with high precision will help to determine the total charm cross section which is a central input for the precise quantification of regeneration (Chapters 5 and 7).

5. Formation of hadrons and light nuclei from a dense partonic system

The question of how a collective partonic system of many degrees of freedom evolves into the hadronic phase and produces colour singlet hadrons and light nuclei is essential for a complete dynamical understanding of nucleus-nucleus collisions. Recent LHC measurements in protonproton and proton-nucleus collisions suggest that also in small-system hadronic collisions the hadronization process may be modified with respect to elementary e^+e^- collisions. Future measurements at the LHC will enable comparative and multi-differential studies of these modifications with unprecedented precision, for both the heavy-flavour sector (D_s^+, B_s^0) , charm and beauty baryons, see Chapter 5) and for light nuclei and hyper-nuclei (Chapter 3). For open heavy flavour, these measurements are also crucial to disentangle the role of mass-dependent radial flow and of recombination, as well as to constrain the parameters of hadronization in the models that are used to estimate QGP properties like the heavy-quark diffusion coefficients. For light nuclei, precise measurements of nuclei and hyper-nuclei with mass numbers 3 and 4 as well as possible observation of exotic baryonic states will address the question whether their production is dominated by coalescence of protons, neutrons and Λ baryons or by statistical hadronization of a partonic system. These measurements, in addition to that of high-momentum deuteron production, also have important astrophysical implications (dense compact stars as well as dark matter searches in the Cosmos).

1.3 Developing a unified picture of QCD collectivity across system size

As discussed at the beginning of this chapter, recent LHC discoveries of signatures of collectivity in proton-nucleus and in proton-proton collisions question common beliefs. For the smallest collision systems, these measurements indicate that more physics effects are at work in multi-particle production than traditionally assumed in the modelling of proton-proton collisions. For the larger collision systems (proton-nucleus and nucleus-nucleus collisions), they question whether the origin of signatures of collectivity is solely (perfect) fluid dynamical, given that these signatures persist in proton-proton collisions. This raises important qualitative questions like: What is the smallest length scale on which QCD displays fluid-dynamic behaviour? Is there a non-vanishing characteristic mean free path for the production of soft and intermediate $p_{\rm T}$ hadrons, and if so, is it smaller or larger than the proton diameter? What are the novel physics concepts with which underlying event simulations in proton-proton collisions need to be supplemented (e.g. in multi-purpose event generators) to account for the totality of observed phenomena? While some of these questions sound technical, it needs to be emphasized that the size of dissipative properties of QCD matter, such as its shear viscosity, are quantitatively related to the presence or absence of intrinsic length scales such as a mean free path. Any systematic experimental variation of the system size therefore relates directly to a search for intrinsic length scales that determine the dissipative properties of hot QCD matter. Within the present report, we identify in particular the following future opportunities for an improved understanding of the system-size dependence of collective phenomena (Chapter 9):

1. Flow measurements in pp and p-A systems: Onset and higher-order correlations

While flow signals have been established in smaller collision systems in recent years, their detailed characterization lacks behind the state of the art achieved in nucleus–nucleus collisions. Future measurements will allow for characterizing higher-order cumulants in largely non-flow suppressed

multi-particle correlations, and test whether there is a system-size dependence in the characteristic correlations between different flow harmonics v_n , or the characteristic reaction plane correlations.

2. Flow of heavy flavour and quarkonium in smaller systems

Is there a minimal system size needed to transport heavy quarks within a common flow field? Given that the local hard production of heavy flavour is expected to be independent of any collective direction, precision measurements of heavy-flavour flow in pp and p–A collisions will provide decisive tests of heavy quark transport, thus experimentally addressing the question of how QCD flow field build up efficiently and on short length and time scales.

3. Strangeness production as a function of system size

One of the recent surprising LHC discoveries that is not accounted for in traditional models of minimum-bias pp collisions but that may be accounted for in a thermal picture is the smooth increase of strangeness with event multiplicity across system size. We discuss in detail how future measurement at the LHC, such as the study of strange D-mesons or baryons, can extend the systematics underlying this observation (Chapters 5 and Section 9.7).

4. Searching for the onset/existence of energy-loss effects in small systems

All dynamical models of collectivity involve final-state interactions. This implies the existence of jet-medium final-state interactions and, a fortiori, the existence of parton energy-loss effects. The latter have not been identified experimentally, yet. In this report, we discuss novel opportunities to test for their existence, including tests in future p–Pb collisions, as well as opportunities specific to O–O collisions.

5. Searching for the onset/existence of thermal radiation in small systems

If the collectivity observed in smaller collision systems is due to final-state isotropization and equilibration phenomena, it must be accompanied by thermal radiation. The search for the corresponding conceptually clean electromagnetic signatures, such as thermal dilepton and photon production in p–Pb collisions, is an important part in developing a unified picture of QCD collectivity. This report discusses the experimental opportunities in light of experimental upgrades.

1.4 Nuclear parton densities and search for non-linear QCD evolution

Future experiments at the LHC offer a variety of opportunities for precision measurements with nuclear beams. Here, we highlight three opportunities that are clearly related to the main physics challenges of the heavy-ion programme (Chapter 10):

1. Precise determination of nuclear PDFs at high Q^2

As high-momentum transfer processes have a short space-time scale, they are not affected by the long-wavelength particle excitations of the QCD matter in nuclear collisions. This implies that the primary production rates of hard processes in nucleus–nucleus collisions are determined perturbatively, whereas their medium modifications arise from traversing a dense QCD matter of considerable spatial extent and considerable colour-field strength. For a dynamical understanding of jet quenching, control over primary-production rates is indispensable and this necessitates the knowledge of nuclear parton distribution functions. Global nPDF-fits that reflect the current state of the art of nuclear parton distributions could be improved at the LHC in the near future at high Q^2 and $x \sim 10^{-3}$ – 10^{-2} in particular with high-precision W, Z and dijets measurements in p–Pb and Pb–Pb collisions (Sections 10.3 and 10.4). High-luminosity Ar–Ar collisions would enable for the first time using top quarks to constrain nuclear PDFs at very high Q^2 and large $x \sim 10^{-2}$ – 10^{-1} and would contribute to constraining experimentally the nuclear mass number A dependence of nuclear PDFs.

2. Constraining nuclear PDFs at low Q^2

Drell-Yan and photon measurements (Section 10.3) in p–Pb collisions could provide significantly improved constraints on the nuclear parton distribution functions at low Q^2 and low $x \sim 10^{-5}$ – 10^{-3} , where nuclear effects are larger. In addition, although so far global PDFs do not use measurements from ultra-peripheral collisions (UPC), it is thought that quarkonia and dijet production in UPC can constrain nuclear PDFs in the future. This report identifies opportunities for the corresponding measurements (Section 10.2).

3. Access to non-linear QCD evolution at small-x

The scale-dependence of parton distribution functions is known to obey linear QCD evolution equations within a logarithmically wide range in Q^2 and $\ln x$. However, where partonic density in the incoming hadronic wave function are not perturbatively small, qualitatively novel non-linear density effects are expected to affect the QCD evolution. For Q^2 smaller than a characteristic saturation scale $Q_s^2(x)$, these effects are dominant, and as the saturation scale $Q_s^2(x)$ increases with decreasing x, one expects on general grounds at sufficiently small $\ln x$ a qualitatively novel saturation regime in which non-linear QCD evolution occurs at perturbatively large Q^2 . Future measurements at LHC will provide novel test for these saturation effects with previously unexplored measurements (Section 10.3). The larger centre-of-mass energy of the HE-LHC would extend the small-x coverage by an additional factor two. Measurements of relevance include in p–Pb collisions dilepton and photon production at small-x and forward measurements of dihadron and dijet correlations. The perspectives for such measurements are discussed in Chapter 13.

1.5 Physics performance studies by ALICE, ATLAS, CMS and LHCb

The performance results presented in this report were obtained by experimental groups within the four Collaborations. Two types of results are included: *simulation* studies used full or fast simulations of the detector setups for Run 3 and/or Run 4; *projection* studies are based on existing measurements where their uncertainties have been reduced as expected with the future detectors and data samples. The impact of the detector upgrades on the various observables is discussed in the corresponding chapters and in more detail in the referenced documents by the Collaborations. The integrated luminosities used for the physics studies are summarised in Chapter 12.

2 Heavy-ion performance of LHC, HL-LHC and HE-LHC

Coordinator: John M. Jowett (CERN)

Contributors: R. Bruce (CERN), M. Schaumann (CERN), M.A. Jebramcik (Johann-Wolfgang-Goethe Universität, Frankfurt & CERN)

2.1 Heavy-ion performance of LHC in Runs 1 and 2

The 2018 Pb–Pb run of the LHC brought Run 2 to an end and launched the hardware upgrades to the collider, and to the ALICE experiment, that should allow the full "HL-LHC" heavy-ion performance to be delivered from 2021 onward. Beyond pp collisions, the 2004 LHC Design Report [1], specified only Pb–Pb collisions with a peak Pb–Pb luminosity of $L = 1 \times 10^{27} \text{ cm}^{-2} \text{ s}^{-1}$. Now, much of the upgraded performance is already in hand. Not only has that peak Pb–Pb luminosity goal been exceeded by a factor of more than 6, but the p–Pb collision mode—an upgrade beyond the initial design whose feasibility was widely doubted—has yielded similarly high luminosity in multiple operating conditions (see Ref. [2] and references therein). Table 1 summarises the main parameters of the runs to date. Additionally, in 2017, the LHC has collided beams of Xe nuclei [3], providing many new results and demonstrating the potential for colliding lighter species. The goal for 2018 was to complete the accumulation of an integrated Pb–Pb luminosity of 1 nb⁻¹ to each of the ALICE, ATLAS and CMS experiments and this was substantially exceeded. The LHCb experiment also received over 0.26 nb^{-1} .

2.2 Pb–Pb luminosity in Run 3 and Run 4 (HL-LHC)

The High Luminosity LHC (HL-LHC) is an upgrade of the LHC to achieve instantaneous pp luminosities a factor of five larger than the LHC nominal value. Its operational phase is scheduled to start in LHC Run 4, in the second half of the 2020s, for the pp physics programmes described in the other chapters of this report. The HL-LHC project also includes hardware upgrades of the present LHC that will allow the LHC to operate with potential peak Pb–Pb luminosities an order of magnitude larger than the nominal [1]. These upgrades will be completed during Long Shutdown 2 and can already be exploited in Run 3, starting in 2021. Upgrades to the heavy-ion injector chain, in the framework of the LHC Injectors Upgrade project will increase the total stored intensity of heavy-ion beams and will also be completed for Run 3. Finally, the ALICE experiment will be upgraded to accept higher peak luminosity.

The heavy-ion performance of the LHC will be similar in Run 3 and in Run 4. Therefore, the two Runs are discussed together in this report in terms of their contribution to the HL-LHC heavy-ion physics programme. To achieve the performance parameters given in the last column of Table 1 a detailed specification of the requirements on the beams at LHC injection has been given [5] and later updated in Ref. [6]. In a typical one-month Pb–Pb run, this will yield an integrated luminosity of 3.1 nb^{-1} . The necessary single-bunch intensities have already been attained but an implementation of slip-stacking in the SPS will be required to obtain a basic bunch spacing of 50 ns and store over 1200 Pb bunches in each LHC ring. The necessary upgrades of the SPS RF system will be implemented during LS2 and it is planned to commission this new mode of operation in 2021.

2.2.1 Secondary beams from the IPs

Ultra-peripheral electromagnetic interactions of Pb nuclei lead to copious lepton-pair production. Most of this is innocuous except for the (single) bound-free pair production (BFPP1):

$${}^{208}\text{Pb}^{82+} + {}^{208}\text{Pb}^{82+} \longrightarrow {}^{208}\text{Pb}^{82+} + {}^{208}\text{Pb}^{81+} + e^+, \tag{1}$$

in which the electron is bound to one nucleus. As extensively discussed in e.g. Refs. [7–9] and elsewhere, the modified nuclei emerge from the collision point, as a narrow secondary beam with modified magnetic rigidity, following a dispersive trajectory that impacts on the beam screen in a superconducting magnet

Table 1: Representative simplified beam parameters at the start of the highest luminosity physics fills, in conditions that lasted for > 5 days, in each annual Pb–Pb and p–Pb run (Ref. [2] and references therein). The original design values for Pb–Pb [1] and p–Pb [4] collisions. and future upgrade Pb–Pb goals are also shown (in these columns the integrated luminosity goal is to be attained over the 4 Pb–Pb runs in the 10-year periods before and after 2020). Peak and integrated luminosities are averages for ATLAS and CMS (ALICE being levelled). The smaller luminosities delivered to LHCb from 2013–2016 and in the minimum-bias part of the run in 2016 are not shown. Emittance and bunch length are RMS values. Single bunch parameters for p–Pb or Pb–p runs are generally those of the Pb beam. The series of runs with $\sqrt{s_{NN}} = 5.02 \text{ TeV}$ also included pp reference runs, not shown here. Design and record achieved nucleon-pair luminosities are \boxed{boxed} , and some key p–Pb parameters are set in red type, for easy comparison. The upgrade peak luminosity is reduced by a factor $\simeq 3$ from its potential value by levelling.

Quantity	"des	ign"	achieved				upgrade	
Year	(2004)	(2011)	2010	2011	2012-13	2015	2016	≥2021
Weeks in physics	-	-	4	3.5	3	2.5	1, 2	-
Fill no. (best)			1541	2351	3544	4720	5562	-
Species	Pb–Pb	p–Pb	Pb–Pb	Pb–Pb	p–Pb	Pb–Pb	p–Pb	Pb–Pb
Beam energy $E[Z \text{TeV}]$	7	7	3	.5	4	6.37	4,6.5	7
$\begin{array}{ll} \mbox{Pb} & \mbox{beam} & \mbox{energy} \\ E[A{\rm TeV}] \end{array}$	2.7	76	1.	38	1.58	2.51	1.58,2.56	2.76
Collision energy $\sqrt{s_{NN}}$ [TeV]	5.52	8.79	2.	51	5.02	5.02	5.02 ,8.16	5.52
Bunch intensity $N_b [10^8]$	0.	7	1.22	1.07	1.2	2.0	2.1	1.8
No, of bunches k_b	592		137	338	358	518	540	1232
Pb norm. emittance $\epsilon_N [\mu \mathrm{m}]$	1.	5	2.	2.0	2.	2.1	1.6	1.65
Pb bunch length σ_z m 0.08		0.07–0.1					0.08	
$\beta^{*} \mathrm{[m]}$	0.	5	3.5	1.0	0.8	0.8	10, 0.6	0.5
Pb stored energy MJ/beam	3.8	2.3	0.65	1.9	2.77	8.6	9.7	21
$\frac{\text{Luminosity}}{L_{\text{AA}} \left[10^{27} \text{cm}^{-2} \text{s}^{-1}\right]}$	1	150	0.03	0.5	116	3.6	850	6
$\frac{\mathrm{NN}}{L_{\mathrm{NN}}} \frac{\mathrm{luminosity}}{[10^{30} \mathrm{cm}^{-2} \mathrm{s}^{-1}]}$	43	31	1.3	22.	24	156	177	260
Integrated luminos- ity/experiment $[\mu b^{-1}]$	1000	10^5	9	160	32000	650	1.9×10^5	10^4
Int. NN lumi./expt. $[nb^{-1}]$	43000	21000	380	6700	6650	28000	40000	4.3×10^5

in the dispersion suppressor (DS) downstream. These secondary beams emerge in both directions from every interaction point (IP) where ions collide. Each carries a power of

$$P_{\rm BFPP} = L\sigma_{\rm BFPP}E_b,\tag{2}$$

where L is the luminosity and $\sigma_{\rm BFPP} \simeq 276$ b is the cross-section at the 2015/18 run energy of $E_b = 6.37Z$ TeV. These losses carry much greater power than the luminosity debris (generated by the nuclear collision cross-section of 8 b) and can quench magnets and directly limit luminosity. With a peak luminosity of $L = 6.1 \times 10^{27} \,\mathrm{cm}^{-2} \mathrm{s}^{-1}$ each secondary beam carries $P_{\rm BFPP} \lesssim 120$ W, which is more than enough to quench an LHC dipole as demonstrated in 2015 [10].

To reduce the risk of quenching these magnets, orbit bumps were implemented around the impact

locations in IP1 and IP5 in order to move the losses out of the dipole and into the adjacent connection cryostat ("missing dipole" in the DS) that does not contain a superconducting magnet coil and therefore is less likely to quench. This technique was first used in 2015. It was almost fully proved in 2018 when the ATLAS and CMS Pb–Pb luminosities were sustained at values very close to the nominal levelling values for Runs 3 and 4. Beam-loss monitor thresholds were set, based on the measured quench level in 2015 and it was clear that there was sufficient margin for still higher luminosity. In IP2, the method of orbit bumps alone is not applicable with present optics and layout. It is therefore foreseen to install an additional collimator in the connection cryostat on the outgoing beam on each side of IP2. In combination with this, orbit bumps will then be deployed to steer the BFPP beams onto the collimators. The installation will take place in LS2 in order to allow the HL-LHC design luminosity for ALICE (corresponding to a hadronic event rate of $50 \, \rm kHz$) in subsequent runs.

2.2.2 Collimation and intensity limit

While the LHC stores unprecedented beam energies, superconducting magnets are needed to bend and focus these beams, most of which are operated at 1.9 K. A loss of a tiny fraction of the beam is enough to induce a magnet quench, and it is therefore vital to avoid any uncontrolled beam losses. To safely intercept losses and provide protection of the magnets, the LHC uses a multi-stage collimation system [11–14]. During the first two runs of the LHC, this system has shown a very good performance with proton beams [15–18] and ion beams [19, 20].

LHC collimation is much less efficient with heavy-ion beams than with protons, since ions have a high cross section for undergoing nuclear fragmentation inside the primary collimators [21]. The angular offsets of the out-scattered fragments are frequently not large enough to reach the secondary collimators in the straight collimation insertion (IR7). At the same time these fragments have a magnetic rigidity different from the main beam, so that they risk being lost where the dispersion starts to rise in the first few dipoles of the DS. This was the most critical beam loss location during the Pb ion runs in Run 1 and Run 2, with a local cleaning inefficiency of about a factor 100 worse than for protons [19, 20]. Therefore, even though the total stored beam energy is about a factor 10 lower with Pb ions than with protons, collimation of heavy ions is critical. Still, ion collimation has worked well in the LHC and did not introduce operational bottlenecks so far. However, extrapolations of the losses in the DS from a 2015 experimental tests to Run 3 and HL-LHC show that, if nothing is done, the total stored Pb beam energy is limited to around 10 MJ, if drops of the instantaneous beam lifetime down to 12 minutes are assumed [22]. At the same time, it is foreseen to increase the stored Pb beam energy to about 24 MJ. To alleviate this limitation and safely intercept the losses, it is planned to install additional collimators, called TCLDs, in the dispersion suppressors [23–25]. On the other hand, the LHC was successfully operated with Pb beams containing over 13.5 MJ each in 2018 thanks to good control of beam lifetimes. In order to make space for the TCLDs, standard 8.3 T LHC dipoles will be replaced by an assembly consisting of two shorter higher-field 11 T dipoles with the TCLD in between [25]. The solution that gives the best simulated Pb cleaning efficiency uses two TCLDs per side of IR7. However, this is not possible within tight constraints of long shutdown 2 and the HL-LHC project, and the baseline is therefore to install one TCLD per side. If this turns out to be a real limitation, it could be considered at a later stage to install a second TCLD. As an alternative and complementary alleviation method, it is under study whether crystal collimation could help in reducing the losses in the DS. In this collimation scheme, bent crystals are used instead of the standard LHC primary collimators [26]. Incoming beam particles follow the curvature of the crystal planes, the so-called channelling, and exit with a significant angular kick. They can then be efficiently steered onto an absorber. Nuclear interactions inside the the channels of well-aligned crystals are significantly suppressed. Initial experiments using an LHC test installation [27] have shown very promising results with Xe and proton beams [28]. Channelling has very recently also been observed with Pb beams in 2018 and potential improvements of the collimation system are presently being assessed experimentally. Studies with Pb beams are not yet conclusive but it is hoped that this will be further clarified by analysis of data taken during the 2018 Pb ion run.

Collimation of lighter ion species has not yet been studied in detail, although some first simulations are presented in Ref. [19]. Results for Ar and Xe beams show that the amount of expected losses in the DS is similar to Pb but the longitudinal loss distribution changes. The fractional change in magnetic rigidity for every lost nucleon in the collimators is larger for light ions, and it is hence expected that outscattered fragments have larger effective energy deviations and are lost more upstream. It is thus likely that the TCLD should help significantly also for lighter ions, although comparative studies on intensity limits for different ion species still remain to be done.

2.3 Proton-lead operation in Run 3 and HL-LHC

Within colliding nuclei, with charges Z_1 , Z_2 and nucleon numbers A_1 , A_2 , in rings with magnetic field set for protons of momentum p_p^{-1} , the colliding nucleon pairs will have an average centre-of-mass energy

$$\sqrt{s_{_{\rm NN}}} \approx 2c \, p_p \sqrt{\frac{Z_1 Z_2}{A_1 A_2}} \approx 2c \, p_p \begin{cases} 1 & \rm pp \\ 0.628 & \rm p-Pb \\ 0.394 & \rm Pb-Pb \end{cases}$$
(3)

and a central rapidity shift in the direction of the (Z_1, A_1) beam

$$y_{\rm NN} \approx \frac{1}{2} \log \left(\frac{Z_1 A_2}{A_1 Z_2} \right) \approx \begin{cases} 0 & \text{pp, Pb-Pb} \\ 0.465 & \text{p-Pb} \\ -0.465 & \text{Pb-p} \end{cases}$$
 (4)

We present parameters for operation at the nominal LHC momentum $p_p c = 7 \text{ TeV}$ extrapolating from the experience of the last p-Pb run in 2016.

The injection and ramp of protons and lead ions with equal magnetic rigidity leads to moving longrange beam-beam encounters in the four interaction regions of the LHC. These beam-beam encounters were one of the reasons why the feasibility of p–Pb operation in the LHC was initially questioned. This effect has been proven small in the LHC and calculations have this will remain true for the HL-LHC era despite larger bunch numbers and higher proton bunch intensities. The dynamic range of the interlock strip-line BPMs, common for the lead and proton beam, limited the proton intensity to $N_b < 5 \times 10^{10}$ protons per bunch during Run 1. Gating the stripline BPM read-out appropriately removed this constraint a few days before the end of the 2016 run. The higher proton intensity of $N_b = 2.8 \times 10^{10}$ protons per bunch resulted in increased luminosities at the IPs but also led to the substantial deposition of collision debris from the Pb beam in the dispersion suppressors at ATLAS and CMS risking a beam dump [29]. The collision debris collimators (TCLs), which could have intercepted emerging fragments from the IPs, were not commissioned at tighter settings for the 2016 p–Pb run. Appropriate TCL settings are expected to neutralise these fragments and should allow for higher peak luminosities in the future.

A potential p–Pb run during Run 3 and beyond will greatly benefit from the longitudinal slip stacking in the SPS and the small $\beta^* = 0.5 \text{ m}$ in three experiments. The proton intensity cannot be pushed to values much larger than the maximum achieved in 2016 as bunches colliding in multiple IPs and especially in ATLAS and CMS will approach the interlock BPM threshold of 2×10^9 charges per bunch too quickly if the luminosities of ATLAS and CMS are not levelled. This would lead to an undesirable early beam dumps while ALICE is still levelled. In order to predict the potential performance of a future p–Pb run, the expected Pb–Pb filling pattern [6] is used providing 1136 collisions in ATLAS/CMS, 1120 collisions in ALICE and 81 collisions in LHCb. This approximation is made since the proton injection should be flexible enough to reproduce most of the respective Pb pattern. This calculations assumes $N_b = 3 \times 10^{10}$ protons per bunch and ALICE being levelled to the instantaneous

¹Conditions imposed by the two-in-one magnet design of the LHC.



Fig. 1: Evolution of the instantaneous luminosity in the LHC experiments during a p-Pb fill. At around 6.1 h (dashed line), the interlock BPM threshold is reached for some bunches, limiting the fill length.

luminosity of $L_{AA} = 5 \times 10^{29} \text{ cm}^{-2} \text{s}^{-1}$. $L_{AA} = 5 \times 10^{29} \text{ cm}^{-2} \text{s}^{-1}$ ATLAS and CMS are not luminosity levelled in this scenario since the loss of integrated luminosity for ATLAS and CMS outweighs the marginal gain for ALICE. A simulation of the beam evolution based on ordinary differential equations including intra-beam scattering and radiation damping leads to a luminosity evolution in the different IPs as displayed in Fig. 1.

At around 6.1 h, the bunch intensity of the bunches colliding in ATLAS, ALICE and CMS drop below the interlock BPM threshold ultimately limiting the fill length, even though the potential levelling time for ALICE has not been reached. Detailed engineering of the filling scheme might avoid this. Key results from the beam evolution study are listed in Tab. 2. The expected peak luminosity in ATLAS and CMS is at around $L_{AA} = 17.4 \times 10^{29} \text{ cm}^{-2} \text{s}^{-1}$, i.e., roughly a factor 2 larger than in 2016. The integrated luminosity in ATLAS and CMS are expected to approach 0.7 pb⁻¹ outperforming the 2016 integrated luminosity by a rough factor 3.5. Since the nominal HL-LHC normalised proton emittance of $\epsilon_N = 2.5 \,\mu\text{m}$ is assumed, the actual performance may exceed these predictions since normalised proton emittances in the range of $\epsilon_N = 1.3 \,\mu\text{m}$ have already been achieved.

2.4 Colliding lighter nuclei at HL-LHC

The bunch intensity limits in the injectors depend largely on the ion charge which changes at the various stripping stages which must be optimised for space-charge limits, intra-beam scattering, efficiency of electron-cooling, beam losses on residual gas and other effects in the ion source, Linac4, LEIR, the PS and SPS. Given the uncertainties, the deliverable intensity for other species can only be determined after sufficient time spend commissioning and empirically optimising the many parameters and operating modes of the whole injector chain. To simplify present considerations, we postulate a simple form relating the number of ions per bunch, N_b , to the well-established value ($N_b(82,208)=1.9 \times 10^8$) for Pb beams

$$N_b(Z,A) = N_b(82,208) \left(\frac{Z}{82}\right)^{-p}$$
(5)

Fitting such an expression to the limited information [30] from the few species used for SPS fixed-target in recent years (since the commissioning of the present ECR ion source and LEIR) yields a value of the fit parameter p = 1.9. Beam quality requirements for fixed-target beams are, of course, less stringent than for injection into the collider. Fitting to the first commissioning of Xe beams for the LHC [3], on the other hand, gives a much less optimistic p = 0.75. Although this was the only occasion where any

Table 2: Key parameters and results of the p–Pb beam evolution calculation. A turn-around time, i.e., the time between Beam Dump and Stable Beams, of 3h and an operational efficiency factor of 0.5 is assumed. The final result was scaled down by additional 5% to take potential deviations of the proton filling pattern into account. The time the first bunches need to reach the interlock BPM threshold is used as the fill time.

Species	р	Pl	0
Beam energy $E[Z \text{ TeV}]$		7	
Collision energy $\sqrt{s_{_{\rm NN}}}$ [TeV]	8	8.78	
Bunch intensity $N_b [10^8]$	300	1.	8
No, of bunches k_b	1232	123	32
Norm. emittance $\epsilon_N [\mu m]$	2.5	1.6	55
Bunch length σ_{z} [m]	0.09	0.0)8
Fill time $t_{\rm fill}$ [h]		6.1	
IP	ATLAS/CMS	ALICE	LHCb
β^* [m]	0.5	0.5	0.5
Colliding bunches k_c	1136	1120	81
Luminosity $L_{AA} [10^{29} \text{cm}^{-2} \text{s}^{-1}]$	17.4	5.0	1.2
NN luminosity $L_{\rm NN} [10^{31} {\rm cm}^{-2} {\rm s}^{-1}]$	36.3	10.4	2.6
$\int_{\text{month}} L_{AA} \text{ dt } [\text{nb}^{-1}]$	674	328	41.1
$\int_{\text{month}} L_{\text{NN}} dt [\text{pb}^{-1}]$	140	68	8.5

other species than Pb was delivered to the collider, only the simplest version of the injection scheme was used and it is clear that, given time, significantly higher intensities could be achieved. We consider that $1.5 \le p \le 1.9$ corresponds to a representative range of possibilities that could be realised in fully-prepared future operation.

In addition, we make a number of simplifying assumptions to allow a simplified, yet meaningful, comparison between species

- The geometric transverse beam emittances at the start of collisions will be equal to those of Pb beams [5]. This is justified, at least at the level of the LHC, since the scaling of intra-beam scattering with N_b , Z and A, given by the parameter $f_{\rm IBS}/({\rm m~Hz})$ is generally smaller than for Pb as long as $p \leq 1.9$. A similar scaling should hold in the injectors such as the SPS where intrabeam scattering may also blow up the emittances. This ignores possible space-charge limits in the injectors which should also be considered once the appropriate stripping schemes and charge states have been defined.
- Same filling scheme and number of bunches.
- No luminosity-levelling in any experiment.
- Fill length optimised for intensity evolution dominated by luminosity burn-off.
- Equal operational efficiency of 50%. Following conventional practice for HL-LHC, the integrated luminosity for a 1-month run is estimated assuming back-to-back ideal fills of optimal length and a turn-around time of 2.5 h between the end of one fill and the resumption of "Stable Beams" for collisions in the next. The operational efficiency factor is then applied as a way of taking into account the time needed for commissioning, intensity ramp-up, faults and other non-availability of LHC and injector systems.

The parameters are estimated using analytical approximations unlike the more elaborate simulations used in Section 2.2. Together with the assumption that there is no luminosity levelling, these lead to a higher estimate of integrated luminosity in a one-month run. Nevertheless they can be used as a guide to the

	$^{16}O^{8+}$	$^{40}{\rm Ar}^{18+}$	$^{40}Ca^{20+}$	$^{78}{ m Kr}^{36+}$	129 Xe ⁵⁴⁺	$^{208}\text{Pb}^{82+}$
γ	3760.	3390.	3760.	3470.	3150.	2960.
$\sqrt{s_{\rm NN}}$ /TeV	7.	6.3	7.	6.46	5.86	5.52
$\sigma_{\rm had}/{\rm b}$	1.41	2.6	2.6	4.06	5.67	7.8
$\sigma_{ m BFPP}/ m b$	2.36×10^{-5}	0.00688	0.0144	0.88	15.	280.
$\sigma_{ m EMD}/ m b$	0.0738	1.24	1.57	12.2	51.8	220.
$\sigma_{ m tot}/ m b$	1.48	3.85	4.18	17.1	72.5	508.
N_b	6.24×10^{9}	1.85×10^9	1.58×10^{9}	6.53×10^{8}	3.56×10^8	1.9×10^8
$\epsilon_{\rm xn}/\mu{ m m}$	2.	1.8	2.	1.85	1.67	1.58
$f_{\rm IBS}/({\rm m~Hz})$	0.0662	0.0894	0.105	0.13	0.144	0.167
$W_b/{ m MJ}$	68.9	45.9	43.6	32.5	26.5	21.5
$L_{ m AA0}/ m cm^{-2}s^{-1}$	1.46×10^{31}	1.29×10^{30}	9.38×10^{29}	1.61×10^{29}	4.76×10^{28}	1.36×10^{28}
$L_{ m NN0}/ m cm^{-2}s^{-1}$	3.75×10^{33}	2.06×10^{33}	1.5×10^{33}	9.79×10^{32}	7.93×10^{32}	5.88×10^{32}
$P_{\rm BFPP}/{\rm W}$	0.0031	0.179	0.303	5.72	43.4	350.
$P_{\rm EMD1}$ /W	4.98	16.5	16.9	40.5	76.7	141.
$ au_{ m L0}/ m h$	16.4	21.3	23.	13.5	5.87	1.57
$T_{\rm opt}/h$	9.04	10.3	10.7	8.23	5.42	2.8
$\langle L_{\rm AA} \rangle \ {\rm cm}^{-2} {\rm s}^{-1}$	8.99×10^{30}	8.34×10^{29}	6.17×10^{29}	9.46×10^{28}	2.23×10^{28}	3.8×10^{27}
$\langle L_{\rm NN} \rangle \ {\rm cm}^{-2} {\rm s}^{-1}$	2.3×10^{33}	1.33×10^{33}	9.87×10^{32}	5.76×10^{32}	3.71×10^{32}	1.64×10^{32}
$\int_{\text{month}} L_{\text{AA}} \text{dt/nb}^{-1}$	1.17×10^4	1080.	799.	123.	28.9	4.92
$\int_{\text{month}} L_{\text{NN}} \text{dt/pb}^{-1}$	2980.	1730.	1280.	746.	481.	213.
$R_{\rm had}/{\rm kHz}$	2.07×10^4	3340.	2440.	653.	270.	106.
μ	1.64	0.266	0.194	0.0518	0.0215	0.00842

Table 3: Parameters and performance for a range of light nuclei with a moderately optimistic value of the scaling parameter p = 1.5 in (5).

relative gain factors in integrated nucleon-nucleon luminosity by changing from Pb to a lighter nucleus.

2.5 Short run for p–O

As discussed in Section 11.3, a short p–O collision run is of interest for cosmic-ray physics. This could be scheduled with rapid set-up in LHC on the successful model of the 2012 p–Pb run which was later re-used in the 2017 Xe–Xe run [3]. Each of those runs took about 16 h of LHC operation time, including set-up and physics data-taking.

Because oxygen is used as the carrier gas in the CERN heavy-ion source, the idea has been mooted that it may be possible to switch from Pb to ${}^{16}O^{8+}$ beams for the LHC, and back, somewhat more rapidly than other species. However it appears that the process is still not quick enough to allow, say, a short p–O run at the end of one of the annual Pb–Pb runs. A more practical possibility would be to schedule the run earlier in the year in order to provide time for the source to be switched back to Pb operation afterwards for the one-month run later in the year. Commissioning of the O beam in the injectors for single-bunch injection into the LHC would need to be scheduled in parallel with p-p operation in the weeks before the p–O run.

2.6 Heavy-ion performance of HE-LHC

Heavy-ion operation of HE-LHC awaits a fully detailed study. First results were presented in [31]. Since the HE-LHC would occupy the same tunnel as the LHC, one can, for the moment, assume the same injected beams as HL-LHC. Future possible upgrades to the injectors might improve this. The total

	$^{16}O^{8+}$	$^{40}{\rm Ar}^{18+}$	$^{40}Ca^{20+}$	$^{78}{ m Kr}^{36+}$	129 Xe ⁵⁴⁺	$^{208}\text{Pb}^{82+}$
γ	3760.	3390.	3760.	3470.	3150.	2960.
$\sqrt{s_{\rm NN}}$ /TeV	7.	6.3	7.	6.46	5.86	5.52
$\sigma_{ m had}/ m b$	1.41	2.6	2.6	4.06	5.67	7.8
$\sigma_{\rm BFPP}/{\rm b}$	2.36×10^{-5}	0.00688	0.0144	0.88	15.	280.
$\sigma_{\rm EMD}/{\rm b}$	0.0738	1.24	1.57	12.2	51.8	220.
$\sigma_{\rm tot}/{\rm b}$	1.48	3.85	4.18	17.1	72.5	508.
N_b	1.58×10^{10}	3.39×10^{9}	2.77×10^{9}	9.08×10^8	4.2×10^8	1.9×10^8
$\epsilon_{\rm xn}/\mu{\rm m}$	2.	1.8	2.	1.85	1.67	1.58
$f_{\rm IBS}/({\rm m~Hz})$	0.168	0.164	0.184	0.18	0.17	0.167
$W_b/{ m MJ}$	175.	84.3	76.6	45.2	31.4	21.5
$L_{\rm AA0}/{\rm cm}^{-2}s^{-1}$	9.43×10^{31}	4.33×10^{30}	2.9×10^{30}	3.11×10^{29}	6.66×10^{28}	1.36×10^{28}
$L_{\rm NN0}/{\rm cm}^{-2} s^{-1}$	2.41×10^{34}	6.93×10^{33}	4.64×10^{33}	1.89×10^{33}	1.11×10^{33}	5.88×10^{32}
$P_{\rm BFPP}/{\rm W}$	0.0199	0.601	0.935	11.	60.6	350.
$P_{\rm EMD1}/{\rm W}$	32.	55.6	52.2	78.3	107.	141.
$\tau_{\rm L0}/{\rm h}$	6.45	11.6	13.1	9.74	4.96	1.57
$T_{\rm opt}/h$	5.68	7.62	8.08	6.98	4.98	2.8
$\langle L_{\rm AA} \rangle \ {\rm cm}^{-2} {\rm s}^{-1}$	4.54×10^{31}	2.45×10^{30}	1.69×10^{30}	1.68×10^{29}	2.95×10^{28}	3.8×10^{27}
$\langle L_{\rm NN} \rangle \ {\rm cm}^{-2} {\rm s}^{-1}$	1.16×10^{34}	3.93×10^{33}	2.71×10^{33}	1.02×10^{33}	4.91×10^{32}	1.64×10^{32}
$\int_{\text{month}} L_{\text{AA}} \text{dt/nb}^{-1}$	5.89×10^4	3180.	2190.	218.	38.2	4.92
$\int_{\text{month}} L_{\text{NN}} \text{dt/pb}^{-1}$	1.51×10^4	5090.	3510.	1330.	636.	213.
$R_{\rm had}/{\rm kHz}$	1.33×10^5	1.12×10^4	7540.	1260.	378.	106.
μ	10.6	0.893	0.598	0.1	0.03	0.00842

Table 4: Parameters and performance for a range of light nuclei with an optimistic value of the scaling parameter p = 1.9 in (5).

integrated luminosity obtainable per fill, summed over all experiments, is bounded by the total intensity divided by the burn-off cross-section and will therefore be similar to the HL-LHC. The same can be said for time taken to inject a fill. Only a modest increase in integrated luminosity, given by somewhat shorter times spent in collision and, most likely, a reduction in the number of experiments, can be envisaged. For purposes of this report, we estimate that the integrated luminosity obtained by each of two experiments in a one-month run will be of order 6 nb^{-1} . However, the BFPP power with Pb–Pb collisions will be very high and this could be a strong argument for running with somewhat lighter species. In that case, one can expect the luminosity to scale similarly to HL-LHC in Sect. 2.4.

3 Light flavour sector: (anti-)(hyper-)nuclei and fluctuations of conserved charges

Coordinator: Francesca Bellini (CERN)

Contributors: M. Arslandok (Heidelberg University), N. K. Behera (Inha University), R. Bellwied (Houston University), K. Blum (CERN and Weizmann Institute of Science), A. Borissov (Münster University), P. Braun-Munzinger (EMMI/GSI Helmholtzzentrum für Schwerionenforschung GmbH), B. Doenigus (Frankfurt University), L. Fabbietti (TU Munich), S. Floerchinger (Heidelberg University), A.P. Kalweit (CERN), R. Lea (University and INFN, Trieste), A. Mastroserio (Foggia University and INFN, Bari), A. Ohlson (Heidelberg University), V. Okorokov (National Research Nuclear University MEPhI, Moscow), S. Piano (INFN Trieste), M. Puccio (University and INFN, Torino), K. Redlich (University of Wroclaw and EMMI/GSI Helmholtzzentrum für Schwerionenforschung GmbH), A. Rustamov (NNRC Baku, GSI Helmholtzzentrum für Schwerionenforschung GmbH), J. Stachel (Heidelberg University), A. Timmins (University of Houston), S. Trogolo (University and INFN, Torino).

3.1 Introduction

The analysis of the data collected at the LHC during Run 1 and Run 2 has consolidated our understanding of a standard model for the production of light-flavour hadrons (containing u, d and s quarks) in heavy-ion collisions: particle chemistry (described by integrated particle yields) is well described by the thermalstatistical model [32, 33] and kinetic equilibrium (reflected in the $p_{\rm T}$ -dependence of particle production) is well described by a common radial expansion governed by hydrodynamics [34,35]. While the physics of light-flavour particles is often perceived as not statistics hungry, the unprecedented large integrated luminosities expected in Run 3 and Run 4 at the LHC offer a unique physics potential. Despite containing only u, d and s valence quarks, light (anti-)(hyper-)nuclei are very rarely produced because of their composite nature and very large mass. Their study will enormously profit from the significant increase in luminosity for heavy-ion collisions expected in the years 2021 until 2029. The same holds true for the study of event-by-event fluctuations of the produced particles, which is closely linked to the production of light (anti-)(hyper-)nuclei in the scenario of a common chemical freeze-out determining light-flavour hadrons and (hyper-)nuclei abundances. If, as indicated by the recent experimental findings [33], the thermal-statistical approach is the correct model to describe (anti-)(hyper-)nuclei production, the chemical freeze-out temperature is most precisely determined by measurements of light (anti-)(hyper-)nuclei as they are not subject to feed-down corrections from strong decays [32]. This is the same temperature at which event-by-event fluctuations of conserved quantities are compared to lattice QCD (lQCD). The physics of light (anti-)(hyper-)nuclei and exotic multi-quark states together with the related observables that will become experimentally accessible in Pb–Pb collisions at the LHC Runs 3 and 4 are discussed in Sec. 3.2. In Sec. 3.3, measurements of fluctuations of particle production and conserved charges are discussed as they give experimental access to fundamental properties of the QCD phase transition at $\mu_{\rm B}$ and allow for direct comparison with lQCD calculations.

In small collision systems (pp, p–Pb), measurements of light-flavour hadrons provide fundamental input to the study of particle production mechanisms and collectivity across systems, as discussed in Ch. 9. At the same time, the physics programme with pp and p–Pb collisions in Runs 3 and 4 will open the possibility for system-size dependent studies of (anti-)nuclei production and for precision measurements of the hyperon-nucleon potentials. The physics case for these measurements in small colliding systems is motivated in this chapter in Sec. 3.2.6, as well as the implications of the findings at the LHC for astrophysics and searches for dark matter in space-based experiments.

3.2 (Anti-)(hyper-)nuclei production

3.2.1 Testing thermal production and nucleon coalescence models

The production of light (hyper-)nuclei and their anti-matter counterparts is modeled within the scenarios of thermal-statistical hadronisation and nucleon coalescence. In the thermal-statistical approach [32,

36], particles are produced from a fireball in thermal equilibrium with temperatures of the order of $T_{\rm chem} \approx 156$ MeV that are near the temperature of the QCD phase transition boundary, as predicted by lQCD calculations [37, 38]. The yields of the produced objects depend on the chemical freeze-out temperature $T_{\rm chem}$ (when inelastic collisions cease) and the mass m of the object, and approximately scale as $dN/dy \propto \exp(-m/T_{\rm chem})$. Thermal-statistical models have been successful in describing light-flavour particle production across a wide range of energies in nucleus-nucleus collisions [32, 33]. Due to their large mass, light (anti-)(hyper-)nuclei are particularly sensitive to $T_{\rm chem}$ and since they are not affected by feed-down from higher mass states [32], the measurement of their production constitutes a precision test for the thermal model.

In the coalescence scenario, composite objects are formed at kinetic freeze-out by coalescence of nucleons that are close in configuration and momentum space [39–44]. Calculations of the coalescence probability based on a density matrix approach [44] require the knowledge of the nucleus wave function and identify the volume of the particle source as the homogeneity volume that can be extracted via Hanbury-Brown–Twiss interferometry [45]. The size of the (hyper-)nucleus is identified with the size parameter of its wave-function, which is related to the (measurable) rms of the charge distribution by simple relations [42, 46].

While there are several theory groups working on the calculation of the expected coalescence [44, 47–50] and thermal production rates [36,51,52], predictions reported in Fig. 2 rely on the study presented in [46], which contrasts the two production scenarios. In order to distinguish them, a measurement of the coalescence parameter for (anti-)(hyper-)nuclei that differ by mass, spin and size as a function of source volume (or source radius) is proposed. The coalescence parameter B_A is defined as

$$E_{A} \frac{\mathrm{d}^{3} N_{A}}{\mathrm{d} p_{A}^{3}} = B_{A} \left(E_{\mathrm{p,n}} \frac{\mathrm{d}^{3} N_{\mathrm{p,n}}}{\mathrm{d} p_{\mathrm{p,n}}^{3}} \right)^{A} \Big|_{\vec{p}_{\mathrm{p}} = \vec{p}_{\mathrm{n}} = \frac{\vec{p}_{A}}{A}},$$
(6)

where $p_{p,n}$ are the momenta of the proton and neutron and $E_{p,n}$ their energies. In the coalescence model (black curves in top panels of Fig. 2), the coalescence parameter is determined analytically. The thermal model predicts p_T -independent particle yields at a given T_{chem} , therefore a Blast-Wave (BW) model is used in [46] to describe the p_T -dependence of (hyper-)nuclei and nucleon production. With the p_T spectra of (hyper-)nuclei and protons obtained in this way, Eq. 6 is used to extract B_A (dashed blue curve in top panels of Fig. 2). Similarly, the coalescence parameter is obtained experimentally from Eq. 6 using the measured (hyper-)nucleus and proton p_T distributions as input. It is considered that for BW, little energy dependence of the fit parameters is observed in Pb–Pb collisions from $\sqrt{s_{NN}} = 2.76$ to 5.02 TeV. The thermal model yields only depend on temperature and no collision energy dependence of the temperature is expected in the LHC energy range. The size of the source can be sampled by means of multiplicity- and centrality-differential measurements.

The particle with the strongest sensitivity to the production mechanism appears to be the hypertriton (a pAn bound state) with its large charge rms radius of about 10 fm, for which the coalescence and the thermal model predictions differ by up to three orders of magnitude as a function of the source radius. While the hypertriton seems to be largely suppressed with respect to ³He (pnn), the ⁴_AH (ppAn) is predicted to have only a slightly lower coalescence probability with respect to ⁴He (ppnn). Moreover, for small *R*, i.e. in small systems as those formed in pp and p–Pb collisions, ³_AH is predicted by coalescence to be suppressed by about a factor of 100 with respect to ³He. These considerations motivate systematic multi-differential measurements of A = 3 and A = 4 nuclei and hyper-nuclei as a function of multiplicity and from small (pp, p–Pb) to large systems (Pb–Pb) to test the validity of the coalescence picture as opposed to thermal production.

With an integrated luminosity $L_{\text{int}} = 10 \text{ nb}^{-1}$ in Pb–Pb collisions in Runs 3 and 4, B_A for ³He, ³_AH and ⁴He can be measured in ALICE in up to ten centrality classes with a statistical precision lower than 5%, 10% and 20%, respectively. The projected relative statistical uncertainties on B_A (σ_{stat}/B_A) for (hyper-)nuclei with A > 2 are reported in the central row of panels of Fig. 2. These uncertainties



Fig. 2: Top: comparison of predictions for the coalescence parameters for (hyper-)nuclei with A = 3, 4 from the Blast-Wave + GSI-Heidelberg thermal statistical model and nucleon coalescence as a function of the radius (*R*) of the particle emitting source. For each (hyper-)nucleus, the radius *r* considered by the coalescence model is reported in the legend. For ${}^{3}_{\Lambda}$ H (${}^{4}_{\Lambda}$ H), two values of the radius are considered: the lower value represents the average separation of the three (four) constituents, whereas the larger *r* corresponds to the average separation between the Λ and the deuteron (triton) core. See [46] for full details on the models. Middle: projection of the relative statistical uncertainty achievable with a minimum bias Pb–Pb integrated luminosity of L_{int} = 10 nb⁻¹ and the upgraded ALICE detector (in red) compared to the relative statistical uncertainty of the Run 2 measurements (in black). Bottom: significance in the discrimination between the two models, assuming 10% and 20% systematic uncertainty in addition to the statistical uncertainty expected with L_{int} = 10 nb⁻¹. For ${}^{3}_{\Lambda}$ H, the coalescence prediction considered is for r = 6.8 fm (corresponding to the black continuous lines in the top panel).

have been estimated by scaling the significance of the nuclei and hyper-nuclei spectra measurements in Pb–Pb collisions at $\sqrt{s_{NN}} = 5.02$ TeV [53, 54] to the expected integrated luminosity of Runs 3 and 4 and assuming thermal production for the states with A = 4. The uncertainties on the proton spectra are negligible already in the existing measurements.

The experimental discrimination power between the models has been extracted as $(B_A^{therm} - B_A^{coal})/\sigma$, where $\sigma = \sqrt{\sigma_{\text{stat}}^2 + \sigma_{\text{sys}}^2}$, and is reported in the lower panels of Fig. 2. Relative systematic uncertainties $\sigma_{\text{sys}}/B_A = 10\%$ and 20% have been considered, to be compared with a typical 15% uncertainty of the Run 1 and 2 measurements. Measurements of ${}^3_{\Lambda}$ H allow for a 10 σ discrimination between models, even in a pessimistic scenario in all centralities. The discrimination power rises above the 10 σ level for ⁴He in semi-central and peripheral collisions.

3.2.2 Light (anti-)(hyper-)nuclei observables in Runs 3 and 4

Measurements of (anti-)(hyper-)nuclei and exotic QCD bound states require large event samples collected with a minimum-bias trigger, as well as high tracking precision for the separation of secondary ver-



Fig. 3: Left: (raw) yield of anti-nuclei in the $2 < p_T < 10 \text{ GeV}/c$ interval, detectable in 0–10% central Pb–Pb collisions with the ALICE detector as a function of the minimum bias luminosity. Right: Projected significance of anti-hyper-nuclei measurements in central Pb–Pb collisions in Runs 3 and 4 with ALICE as a function of the integrated minimum-bias luminosity. In both panels, the arrow represents the minimum bias Pb–Pb luminosity anticipated for the end of Run 2. The dashed vertical line marks the projections with $L_{\text{int}} = 10 \text{ nb}^{-1}$. The bands represents the uncertainty on model prediction for the yield (see text for details).

tices and charged-hadron (light nucleus) identification. The upgraded ALICE detector after LS2 [55–58] fulfills these requirements, developing further the potential already explored in Runs 1 and 2. The yields of (hyper-)nuclei (d, ³He, ⁴He, ³_{Λ}H, ⁴_{Λ}H, ⁴_{Λ}He) and their anti-particles in Pb–Pb collisions at the LHC in Runs 3 and 4 have been estimated for measurements with ALICE. The detectable yield and significance for (anti-)(hyper-)nuclei have been estimated for 0–10% central Pb–Pb collisions considering the acceptance and detection efficiency in the nominal magnetic field of the ALICE detector (B = 0.5 T). These projections are reported for anti-particles in Fig. 3 as a function of the minimum-bias integrated luminosity. The detectable particle and anti-particle yields are equivalent in the considered $p_{\rm T}$ range. All projections have been extracted in the $2 < p_{\rm T} < 10 \text{ GeV}/c$ range, where the lower limit is given by the $p_{\rm T}$ down to which nuclei with A \geq 3 can be reconstructed without ambiguity in ALICE. In a scenario in which ALICE will take data with a central-barrel low-field configuration (B = 0.2 T), it will be possible to extend the low- $p_{\rm T}$ limit for (anti-)nuclei identification down to 1 GeV/c, increasing the expected number of detectable light (anti-)(hyper-)nuclei (by about 20% for ³He). In Fig. 3, the bands indicate the uncertainty on the yield (significance) associated with different model predictions: the central line is obtained assuming statistical-thermal production with $T_{\rm chem}$ = 156 MeV [36], the upper line is the yield (significance) assuming thermal production at $T_{\rm chem}$ = 158 MeV, and the lower one using for the yields the expectation from coalescence (see Sec. 3.2.1). The arrow represents the recorded luminosity at the end of the LHC Run 2. It has to be noted that for this study, the geometry of the ALICE Inner Tracking System (ITS) in Run 2 has been considered. The new geometry and acceptance of the upgraded ITS system [55] are expected to increase the detection efficiency by up to 20%.

The expected yield per unit of rapidity at mid-rapidity are reported for \overline{d} , ${}^{3}\overline{He}$ and ${}^{4}\overline{He}$ in left panel of Fig. 3. With $L_{int} = 10 \text{ nb}^{-1}$ recorded with the nominal magnetic field, a measurement of the elliptic flow (v_2) of 3 He and 3 H (and anti-nuclei) in Pb–Pb collisions will become feasible with ALICE with a statistical precision better than 5% in the 2–10 GeV/*c* transverse momentum range in at least

eight centrality intervals. Elliptic flow measurements for anti-nuclei provide a powerful independent test of coalescence scenarios as already demonstrated with deuterons [59] and might provide an indirect assessment of the neutron flow comparing the ³He and ³H results. In addition, the large data sample that will be collected for light anti-nuclei will lead to the first precise measurements of the mass of light anti-nuclei with A = 3, by means of the Time-Of-Flight detector [60]. This measurement will make it possible to test Charge Symmetry Breaking (CSB) in the anti-nuclei sector due to the differences in the up and down quark masses and due to electromagnetic effects [61]. The differences in A = 3 systems are extensions of the neutron-proton difference. Although the mass difference for the lightest "mirror pair" with A = 3 (i.e. ³H, ³He), is well known (at the level of O(eV) [62]), no measurement has been performed in the anti-matter sector and will be accessible with $L_{int} = 10 \text{ nb}^{-1}$.

In the right panel of Fig. 3, the expected significance of anti-hyper-nuclei measurements in central Pb–Pb collisions is reported as a function of the minimum bias integrated luminosity. For each species, the decay channels with the minimum number of charged particles in the final state and with the highest detection efficiency in ALICE have been considered for this study, as reported in the legend of Fig. 3. The study of other decay channels, e.g. the three body decay of ${}^3_{\Lambda}$ H that has larger theoretical branching ratio with respect to the 2-body decay [63], but lower detection efficiency in ALICE, will be also carried out profiting from the large integrated luminosity. Considering the thermal model predictions at $T_{\text{chem}} = 156$ MeV, the expected significance of ${}^3_{\Lambda}\overline{\text{H}}$, ${}^4_{\Lambda}\overline{\text{H}}$ and ${}^4_{\Lambda}\overline{\text{He}}$ at $L_{\text{int}} = 10 \text{ nb}^{-1}$ is 100, 7 and 5, respectively. The collected sample will enable very precise measurements of the production of ${}^3_{\Lambda}\overline{\text{H}}$ and ${}^3_{\Lambda}\overline{\text{H}}$ and the first ever measurement of their elliptic flow as a function of p_{T} . The discovery of ${}^4_{\Lambda}\overline{\text{H}}$ and ${}^4_{\Lambda}\overline{\text{He}}$ will be in reach with $L_{\text{int}} = 10 \text{ nb}^{-1}$ at the end of Run 4.

3.2.3 The hypertriton lifetime

The experimental measurement of the Λ separation energy of the ${}^{3}_{\Lambda}$ H, $B_{\Lambda} = 0.13 \pm 0.05$ (stat.) ± 0.04 (syst.) MeV [64], led to the hypothesis that the lifetime of the hypertriton is equal to or only slightly below the free Λ lifetime $\tau(\Lambda) = 263.2 \pm 0.2$ ps [65]. Three different experimental techniques have been used to tackle this question: photographic emulsion, He bubble chambers, and counter experiments. The average for the emulsion experiments is 203^{+40}_{-31} ps [66], for the He bubble chambers is 193^{+15}_{-13} ps [66], and for the combination of both visualizing techniques is 193^{+14}_{-13} ps [66]. The most recent results, 181 $^{+54}_{-39}\pm 33$ ps and $142^{+24}_{-21}\pm 29$ ps, have been obtained with the counter technique in heavy-ion collisions by the ALICE [67] and STAR [68] experiments, respectively. This technique is currently the one with the highest precision (14-16%) and the weighted average of heavy-ion experiments results is 185^{+28}_{-23} ps [66]. However, the few existing theoretical calculations point in the direction of the hypothesis mentioned at the beginning of this section. The first theoretical determination of $\tau(^{3}_{\Lambda}$ H) (by Dalitz and Rayet, [69]) ranged from 239.3 to 255.5 ps. More recent calculations from Congleton [70] and Kamada [63] estimated values of 232 ps and 256 ps, respectively. The deviation of the experimental results from the theoretical calculations and the free Λ lifetime, by more than 2σ , is known as the "hypertriton lifetime puzzle".

With the expected integrated Pb–Pb luminosity at the end of the LHC Run 4, the statistical uncertainty on the lifetime will be reduced down to 1%. In parallel, a reduction of the systematic uncertainty ($\sim 10\%$ in the most recent ALICE measurements), will be achieved with the upgraded ALICE ITS that will allow for a reduction of the uncertainty due to tracking and material budget. To improve even further down the control on the systematic uncertainty, a better understanding of the corrections for the absorption in the material will be crucial.

3.2.4 Σ -hypernuclei

In addition to measurements of Λ -hypernuclei, also the search for Σ -hypernuclei is to be considered with the luminosities forseen for the LHC Runs 3 and 4. Theory calculations for the Σ NN system suggest the presence of a near-threshold narrow (~2 MeV wide) quasi-bound state in the I = 1 and J = 1/2

configuration, where the possible isospin and spin states are 0, 1, 2 and 1/2, 3/2, respectively [71]. Among Σ -hypernuclei, only the $\frac{4}{\Sigma^0}$ He bound state has been observed so far, using the ${}^4\text{He}(K^-, \pi^-)$ reaction [72]. When the Σ^0 hyperon is bound inside a nucleus, the electromagnetic decay is dominated by the conversion reaction $\Sigma^0 N \to \Lambda N$, thus the partial width of electromagnetic decay is expected to be reduced substantially. However, for the I = 2 state the conversion reaction is not allowed and the electromagnetic decay becomes prominent. Experimental searches for Σ -hypertriton bound states will also profit from the Pb–Pb data-taking programme of the LHC Runs 3 and 4 to exploit the strong decay $\frac{3}{\Sigma^0} \text{H} \left(\frac{3}{\Sigma^0} \overline{\text{H}}\right) \to \Lambda (\overline{\Lambda}) + d$ and the decay $\frac{3}{\Sigma^0} \text{H} \left(\frac{3}{\Sigma^0} \overline{\text{H}}\right) \to \frac{3}{\Lambda} \text{H} \left(\frac{3}{\Lambda} \overline{\text{H}}\right) + \gamma$ following a similar strategy to the detected electromagnetic decay of $\Sigma^0 (\overline{\Sigma}^0) \to \Lambda (\overline{\Lambda}) + \gamma$ [73]. The signal of hypertriton can be reconstructed in ALICE as discussed in Sec. 3.2.2. The soft photon can be identified by exploiting the conversion into electron pairs in the detector material of the ITS and Time Projection Chamber $(X/X_0 \approx 9 \%$ considering the upgraded ITS and the TPC together), covering the pseudorapidity range $|\eta| < 0.8$, over the full azimuth ($\Delta \varphi = 2\pi$) [74]. Alternatively, the photon can be detected in the PHOS calorimeter, but with limited acceptance of $\Delta \varphi = 100^\circ$ and $|\eta| < 0.12$ [74]. The search for Σ -hypernuclei via electromagnetic decay will be carried out in ALICE profiting from the expected detector performance and detection of about 10^5 hypertriton candidates (see Fig. 3) in 0–10% central Pb–Pb collisions at $L_{\text{int}} = 10 \text{ nb}^{-1}$.

3.2.5 Exotic QCD bound states

At LHC energies, potential QCD bound states that have more complex structures such as pentaquarks, tetraquarks, hadron molecules or dibaryon states could be produced. In particular, the possibility to detect and measure $f_0(980)$, N(1875), N \equiv , N Ω and N Λ_c in heavy-ion collisions with the unprecedented statistics of the LHC Runs 3 and 4 programme has been investigated. The advanced capabilities of the ALICE experiment in terms of hadron identification, including topological reconstruction of weak decays, are particularly suited for these studies.

The per-event yields of these states $(dN/dy)_{th}$) predicted by quark- and hadron-coalescence models [47] and the statistical-thermal model [32] are reported in Tab. 5.

The total number of signals (S_{raw}) detectable in ALICE with a minimum bias Pb–Pb integrated luminosity of 10 nb⁻¹ have been estimated assuming the same detector performance as in Run 2 [75,76]. The significance is defined as $\frac{S}{\sqrt{S+B}}$, where S and B are the integrals of the signal and background distributions, respectively, in a $\pm 3 \sigma$ window centered at the nominal mass from [65]. σ is $\Gamma/2.35$, where Γ is the resonance width taken from [65]. The significance for f₀(980) and N(1835) was extracted assuming a combinatorial background in the invariant mass range under study. Such combinatorics was computed based on particle species that can populate the invariant mass distribution, making use of the corresponding momentum distribution as measured in ALICE in Run 2 (e.g. individual primary charged pions paired as candidates for the f₀(980) $\rightarrow \pi^+ + \pi^-$ channel) and assuming a uniform distribution in φ and η^2 . The resulting significance is reported in the last row of Tab. 5 for the most pessimistic scenario, in which production occurs via quark coalescence, and the most optimistic scenario, corresponding to thermal production.

Measurements of $f_0(980)$ and N(1875) will be feasible in Runs 3 and 4 and will shed light on the highly-debated nature of the states (hadrons or hadronic molecules). In particular, the N(1875) can be considered a molecular bound state and at the same time the strange partner of the recently discovered pentaquark P_c [78]. Because the structure of exotic states is related to the fundamental properties of Quantum Chromodynamics (QCD), their observation can provide new insights on the properties of QCD at finite temperature and density, for instance that tetraquark condensation may lead to a second chiral phase transition [47]. Several possible states have been studied and predictions are available on the

²An additional factor is introduced if the decay particle is reconstructed via invariant mass, since the candidate may belong also to the background.

expected yields at LHC energies [47]. Among the possible dibaryon bound states, the N Ω , N Ξ and N Λ_c look promising in terms of detection feasibility. Their detection and baryon-baryon correlations will be useful for hyperon correlation studies, providing new insights into the baryon-baryon attractive potential as well as upper limits on the formation of such bound states in central heavy-ion collisions.

3.2.6 Implications of anti-nuclei measurements for cosmic-ray physics and dark-matter searches

The HL-LHC physics program with pp and p–Pb collisions will allow for precision measurements of anti-nuclei production and related observables that have implications for cosmic-ray physics and dark-matter searches. Cosmic-ray (CR) anti-nuclei \overline{p} , \overline{d} , and ${}^{3}\overline{\text{He}}$ have long been considered as probes of new physics, such as dark matter annihilation [79–90]. Detecting these particles is one of the main goals of various CR experiments (e.g. AMS-02 [91,92], GAPS [93,94], BESS-Polar [95]).

The galaxy produces CR anti-nuclei as secondaries, due to collisions of CR protons and helium with interstellar matter. Information from accelerator experiments is essential for the theoretical description of the background constituted by these secondary anti-nuclei. The flux of secondary anti-nuclei can be calculated with only minor sensitivity to the details of CR astrophysics. The point is to use secondary to-secondary flux ratios, where astrophysical uncertainties largely cancel. Secondary \overline{p} , \overline{d} , and ${}^{3}\overline{\text{He}}$ are all formed dominantly by the same set of reactions. Using this basic fact, an explicit prediction for the locally observable flux of secondary \overline{d} , relative to the flux of secondary \overline{p} can be derived [96–98]:

$$\frac{J_{\overline{d}}(\mathcal{R})}{J_{\overline{p}}(\mathcal{R})} = \frac{\int d\epsilon J_{p}(\epsilon) \frac{d\sigma_{pp\to\overline{d}}(\epsilon,\epsilon_{\overline{d}})}{d\epsilon_{\overline{d}}}}{\int d\epsilon J_{p}(\epsilon) \frac{d\sigma_{pp\to\overline{p}}(\epsilon,\epsilon_{\overline{p}})}{d\epsilon_{\overline{p}}} + \left(\sigma_{\overline{d}}(\epsilon_{\overline{d}}) - \sigma_{\overline{p}}(\epsilon_{\overline{p}})\right) J_{\overline{p}}(\mathcal{R})}.$$
(7)

Here $J_{\overline{d}}(\mathcal{R})$ is the predicted \overline{d} flux, given at magnetic rigidity $\mathcal{R} = p/Z$, where p is the momentum and Z is the electric charge. $J_{\overline{p}}(\mathcal{R})$ is the (already well-measured [99]) \overline{p} flux at the same rigidity, and $J_{p}(\epsilon)$

Table 5: Properties and yields of exotic states in 0–10% central Pb–Pb collisions at $\sqrt{s_{\rm NN}} = 5.5$ TeV. Theoretical predictions of yields per event, $(dN/dy)_{\rm th}$, are given in three different scenarios: quark- and hadron-coalescence [47], and thermal model [36]. S_{raw} represents the total detectable yield at the Pb–Pb luminosity of 10 nb⁻¹, taking into account the branching ratios (B.R.) of the decay channels considered and assuming the ALICE detector performance as in Run 2 [75,76]. For f₀(980), a K $\overline{\rm K}$ state and a decay into K $\overline{\rm K}$ with B.R. = 10⁻³ is assumed for hadron coalescence [†]. A tetraquark state is assumed for quark coalescence and a decay into $\pi\pi$. The same decay channel is assumed for the thermal production case. Masses are from [77].

	Model	f ₀ (980)	N(1875)	NΞ	NΩ	$\mathrm{N}\Lambda_c$
Structure		$qq\bar{q}\bar{q}$ or $K\overline{K}$	hadron molecule	dibaryon	dibaryon	dibaryon
	q-coal.	5.4×10^{-2}	-	-	1.8×10^{-3}	1.5×10^{-3}
$\left(\frac{\mathrm{d}N}{\mathrm{d}u}\right)_{\mathrm{th}}$	h-coal.	3.2 †	-	-	1.6×10^{-3}	$5 imes 10^{-3}$
	thermal	10	$3 imes 10^{-1}$	$8.7 imes 10^{-3}$	$5.7 imes 10^{-3}$	$4 imes 10^{-3}$
Decay channel		$\pi\pi$ / K $\overline{ m K}$	$\Sigma^* (\to \Lambda \pi) \mathbf{K}$	$\Xi \to \Lambda \pi$	$\Omega \to \Lambda K$	$\Lambda_c \rightarrow \pi Kp + \Lambda_c \rightarrow K^0_S p$
B.R. (%)		dominant / seen [†]	unknown (87)	99.9	67.8	6.2 + 1.58
Mass (MeV/ c^2)		990	1850 - 1920	-	-	-
Width (MeV/ c^2)		10 - 100	120 - 250	-	-	-
	q-coal.	$1.8 imes 10^8$	-	-	6.2×10^4	$1.5 imes 10^4$
S_{raw}	h-coal.	$6.4 imes10^{6}$ †	-	-	$5.5 imes 10^4$	$5.1 imes 10^4$
	thermal	$3.6 imes 10^{10}$	$5.5 imes 10^7$	$6.7 imes 10^5$	$1.9 imes 10^5$	$4.1 imes 10^4$
	q-coal.	130-3.5	-	-	-	-
$\frac{S}{\sqrt{S+B}}$	h-coal.	-	-	-	-	-
v S + B	thermal	2600-70	520-360	-	-	-

is the proton flux [100] at energy ϵ . $\frac{d\sigma_{pp\to\bar{x}}(\epsilon,\epsilon_{\bar{x}})}{d\epsilon_{\bar{x}}}$ and $\sigma_{\bar{x}}(\epsilon_{\bar{x}})$ are the inclusive production and inelastic cross sections, respectively, with x = d, p. The particle energy $\epsilon_{\bar{x}}$ for a nucleus with mass number A is evaluated at \mathcal{R} : $\epsilon_{\bar{p}} = \sqrt{\mathcal{R}^2 + A^2 m_p^2}$. To describe ³He we use an analogous expression to Eq. (7), adding the production of \bar{t} which decays to ³He. More details, including the relation of the differential cross section appearing in Eq. (7) to the Lorentz-invariant differential cross section measurable at the LHC, can be found in [98].

The cross section for producing an anti-nucleus can be parameterized in terms of the anti-proton cross section, using the coalescence factor B_A : $(\epsilon_A d\sigma/d^3 p)_{pp \to A} = B_A/\sigma_{pp}^{A-1}[(\epsilon_{\overline{p}} d\sigma/d^3 p)_{pp \to \overline{p}}]^A$, where σ_{pp} is the total inelastic pp cross section. Here, for simplicity, threshold effects are omitted [98, 101, 102]. Using Eq. (7), and plugging in the coalescence factors experimentally obtained at the LHC [103], the predicted flux ratios can be obtained. Secondary CR production is dominated by the low p_T region. As a result, the impact on the CR flux, due to p_T -dependent B_A , can be factored out to good approximation, allowing us to derive simple approximate formulae ³ [98]:

$$\frac{J_{\overline{d}}(\mathcal{R})}{J_{\overline{p}}(\mathcal{R})}|_{\mathcal{R}=100\text{GV}} \approx 4 \times 10^{-4} \left(\frac{B_2}{1.5 \times 10^{-2} \text{ GeV}^2}\right),\tag{8}$$

$$\frac{J_{^{3}\overline{\text{He}}}(\mathcal{R})}{J_{\overline{p}}(\mathcal{R})}|_{\mathcal{R}=100\text{GV}} \approx 2 \times 10^{-7} \left(\frac{B_{3}}{1.5 \times 10^{-4} \text{ GeV}^{4}}\right), \tag{9}$$

where, for CR studies, the B_2 and B_3 values should be read from the average yield in the range $p_T/A = (0 - 0.5) \text{ GeV}/c$ in the accelerator analysis. The precision requirements (O(10%)) on the flux ratio determination for the astrophysical applications discussed here will be matched by measuring B_2 and B_3 in the lowest p_T bin with a relative systematic uncertainty of the order of 10% [103]. The latter largely dominates the statistical uncertainty that is expected to be of O(0.1%) already with $L_{int} = 6 \text{ pb}^{-1}$ in pp collisions at $\sqrt{s} = 5.5$ TeV. Moreover, a first measurement of B_4 in pp collisions will be achievable in the same sample. The statistical precision on B_4 can be lowered to the 10% level if a luminosity of 200 pb⁻¹ in pp collisions at $\sqrt{s} = 14$ TeV can be inspected with a dedicated trigger for (anti-)nuclei.

It is important to note that the B_A measurement [103] performed by ALICE during the LHC Run 1 was confined to mid-rapidity, |y| < 0.5. Possible y dependence of the coalescence factor B_A at y = O(1), or variation of the \overline{p} differential cross section with respect to current parameterisations [104], would affect the prediction in Eqs. (8-9). It would be an important task of future LHC measurements to test these effects. Similarly important, albeit – possibly – beyond the reach of the LHC, would be to study the low $\sqrt{s} = O(10)$ GeV behaviour of B_A [98].

3.2.7 Implications of anti-nuclei measurements and hyperon-nucleon correlations for neutron star physics

Although the neutron star crust is composed of neutrons, within the innermost core hyperons could be present [105]. Whether or not this scenario holds true depends on the two- and three-body hyperon nucleon interactions (YN and YNN) that are still only rather scarcely constrained experimentally.

At present the mass range for observed neutron stars is about $(0.9 - 3.0)M_{\odot}$ within errors [106], where M_{\odot} stands for the solar mass. The equation of state (EoS) of neutron stars is constrained by the massradius relationship, in particular, the maximum mass (M_{max}) . An EoS with "conventional" $(N+\pi)$ degrees of freedom provides M_{max} invariably above $2M_{\odot}$ [107–109]. However, adding the Λ hyperon in the hadronic basis softens the EoS and, as a consequence, significantly reduces M_{max} . The solution of this so-called "hyperon puzzle" is non-trivial, and is presently the subject of very active research.

Thanks to the large yields of free hyperons and exotic (anti-)hyper-nuclei that can be produced in collider experiments and the excellent particle identification capabilities of the ALICE experiment, the

³Note that the rigidity $\mathcal{R} = 100$ GV refers to the CR experiment rest frame, which is boosted w.r.t. the proton-proton collision centre of mass frame. In the proton-proton collision centre of mass frame, the anti-nuclei are formed close to threshold.



Fig. 4: Expected $p\Xi^- + \bar{p}\Xi^+$ correlation for pp collisions at $\sqrt{s} = 5.5$ TeV and 4×10^{11} minimum bias events, corresponding to $L_{\text{int}} = 6 \text{ pb}^{-1}$. Only statistical errors have been considered.

upcoming experimental campaigns in Runs 3 and 4 at the LHC offer a unique opportunity to quantitatively characterise hypermatter under controlled (laboratory) conditions and infer on the equation of state of compact objects as neutron stars.

One of the crucial element to solve the "hyperon puzzle" is the determination of the Λ NN threebody forces. Calculations show that with a parameterization of these forces compatible with the hypernuclear binding energies, the admixture of Λ 's in neutron star matter gets strongly reduced such that the pressure to support a $2M_{\odot}$ neutron star can be maintained [110]. The observation of neutron-rich Λ hyper-nuclei like $^{4}_{\Lambda}$ H etc. at colliders could be very promising for studying the effects of the three-body Λ NN forces in dense strongly interacting matter since a precise knowledge of light neutron-rich hypernuclei energy level structure could imply far-reaching consequences for dense strange stellar matter properties.

Another promising way to contribute to the understanding of the hyperon puzzle is to pin down the hyperon-nucleon two-body interaction for hyperons such as Σ^- and Ξ^- . These hyperons can also be produced within neutron rich matter $(n + n \rightarrow \Sigma^- + p, \Lambda + n \rightarrow \Xi^- + p)$ depending on their interaction with the surrounding neutrons. Some models assume a repulsive $p\Sigma$ interaction and postulate that $\Sigma^$ can appear in neutron rich matter only starting from baryon densities $\rho \simeq 4\rho_0$ (where ρ_0 is the density of standard nuclear matter) [111]. For Ξ , no reliable experimental information about the interaction is available. Recent studies [112] showed that the femtoscopy technique applied to pp and p–Pb collisions at LHC energies are particularly suited to study the final state interaction between nucleons and strange baryons (e.g.: Λ -p) and between two strange baryons (e.g.: Λ - Λ). Indeed, small colliding systems such as pp and p–Pb lead to hadron sources of rather small dimensions, of the order of 1 fm, in the range where the strong interaction is mostly evident. Also, the production mechanism of hadrons in minimum bias pp and p–Pb collisions at LHC energies. This allows for a more precise study of the hadronhadron interactions.

Among the quantitative results obtained by ALICE in Run 2 at the LHC is the first observation of the attractive $p\Xi^-$ interaction. Figure 4 shows the expected $p\Xi$ correlation for the Run 3 pp sample as a function of the relative momentum k*. The projection is obtained on the basis of the current prediction by the HAL-QCD lQCD group [113, 114] that is in agreement with the Run 2 results. The clear deviation from the Coulomb-only correlation function shows the effect of the strong attractive interaction and the expected statistics will allow for a quantitative determination of the scattering parameters and the test of

different hadronic models [115, 116]. The investigation will also be extended to the Σ^0 hyperon, since in Run 3 and 4 we expect a total of 500,000 p Σ^0 pairs to be used to study the femtoscopy correlation.

In summary, massive neutron stars with $M \sim 2M_{\odot}$ are very intriguing recent observations in relativistic astrophysics. An improved account of the two-body YN interaction, the three-body Λ NN forces, and the contribution of multi-strange hyperons in the EoS is crucially important for more realistic description of compact astrophysical objects, in particular neutron and hybrid stars. The measurement of hyper-nuclei and hyperon correlations with the HL-LHC project are suggested as a promising tool for astrophysical applications.

3.3 Fluctuations of conserved charges

3.3.1 Physics introduction and observables

In the phase diagram of strongly interacting matter at zero net baryon density, the presence of a chiral phase transition between hadronic matter and a QGP has been conjectured [117], and arguments have been presented [118,119] in lQCD that the transition, for vanishing light quark masses, is of second order and belongs to the O(4) universality class. Due to the small but finite physical quark masses, in lQCD a rapid crossover is found [120–124] which, however, exhibits pseudo-critical features due to the smallness of the u- and d-quark masses and the proximity of the crossover region to the O(4) line [118, 125].

In general, fluctuations can be linked to critical behaviour associated with a phase transition, and it has been pointed out that fluctuations of conserved charges in heavy-ion collisions can provide an experimental observable to test for critical behaviour in the phase diagram of strongly interacting matter [126–129]. These fluctuations can be related to susceptibilities, specifically to the derivatives of the pressure with respect to the chemical potentials corresponding to the conserved charges. Here, the relevant 'charges' are baryon number B, strangeness S, and electrical charge Q, and the corresponding chemical potentials are μ_B , μ_S , and μ_Q . The susceptibilities are defined (see e.g. [129, 130]) in terms of dimensionless normalized chemical potentials $\hat{\mu}_X \equiv \mu_X/T$ as

$$\chi_{ijk}^{BQS}(T) = \left. \frac{\partial P(T,\hat{\mu})/T^4}{\partial \hat{\mu}_B^i \partial \hat{\mu}_Q^j \partial \hat{\mu}_S^k} \right|_{\hat{\mu}=0} \,. \tag{10}$$

The generalized susceptibilities can be computed in IQCD at vanishing chemical potential, exactly the conditions probed by experiments at the LHC. Within the Grand Canonical Ensemble (GCE), these generalized susceptibilities can be related to experimental measurements of the fluctuations of particle multiplicities, such as the net number of baryons. For instance, a measurement of higher moments or cumulants of net baryon number in relativistic nuclear collisions can be directly related [131–136] to theoretical predictions from IQCD or from more phenomenological models of the chiral phase transition [137,138] to shed light on the possible critical behaviour near the QCD phase boundary. For a distribution of the net baryon number, $\Delta N_B = N_B - N_{\overline{B}}$, with moments defined as

$$\mu_i = \langle (\Delta N_B - \langle \Delta N_B \rangle)^i \rangle, \tag{11}$$

the cumulants κ_i can be directly linked to the generalized susceptibilities such as

$$\kappa_2 = \mu_2 = V T^3 \chi_2^B \tag{12}$$

$$\kappa_3 = \mu_3 = VT^3 \chi_3^B \tag{13}$$

$$\kappa_4 = \mu_4 - 3\mu_2^2 = VT^3\chi_4^B.$$
 (14)



Fig. 5: Ratio of sixth to second-order baryon number susceptibilities from IQCD. The left-hand figure is from [127]. The right-hand figure is calculated from recent IQCD data on sixth and second order susceptibilities from [139].

In the O(4) universality class, a singular contribution to the pressure shows up for higher order moments. More specifically, at vanishing chemical potential, all odd susceptibilities of the net baryon number vanish. In addition, in the O(4) universality class, the second- and fourth-order susceptibilities remain finite at the phase transition temperature at $\mu_B = 0$ in the chiral limit, implying that only sixthand higher-order susceptibilities diverge [126, 128]. Thus, for physical quark masses and at $\mu_B = 0$, only higher order cumulants κ_n with $n \ge 6$ can exhibit O(4) criticality, whereas at finite μ_B this is already the case for κ_n with $n \ge 3$.

Sensitivity to chiral criticality due to the vicinity of the O(4) line at $\mu_B = 0$ is borne out in phenomenological models as is shown in [128, 137], and in IQCD predictions [127, 139], by strong deviations of χ_6^B/χ_2^B from unity as shown in Fig. 5.

We note that a convenient baseline for the cumulants of multiplicity distributions and fluctuations of produced particles in relativistic nuclear collisions can be obtained in the framework of the hadron resonance gas (HRG) [131, 139–142]. In this model, uncorrelated Poissonian fluctuations of baryon and anti-baryon multiplicities result in a Skellam distribution for the net baryon number, in which the higher moments and cumulants can all be related to the first moments in the following way [141, 143, 144]:

$$\kappa_n (N_B - N_{\overline{B}}) = \langle (N_B + (-1)^n N_{\overline{B}}) \rangle \tag{15}$$

For zero net baryon number then all odd cumulants vanish and all even cumulants are identical.

Measuring such cumulants with precision poses a formidable experimental challenge due to the requirement of very large data sets (> 10^9 events of a particular event or centrality class) with superb control of systematic uncertainties. As a first physics case to consider along this line, the impact of measuring the distribution of net protons as a proxy for net baryons needs to be studied further. We note that, at LHC energy and low transverse momentum, particle production near mid-rapidity takes place mostly in gluonic processes, implying that isospin asymmetries, as in the colliding nuclei, are absent. As a consequence, the production yields of protons and neutrons should be very close. For light nuclei this isospin symmetry has been checked experimentally, albeit with significant uncertainties. In addition, non-critical contributions to the cumulants from volume fluctuations and global baryon number conservation [132, 144, 145] need to be evaluated and the data corrected accordingly. Furthermore, in particular for comparison to lQCD predictions, care needs to be taken to keep experimental cuts such as in $p_{\rm T}$ to a minimum insofar as such cuts cannot be introduced in lQCD [146, 147].

Two-particle correlations with net baryons can also be used to explore transport properties of the hydrodynamic evolution. The baryon diffusion constant D is a fundamental transport property of the

quark-gluon plasma, similar to shear viscosity η or bulk viscosity ζ . It characterizes the mobility of baryon number, and is predicted to be finite at the LHC despite the fact that $\mu_B \sim 0$. A two-particle correlation function as been proposed [148], which explores correlations of net-baryon fluctuations as a function of separations in azimuthal angle and rapidity, and can provide experimental constraints on the diffusion coefficient D. As $\mu_B \sim 0$ at the LHC, such an analysis has yet to be carried out in Run 1 and 2 data since it is statistically challenging, and will be greatly aided by the increase by about a factor 100 in the Pb–Pb integrated luminosity foreseen for Runs 3 and 4.

3.3.2 State of the art experimental measurements and present limitations

Net proton fluctuations measured by the ALICE experiment and in the STAR beam energy scan program provide interesting and stimulating results. The measurements at STAR [149] complement the corresponding measurements from ALICE, which will make it possible to pin down the global structure of the phase diagram of strongly interacting matter in a wide range of temperatures and net-baryon densities. However, before drawing firm conclusions by confronting theoretical calculations with data, non-dynamical contributions stemming from unavoidable fluctuations of participant nucleons and overall baryon number conservation have to be subtracted from the experimental measurements. Both of these non-dynamical contributions, which exist neither in IQCD nor in the HRG model, lead to deviations from the baseline as defined in Eq. 15. Indeed, the acceptance dependence of the second-order cumulants of net-protons measured by ALICE [150] exhibits deviations from the non-critical (Skellam) baseline. However, these deviations were explained by global baryon number conservation [144, 145, 150], which, in accordance with the experimental findings, decreases the amount of fluctuations with the increasing acceptance. This is the first experimental verification of the IQCD predictions for the second-order cumulants of net-baryon distributions. This also serves as a strong support of the HRG model, in that experimental measurements of the second cumulants of net-protons do not show any evidence of criticality and actually coincide with the second cumulants of the Skellam distribution. In order to probe critical phenomena, higher cumulants beyond the second order have to be addressed.

As mentioned in the previous section, even at vanishing net-baryon densities, IQCD and other theoretical calculations such as Polyakov-loop extended Quark- Meson model (PQM) [137] predict critical fluctuations encoded in the deviations of net-baryon κ_4/κ_2 and κ_6/κ_2 from unity. Moreover, at the pseudo critical temperature of about 156 MeV the magnitudes of κ_4/κ_2 and κ_6/κ_2 are predicted in Ref. [137] to be 0.5 and -0.39, respectively. Similar values of κ_6/κ_2 are quoted in different IQCD calculations as presented in Fig 5. These numbers, shown in Fig. 6, do not take into account experimental artefacts such as global net-baryon number conservation and unavoidable fluctuations of participating nucleons from event to event. Also shown are the values of of κ_4/κ_2 and κ_6/κ_2 after accounting for these non-dynamical effects using the procedure in Refs. [144, 145]. Even after accounting for participant fluctuations and global baryon number conservation we observe deviations in κ_4/κ_2 and κ_6/κ_2 from unity, although they are somewhat reduced. This motivates our experimental program of measuring higher moments of net-proton distributions at the LHC energies. Also, fluctuation measurements are underway in the strange baryon sector to approach measurements of net baryon number fluctuations. All this will be greatly helped by the anticipated dramatic increase in statistics in Runs 3 and 4.

3.3.3 Projections for HL-LHC

As discussed above, precise studies of the higher-order cumulants of particle multiplicity distributions are needed to verify theoretical predictions. In this section we estimate the statistics needed to address these measurements with the ALICE experiment. For this purpose two distinct Monte Carlo simulations have been developed. In the first approach, following recent developments in [151], the probability distribution function of net-protons is approximated by a superposition of two Gaussian distributions which has four free parameters. Using the experimentally measured second cumulant of net-protons for 0-5 % most central Pb–Pb collisions [150] and the κ_4/κ_2 and κ_6/κ_2 ratios from the PQM model [137],



Fig. 6: κ_4/κ_2 and κ_6/κ_2 as calculated within PQM [137] model (open symbols). After taking into account contributions from participant nucleon fluctuations and global baryon number conservation [144, 145], the deviations from unity decrease (closed symbols).

absolute values for κ_4 and κ_6 were obtained first. These values were adjusted to account for fluctuations from participant nucleons in 0-5% most central Pb–Pb collisions and global baryon number conservation [144, 145]. Finally, the free parameters of the double Gaussian distribution were fixed using the expected values of κ_1 , κ_2 , κ_4 and κ_6 , where κ_1 equals zero by definition. The event-by-event net-proton number was sampled from the double Gaussian function thus generating the net-proton distribution for a given number of events. In the second approach the probability distribution functions of protons and anti-protons are calculated separately by exploiting the Pearson curve method [152]. This approach also needs four measurements as inputs, which are taken as the first four cumulants of the proton and antiproton multiplicities measured by ALICE [153]. The net-proton distribution for a given number of events is constructed by sampling the obtained proton and anti-proton probability distribution functions. In each approach, the resulting statistical uncertainties are obtained using the subsample method.

The obtained results for κ_4/κ_2 and κ_6/κ_2 and their corresponding statistical uncertainties are shown in Fig. 7 as a function of the simulated event statistics. The dashed red lines correspond to the input values predicted by PQM calculations of critical fluctuations (CF) and assuming a double Gaussian net-proton distribution, while the green dashed lines come from the Pearson curve method based on the lower-order cumulants measured by ALICE. As expected, with increasing statistics both κ_4/κ_2 and κ_6/κ_2 approach their nominal values. The statistics necessary to measure these cumulants are presented in the bottom panels of Fig. 7, where the deviations of the expected values from unity are quantified in units of the magnitudes of the statistical uncertainty (σ). As seen from the left panel of Fig. 7, for κ_4/κ_2 already 10 million events are sufficient to distinguish the expected critical fluctuations signal from unity with a statistical significance of 4σ . Similar conclusions are obtained with the Pearson curve method. Several times this amount of data has already been recorded by ALICE, and the expected statistics in Runs 3 and 4 will make it possible to measure κ_4/κ_2 with unprecedented precision.

For κ_6/κ_2 , however, significantly larger event sample is needed. As seen from the right panel of Fig. 7, more than 5 billion 0-5 % central events generated with the double Gaussian approach are needed in order to observe statistically significant deviations from unity in favor of the critical values indicated

with the red dashed line. These would correspond to a minimum bias Pb–Pb integrated luminosity of 12.5 nb⁻¹ in Runs 3 and 4. Results obtained with the Pearson curve method indicate that more than 200 million 0-5 % central events (corresponding to a minimum bias Pb–Pb integrated luminosity of 0.5 nb⁻¹) would be sufficient in order to claim a significant deviation from unity in favour of the corresponding expected value. This difference in the estimation of the required statistics for κ_6/κ_2 comes mainly from the different baseline values of -1.43 and -0.27 used in the Pearson and double-Gaussian methods, respectively. In addition, the value of κ_2 used in the Pearson method is about two times smaller than measured in the experiment and used in the double Gaussian method. Track reconstruction and particle identification efficiency in the fiducial acceptance in η and $p_{\rm T}$ efficiencies, which would increase the required number of events for a given statistical precision, are not included in the study presented here because they depend on the details of the analysis. Considering that these efficiencies are expected to range from 60% to 80%, we conclude that the Pb–Pb integrated luminosity of 13 nb⁻¹ foreseen in Runs 3 and 4 (see Ch. 12) will be sufficient to probe the critical phenomena contained in κ_6/κ_2 .



Fig. 7: Simulated values of κ_4/κ_2 (left panel) and κ_6/κ_2 (right panel) as functions of the generated number of events. Full symbols represent results obtained with the double Gaussian approach adjusted to reproduce critical fluctuations (CF) predicted in the PQM model [137]. Open symbols are obtained with the Pearson Curve Method [152].

4 Flow and Correlations

Coordinators: Soumya Mohapatra (Nevis Labs, Columbia University)

Contributors: A.F. Dobrin (CERN), S. Floerchinger (Heidelberg University), P. Huo (Stony Brook University), W. Li (Rice University), J. Noronha-Hostler (Rutgers University), V. Okorokov (National Research Nuclear University MEPhI), B. Schenke (Brookhaven National Laboratory), M. Sievert (Rutgers University), A. Trzupek (IFJ PAN, Kraków), U.A. Wiedemann (CERN)

4.1 Introduction

It is particularly interesting to study the macroscopic properties of the QGP fluid because - at least conceptually - they are fully fixed by the microscopic properties of a renormalizable, fundamental quantum field theory, namely QCD. One key question is how macroscopic properties of QGP arise from its microscopic interactions. Many theoretical methods ranging from perturbative to non-perturbative techniques are being developed to understand this in detail and one can expect that the insights gained here will be valuable for many related problems in fields ranging from condensed matter theory to cosmology in the future. Many different fronts of research are being explored at the moment. This ranges from conceptual questions on how to consistently formulate relativistic fluid dynamics or how to solve quantum field theory in non-equilibrium situations to very concrete practical questions about the thermodynamic and transport properties (such as viscosities or conductivities) of the QGP. The description of the initial state - prior to OGP formation - and the mechanism by which the products of the collision rapidly thermalize are also under investigation. Besides the role of strong interactions, also electromagnetic interactions and in particular the role of magnetic fields are being explored. Other fronts of research concern the role of quantum anomalies, chirality and vorticity or the dependence of collective behavior on system size (nucleus-nucleus versus proton-nucleus and proton-proton collisions), on centrality and collision energy, the initial state directly after the collision, or various types of fluctuations. These challenges are discussed in more detail in this section.

4.2 Review of current status of theory on bulk and flow observables

4.2.1 QCD Equation of State

The QCD equation of state, accessible in high-energy collisions (and in the region around mid-rapidity) is one that has vanishing baryon chemical potential. It has been established for some time that it features a crossover transition to a chirally symmetric quark gluon plasma [120]. Most recent lattice calculations [154] have determined the cross-over temperature to be $T_c \simeq 156.5 \pm 1.5 \,\mathrm{MeV}$. Recent efforts are also exploring the equation of state at finite μ_B , which at LHC would have relevance mainly at very forward rapidities. Here, because of the fermion sign problem in lattice QCD, methods like Taylor expansion [155–158] or imaginary chemical potentials [159–163] have to be used. To employ lattice QCD based equations of state in hydrodynamic calculations, they need to be matched to a hadron resonance gas model at low temperatures. Various equations of state [37, 164, 165], using different lattice data and different matching conditions have been used in simulations. A comparison of some of them can be found in [166], where the sensitivity of observables to the choice of equation of state was studied. For a recent theoretical proposal on how to gain experimental sensitivity on the equation of state, see ref. [167]. Currently available data are consistent with the lattice QCD equation of state, however with an $\sim 50\,\%$ error. To reduce the uncertainty, measurements of particle spectra over a wide range of colliding energies using a single detector with good particle identification, especially at low transverse momentum, would be needed. Another possibility to constrain the equation of state from experimental data would be to extend state of the art Bayesian techniques [168] to include free parameters describing the equation of state and fit them along with other free parameters such as shear and bulk viscosities.
4.2.2 Shear and bulk viscosities of hot nuclear matter

Ideal fluid dynamics has been very successful in describing a variety of bulk observables in heavy ion collisions [169–171], indicating early on that the shear and bulk viscosities of the produced matter cannot be large. Calculations in the strong coupling limit using gauge gravity duality have found a value of $\eta/s = 1/4\pi$ for an N = 4 Supersymmetric Yang-Mills quantum system [172, 173]. This value was significantly smaller than the η/s obtained in perturbative QCD calculations, which were however, beset by significant uncertainties, mainly resulting from uncertainties in the relevant scales [174]. Recently, such perturbative calculations have been extended to include next-to-leading order corrections and a significant reduction compared to the leading order result was found [175]: At temperatures of the order of the QCD transition the η/s obtained using NLO corrections is smaller by a factor of 5 compared to the LO result, and reaches values of approximately $2/4\pi$. Extractions of transport coefficients from lattice QCD calculations [176–178] are extraordinarily hard, because a numerically difficult analytic continuation from imaginary to real times is necessary.

There are also several theoretical indications that bulk viscosity could play an important role in the QGP to hadron gas transition region (see [179] and references therein). Perturbative calculations have shown that the simple estimate $\zeta \approx 15\eta(1/3 - c_s^2)^2$ [180] is parametrically correct for QCD [181], where $(1/3 - c_s^2)$ is the deviation from conformal symmetry. Lattice calculations using the Kubo formula yield large values of ζ/s (~ 1) around T_c [182, 183] with large uncertainties [184]. The calculations also show a fast drop of ζ/s with increasing T. Parametrizations of the bulk viscosity over entropy density's temperature dependence were performed in [185] with input from [182] for the QGP phase and [186] for the hadronic phase. Similar to the case of shear viscosity, bulk viscosity over entropy density ratios have been determined also in holographic models that are supplemented with non-conformal features [187, 188]. In this approach, the ζ/s reaches a much lower peak value ~ 0.05 at temperatures slightly above T_c [188]. It remains to be understood to what extent this provides semi-quantitative guidance for the value in QCD.

Apart from theoretical calculations of the shear and bulk viscous properties, one may also constrain them by means of fluid dynamic simulations and comparison to experimental data [189, 190]. This method suffers mainly from uncertainties in the initial state (see also Section 4.2.3) and has an uncertainty of approximately a factor of 3 (for η/s) at this point. Some of the latest constraints come from simulations using the IP-Glasma initial state [191, 192], the EKRT model [193] and Bayesian analyses employing the Trento initial state model [168]. In terms of the Viscous corrections to the distribution function at freeze-out, the low- $p_{\rm T}$ range of the spectrum is more sensitive to the bulk viscosity than to the shear viscosity [194, 195]. Consequently, the uncertainties resulting from bulk viscous corrections are typically larger than for shear when studying $p_{\rm T}$ integrated observables. Precise measurements of the low- $p_{\rm T}$ spectra and mean- $p_{\rm T}$ in different collision systems will help in lowering the current uncertainties, specifically in the extraction of ζ/s . In order to disentangle features of the initial state and medium properties, it might be useful to study additional collision systems such as Ar–Ar or O–O and to perform a more detailed global analysis including refined data on harmonic flow coefficients for identified particles that become available during Run 3 and 4 (see Section 4.3.1).

4.2.3 Initial conditions

Modelling the exact geometry and initial conditions for a fluid dynamic description of heavy ion collisions is not a simple task, because it involves non-perturbative physics. The available descriptions for the initial state thus range from very simplistic models that assign deposited energy densities based on the wounded nucleons or binary collisions determined in a Monte-Carlo Glauber prescription, to classical effective theories of QCD that are valid in the high energy limit. The major ingredient that needs to be provided by an initial state model is the geometry of the interaction region in the plane transverse to the beam. It is entirely dominated by the positions of incoming nucleons whose fluctuations also play an important role. Initial conditions for hydrodynamic simulations have to provide, in principle, all components of the energy momentum tensor as a function of spatial position (and initial conditions for other conserved charges, if considered). This includes, apart from the always included energy density distribution, the initial flow as well as initial viscous corrections. Initial flow is included in many recently developed models, that either assume free streaming [168], including Yang-Mills evolution, which is close to free streaming [196] or an initial flow distribution motivated by strong coupling calculations [197]. Initial viscous corrections are often set to zero. Only a few works [168, 198, 199] include the full viscous stress tensor provided by the initial state description.

Since the initial state models that provide the entire $T^{\mu\nu}$ all switch from essentially a freely streaming system to strongly interacting hydrodynamics at a fixed time τ , that transition is somewhat abrupt and unphysical. To improve over this situation an intermediate step using effective kinetic theory has been introduced [200, 201]. This procedure allows for a somewhat smoother matching but has yet to be used in full fledged hydrodynamic simulations. A first study that matches full kinetic theory to full viscous fluid dynamics indicates that the intermediate kinetic transport formulation becomes more important with decreasing system size [202].

As already discussed in Section 4.2.2, the choice of initial state has a significant effect on the extraction of transport coefficients. A more compact initial state and the presence of initial flow lead to a larger transverse flow, which requires a larger bulk viscosity to compensate for it and produce agreement with experimental data [199]. In addition the initial flow also affects the value of the extracted shear viscosity. Also, the models' eccentricities will affect the final momentum anisotropies, influencing the extracted shear viscosity to entropy density ratio. Two possible attempts to solve this problem have been pursued: the first is to perform a combined Bayesian analysis of all parameters [168], including those of the initial state, to find the best fit for all transport coefficients along with the initial state description. The second is to constrain an initial state description as well as possible using data from experiments other than heavy ion collisions e.g. e^{-p} scattering data, which will hopefully be extended to e^{-A} in a future electron ion collider facility. As mentioned above, at the moment the two approaches lead to some similar features of the initial state (product of thickness functions, presence of subnucleon structure), but also show discrepancies (size of the nucleon and sub-nucleon scales along with the size of the extracted bulk viscosity). In the near future, flow measurements in light ion collisions such as Ar-Ar, O-O etc. can also provide independent experimental constraints on initial conditions (see Section 4.3.2). Similarly, new flow observables can also help constrain the initial conditions. In particular the measurement of flow fluctuations has provided constraints on initial geometry models, in both the approaches discussed above. Analogously, the more recent studies of longitudinal flow fluctuations [203-205] and their extensions in Run 3 and 4 (see Section 4.3.3) can provide additional constraints.

4.2.4 Response functions

In a fluid dynamic description of heavy ion collisions, one can understand the azimuthal harmonic flow coefficients v_n as a response to deviations of the initial state from an azimuthally isotropic form. Mathematically, one can formulate this in terms of response functions that describe how the solution of the fluid dynamic evolution equations, as well as resulting experimental observables such as azimuthal particle distributions, get modified when the initial values of the fluid fields are changed [206–208]. In the simplest implementation, linear response functions describe the linear response of flow coefficients to eccentricities $v_n \sim \epsilon_n$, while the quadratic response functions describe terms like $v_n \sim \epsilon_a \epsilon_b$ where symmetry reasons imply $|n| = |a \pm b|$ [209,210]. Response functions can not only be used to study deviations from azimuthal rotation symmetry but also for deviations from (approximate) Bjorken boost invariance, vanishing baryon number density, for electric fields and so on. Quite generally, response functions carry interesting information about fluid properties such as thermodynamic and transport properties. Where the response functions are known, one can infer properties of the initial state by reverse engineering. Experimentally, one can constrain properties of response functions indirectly via measurements of various

particle correlation functions. It is particularly interesting to compare situations with strong deviations from a symmetry (such as peripheral collisions for the case of azimuthal rotation invariance) to situations with small deviations (such as central collisions) in order to differentiate between linear and non-linear response. For existing experimental work in this direction see [211–213] and for an example of a recent further going theoretical proposal see [214].

Detailed comparison of flow observables between experiment and theoretical calculations, especially regarding the dependence on external parameters like system size and collision energy as well as differential information such as on centrality, or particle identification will be helpful to make further progress in constraining response functions. Improvements in particle identification, transverse momentum and longitudinal coverage in Run 3 and 4 will be particularly valuable to this end.

4.3 Experimental constraints from Run 3 and 4

Since measurements of flow and correlations provide arguably the most direct manifestations of collectivity, they play naturally a central role in the scientific programme of exploring finite temperature QCD via collectivity. At the HL-LHC, much more stringent tests of the collective dynamics in nucleus– nucleus collisions will be possible. These will constrain QGP medium properties and initial conditions, as discussed in the previous section, in great detail. In the following, these newly arising opportunities are illustrated with a set of physics performance studies exploiting: 1) high-statistics particle-identified flow measurements, 2) system-size dependence of flow, and 3) longitudinal flow fluctuations.

4.3.1 Identified particle v_n



Fig. 8: ALICE projections for v_2 (left) and v_3 (right) of π^{\pm} , $p+\overline{p}$, $\Xi+\overline{\Xi}$, $\Omega+\overline{\Omega}$, and the ϕ -meson in the 10–20% centrality interval for an integrated luminosity of 10 nb⁻¹. Error bars (shaded boxes) represent the projected statistical (systematic) uncertainties.

In a fluid dynamic picture of A–A collisions, hadrons decouple from the fluid at late times, when the density of the system is sufficiently low and the mean-free path is sufficiently large so that a fluid dynamic description ceases to apply. Therefore, it is a direct consequence of this late time-scale, that the PID-dependence of v_n measurements tests a limitation of fluid dynamic behavior. Do all hadron species emerge from the same common flow field? Or can one see at the higher accuracy of future v_n measurements particle specific differences in the decoupling which are related to the differences in hadronic cross sections? Quantitative questions of this kind will allow one to better constrain how significant the hadronic stage of A–A collisions contribute to v_n , and since the viscous properties of QCD change significantly in the hadronic stage, this is of direct relevance for extracting information about viscous transport coefficients with higher precision. The same class of improved PID-sensitive v_n measurements is also important for testing different dynamical pictures of hadronization. In particular since fragmentation is expected to be the dominant hadronization mechanism at high $p_{\rm T}$ while recombination becomes relevant at lower $p_{\rm T}$, extending these PID measurements with precision over the largest possible transverse momentum range will be of interest. To this end, figure 8 shows projections from the ALICE collaboration for the v_2 and v_3 of several light-flavor species, that are expected for an integrated luminosity of 10 nb⁻¹ expected in Run 3 and 4. The projected statistical uncertainties are typically negligible over the entire $p_{\rm T}$ range and in most cases the systematic uncertainties are quite small as well. These measurements will be much more precise compared to those in Run 1 and 2, and will allow for the recombination/fragmentation descriptions to be examined with unprecedented accuracy. Similar projections for heavy-flavor particles are discussed further in Chapter 5 together with their physics implications.

4.3.2 System size dependence



Fig. 9: Left panels: comparisons of the centrality dependence of the v_n measured in Pb–Pb collisions at $\sqrt{s_{NN}}$ =5.02 TeV to Xe–Xe measurements. The plots are for the 0.5–5 GeV/*c* p_T interval. From top to bottom each row corresponds to a different harmonic order *n*. The ratios are compared to theoretical predictions from Ref. [215]. Also shown are theory calculations of the ratios for Ar–Ar and O–O. ATLAS Data taken from Ref. [216].

In Pb–Pb collisions, previous v_n measurements at the LHC have accessed the system size dependence of flow via the centrality dependence. However, this centrality dependence is biased by a strong variation of the spatial eccentricity of the nuclear overlap. While modelling allows one to control this eccentricity dependence to some extent, studying smaller collision systems at highest centrality (i.e. impact parameter close to zero) remains conceptually the cleanest way of establishing the system size dependence of flow. It also provides a way of disentangling the event-averaged spatial eccentricity from the event-by-event eccentricity fluctuations, and can thus contribute significantly to constraining the initial condition from which collective behaviour emerges. These are important motivations for studying soft multi-particle production and its transverse asymmetries in the collision of lighter nuclei.

Figure 9 shows ATLAS comparisons of the v_n in Xe–Xe and Pb–Pb collisions as a function of centrality (left panels) and their ratios (right panels). Also shown for comparison in the right panels are theoretical predictions for the ratios from Ref. [215]. It is seen that in most central collisions the ratio v_n (Xe–Xe)/ v_n (Pb–Pb) is larger than unity for n = 2 and 3. With decreasing centrality the ratios for all harmonics show a decreasing trend. These trends can be explained as follows: Xe-Xe being a smaller system than Pb-Pb, the effect of fluctuations is more important. The fluctuations increase the initial eccentricities of the collision geometry and therefore enhance the $v_{\rm n}$. However, because Xe–Xe is a smaller collision system the viscous effects (which suppress the v_n) are larger, and play a bigger role with decreasing centrality and increasing harmonic order. In most central events, the effect of the increased fluctuations wins for v_2 . But with increasing harmonic order and/or decreasing centrality, eventually the viscous effects reduce the v_n compared to Pb-Pb. These observations indicate the ability of such cross-system v_n measurements to be very sensitive to initial conditions of the heavy ion collision as well as the transport coefficients of the QGP. The measured ratios for the v_n (Xe–Xe)/ v_n (Pb–Pb) are qualitatively reproduced by the theory predictions from Ref. [215]. In order to illustrate the sensitivity of such models to the collision species, predictions are also presented for Ar–Ar and O–O collisions. The predictions show considerably larger variation of the centrality dependence of the v_n going from Xe–Xe to Ar-Ar to O-O, as compared to the variation going from Pb-Pb to Xe-Xe. Given such strong trends in the theory predictions, performing $v_{\rm n}$ measurements in light ion species such as Ar–Ar and O–O in Run 3 can provide strong constrains on the theoretical models.

Furthermore, there has been much work in studying long-range correlations observed in p-A, d-A, ³He–A, and more recently in pp collisions (see Chapter 9). Measuring flow in medium and light ions would allow for a continuous study of how collective phenomena vary from large (Pb–Pb) to small (p–A and pp) systems.

4.3.3 Longitudinal flow fluctuations

The characterization of how the longitudinal scales at which the symmetry planes Ψ_n associated to v_n decorrelate, and how this relates to the variation of the signal strength v_n with rapidity is still far from the state of the art reached in v_n measurements at mid-rapidity. However, any deviation from the simple picture of a rapidity-independent (Bjorken-like) longitudinal dynamics directly impacts our understanding of the time evolution of matter in all rapidity windows, including the well studied mid-rapidity one. Multiple recent measurements at the LHC indicate the presence of considerable longitudinal dynamics. These include measurements from CMS of event-plane decorrelation in p–Pb and Pb–Pb collisions [203, 204] and from ATLAS on flow-decorrelations [205] and forward-backward multiplicity fluctuations [217]. It is therefore important that experiments at the HL-LHC will parallel higher precision measurements at mid-rapidity with improved experimental control over the longitudinal evolution.

In the ATLAS measurements in Ref. [205], the flow decorrelation is quantified by constructing a correlator $r_{n|n:1}$ defined as:

$$r_{n|n;1}(\eta) = \frac{\langle \mathbf{v}_n(-\eta) \mathbf{v}_n^*(\eta_{\text{ref}}) \rangle}{\langle \mathbf{v}_n(\eta) \mathbf{v}_n^*(\eta_{\text{ref}}) \rangle},\tag{16}$$

where \mathbf{v}_n is the normalized flow vector, and η_{ref} is the reference pseudo-rapidity [205]. The correlator, $r_{n|n;1}$, measures the relative difference between flow $v_n e^{in\Psi_n}$ at η and $-\eta$. If flow were boost-invariant, then $r_{n|n;1}$ would equal unity. However any difference in the η dependence of the flow magnitude v_n and the event plane angle Ψ_n will lead to $r_{n|n;1}$ become smaller than unity. The ATLAS measurements of $r_{2|2;1}$ over the 0–2.5 η range are shown in Figure 10 by the markers. It is observed that the $r_{2|2;1}$ is significantly smaller than unity in central collisions, which indicates stronger flow decorrelation. For



Fig. 10: ATLAS projections of the flow-decorrelation observable $r_{2|2;1}$ as a function of η (lines). The markers indicate the present measurements from Ref. [205]. The left and right panels show projections for the 0–5% and 20–30% centrality intervals, respectively. The width of the projection bands indicates the expected statistical uncertainty.

a given centrality, $r_{2|2;1}$ decreases faster at low p_T that at higher- p_T . In the 20–30% mid-central $r_{2|2;1}$ decreases linearly with η , however in the 0–5% most central collisions there are indications that the decorrelations are possibly quadratic.

Repeating this measurement in Run 4 will lead to significant improvements due to increased luminosity and especially due to increased tracking acceptance in η to ± 4 units. Figure 10 also shows the ATLAS projections for $r_{2|2;1}$ made for Run 4, indicated as dashed lines. The ATLAS tracking acceptance in Run 4 will extend the η range to ± 4 units, but the projected measurement is made to ± 3.5 units, in order to leave a gap between the ITk and the region of the forward calorimeter in which the reference measurement is made ($4.4 < |\eta| < 4.9$). The projections are made by fitting the existing data with a linear function for the 20–30% centrality range and with a quadratic function for the 0–5% centrality range. It is seen that with the increased η acceptance the non-linearity in the flow decorrelation can be studied in much more detail.

The longitudinal flow-decorrelation observables are sensitive to the event by event fluctuations of the initial energy density profile in the longitudinal direction. Thus precise measurement of these decorrelations should give a better understanding of the initial conditions along the longitudinal direction and in the development of full three-dimensional viscous hydrodynamic models. These would in turn result in a more accurate estimation of η/s .

4.4 Vorticity and polarization

An interesting open question for relativistic fluids is to what extent the spin degrees of freedom thermalize locally and to what extend spin polarization results as a consequence of the fluid motion. Intuitively, one might expect that spin aligns locally with the rotational motion of the fluid as measured by vorticity, corresponding to the curl of the fluid velocity.

The relativistic generalization of the non-relativistic fluid vorticity is not unambiguous, however. The vorticity of a fluid in local equilibrium is characterized by the so-called thermal vorticity tensor, corresponding to $\omega_{\mu\nu} = \frac{1}{2}(\nabla_{\nu}\beta_{\mu} - \nabla_{\mu}\beta_{\nu})$ where $\beta_{\mu} = u_{\mu}/T$ is the ratio of fluid velocity to temperature [218]. This thermal vorticity includes contributions from global rotational motion, local fluid acceleration, and temperature gradients. It has been argued that this thermal vorticity should lead to local spin polarization. If this holds at chemical freeze-out, one should be able to find traces of the thermal vorticity in the spin polarization of particles and resonances, such as $\Lambda(\overline{\Lambda})$ particles.

Spin polarization is in this picture closely tied to angular momentum of the expanding fireball. For non-central events, the angular momentum of the produced matter is perpendicular to the event plane. Via the spin-vorticity coupling mechanism, this leads to a global polarization in the transverse plane aligning with the global angular momentum (also known as the "transverse polarization"). This global transverse polarization has recently been observed in the measurement of Λ spin polarization at RHIC [219]. For this global effect following global angular momentum, one expects a decreasing magnitude with increasing collision energy and the effect is expected to be relatively small at LHC energies.



Fig. 11: ALICE projections for the Global hyperon polarization in Pb–Pb collisions at $\sqrt{s_{\rm NN}} = 2.76$ TeV for an integrated luminosity of 10 nb⁻¹ (blush symbol), together with the present measurements (orange symbol) compared to analogous measurements at various collision energies from the STAR collaboration [219, 220] (blue and red symbols). The blue line indicates the prediction for the maximum value at the LHC [221]. The inlay plot shows a zoomed in version of the plot around the ALICE measurement and Run 3 and 4 projection, together with the prediction for the maximum value at the LHC. The points for $\overline{\Lambda}$ are slightly shifted along the horizontal axis for visibility. Error bars (open boxes) represent the statistical (systematic) uncertainties.

Figure 11 shows the energy dependence of the global transverse polarization of Λ and $\overline{\Lambda}$ for semicentral heavy ion collisions. The RHIC results show the decrease of polarization with increasing $\sqrt{s_{\text{NN}}}$. The preliminary ALICE data point at $\sqrt{s_{\text{NN}}} = 2.76$ TeV is consistent with zero within 1σ statistical/systematic uncertainties. However it is also consistent with the predicted maximum value (blue line) within $\sim 2\sigma$ statistical/systematic uncertainties. But the ALICE upgrade projection at twice large collision energy, (assuming zero signal) shows that the polarization in Run 3 and 4 can be measured with very high precision. Therefore the study of global polarization of Λ and $\overline{\Lambda}$ within HL–LHC project allows the unambiguous conclusion with regard of the values of this physics quantity in the TeV-energy domain.

In addition to the transverse polarization, an azimuthal-dependent, longitudinal polarization (in

the direction of the beam pipe) has also been predicted and recently observed at RHIC [222]. This is mainly a consequence of an azimuthal dependence of local acceleration and temperature gradient (e.g., the elliptic flow), which could lead to an elliptic modulation of longitudinal spin polarization in non-central collisions. Unlike the global transverse polarization, this longitudinal polarization effect has a much weaker dependence on collision energy from RHIC to the LHC [223], mainly because the anisotropic flow has a weak collision energy dependence. With increased data sample and upgraded detectors covering a wider rapidity range in the HL–LHC era, there will be exciting opportunities for precision study of the Λ polarization and to map out the dependence on variables such as azimuthal angle, rapidity, transverse momentum.

4.5 Chiral Magnetic Effect

An important property of the strong interaction which is potentially observable in heavy-ion collisions is parity violation. Although it is allowed by quantum chromodynamics (QCD), global parity violation is not observed in strong interaction. However, QCD predicts the existence of topologically non-trivial configurations of the gluonic field, instantons and sphalerons, which might be responsible for local parity violation in microscopic QCD domains at finite temperature [224–227]. The P- and CP-odd interactions between quarks and such fields with non-zero topological charge [228] change the quark chirality, breaking parity symmetry by creating an imbalance between the number of left- and right-handed quarks. Furthermore, an extremely strong magnetic field is expected to be produced in heavy-ion collisions [229,230] (of the order of 10^{19} Gauss at the LHC) because the charges of initial ions add coherently. This strong magnetic field aligns the spins of the positively (negatively) charged quarks in the direction parallel (anti-parallel) to magnetic field orientation. Moreover, right-handed (left-handed) quarks have their direction of momentum parallel (anti-parallel) to the spin orientation. The spin alignment coupled with the local imbalance between the number of left- and right-handed quarks leads to the development of a quark current. The current moves the positively charged quarks along its direction and the negatively charged quarks in the opposite direction. This implies a charge separation along the direction of the magnetic field, which is on average perpendicular to the reaction plane, a phenomenon called Chiral Magnetic Effect (CME) [231–234].

The sign of the topological charge can give rise to a positive or negative current in the magnetic field direction with equal probability. Therefore, the charge separation averaged over many events is zero. This makes the observation of the CME experimentally difficult and possible only via azimuthal particle correlations, which introduces a large flow related background into the measurements.

The three-particle correlator $\gamma_{\alpha\beta} = \langle \cos(\varphi_{\alpha} + \varphi_{\beta} - 2\Psi_2) \rangle$ [236], where φ_{α} is the azimuthal angle of the particle of charge α and Ψ_2 is the second harmonic symmetry plane angle, was proposed to measure charge-dependent azimuthal correlations. This correlator eliminates correlations independent of symmetry plane orientation, suppressing background contributions at the level of $\sim v_2$. However, the interpretation of the experimental results is complicated by the remaining background (e.g. local charge conservation (LCC) coupled with elliptic flow [237,238]). Recent observation of similar charge-dependent azimuthal correlations in p–Pb (where the CME is not expected) and Pb–Pb collisions [239] indicates the $\gamma_{\alpha\beta}$ correlator be dominated, if not all, by the background effect. The ALICE [235] and CMS [240] collaborations have used the Event Shape Engineering (ESE) technique [241] to estimate the CME fraction to the charge dependence of $\gamma_{\alpha\beta}$, $f_{\rm CME}$, in Pb–Pb collisions. ALICE extracted $f_{\rm CME}$ by relating measurements of the charge dependence of $\gamma_{\alpha\beta}$ from the ESE analysis to CME signal expectations from various initial state model calculations including a magnetic field. It has been assumed that the CME signal is proportional to $\langle |B|^2 \cos(2(\Psi_{\rm B} - \Psi_2)) \rangle$, where |B| and $\Psi_{\rm B}$ are the magnitude and direction of the magnetic field, respectively. Within current experimental uncertainties, the CME signal contribution to the $\gamma_{\alpha\beta}$ correlator is consistent with zero.

Figure 12 shows the upper limit on $f_{\rm CME}$ at 95% confidence level for the 20–30% centrality interval reported by the ALICE collaboration together with expectations for $f_{\rm CME} = 0.164$ (current



Fig. 12: ALICE projections for the upper limit on the CME fraction at 95% confidence level as a function of the number of events in the 20–30% centrality interval. The right-most projection point corresponds to an integrated luminosity of 10 nb⁻¹. The result reported by the ALICE collaboration [235] is shown together with expectations for $f_{\rm CME} = 0.164$ (current estimate) and $f_{\rm CME} = 0$ (null hypothesis). The shaded boxes denote variations from various initial state models (see text for details).

estimate) and $f_{\rm CME} = 0$ as a function of the number of events. The shaded boxes denote variations due to different estimates of the magnetic field from the investigated models. The ALICE upgrade projection indicates that stringent constraints for the CME contribution to the charge dependence of $\gamma_{\alpha\beta}$ can be achieved at a level of less than 1% with the expected HL–LHC statistics.

One key ingredient needed for the observation of the CME is the strong magnetic field in the QGP medium. It is important to establish direct evidence for this field and determine its strength, which will help significantly constrain theoretical predictions on the magnitude of the CME signal. Measurement of the pseudorapidity-odd component of directed flow, v_1^{odd} , separately for positive and negative charged particles has been proposed as a probe to the magnetic field [230]. Any difference will indicate the presence of induced electromagnetic currents and will allow to estimate the magnitude of the effect. It will also provide information on the electric conductivity of the QGP medium.

Figure 13 shows the charge difference of v_1^{odd} , $\Delta v_1^{\text{odd}} = v_1^{\text{odd}+} - v_1^{\text{odd}-}$, as a function of pseudorapidity measured by the ALICE collaboration in Pb–Pb collisions at $\sqrt{s_{\text{NN}}} = 5.02$ TeV [242] together with a linear fit. A hint of a charge-dependent difference is observed and quantified by the slope k with a total significance of 2.6 σ . This difference, which differs both in magnitude and sign compared to predictions for π^{\pm} at $\sqrt{s_{\text{NN}}} = 2.76$ TeV and similar $\langle p_{\text{T}} \rangle$ [230], needs to be measured with better precision. This will be achieved with the large data sample expected at the HL–LHC which will be sensitive to a difference as small as 5×10^{-5} (about three times smaller that the current measurement), as reported by the ALICE upgrade projection in Fig. 13. Furthermore, similar measurement can also be performed in the heavy flavor sector, e.g., for D⁰ and $\overline{D^0}$ meson directed flow [243], this is further discussed in Chapter 5. Heavy flavor quarks have the advantage of being produced at a very early stage, and thus potentially have a better sensitivity to the magnetic field at its evolution at early time.



Fig. 13: Charge difference of v_1^{odd} as a function of pseudorapidity measured by the ALICE collaboration in Pb–Pb collisions at $\sqrt{s_{\text{NN}}} = 5.02 \text{ TeV}$ [242] (red symbols) and the projection for a 5×10^{-5} difference [230] from 10 nb⁻¹ (black symbols) together with linear fits (dashed lines). Error bars (open boxes) represent the statistical (systematic) uncertainties.

4.6 Summary

The measurements of inclusive hadron v_n by traditional methods such as two-particle correlations, eventplane/scalar-product methods, multi-particle cumulants etc. have been performed with high precision by the ALICE, ATLAS and CMS experiments at the LHC. These inclusive hadron v_n measurements are not statistically limited across most of the centrality- $p_{\rm T}$ phase space and further improvement in the measurements is not a high priority for Run 3 and 4. However, in the case of identified hadrons the increased statistics will lead to further improvement in the v_n measurements. This is true for both light hadrons such as pions, protons, ϕ -mesons as shown in Figure 8, as well as for heavy-flavor particles such as D⁰, D^{\pm} , J/ψ , Υ which are discussed in Chapters 5 and 7, respectively. Significant improvements are expected in measurements of longitudinal flow fluctuations, which have only been briefly investigated in Run 1 and 2. These are largely driven by the increases η acceptance of the ATLAS and CMS tracking detectors in Run 4, the acceptance is planned to reach ± 5 units. The study of longitudinal flow fluctuations will allow comparisons to predictions of 3+1D hydro models. Flow measurements in light ions such as Ar-Ar and O-O, will lead to stronger constraints on theoretical models describing different stages of a heavy ion collision – initial conditions, equation of state, transport coefficients etc. This is difficult presently, as flow observables are dependent on all of these, so it becomes difficult to constrain any one of these without full knowledge of the others. Flow measurements across a variety of colliding species will provide independent data that will improve our understanding of the different stages of heavy ion collisions. Further physics motivations for colliding light ions are discussed in Chapter 11.1.

Other observables related to collective phenomena where current measurements are statistics limited and are expected to improve considerably are related to effects of vorticity and magnetic fields. The current measurements of Λ polarization from ALICE are statistics limited and consistent with both the null hypothesis as well as with the theoretically predicated value. The ALICE projections for Λ polarization in Run 3 and 4 show that the measurements will have significantly smaller statistical uncertainties and will differentiate between the null and predicted values. ALICE and CMS have measured the fraction of the three-particle correlator $\gamma_{\alpha\beta}$ that arises from CME effects: $f_{\rm CME}$. The measured $f_{\rm CME}$ by ALICE is consistent with zero but due to large uncertainties its upper limit at 95% CL can be as large as ~0.5. ALICE projections for Run 3 and 4 show that the $f_{\rm CME}$ can be determined with a precision of better than 1%.

5 Open heavy flavour

Coordinators: Elena Bruna (INFN Torino) and Gian Michele Innocenti (Massachusetts Institute of Technology, CERN)

Contributors: J. Aichelin (SUBATECH, CNRS/IN2P3, IMT Atlantique, Université de Nantes), S.A. Bass (Duke University), C. Bedda (Utrecht University), A. Beraudo (INFN Torino), G.E. Bruno (Politecnico di Bari and INFN), Z. Citron (Ben-Gurion University of the Negev), Z. Conesa del Valle (CNRS/IN2P3, Université Paris-Sud, Université Paris-Saclay, Orsay), A. Dainese (INFN Padova), A. Dubla (GSI Helmholtzzentrum für Schwerionenforschung GmbH), F. Fionda (University of Bergen), P.B. Gossiaux (SUBATECH, CNRS/IN2P3, IMT Atlantique, Université de Nantes), V. Greco (INFN Catania, Università di Catania, INFN-LNS), F. Grosa (Politecnico di Torino and INFN), Y.-J. Lee (Massachusetts Institute of Technology), J. Margutti (Utrecht University), V. Minissale (INFN-LNS, Catania), A. Mischke (Utrecht University), S. Mohapatra (Columbia University, New York), C.C. Peng (Purdue University), M. Peters (Massachusetts Institute of Technology), S. Plumari (Università di Catania), F. Prino (INFN Torino), A. Rossi (Padova University and INFN), J. Sun (Tsinghua University, Beijing), C. Terrevoli (Padova University and INFN), B. Trzeciak (Utrecht University), A. Uras (Université de Lyon, CNRS/IN2P3, IPN-Lyon), L. Van Doremalen (Utrecht University, Utrecht), I. Vitev (Los Alamos National Laboratory), J. Wang (Massachusetts Institute of Technology), T.-W. Wang (Massachusetts Institute of Technology), M. Winn (LAL, now DPhN, CEA/IRFU), R. Xiao (Purdue University), Y. Xu (Duke University)

This chapter is dedicated to the memory of our colleague Andre Mischke.

5.1 Perspectives for heavy-flavour observables in LHC Run 3 and 4

Charm and beauty quarks are produced in hard scattering processes occurring in the early stage of heavyion collisions. They subsequently traverse the QGP medium and interact with its constituents through inelastic (gluon radiation) and elastic (or collisional) processes. These interactions may lead to the thermalisation of low-momentum heavy quarks, which would thus take part in the expansion and hadronisation of the medium. For these reasons, heavy-flavour hadrons provide information on all stages of the system evolution and they uniquely probe the quark-mass dependence of the QGP inner workings (see Refs. [244–246] for recent reviews).

Many experimental observations from RHIC and LHC showed evidence that charm and beauty quarks interact strongly with the QGP and that beauty quarks lose less energy at low transverse momentum compared to charm quarks [247, 248]. While data are becoming more and more precise to start imposing constraints on theoretical calculations, there are still several unresolved questions: What are the microscopic mechanisms that drive heavy-flavour interaction and diffusion in the QGP, and what are their implications for the QCD matter structure? What is the relative relevance of collisional and radiative processes? Can the same QCD process can describe both the heavy-quark interaction with the strongly coupled plasma and the mechanisms of hadronisation?

The Run 3 and 4 of the LHC will open a new precision era for heavy-flavour measurements in heavy-ion collisions that will address the above questions. With the upgrades of the machine and of the tracking detectors, the higher accumulated statistics and higher precision will make it possible to quantify the properties of the QGP with heavy-flavour probes. This high-precision era will also make new and more differential observables accessible for the first time. The key measurements that are expected to have a strong impact on the characterisation of the QGP with heavy-flavour observables are discussed in this chapter and summarized below.

- Nuclear modification factor and flow harmonics: these measurements for particles with charm and beauty in the large kinematic range covered by combining the different LHC experiments will put the strongest constraints on the transport coefficients of the QGP, clarifying the microscopic mechanisms governing the interactions of heavy quarks with the medium, and quantifying their degree of thermalisation.

- Strange D and B mesons, charm and beauty baryons: currently limited by statistics, these measurements will help to quantify not only the degree of thermalisation of heavy quarks, but also the contribution of recombination with lighter quarks to the hadronisation process. They are also sensitive to the mass scaling of the hyrodynamical flow in the heavy-flavour sector.
- Heavy-flavour correlations and jet observables: they will provide new insights on the parton mass effects in parton showers, on the redistribution of the radiated energy, and on the role of collisional and radiative energy loss.

5.2 Impact of detector upgrades on heavy-flavour measurements

The upgrades of the four large LHC experiments during LS2 and LS3 will strongly enhance their performance for open heavy-flavour measurements. The detector improvements that will have the largest impact are the new silicon trackers, with higher granularity and precision, as well as extended pseudorapidity coverage. Brief descriptions of these improvements are reported in the following.

- ALICE. The new Inner Tracking System [249], which will be installed during LS2, is composed of seven layers of pixel detectors with an intrinsic spatial precision of about $5 \times 5 \ \mu m^2$ and a material thickness of 0.3% of the radiation length in the innermost layers. The track pointing resolution will be improved by a factor 3 in the direction transverse to the beam line and by a factor 5 in the longitudinal direction, down to values of about 20 μm for tracks with $p_T = 1 \ \text{GeV/c}$. A Muon Forward Tracker [250], composed of 5 disks of pixel detectors with the same spatial resolution as the Inner Tracking System, will instrument the region $2.5 < \eta < 3.6$, in front of the muon spectrometer, enabling the separation from the primary vertex of single and dimuons from D, B and J/ ψ decays. The upgraded TPC with GEM-based readout chambers, together with readout upgrades of several other detectors and with a new Online-Offline computing system, will enable the full recording of Pb–Pb interactions with a minimum-bias trigger at a rate of 50 kHz, which is 50-fold larger than for the present apparatus.
- ATLAS. The Inner Tracker (ITk) [251] will be an all-silicon tracker composed of pixels and strips installed during LS3 for ATLAS phase II. The ITk will provide charged-particle tracking acceptance for $|\eta| < 4$. The performance of the ITk in Pb–Pb collisions is expected to be comparable to pp collisions. The High Granularity Timing Detector [252] has been proposed to complement the spatial information of the ITk with timing information. These detectors will improve jet reconstruction capabilities, and in particular tagging of heavy-flavour jets, as well as all studies using charged particles.
- CMS. The following upgrades scheduled for LS3 will largely enhance the performance for heavy-flavour measurements, in particular in the low-momentum region [253]. The upgraded inner tracker will cover a large acceptance up to $|\eta| < 4$ [254]. The improved L1 and DAQ rate (up to 60 GB/s) will allow more sophisticated triggers and to record a larger number of minimum-bias triggered events. In addition, the proposed MIP Timing Detector [255] with a radius of 1.16 m and a time resolution of ≈ 30 ps could provide, in conjunction with other detectors, proton, pion and kaon separation in the interval $0.7 < p_T < 2$ GeV/c in $|\eta| < 1.5$.
- LHCb. The experiment is preparing to run at five times larger instantaneous luminosities in pp collisions, processing the full event rate with a software trigger and preserving or exceeding the present performance. All tracking detectors will be upgraded during LS2 [256,257]. Most notably for heavy-flavour observables, the active area of the upgraded Vertex Locator, the pixel detector replacing the present silicon strip detector, will move as close as 5.1 mm to the nominal beam spot. The larger granularity for the majority of phase space will improve the performance in Pb–Pb collisions whereas proton-induced reactions will result in average in lower detector occupancies than the standard pp running.

5.3 Nuclear modification factor and collective flow

The standard observable used to study the medium effects on heavy-flavour meson production is the nuclear modification factor (R_{AA}), defined as the ratio of the Pb–Pb yield to the pp cross-section scaled by the nuclear overlap function. In the view of the pQCD-based models, heavy quarks interact with the medium constituents via radiative and collisional processes. While radiative interactions only lead to energy loss, collisional ones can also result in an increase of the heavy-quark momentum. The dead-cone effect [258] is expected to reduce small-angle gluon radiation of heavy quarks when compared to both gluons and light quarks. At low $p_{\rm T}$, the production rate of heavy-flavour mesons in heavy-ion collisions is sensitive to the elastic energy loss of the heavy quark in medium, the nuclear shadowing effect in the initial state, and the recombination of the heavy quark with light quarks at the hadronization stage. At high $p_{\rm T}$, the nuclear modification factor is sensitive to the medium-induced radiative energy loss of heavy quarks. Precise measurements of the $R_{\rm AA}$ thus provide insights on the momentum dependence of heavy quark energy loss, and provide important tests of QCD predictions, in particular for the expected flavour and mass dependence of the energy loss processes.

Another interesting observable is the azimuthal anisotropy of open heavy flavour production, which can be characterized by the Fourier coefficients v_n in the azimuthal angle (φ) distribution of the heavy-flavour hadron yield with respect to the reaction plane in non-central Pb–Pb collisions. At low p_T , the v_2 measurements can provide important insights into the mechanisms of interaction of heavy quarks with the medium and on their strength (as discussed in more details in Sec. 5.3.2). Heavy quarks are indeed expected to acquire a positive v_2 mostly as a consequence of their interaction with the light quarks of the medium. Measurements of elliptic flow of heavy hadrons are also sensitive to hadronisation processes. In particular, they can be used to study the relevance of heavy-flavour recombination (see Sect. 5.4.1) in which heavy-quarks can acquire additional v_2 by combining with light quarks at the hadronisation stage. At high p_T , v_2 of heavy-flavour hadrons is sensitive to the path-length dependence of heavy quark energy loss. The simultaneous description of R_{AA} and v_2 for heavy-flavour hadrons is still challenging for most of the theoretical calculations, because it entails accurate modelling of the initial heavy-quark production and its modification in nuclei, of the medium and its expansion, of the various quark-medium interaction mechanisms and of the possible modification of hadronisation processes.

5.3.1 Experimental performance of the ALICE, ATLAS and CMS experiments

Figure 14 shows the projected performance for the R_{AA} of several heavy-flavour hadrons or decay channels with $L_{int} = 10 \text{ nb}^{-1}$. The left panel presents the projection of charged particles, D^0 , B^+ and non-prompt J/ ψ from b-hadron decay which can be measured by CMS. The right panel shows the ALICE simulation results for D^0 , non-prompt J/ ψ , non-prompt D^0 , $B^+ (\rightarrow D^0 \pi^+)$ —in addition, the $B^0 \rightarrow D^{*+} \pi^-$ reconstruction was studied by ALICE and it provides an alternative channel for the study of the beauty meson R_{AA} with a significance of larger than 5σ at $p_T > 3 \text{ GeV}/c$. With the high luminosity and the Inner Tracking System Upgrade in ALICE, the R_{AA} of light hadrons, charm hadrons and beauty hadrons can be clearly separated in a wide kinematic range.

Figure 15 shows the projected performance for v_2 of charm hadrons with $L_{int} = 10 \text{ nb}^{-1}$. The left panel shows the projection for D^0 in CMS, with the charged particle v_2 also shown for comparison. The right panel presents the projection for D^0 , D_s and Λ_c in ALICE [249]. Precise measurements of charm hadron v_2 will allow the study of the thermalization of heavy quarks and the wide kinematic range allows to get insights on different process, as coalescence hadronization and energy loss. Figure 16 (left) shows the projected performance for v_2 of B⁺ mesons, non-prompt D⁰ and non-prompt J/ ψ . These will be the first precise measurements of B meson elliptic flow at the LHC. Heavy-flavour flow will also be measured with high precision using decay electrons and muons. As an example the projection for the measurement with muons by ATLAS is shown in Fig. 16 (right) [260].



Fig. 14: Nuclear modification factors of charged particles, D^0 , B^+ and non-prompt J/ ψ in CMS [259] (left). R_{AA} of D^0 , non-prompt J/ ψ and non-prompt D^0 in ALICE in central Pb–Pb collisions for $L_{int} = 10 \text{ nb}^{-1}$ [249] (right).



Fig. 15: v_2 of charged particles and D⁰ in CMS [259] (left), charm hadrons (D⁰, D_s, Λ_c) in ALICE (right) in Pb–Pb collisions with $L_{int} = 10 \text{ nb}^{-1}$ [249].

5.3.2 Constraining the heavy-quark diffusion coefficient $2\pi TD_s$

Many theoretical efforts have been recently undertaken to understand the properties of the QGP medium and the interaction between heavy quarks and the medium constituents, see Refs. [244–246] for recent reviews. Although the interaction mechanism can widely vary among different theoretical models, the reduction to a few transport coefficients allows one to compare these models and evaluate different microscopic pictures. Most of the present theoretical models explain the interactions of heavy quarks as dominated by collisional (elastic) processes in the low transverse momentum region (up to about 5– 10 GeV/c) and by radiative energy loss (inelastic process with gluon radiation off the heavy quark) at higher $p_{\rm T}$.

The extraction of the heavy-quark spatial diffusion coefficient, which is one of the main QGP



Fig. 16: v_2 of non-prompt D⁰, non-prompt J/ ψ and B⁺ (\rightarrow D⁰) in ALICE (left) [249] and of heavy-flavour decay muons in ATLAS (right) [260] in Pb–Pb collisions for $L_{\text{int}} = 10 \text{ nb}^{-1}$.



Fig. 17: Left: Normalized χ^2 as a function of spatial diffusion coefficient $(2\pi TD_s)$ for different experimental precision levels from the Catania Fokker-Plank transport model [262, 263]. Right: Coefficient range (90% credibility region) for $2\pi TD_s$ as a function of T/T_c for different experimental precision levels, estimated by a model-to-data Bayesian analysis using a modified Langevin framework [264].

properties regulating the strength of collisional processes, is considered here to illustrate the impact of the high-precision measurements in Run 3 and Run 4. In particular, the $p_{\rm T}$ -dependence of this coefficient provides important constraints on the weakening of the interaction strength with increasing $p_{\rm T}$. For illustration, the extracted values of D_s is considered at a fixed $p_{\rm T}$ value. The heavy-quark spatial diffusion coefficient D_s in the QGP is related to the relaxation (equilibration) time of heavy quarks $\tau_{\rm Q} = \frac{m_{\rm Q}}{T}D_s$, where $m_{\rm Q}$ is the quark mass and T is the medium temperature [261].

Figure 17 shows the constraining power of future experimental measurements of R_{AA} and v_2 on the heavy quark diffusion coefficient $(2\pi TD_s)$ using two different transport models: Catania model with Fokker-Plank equation [262, 263] on the left and a modified Langevin framework on the right [264]. The left figure presents a normalized χ^2 as a function of spatial diffusion coefficient by comparing the



Fig. 18: Left: projection of the expected ratio of D^0 -meson v_2 in the 10% events with larger (smaller) q_2 with respect to the unbiased one as a function of p_T for the 30–50% centrality class. The modification of the D^0 -meson v_2 was assumed to be equal to that measured for the charged particles in Pb–Pb collisions at $\sqrt{s_{NN}} = 2.76$ TeV in the the 30–40% centrality class (superimposed for comparison) [265]. Right: projection of the expected ratio of D^0 -meson p_T -differential yield in the 10% events with larger q_2 with respect to the unbiased one, estimated considering the prediction provided by the POWLANG model [266].

model calculation [262, 263] of D-meson R_{AA} in Pb–Pb collisions at 5.02 TeV in a single centrality class (0–10%). The cases of the present experimental uncertainties (2015 Pb–Pb sample) and of these uncertainties reduced factors of two or five are considered for illustration. In the projections for $L_{int} = 10 \text{ nb}^{-1}$ shown in the previous section, the D-meson R_{AA} uncertainties are reduced by a factor between two and five, depending on p_T , with respect to the present measurements. Considering the $2\pi T D_s$ range with $\chi^2_{R_{AA}}/\text{n.d.f} < 1.5$ (corresponding to 85% confidence), it is found that by reducing the present experimental uncertainty by a factor two or five, the uncertainty on the estimation $2\pi T D_s$ would be also reduced almost to 50% or 20%, respectively. The right panel presents the diffusion coefficient as a function of temperature, which is estimated using a Bayesian calibration on D-meson R_{AA} and v_2 in Pb–Pb collisions at 5.02 TeV for different centralities. The $2\pi T D_s$ shows a positive temperature dependence with the minimum value around T_c . Such behaviour is consistent with the Bayesian estimation for shear viscosity η/s . The potential improvement with Run 3 and Run 4 measurements is estimated using the R_{AA} and v_2 projections by ALICE and CMS shown in the previous section and it is shown by the red band. With these future experimental measurements, the diffusion coefficient around T_c could be constrained with an uncertainty of about 30–50% of the present one.

5.3.3 D-meson analyses with Event Shape Engineering

Further insight into the dynamics of heavy quarks in the medium can be obtained from measurements of the yield and elliptic flow of heavy-flavour particles with the Event Shape Engineering (ESE) technique [241]. This technique consists of selecting events with the same centrality but different magnitude of the average bulk elliptic flow and therefore initial-state geometry eccentricity. The analyses with ESE will allow us to investigate the correlation between the flow coefficients of heavy-flavour hadrons and soft hadrons, to study the interplay between elliptic and radial flow, and to further constrain the path-length dependence of the energy loss suffered by the heavy quarks in the QGP. A first analysis was published by ALICE using 2015 Pb–Pb data [267]. In the left-hand panel of Fig. 18, the prospects for the measurement of the D⁰-meson v_2 with the ESE technique in the 30–50% centrality class with $L_{int} = 10 \text{ nb}^{-1}$ are shown. In particular, the ratio between the v_2 of D mesons in the 10% of the events with larger (smaller) elliptic flow of the bulk (quantified through the magnitude of the so-called reduced flow vector q_2) and the v_2 in all the collisions in the considered centrality class is reported as a function of p_T . It is compared to the current measurement of the same ratio for charged particles, which is dominated by light-flavour hadrons. The expected statistical uncertainties for the D⁰-meson v_2 in the 0–10% of events with larger (smaller) q_2 are of the order of about 1–2% in the interval $1 < p_T < 8$ GeV/c. This will allow us to resolve a possible difference of a few percent in the response of the v_2 to the ESE selection between the D⁰ mesons and the light hadrons, providing new insight on the coupling of the charm quark with the medium constituents and on its degree of thermalisation. The performance for the measurements of D⁰-meson p_T -differential yield in event-shape classes is displayed in the right-hand panel of Fig. 18. The expected performance will provide a sensitivity of a few percent for the modification of the D-meson p_T spectra in events with small (large) initial geometrical anisotropy. This will open the way for precise studies on the interplay between the initial geometrical anisotropy (the collective flow of the bulk) and the heavy-flavour radial flow and energy loss.

5.4 Studies of heavy-quark hadronisation

5.4.1 Hadronisation mechanisms for c and b quarks

The hadronisation mechanisms belong to the non-perturbative domain of QCD and a first-principle description of these processes is still missing, for both light and heavy flavours. However, from the study of charm dynamics in nucleus–nucleus collisions in the last decade there is a general consensus that the details of hadronisation have a large effect on both the heavy-flavour observables R_{AA} and v_2 [246, 268, 269]. This section introduces the two main microscopic hadronisation mechanisms for the production of heavy-flavour hadrons: fragmentation and coalescence (also denoted as recombination). Fragmentation is one of the most common approaches for the calculation of inclusive hadron production and it is appropriate for high-momentum partons emerging from initial hard processes, where high-momentum quarks fragment directly and independently into high-momentum hadrons. Independent fragmentation has also been widely applied at low momentum in e^+e^- , ep and pp collisions. On the other hand, coalescence is expected to dominate in the low-momentum regime in nucleus–nucleus collisions, where partons are abundant and heavy quarks can hadronise by recombination with light quarks [270–272]. Recent measurements at the LHC indicate that fragmentation may not be sufficient to describe charm quark hadronisation at low momentum in pp and p–Pb collisions, at least for what concerns baryon production [273, 274].

The hadron momentum spectra produced by heavy-quark fragmentation are given by:

$$\frac{\mathrm{d}N_{\mathrm{had}}}{\mathrm{d}^2 p_{\mathrm{T}} \,\mathrm{d}y} = \sum \int \mathrm{d}z \frac{\mathrm{d}N_{\mathrm{fragm}}}{\mathrm{d}^2 p_{\mathrm{T}} \,\mathrm{d}y} \frac{D_{\mathrm{had/Q}}(z,Q^2)}{z^2} \tag{17}$$

where $z = p_{had}/p_Q$ is the fraction of quark momentum carried by the hadron, $Q^2 = (p_{had}/2z)^2$ is the momentum scale for the fragmentation process and $D_{had/Q}$ is the fragmentation function.

In the basic coalescence model developed in Refs. [275–281] and used here for illustration, the spectrum of heavy-flavour hadrons formed by coalescence of heavy-light quarks can be written as

$$\frac{\mathrm{d}^2 N_{\mathrm{had}}}{\mathrm{d}p_{\mathrm{T}}^2} = g_{\mathrm{had}} \int \prod_{i=1}^n \frac{\mathrm{d}^3 p_i}{(2\pi)^3 E_i} p_i \cdot \mathrm{d}\sigma_i \cdot f_{q_i}(x_i, p_i) f_{\mathrm{had}}(x_1 \dots x_n, p_1 \dots p_n) \,\delta^{(2)} \left(p_{\mathrm{T,had}} - \sum_{i=1}^n p_{\mathrm{T},i} \right)$$
(18)

where $d\sigma_i$ denotes an element of a space-like hypersurface and f_{q_i} are the quark (antiquark) phase-space distribution functions for *i*-th quark (antiquark), while g_{had} is the statistical factor to form a colourless hadron from quarks and antiquarks with spin 1/2. An alternative approach proposes resonance formation in quark-antiquark scatterings as the main production channel of D mesons through coalescence. In this approach, the same QCD process is responsible for both the interaction of charm quarks with the strongly-coupled medium and for the process of D-meson formation through recombination [282, 283]. Hadronisation via coalescence leads to a modification of the relative abundance of the various heavy-flavour hadron species produced. The most striking effect is an enhancement of the baryon-tomeson ratios for heavy-flavour hadrons. First studies of these ratios [279, 281] indicated a significant change in the relative abundances of the heavy-flavour hadron species, and in particular a ratio of Λ_c/D^0 close to unity, which is nearly an order of magnitude larger than what predicted by the fragmentation process implemented in the PYTHIA event generator. This value is also much larger than the prediction of the statistical hadronisation model [284], in which the hadronisation occurs by recombination of an equilibrated system of quarks and the hadron abundances are mainly determined by their masses. In addition, hadronisation via coalescence in a strangeness-rich Quark-Gluon Plasma is predicted to lead to a large enhancement in the production of strange heavy-flavour hadrons, like D_s and B_s [285–287]. Hints of an enhancement of the D_s/D⁰ and Λ_c/D^0 ratios in nucleus–nucleus collisions have recently been reported by ALICE at the LHC [288, 289] and by STAR at RHIC [290].

The hadronisation by coalescence plus fragmentation also affects the $p_{\rm T}$ distribution of heavyflavour hadrons. For D mesons the effect can be roughly seen as a shift in $p_{\rm T}$ of about 1.0–1.5 GeV/c in the region of $p_{\rm T}$ of 1.5–6 GeV/c, resulting in a significant enhancement of $R_{\rm AA}(p_{\rm T})$. The degree of these enhancements for D mesons, however, is significantly different among the different implementations of the hadronisation by coalescence and it is nearly unknown for heavy-flavour baryons, like $\Lambda_{\rm b}$ and $\Lambda_{\rm c}$.

Finally, the coalescence process leads to a significant enhancement of the $v_2(p_T)$ in the intermediate p_T region that, for D mesons, amounts to about 20–40% depending on the specific modelling. The Λ_c is expected to acquire instead a much larger enhancement of at least a factor of two for the v_2 . Therefore, a combined measurement of the Λ_c/D and Λ_b/B ratios and of the baryon elliptic flow would impose strong constraints on the hadronisation mechanism and lead to a better determination of the diffusion transport coefficient of heavy quarks (see Section 5.4.3).

5.4.2 Measurement performance studies (ALICE and CMS)

Simulation studies for the measurement of the D_s , Λ_c and Λ_b production were carried out by the ALICE Collaboration [249] and are updated in the present document. A projection of the performance for the B_s^0 meson by the CMS Collaboration is also reported [259]. The D_s , Λ_c and Λ_b ($\rightarrow \Lambda_c \pi$) reconstruction strongly benefits from the improved track spatial resolution of the ALICE Inner Tracking System Upgrade, because they have small mean proper decay lengths (e.g. about 60 μ m for Λ_c) and large combinatorial backgrounds. The Λ_c , Λ_b and B_s^0 , in particular, require very large integrated luminosities, because the decay branching ratios are very small (e.g. about 3×10^{-4} for $\Lambda_b \rightarrow \Lambda_c(\rightarrow pK\pi)\pi$) and the combinatorial background is very large, in case for the Λ_c , which has a small separation from the interaction vertex and lower invariant-mass than the b-hadrons. In [259] it has been shown that the statistical uncertainty in the lowest accessible p_T intervals for these hadrons would increase above 20–30% with integrated luminosity significantly lower than 10 nb⁻¹.

Figure 19 shows the performance for the R_{AA} of D_s (left) and B_s^0 (right), compared with the corresponding non-strange mesons. The predicted difference between the D^0 and $D_s R_{AA}$ will be measured very precisely and the difference in the beauty sector could be observed with a significance of about 3 σ . The measurements were studied only for $p_T > 2$ and 8 GeV/*c*, respectively, but an extension to lower p_T is considered within reach.

Figure 20 shows the performance for the charm and beauty baryon-to-meson ratios as they can be measured by ALICE with $L_{int} = 10 \text{ nb}^{-1}$ [249]. The measurements are compared with predictions based on various mechanisms for heavy-quark recombination in the medium [279,281]. Figure 15 (right) shows the performance for the elliptic flow coefficient v_2 of D⁰, D_s and A_c in semi-central Pb–Pb collisions [249]. The precision of the D_s v_2 should be sufficient to enable a significant comparison with D⁰ and with model calculations, in which the observable is found to be sensitive to the interactions of D mesons in the hadronic phase that characterises the late stages of the collision [286]. Both measure-



Fig. 19: Measurement performance projections for the nuclear modification factor R_{AA} of D_s (left) and B_s^0 (right) mesons in Pb–Pb collisions ($L_{int} = 10 \text{ nb}^{-1}$). The ALICE study for D_s is based on full simulation [249]. The CMS projection is based on scaling of uncertainties from existing measurements [259, 291].



Fig. 20: ALICE measurement performance for the Λ_c/D^0 (left) and Λ_b/B^+ ratios in central Pb–Pb collisions ($L_{\text{int}} = 10 \text{ nb}^{-1}$), based on studies from [249].

ments cannot be extended to the very-low-momentum region, where the separation of the heavy-flavour secondary vertex from the primary vertex is small. This limitation motivates studies for a further improvement of the ALICE inner tracker during LS3 [292]. A more precise measurement would open the possibility to test in the charm sector some features at present only observed for the v_2 of light-flavour hadrons: the mass scaling at low p_T and the baryon-meson grouping at high p_T .

5.4.3 Impact of hadronisation models on QGP characterisation

The hadronisation mechanism of heavy quarks is important for the description of the measured heavy-flavour R_{AA} and v_2 at RHIC and LHC energies. In particular, it has been recognized that recombination



Fig. 21: Illustration of charm diffusion coefficient $2\pi TD_s(T)$ estimate from D meson R_AA data using different hadronisation assumptions in the Catania model [262, 263] as a function of temperature. The blue band refers to fit with fragmentation only, the red band to the fit with coalescence plus fragmentation. In the left panel the D_s estimation obtained with the current experimental uncertainties is shown, while in the right panel the estimation considering the case with more precise measurements as expected with $L_{\text{int}} = 10 \text{ nb}^{-1}$ is shown.

play a dominant role in describing simultaneously both D meson R_{AA} and v_2 [269,293–295]. Moreover, as discussed in Section 5.4.1, the charm baryon-to-meson ratio Λ_c/D^0 measured at RHIC and LHC is not consistent with a fragmentation only scenario [279,281]. Therefore, a combined study including also the heavy baryon-to-meson ratio provides further information to solve the ambiguity on the recombination fraction. In the following, the sensitivity of the OGP characterisation to the recombination fractions described in Section 5.4.1 is illustrated. To estimate how different model implementations of the hadronisation mechanisms can affect the extraction of the charm quark diffusion coefficient $2\pi TD_s$, a global quantitative χ^2 analysis was carried out by comparing experimental data on the D meson R_{AA} with theoretical results obtained using the Fokker-Planck equation under a standard bulk medium evolving hydrodynamically with $\eta/s = 0.1$ [296, 297]. The model developed in [262, 263] with diffusion and drag coefficients related by the fluctuation-dissipation theorem was used, considering two different implementations of the hadronisation process: one using only fragmentation of charm quarks to D mesons (with the Peterson function) and another one including a hybrid hadronisation by coalescence plus fragmentation [269, 281]. For the estimate of the temperature dependence of $2\pi TD_s$, a schematic model in which the drag coefficient is parametrized as $\gamma = \gamma_0 (T/T_c)^{\beta}$ was used. The two parameters γ_0 and β were determined by minimizing the $\chi^2/\text{n.d.f.}$ of the model with respect to the measured D meson R_{AA} . The exercise could in principle be repeated using both R_{AA} and v_2 . The spatial diffusion coefficient is directly related to the drag coefficient by $D_s = T/(M \cdot \gamma(p=0))$, where M is the charm quark mass. The left panel of Fig. 21 shows the spatial diffusion coefficient $2\pi TD_s(T)$ estimated from the fit to the present data with $\chi^2/n.d.f < 2.5$. The blue band corresponds to the fragmentation only results and the red band to the coalescence plus fragmentation result. The right panel shows the same calculations obtained with experimental data with smaller error bars on the D-meson R_{AA} as expected with $L_{\rm int} = 10 \text{ nb}^{-1}$. The comparison between the two cases highlights the difference in the estimation of the D_s coefficient obtained with the two different hadronisation mechanism. Clearly, an optimal estimate of the diffusion coefficient requires an accurate description of the hadronisation mechanisms in the model.



Fig. 22: LHCb projection of the azimuthal \overline{DD} correlations in p–Pb and in pp collisions at $\sqrt{s} = 8.8$ TeV for the integrated Run 3 and Run 4 luminosity. The p–Pb panel shows a calculation for two different kinematic selection in p–Pb collisions within the EPOS3-HQ generator. A detailed description of the experimental estimates can be found in Ref. [300].

5.5 Heavy-flavour correlations and jets

5.5.1 Heavy-flavour correlations

Although heavy quarks at the LHC energies are mostly produced in primary hard scatterings between the incoming partons in hadronic collisions, a non negligible fraction of c and b quarks are originated from processes of gluon splitting. At the leading order, $c\bar{c}$ pairs are produced with an azimuthal opening angle of 180°. At the next to leading order, gluon splitting and flavour excitation processes can generate $c\bar{c}$ pairs typically at small opening angles. The role of next-to-leading order production is currently poorly understood even in proton-proton collisions [298,299] and introduces sizeable uncertainties in the models for Pb–Pb collisions. As a result, heavy-ion calculations which use proton-proton generators as baseline can be biased because of the different energy loss of quarks and gluons. Correlations between D and \overline{D} represent a promising way to study the relevance of these mechanisms in different kinematic ranges of the charm quarks. In p–Pb collisions, the study of open heavy-flavour correlations at forward and backward rapidity provides information to test modifications of parton distribution functions in nuclei (see Sect. 10.3.3).

Figure 22 shows the LHCb projections for the azimuthal $D\overline{D}$ correlations at forward rapidity in pp and p–Pb collisions at the same collision energy per nucleon-nucleon-pair of 8.8 TeV. The figure shows only statistical uncertainties (dominant with respect to the systematic uncertainties) as expected with the Run 3 and 4 integrated luminosity. The measurement of the $D\overline{D}$ correlation can be performed in intervals of D meson $p_{\rm T}$, providing differential information to test theoretical models with precision. The experimental projections are compared with predictions obtained with EPOS3-HQ event generator for two different kinematic selections in case of *p*Pb collisions. In particular, the two selected $p_{\rm T}$ ranges, 2-3 GeV/*c* and 3-12 GeV/*c*, yield to significantly different correlation shapes in the calculation. This dynamic change of shape demonstrates the necessity to provide precise quantitative tests of the importance of different production mechanism at the partonic level, the hadronisation and potential medium effects in pp and in p–Pb collisions across different kinematic configurations.

In nucleus-nucleus collisions, $D\overline{D}$ correlations are sensitive observables to discriminate among different mechanisms of in-medium energy loss of heavy-quarks, like radiative and collisional processes. Such measurements, presently challenged by statistical limitations, will greatly benefit from future high-luminosity heavy-ions runs.



Fig. 23: (Left) Nuclear modification factor of D⁰-meson tagged jets in central Pb–Pb collisions: ALICE simulation results compared to preliminary results from the 2015 run. (Right) CMS projection for the distribution of D mesons in jets as a function of the distance from the jet axis for jets of $p_T > 60 \text{ GeV}/c$ in Pb–Pb collisions divided by the distribution in pp collisions.

5.5.2 Heavy-flavour jet measurements

Further insights into the mechanism of parton energy loss in the QGP can be provided by the study of reconstructed heavy-flavour jets. These measurements provide complementary information to the studies of D and B mesons since they enable to better determine the energy of the initial heavy quark and give access to the jet energy profile. By comparing the production of heavy-flavour jets with light jets one can test the expected flavour dependence of the in-medium energy loss in a wide transverse momentum range and study the different energy loss mechanism of quarks and gluons. The study of more differential observables related to the production of heavy-flavour particles in jets, like the fragmentation function and the angular correlations, provides additional constraints into the mechanisms of redistribution of the energy lost by the parton inside the medium.

Figure 23 (left) shows the projections for the nuclear modification factor of D^0 -meson tagged jets in Pb–Pb collisions at $\sqrt{s_{NN}} = 5.02$ TeV with ALICE. The measurement will provide a precise estimation of the suppression of D^0 -tagged jets down to low p_T , opening the way to study possible modifications of charm fragmentation. Future measurements of the fragmentation functions of charmed hadrons in pp collisions will also help to reduce the uncertainties on the charm fragmentation mechanisms, which are currently among the main sources of uncertainties for theoretical calculations that describe heavy-quark production at the LHC.

Figure 23 (right) shows the CMS projection for the distribution of D mesons as a function of the distance from the jet axis for jets of $p_T > 60 \text{ GeV}/c$ in Pb–Pb collisions divided by the distribution in pp collisions. With the precision achievable with the high-luminosity data, one would be able to measure precisely the effect of heavy-quark in-medium energy loss and the redistribution of the energy at large angle with respect to the jet axis. This is expected to provide new constraints on in-medium energy loss calculations. By comparing the ratio measured for D mesons with the one obtained for charged particles, it will be possible to assess the relevance of medium response phenomena, which can induce modification of the jet shape at large angles and are expected to be less relevant for heavy quarks due to their large masses.

Another area of research where the LHC experiments can make key contributions is the study of heavy-flavour jet substructure. New experimental observables related to the inner structure of heavy-

flavour jets, like the splitting functions, can provide insights into the mass dependence of the parton shower in new kinematic regimes. A precise measurement of these observables at the LHC down to low $p_{\rm T}$ would also provide a unique opportunity to further investigate the dead cone effect [258], currently not well understood and constrained.

5.6 Sensitivity to early magnetic fields and vorticity phenomena

Recently, it has been recognized that very strong electric and magnetic fields are created at early times of ultra-relativistic heavy-ion collisions. In Fig. 24 (left), the time dependence of the magnetic (B_y) and electric (E_x) fields in Pb–Pb collisions at $\sqrt{s_{NN}} = 2.76$ TeV with impact parameter b=9.5 fm is presented for a system with electric conductivity $\sigma_{el} = 0.023$ fm⁻¹. The space-time evolution is obtained as solution of the Maxwell's equations as developed in [230]. The magnetic field produced in non-central heavy ion collisions is dominated by the B_y component which induces a current in the xz plane while the time dependence of B_y generates an electric field which is directed in the x direction. The combined effect of both fields is a current in the xz plane.

The presence of early magnetic fields produced in heavy-ion collisions is expected to have an effect on the charm directed flow [243, 301], resulting in a v_1 value larger than that of lighter particles (short charm formation time and therefore sensitive to the maximum magnetic field strength), and opposite for particles with charm and anti-charm (due to the Lorentz force). Also the initial vorticity of non-central collision is expected to affect the directed flow observable. The STAR collaboration recently presented the first measurements of the directed flow v_1 coefficient for mesons. This first observation of non-zero v_1 for charm mesons, larger than that of lighter particles, is in qualitative agreement with theoretical models including both electromagnetic and vorticity effects. The uncertainties on the difference between the v_1 of D⁰ and \overline{D}^0 from STAR are still too large to draw conclusions on the effects of the early magnetic fields. The LHC Run 3 and 4 will enable more precise measurements on the charm directed flow, which will give additional insights into the initial vorticity of the Quark Gluon Plasma and the strength of the electromagnetic fields. Figure 24 (right) shows the precision level for the difference of directed flow v_1 for D⁰ and \overline{D}^0 which ALICE can measure as a function of pseudorapidity in semi-central Pb–Pb collisions at $\sqrt{s_{NN}} = 5.5$ TeV.

5.7 Heavy flavour measurements in small colliding systems

Traditionally, proton-nucleus collisions were considered just as a benchmark to investigate Cold-Nuclear-Matter (CNM) effects in the absence of a deconfined medium. As discussed in the devoted section, recent results suggest to partially reconsider such a paradigm. In particular, the study of soft observables in high-multiplicity p-Pb (and even pp) collisions led to the observation of signals traditionally attributed to the formation of a hot deconfined medium: long-range azimuthal 2-particle correlations [302–305], non-vanishing flow harmonics [306, 307], increasing baryon/meson ratio as a function of $p_{\rm T}$ [308], enhancement of strange-hadron production [309]. Many authors interpreted these observations as indications that the same strongly-interacting medium – with a collective hydrodynamic expansion driven by pressure gradients – supposed to be formed in nucleus-nucleus collisions is also produced in smaller systems. If the hydrodynamic scheme provides a consistent framework describing the above observations, its strong assumptions (e.g. a mean-free-path much smaller than the system size $\lambda_{mfp} \ll L$) look challenging to satisfy for such small systems. Hence, some authors looked for alternative interpretation of the data in terms, for instance, of initial-state effects (Color-Glass-Condensate [310]), of the formation of a system with small parton-parton cross-sections [311], of the presence of anisotropic parton escape mechanisms [312] or of quantum-mechanical interference in the presence of multiple sources of particle production, entailing to reconsider the no-interaction baseline before looking for final-state collective effects [313] .

On the other hand, no signature of jet-quenching or suppression of high- $p_{\rm T}$ particle production was



Fig. 24: Left: Time evolution in the forward rapidity region of the magnetic field B_y and electric field E_x in Pb–Pb collisions at $\sqrt{s_{NN}} = 2.76$ TeV with impact parameter b = 9.5 fm [243]. Right: ALICE projection of the difference of directed flow v_1 for D^0 and \overline{D}^0 as a function of pseudorapidity in semicentral Pb–Pb collisions at $\sqrt{s_{NN}} = 5.5$ TeV.

observed in high-multiplicity proton-nucleus collisions [314]. At first glance this appears in contradiction with the measurements involving soft observables. One should in any case consider that the quenching of jets due to parton energy-loss in the QGP has a strong dependence on the thickness of the crossed matter. On the contrary, if one accepts the hydrodynamic paradigm, the smaller size of the medium with respect to the nucleus-nucleus case would lead to even larger pressure gradients and hence to a larger acceleration of the fluid, compensating its shorter lifetime.

In light of the above findings, as an independent probe, it is clearly of interest to address the study of small systems also through heavy-flavour observables. Due to their large mass c and b quarks are in fact produced in the very first instants in hard processes described by pQCD, at most affected by nuclear modifications of the Parton Distribution Functions (nPDFs) and by an initial transverse-momentum broadening acquired in CNM. It looks then natural to extend transport calculations, developed to describe heavy-flavour propagation through the plasma formed in nucleus-nucleus collisions, also to the case of small systems, assuming as a working hypothesis that also in this case a hot deconfined medium is formed. The theoretical modelling involves the same processes as in the heavy-ion case: the so-called CNM effects (nPDFs and initial $k_{\rm T}$ -broadening) modifying the hard $Q\overline{Q}$ production, the propagation of the heavy quarks throughout the fireball and finally their hadronization in the presence of a hot medium. The only difference is that, in the case of small systems, it is mandatory to include event-by-event fluctuations in the initial conditions.

In order to illustrate the level of precision which future experimental measurements must reach to discriminate among different scenarios of initial and final-state effects, various sets of predictions based on the POWLANG model are reported in the following [316]. In Fig. 25 results of transport calculations for the nuclear modification factor of beauty hadrons in p–Pb collisions at $\sqrt{s_{NN}} = 5.02$ TeV are displayed and compared to experimental projections by CMS for the B-meson $R_{\rm pPb}$ obtained from Run 2 data and projections of LHCb at lower transverse momentum for the B-meson $R_{\rm pPb}$ [300] obtained using as central points the calculations based on Ref. [317]. As can be seen, model predictions are sensitive to the different choices of the transport coefficients (see discussion in [316]) and of the initial conditions (each nucleon-nucleon collision is assumed to deposit some entropy with a Gaussian smearing). In order to answer the question about possible final-state hot-medium effects, experimental measurements should



Fig. 25: Projection for the measurement of the nuclear modification factor of B mesons in p–Pb collisions achievable by CMS [291] and LHCb at $\sqrt{s_{NN}} = 5.02$ TeV and 8.8 TeV respectively. The predictions of the POWLANG model with different choices of the transport coefficients and of the smearing of the initial condition at $\sqrt{s_{NN}} = 5.02$ TeV are shown. Also reported, in magenta, the curve containing only Cold-Nuclear-Matter effects.

be extended to lower $p_{\rm T}$, as planned for the different experiments (see ALICE plans to perform beauty measurements down to low $p_{\rm T}$ in nucleus-nucleus collisions in Sec. 5.3.1). In fact, for $p_{\rm T}$ larger than 10– 15 GeV/c the curves accounting for the transport and hadronization of the heavy quarks in a hot medium (green curves) are very close to the (magenta) one which includes only the effect of the nPDF's and of the initial $k_{\rm T}$ -broadening acquired in CNM, all of them being very close to unity as the experimental projections by CMS. On the contrary, at lower $p_{\rm T}$ the radial flow of the beauty hadrons, acquired in part during the propagation through the hot medium and in part at hadronization via recombination, leads to a depletion of the spectrum at low $p_{\rm T}$ and to an enhancement at intermediate $p_{\rm T}$ which would allow one to distinguish this scenario from the case of pure CNM effects.

As displayed in Fig. 26 the possible production of a hot deconfined medium in proton-nucleus collisions would also leave its fingerprints in the azimuthal distribution of the final hadrons, leading in particular to a non-vanishing elliptic-flow coefficient. Notice how one gets a positive v_2 for charmed hadrons only starting from $p_T \approx 2 \text{ GeV}/c$, in agreement with recent CMS data [318]. Differently from the heavy-ion case, in such small systems it is important how the initial entropy deposition is modelled with varying the smearing parameter (left panel). As displayed in the right panel of Fig. 26, within the framework of the model, an important role is played by hadronization via recombination with light partons from the fireball, whose collective flow enhances the azimuthal asymmetry of the charmed hadrons.

The projected precision with p–Pb integrated luminosities $L_{int} \sim 2 \text{ pb}^{-1}$ (ATLAS and CMS) and $\sim 1 \text{ pb}^{-1}$ (ALICE) has the potential to shed light on the different mechanisms behind the observed anisotropy (see Fig. 27 for D mesons and prompt J/ ψ with CMS [315] and electrons from heavy-flavour hadron decays with ALICE).

Besides the analysis of kinematic distributions (in momentum and angle) of heavy-flavour particles, further interesting information on the onset of possible medium effects may come from the study of the yields of the various charm and beauty hadrons as a function of charged-particle multiplicity going from pp to p–Pb and Pb–Pb, as done in the past for the case of strangeness production. As an example, with high-multiplicity triggers in pp collisions at $\sqrt{s} = 14$ TeV ($L_{int} = 200 \text{ pb}^{-1}$), the ALICE exper-



Fig. 26: Left: The elliptic flow of charmed hadrons in the 0-40% most central p–Pb collisions at $\sqrt{s_{NN}} = 8.16$ TeV. Results of the POWLANG model with different choices of the transport coefficients and of the smearing of the initial condition are shown. Right: a comparison of the results at the level of charm quarks and hadrons. An important fraction of the flow is acquired at hadronization via recombination with light partons from the medium.



Fig. 27: Left: projection of prompt D-meson and prompt J/ ψ elliptic flow as a function of $p_{\rm T}$ in highmultiplicity p–Pb collisions at $\sqrt{s_{\rm NN}} = 8.16$ TeV with CMS [315]. Right: projection of elliptic flow of electrons from heavy-flavour hadron decays as a function of $p_{\rm T}$ in 0–20% central p–Pb collisions at $\sqrt{s_{\rm NN}} = 5.02$ TeV with ALICE.

iment will have the potential to detect D mesons with a precision better than 10% in a wide kinematic range up to about 10–12 times the average charged-particle multiplicity.

6 Jets and parton energy loss

Coordinators: Marta Verweij (Vanderbilt University and RIKEN BNL Research Center)

Contributors: L. Apolinario (LIP and IST Lisbon), R. Bi (Massachusetts Institute of Technology), Z. Citron (Ben-Gurion University of the Negev), Y. Chen (CERN), L. Cunqueiro Mendez (Oak Ridge National Laboratory), D. d'Enterria (CERN), P.M. Jacobs (Lawrence Berkeley National Laboratory), F. Krizek (Academy of Sciences, Prague), Y.-J. Lee (Massachusetts Institute of Technology), M. van Leeuwen (NIKHEF), C. Mcginn (Massachusetts Institute of Technology), G. Milhano (LIP and IST Lisbon, CERN), D.V. Perepelitsa (University of Colorado Boulder), M. Płoskoń (Lawrence Berkeley National Laboratory), M. Rybar (Columbia University), A.M. Sickles (U. Illinois, Urbana-Champaign), M. Spousta (Charles University), K. Tatar (Massachusetts Institute of Technology), J. Wang (Massachusetts Institute of Technology)

6.1 Introduction

The most direct way to measure the structure of matter is the controlled scattering of a beam of probe particles. This approach was used to discover the atomic nucleus, and quarks and gluons, and it is employed today to explore the partonic structure of nucleons and nuclei. However, the partonic phase of the QGP lives for $\sim 10^{-23}$ seconds before breaking up into its hadronic remnants, so that probing it by the scattering of an externally-generated beam is impossible in practical terms. As an alternative, energetic jets arising from high- Q^2 processes in the same nuclear collision that generates the QGP provide internally-generated probes that may be applied for this purpose [319–324].

Jets, observed as collimated sprays of energetic particles, were predicted by Quantum Chromo Dynamics (QCD) to form in high energy collisions. They constitute a substantial part of the background in beyond the Standard Model physics searches and were instrumental in the Higgs boson discovery. While jet evolution in vacuum is well understood, the question of how jets interact with a dense deconfined medium remains an active field of study, that is largely driven in the recent years by the unprecedented experimental capabilities of the RHIC and LHC accelerators and detectors. Understanding from first principles how a jet evolves as a multi-partonic system, spanning a large range of scales (from $\sim 1 \text{ GeV}$ to $\sim 1 \text{ TeV}$) is crucial to quantitatively probe the Quark Gluon Plasma (QGP). The successful description of bulk observables by viscous hydrodynamic calculations with a small viscosity to entropy ratio have led to the standard picture of a strongly coupled plasma. However, due to the property of asymptotic freedom in QCD, the produced matter is expected to behave differently at smaller and smaller distances which can only be accessed with well calibrated probes, namely, QCD jets.

High- Q^2 processes between the partonic constituents of colliding nucleons occur early in the collision. Further interactions of the outgoing partons with the hot and dense QCD medium produced in heavy ion collisions are expected to modify the angular and momentum distributions of final-state jet fragments relative to those in proton-proton collisions. This process, known as jet quenching, can be used to probe the properties of the QGP [319–324]. Jet quenching was first observed at RHIC, BNL [325–335] and then at the CERN LHC [336–346] by studying the redistribution of energy radiated from the parton because of interactions with the QGP. More recent detailed analyses have focused on modifications to the distribution of final-state particles emitted in the parton's shower [347–353].

One of the main goals of the RHIC and LHC heavy ion physics programs is utilization of jets and their decay products, including high p_T hadrons formed by light and heavy quarks, to investigate the QGP properties. A milestone in this program is the extraction of the transport coefficient \hat{q} by the JET Collaboration [354], based on inclusive hadron suppression measurements at RHIC and the LHC. However, this result has significant systematic uncertainties, due both to theoretical issues and to the limited view provided by inclusive hadron suppression measurements into the fundamental processes underlying jet quenching. A more complete picture requires measurements of reconstructed jets and their in-medium modification.

At the LHC, the collision energy is over an order of magnitude larger than at RHIC. Jet production

cross-sections are correspondingly larger, enabling the study of hard processes over a wider kinematic range. Detectors at both facilities have extensive capabilities to study fully-reconstructed jets by grouping the detected particles within a given angular region into a jet, thereby capturing a significant fraction of the parton shower. Jets are a key diagnostic of the QGP, as their interactions with this new state of matter reveal its properties. The interaction with the medium can result in a broadening of the jet profile with respect to vacuum fragmentation. In this case, for a given jet size and a fixed initial parton energy, the energy of the jet reconstructed in heavy ion collisions will be smaller than in vacuum. In the case where the gluons are radiated inside the cone, the jet is expected to have a softer fragmentation and a modified density profile compared to jets in vacuum. Jets may also scatter coherently in the medium, and measurements of jet deflection may provide a direct probe of the micro-structure of the QGP. Fully reconstructed jets provide better theoretical control than high $p_{\rm T}$ hadrons because they are less sensitive to non-perturbative physics and therefore have the potential to provide a better characterization of the QGP. Furthermore, major theoretical and experimental advances were made recently in understanding the evolution of parton showers in a QCD medium with the development of novel jet substructure observables.

In the following sections the expected performance using a total integrated luminosity of 10 nb^{-1} of Pb–Pb data, which is expected for HL-LHC, for a selection of key jet observables will be discussed.

6.2 Out-of-cone radiation

One of the classic observables to measure the out-of-cone radiation due to jet quenching is the jet nuclear modification factor R_{AA} defined as:

$$R_{\rm AA} = \frac{\frac{1}{N_{\rm evt}} \frac{{\rm d}^2 N_{\rm jet}}{{\rm d}p_{\rm T} {\rm d}y}\Big|_{\rm cent}}{T_{\rm AA} \frac{{\rm d}^2 \sigma_{\rm jet}}{{\rm d}p_{\rm T} {\rm d}y}\Big|_{pp}},$$
(19)

where $N_{\rm iet}$ and $\sigma_{\rm iet}$ are the jet yield in Pb-Pb collisions and the jet production cross-section in pp collisions, respectively, both measured as a function of transverse momentum, $p_{\rm T}$, and rapidity, y, and where N_{evt} is the total number of Pb–Pb collisions within a chosen centrality interval. Measurements of the jet $R_{\rm AA}$ at the LHC have shown a suppression of a factor of two in central collisions over a wide range of jet transverse momentum [342, 343, 355]. Figure 28 shows the current precision obtained with 0.5 nb^{-1} and what can be achieved at the HL-LHC with a factor of 20 more data (10 nb^{-1}). Especially at high transverse momentum a strong reduction of the experimental uncertainties is expected, which will allow a detailed study of the momentum dependence of the out-of-cone radiation. The jet R_{AA} is sensitive to various physics mechanisms such as color coherence, flavor dependence of energy loss, and the medium response to the jet. Models incorporating these various physics effects can be confronted with the high precision data from HL-LHC with a goal of determining what the relative contribution of each of these phenomena is. The expected performance is compared with several recent model predictions: the Linear Boltzmann Transport model (LBT) [356], three calculations using Soft Collinear Effective Theory (SCET) [357-360], and the Effective Quenching model (EQ) [361]. The higher precision data will allow tighter constraints on or falsification of theoretical model predictions. In addition to the inclusive jet R_{AA} it is particularly interesting to study the mid- and forward rapidity region separately since it allows to study the interplay between flavor and spectral steepness, and the path-length dependence of jet quenching. The right panel of Fig. 28 shows the improvement in statistical precision in the forward rapidity region. The statistical precision should be sufficient to quantitatively assess the rapidity dependence of the R_{AA} up to a rapidity of |y| = 2.8. Both of these predictions indicate that HL-LHC should bring a definitive understanding of the intriguing features of the jet $R_{\rm AA}$ as seen in the current data.



Fig. 28: Projection of the precision that can be reached for jet R_{AA} at the HL-LHC using calorimeter jets at ATLAS as function of p_{T} (left panel) and rapidity (right panel) [362]. See text for model details.

Parton energy loss can be studied more differentially using boson tagged jets. The bosons (photons or Z^0 bosons) escape the region of the hot dense medium unmodified. This has been confirmed through the absence of significant modification of both photon and Z^0 boson production rates in Pb–Pb collisions relative to expectations from measured cross section in pp collisions by both ATLAS and CMS collaborations [363–366]. However, the partons recoiling from the boson is modified in heavy ion collisions due to interactions with the QCD medium. Furthermore, jets produced opposite to the isolated boson are more likely to originate from quarks, while dijet and hadron+jet correlations usually involve significant gluon contributions. Comparing Z+jet and γ +jet observables to dijets [341, 367] (or hadron+jets [346]) allows to explore the difference between energy loss for quark and gluon initiated jets. Figures 29 and 30 show the expected performance at the HL-LHC for the transverse momentum balance between the jet and the boson. The central values are based on the smoothed data from the previous CMS publications [368, 369]. The systematic uncertainties are reduced by a factor of two with respect to the results with the 2015 Pb-Pb data due to improvements on the jet energy scale and jet energy resolution uncertainties available with the larger data sample at the HL-LHC. The collected number of γ +jet events will also be sufficient to study the path length dependence of jet quenching by performing measurements as a function of angle with the reaction plane for the first time. In addition to the smaller uncertainties due to the enhanced statistics at the HL-LHC, it will also be possible to utilize higher momentum photons and Z^0 bosons allowing the measurement of larger jet energy losses. The LHC experiments also envision extending the jet momentum reach to lower transverse momentum in certain analyses, allowing to recover those heavily quenched jets which are currently not selected for such measurements due to limitation arising from the fluctuating background. A distinct effect due to large backgrounds is that of limited jet energy resolution, which can be improved by using more sophisticated techniques for the background correction as was recently shown in Ref. [370].

6.3 Jet deflection

Angular deflection of the jet relative to its initial direction due to momentum transfer with the medium can be used as a direct probe of the QGP. Jet deflection can be measured by coincidence observables, in which an axis is determined by a hard reference object, and the deflection of the jet recoiling from hard object is measured relative to that axis. Such scattering measurements, carried out over a wide range in energy and resolution scale, can be used to explore the microscopic structure of the QGP. Modification of the rate of rare, large-angle jets with respect to the hard reference object in nuclear collisions compared to the production rate in vacuum may arise from the scattering off quasi-particles (quarks and gluons or composite objects) of the QGP, thereby probing their nature [371]. In addition, the recoil jet distribution at small recoil angles relative to the reference axis (the axis of the hard object selected at the opposite



Fig. 29: (Left Panel) Photon-jet momentum balance $x_{j\gamma}$ distribution for isolated-photon+jets of $p_{\gamma} > 100$ GeV/c and $|\eta_{\gamma}| < 1.44$, $p_{jet} > 30$ GeV/c and $|\eta_{jet}| < 1.6$ in the HL-LHC data (Right Panel). Comparison between the current performance with 0.4 nb⁻¹ of Pb–Pb data collected in 2015 and with HL-LHC data [259].



Fig. 30: (Left Panel) X_{jZ} distribution for Z boson-jet pairs with $p_T^Z > 60$ GeV/c, $p_{jet} > 30$ GeV/c and $|\eta_{jet}| < 1.6$ in the HL-LHC data (Right Panel) Comparison between the current performance with 0.4 nb⁻¹ of Pb–Pb data collected in 2015 and with HL-LHC data [259].

hemisphere) may be modified by soft multiple scattering in the QGP, which can be used to extract the jet transport parameter \hat{q} by comparison to models [372].

Measurements of the angular distribution of jets relative to a reference axis have been reported for either dijet, photon-jet, Z^0 -jet and hadron-jet coincidences, at RHIC [335] and LHC [341, 346, 368, 369, 373]. These current measurements exhibit no significant evidence of in-medium modification of angular distributions, both at small and large angles to the reference axis. While they impose constraints on the magnitude of in-medium scattering effects, their statistical precision is limited. Measurements during HL-LHC will either discover in-medium modification to the recoil jet angular distributions, or improve

these constraints substantially.

There is an intensive ongoing effort to develop analysis tools and calculable approaches that discriminate the various contributions to in-medium jet deflection and shower modification, in both experiment and theory [374] (see Sec. 6.2 and 6.4). In this section we focus solely on jet-centroid deflection measurements, without consideration of the effects of shower broadening or other shower modification (see Sec. 6.4). Measurements of both classes of jet quenching observable must ultimately be interpretable in a single consistent picture, but such an approach is beyond current experimental and theoretical capabilities.

The most significant background to the measurement of medium-induced jet deflection is broadening of the angular difference between the two leading jets due to well-established vacuum QCD effects, in particular Sudakov radiation, which is radiation outside the jet cone that generates a broad peak in the recoil jet angular distribution relative to a reference axis (for example a high momentum hadron) [372,375].

Low-energy jet measurements are expected to experience larger deflection for a given momentum transfer between the jet and medium [376, 377] and are therefore more likely to show large angle deflection. A recent calculation, that includes the effects of vacuum Sudakov radiation and jet-medium interactions based on the few-hard (GLV) or multiple-soft (BDMPS) scattering approaches to jet quenching, finds that the acoplanarity distributions for these different jet quenching pictures differ by a few percent in the range $20 < p_{T,jet} < 40$ GeV/c [377]. This sets the precision required for the observation of medium-induced jet deflection during HL-LHC. Additional theoretical considerations of in-medium p_{T} -broadening can be found in [322, 378].

In light of such considerations, it is necessary to utilize analysis techniques that can attain few percent precision in the measurement of recoil jet angular distributions for low $p_{T,jet}$ and large jet radius R, over the large and complex uncorrelated backgrounds in central Pb–Pb collisions at the LHC. This precision is achievable using the statistical approach to jet background correction [335, 346, 368, 369], in which the discrimination of correlated and uncorrelated recoil jet yield is carried out in a fully data-driven way, at the level of ensemble-averaged distributions. The statistical correction approach has been used to measure the azimuthal distribution for charged jets with R = 0.5 and $40 < p_{T,jet}^{ch} < 60$ GeV/c recoiling from a high- p_T hadron in central Pb–Pb collisions at the LHC [346], and for charged jets with R = 0.5 and $p_{T,jet}^{ch} \sim 10$ GeV/c in central Au–Au collisions at RHIC [335], as well as for photon-jet and Z-jet correlations [368, 369]. We expect that reaching as low as $p_{T,jet} = 10$ GeV/c is likewise achievable at the LHC, with good systematic precision.

The required experimental approach is therefore in hand, and we explore here the statistical precision achievable using it for such measurements during HL-LHC. We utilize the JEWEL event generator [379] for these projections, which incorporates medium-induced interactions of partons propagating in the QGP. Calculations are carried out for central Pb–Pb collisions at $\sqrt{s_{NN}} = 5.02$ TeV with integrated luminosity of 10 nb⁻¹, and pp collisions at $\sqrt{s} = 5.02$ TeV with integrated luminosity of 6 pb⁻¹. The JEWEL calculations for central Pb–Pb collisions are carried out with the "Recoil off" configuration in which the partons from the medium response are neglected.

Figure 31 shows the recoil jet azimuthal angle, $\Delta \varphi$, defined with respect to the reference axis [346] as simulated by the JEWEL event generator. The background-corrected azimuthal distribution of recoil jets recoiling from a high- $p_{\rm T}$ hadron, with the statistics expected by ALICE for central Pb–Pb and pp collisions during the HL-LHC phase is shown. The distribution for central Pb–Pb collisions exhibits an overall yield suppression, corresponding to jet quenching, but also a slight narrowing of the main peak at $\Delta \varphi \sim \pi$ and an enhancement at large deflection angle. The narrowing is characterized by extracting the width of the $\Delta \varphi$ distribution which is 0.204 ± 0.005 in the pp simulation and 0.163 ± 0.001 for the Pb–Pb simulation with JEWEL In order to quantify the difference at large recoil jet deflection angle between pp



Fig. 31: JEWEL simulation of the angular distribution of charged jet yield in the ALICE acceptance for $40 < p_{T,jet}^{ch} < 45$ GeV/c and R = 0.4 recoiling from a high- p_T reference hadron ($20 < p_{T,trig} < 50$ GeV/c), for central Pb–Pb collisions at $\sqrt{s_{NN}} = 5.02$ TeV with 10 nb⁻¹ int. luminosity, and pp collisions at $\sqrt{s} = 5.02$ TeV with 6 pb⁻¹ int. luminosity. The recoil jet azimuthal angle $\Delta \varphi$ is defined with respect to the reference axis. The observable shown is $\Phi(\Delta \varphi)$ which incorporates statistical suppression of uncorrelated background.

and central Pb–Pb collisions, we integrate the $\Phi(\Delta \varphi)$ from $\pi/2$ to a threshold angle $\Delta \varphi_{\text{thresh}}$ [346],

$$\Sigma(\Delta\varphi_{\rm thresh}) = \int_{\pi/2}^{\pi-\Delta\varphi_{\rm thresh}} \Phi(\Delta\varphi) d\Delta\varphi.$$
⁽²⁰⁾

Figure 32 shows $\Sigma(\Delta \varphi_{\text{thresh}})$ for the $\Phi(\Delta \varphi)$ distributions in Fig. 31, together with their ratio. In this calculation, the value of Σ at $\Delta \varphi_{\text{thresh}} = 0$ is around 0.5, which is the yield suppression averaged over the full recoil hemisphere. The ratio grows to $\Sigma \sim 1$ at $\Delta \varphi_{\text{thresh}} = 1.2$, indicating a factor two enhancement in large-angle yield relative to the hemisphere average. The statistics of the measurement are clearly sufficient to measure the effect predicted by this calculation. However, the calculation in [377] predicts a difference of only a few percent in these distributions for GLV-like and BDMPS-like in-medium scattering, which is more difficult to discriminate. The statistical error in the ratio in Fig. 32 is around 5% at $\Delta \varphi_{\text{thresh}} \sim 1$, due predominantly to the statistical precision of the pp distribution.



Fig. 32: Cumulative large-angle yield $\Sigma(\Delta \varphi_{\text{thresh}})$ (Eq. 20) vs. $\Delta \varphi_{\text{thresh}}$ for the pp and central Pb–Pb distributions $\Phi(\Delta \varphi)$ in Fig. 31. See text for details.

6.4 Jet internal structure

The first measurements of jet quenching through full jet reconstruction at the LHC revolutionized our understanding of parton energy loss in a hot and dense medium. Nevertheless, there remains a gap in our understanding of the jet quenching mechanism that could be resolved by measuring the exact properties of the parton evolution through the medium. High statistics of collected jets during HL-LHC will provide a prime opportunity to explore the details of the internal structure of high energy jets that undergo interactions with the QGP. Observables probing the internal structure of jets can be defined using all measured hadrons in a jet or by using subjet techniques selecting only a specific region of the radiation phase space. In the following sections both approaches and their potential will be discussed.

6.4.1 Substructure with hadrons

Inclusive measurements of the longitudinal and transverse momentum distribution of hadrons in inclusive jets have been performed with high accuracy at the LHC [380, 381]. The modification due to jet quenching is studied by comparing the results in pp and Pb–Pb collisions. For certain kinematic selections, for example at large z where the leading particle in the jet is carrying a large fraction of the total jet momentum, the current experimental uncertainties are however large (see left panel of Fig. 33) limiting the constraints on the jet quenching mechanism that can be extracted by comparing data to theoretical models. The expected statistical precision at HL-LHC for the ratio of fragmentation functions in Pb–Pb and pp collisions is shown in Fig. 33. This precision will allow detailed characterization of the excess in yield of hard (large z) and soft (small z) fragments and the suppression in the region between these two excesses providing strong constraints to theoretical models. Measurements of the rapidity dependence of jet observables are of great interest since the fraction of quark- and gluon-initiated jets varies with rapidity. However, current measurements of the fragmentation function are statistics limited and no significant rapidity dependence is observed [380]. The right panel of Fig. 33 shows the ratio of fragmentation functions of high momentum jets for most central collisions with the expected accuracy at the HL-LHC. The projection are compared to the hybrid model [382, 383] which implements energy loss according to the strong coupling description of the radiation of low energy gluons associated with the hot QCD matter which predicts a rapidity-dependent suppression of particle yield at high z.



Fig. 33: Projection of the precision that can be reached for the modification of jet fragmentation function, $R_{D(z)}$, measured in jet $p_{\rm T}$ interval 200 – 251 GeV/c. In the left panel the statistical uncertainty on the measurement with the shaded boxes corresponding to 0.49 nb⁻¹ while the vertical bars are for 10 nb⁻¹. The right panel shows a comparison of $R_{D(z)}$ with a theory model (see text for more details) [362].

When interpreting the modification of inclusive jets one has to realize that by requiring a certain jet momentum range a different sample of partons initiating the jet is selected in pp and Pb–Pb collisions. Incorporation of this effect in model calculations introduces an additional uncertainty limiting the constraints that can be put on a model. This can be overcome by using the rare process of jets recoiling from photons. The expected performance of the radial $p_{\rm T}$ profile in jets recoiling from a high momentum photon at HL-LHC is shown in Fig.34. The central values of the extrapolated spectra are obtained by smoothing the results from [384] by a third order polynomial. The systematic uncertainties shown are obtained by reducing by a factor of two those from the 2015 Pb–Pb data results, considering the expected improvements on the jet energy scale and jet energy resolution uncertainties. The results show that the photon-tagged jet shape could be measured with high precision providing insights about the modification of the jet transverse structure of quark initiated jets in the strongly interacting medium. Figure 35 shows the expected statistical precision of the fragmentation function on photon-tagged events. The larger data sample will enable the measurement for finer centrality selections with respect to the current preliminary results [385] allowing an exploration of the temperature and path length dependence of jet quenching.

6.4.2 Substructure with subjets

Early hard splittings in a parton shower may result in two partons with high transverse momentum that are well separated in angle. Information about these leading partonic components can be obtained by removing the softer wide-angle radiation contributions. This is done through the use of a jet grooming algorithm called "soft drop", an extension of the modified mass drop tagger (mMDT), that attempt to split a single jet into two subjets, a process referred to as "declustering" [386–390]. For a parton shower in vacuum, these subjets provide access to the properties of the first splitting in the parton evolution [391, 392]. Figure 36 shows the expected performance for the momentum sharing fraction, z_g [392], in the HL-LHC phase. The central values of z_g and jet mass are from previous CMS publications [352, 353]. The systematic uncertainties are reduced by a factor of two with respect to the results with 2015 Pb–Pb data due to the expected improvements on the jet energy scale and jet energy resolution uncertainties. While the current data is not precise enough to constrain the medium properties further, the expected luminosity at the HL-LHC will allow more detailed constraints as can be observed from the different results of the



Fig. 34: (Left Panel:) The ratio of measured photon-tagged jet shape in Pb–Pb and pp collisions with the 2015 Pb–Pb data [384]. (Right Panel:) The expected performance of the jet shape ratio in the HL-LHC data, using a third-order polynomial for smoothing the data. [REF to be added when note is public]



Fig. 35: Projection of the statistical precision that can be reached for the ratio of jet fragmentation functions in Pb–Pb and pp collisions, $R_{D(z)}$, of jets recoiling from a photon. The left panel shows the projection for the most central collisions while the right panel for the more peripheral events [362].

BDMPS [393] and SCETg [394] calculations when the medium density (\hat{q} for BDMPS and g for SCETg) is varied. In addition, the expected precision will also provide the ability to distinguish different physical mechanisms and scales relevant for jet quenching as is shown for the role of coherence in Fig. 36 in the HT theoretical calculations [395]. A measurement of the groomed jet mass with the 2015 LHC Pb–Pb data already showed that jet quenching might cause an increase of high mass jets [353]. Figure 37 shows the expected performance for the groomed jet mass at HL-LHC which will allow measuring the high mass region with higher precision.


Fig. 36: Performance of jet splitting function measurement with HL-LHC data in Pb–Pb collisions for two different selections in jet transverse momentum. [259]

6.4.3 Radiation phase space with Lund diagram

Recently, a theoretical representation of the radiation phase space within jets inspired by Lund diagrams [396] has been proposed [374] to study medium modification of the radiation pattern. The socalled Lund jet plane [397] - a portrayal of the internal structure of jets - was designed to build a conceptual connection between manually constructed observables and approaches that use Machine Learning techniques to study QCD jets and/or discriminate between signal and background jets. The diagram is constructed by mapping the available phase-space within a jet to a triangle in a two dimensional (logarithmic) plane that shows the transverse momentum and the angle of any given emission with respect to its emitter. Such a triangular diagram, a representation of the radiation within any given jet, can be created through repeated Cambridge/Aachen declustering.

To demonstrate the potential of future measurements at the LHC we constructed Lund diagrams using the JEWEL Monte Carlo event generator [379]. To study the differences in the Lund diagram due to medium effects the results are compared to a vacuum reference (jets produced in pp collisions). For the simulations the JEWEL generator with the default settings is used without the optional calculation of the so-called medium response retaining the partons / scattering centres that interacted with the jet was not used (i.e. *recoils off* setting of the MC generator was used). The substructure of jets was analysed by reclustering the constituents of the jet with the Cambridge/Aachen (C/A) algorithm as implemented in the FASTJET package [398, 399] for two selections of jet $p_{\rm T}$ 80–120 GeV/c and 200–250 GeV/c.

The Lund diagram density can be constructed experimentally and compared to analytic predictions and parton-shower Monte-Carlo simulations, such as JEWEL. For this purpose a density map of points (emissions) is defined following formulations in [397]:

$$\bar{\rho}(\Delta,\kappa) = \frac{1}{N_{\text{jet}}} \frac{\mathrm{d}n_{\text{emission}}}{\mathrm{d}\ln\kappa\,\mathrm{d}\ln 1/\Delta},\tag{21}$$

where for two clusters 1 and 2 labeled such that $p_{T,1} > p_{T,2}$, $\Delta^2 = (y_1 - y_2)^2 + (\varphi_1 - \varphi_2)^2$ with φ being the azimuthal angle and y the rapidity of a cluster, and $\kappa = \frac{p_{T,2}}{p_{T,1} + p_{T,2}}\Delta$. Figure 38 shows the density $\bar{\rho}$ from the JEWEL simulation without (left panels) and with (right panels) medium effects. The z_g variable which was defined in [392] and studied in heavy-ion collisions [352] is related to the variables in the Lund plane in the following way: $z_g = \kappa/\Delta$ from the first of the entries (1...i...N) in the primary declustering sequence that satisfies $z^{(i)} \geq z_{cut}(\Delta^{(i)})^{\beta}$ [397] resulting in diagonal lines with negative



Fig. 37: Jet mass distribution with grooming setting $(z_{cut}, \beta) = (0.1, 0.0)$ (Upper panels) and $(z_{cut}, \beta) = (0.5, 1.5)$ (Lower panels). [259]

slope in the Lund diagram for a constant value of z_{g} .

The effect of jet quenching on the Lund diagram is quantified by taking the difference between the diagram with and without medium effects as shown in Fig. 39 for the two transverse momentum ranges considered in this study. The average density integrated over $\ln \kappa$ calculated for Pb–Pb (MEDIUM) case shows little deviation from the pp (VACUUM) reference. The most pronounced differences between VACUUM and MEDIUM calculations are visible for the region of $-3 < \ln \kappa < -3$ and large $\ln 1\Delta$ which correspond to the hard-collinear splittings (Region-A), and a band along $\ln 1/\Delta$ for small $\ln \kappa$ (Region-B): $-5 < \ln \kappa < -6$ for the lower $p_{\rm T}$ selection and $-5.5 < \ln \kappa < -7$ for higher $p_{\rm T}$ jets; that corresponds to an enhancement of soft (moderate $\ln 1/\Delta$) and hard collinear splittings (large $\ln 1/\Delta$). These observations are consistent with soft and hard collinear splittings being modified by the medium.

To illustrate the different modifications of the Lund diagram density for the two regions identified in Fig. 39, projections along $\ln 1/\Delta$ are shown in Fig. 40. For Region-A we observe 30%-40% depletion of splittings for the MEDIUM case whereas in Region-B a moderate increase of splittings induced by the medium is visible. The depletion in Region-A is consistent a sample of more collimated jets consistent with previous measurements in heavy-ion collisions [351, 381]. The increase seen in Region-B is consistent with a small in-medium enhancement of splittings with moderate dependency on the angle of the



Fig. 38: The density of points of a Lund diagram for anti- $k_T R = 0.4$ jets for two p_T selections: 80 < p_T < 120 GeV/c in the upper row and 200 < p_T < 250 GeV/c in the lower row. Result of the JEWEL Monte Carlo generator with left column: jets in pp collisions; Right column: jets from Pb–Pb collisions - some with in-medium modifications. Each of the density plots shows curves of the average quantities of the densities over the other axis.

splitting but favoring the soft collinear medium-induced radiation (moderate $ln1/\Delta$).

As discussed in Ref. [374] specific regions in the Lund plane are sensitive to different type of parton splittings. These regions can be identified by selecting the desired area using linear functions $\ln \kappa = \ln 1/\Delta + \ln \frac{1}{p_T t}$, where t is related to the decoherence time (thus formation time). Depending on the selection, different formation times are probed and splittings will occur within or outside the medium. Several arbitrary regions selected by the diagonal lines for constant $\frac{1}{p_{T}t}$ are indicated in Fig. 38. To illustrate the in-medium effects and their dependence on the jet momentum, the $p_{\rm T}$ and the formation (decoherence) time the density of the splittings can be projected along the momentum imbalance z = $p_{T,2}/(p_{T,1}+p_{T,2})$. In the left panel of Fig. 41 we show the relative difference of the splitting density $\Delta \bar{\rho} = (\bar{\rho}_{med} - \bar{\rho}_{vac})/\bar{\rho}_{vac}$ for a selection of $p_T t$. For small $p_T t$ the splitting density is suppressed for the in-medium calculations whereas for large $p_{T}t$ the modification is smaller. This is consistent with the expectation that for large formation times the medium effects should be of smaller magnitude as compared to splittings formed early. As the suppression seen in $\Delta \bar{\rho}$ selected on $p_{\rm T}t$ depends on the jet p_{T} we find similar suppression for high- and low- p_{T} jet selection for substantially different product of the $p_{\rm T}$ and formation time. In particular, for low momentum jets the modifications of $\Delta \bar{\rho}$ for large $p_T t$ is small. To further exploit the formula providing the approximate formation time dependence of the splittings we select two regions of the Lund diagram: "late" t > 10 and "early" t < 10. The "early" region should be dominated by splittings that form within the medium, whereas the "late" splittings are to be dominated by the shower evolution outside the medium of length $L \sim t$. In the right panel of Fig. 41 we present $\Delta \bar{\rho}$ for two selections of decoherence time. As expected, a small dependence on jet $p_{\rm T}$ for "late" and "early" splittings is seen - a similar suppression for "early" splittings independent of



Fig. 39: Result of the JEWEL+PYTHIA MC simulation: MEDIUM-VACUUM difference of the calculations shown in Fig. 38 for two jet p_{T} selections.



Fig. 40: Projections of the lund diagram along the angular separation $\ln 1/\Delta$ of the splittings for the two selections of jet $p_{\rm T}$. In-medium suppression of splittings for moderate $\ln \kappa$ according to JEWEL (left). Enhancement for small $\ln \kappa$ (right).

jet p_T and almost identical $\Delta \bar{\rho}$, with small deviations from unity, for "late" emmissions. The residual differences could be attributed to different fractions of the splittings resolved by the medium, and likely, different impact of non-perturbative effects (such as hadronization) for the two jet p_T selections.

6.5 Opportunities for jet quenching studies with light-ion collisions

The ability of the LHC to collide ions lighter than Pb as discussed in section 2.4 provides an opportunity to enhance the heavy-ion programme with a very large number of rare probes as summarized in Sect. 11.1. Ar–Ar collisions are used as a test case for light ion running, although the optimal choice of ion is still under study. It is clear that due to the larger integrated luminosity obtainable for a given heavyion running period the number of jets produced in Ar–Ar collisions will be significantly larger than in Pb–Pb collisions. The question which will determine the value of the of light-ion running to the study of jets and parton energy loss is to what extent jet suppression effects will be reduced in smaller systems. To address this question, projections for Ar–Ar and Xe–Xe collisions at the LHC are considered using the JEWEL Monte Carlo event generator [379].

The jet nuclear modification factor R_{AA} (discussed in Sect. 6.2) in Ar–Ar collisions is here computed as the ratio of the jet transverse momentum spectrum in medium (Ar–Ar) over that in vacuum (pp)



Fig. 41: Projections of the relative difference of the Lund diagram onto momentum imbalance of the splittings for two selections of jet $p_{\rm T}$. Left: selection of splittings for different $\frac{1}{p_T t}$. Right: selection of splittings for different t.

$$R_{\rm AA}^{\rm jet} = \frac{\left(dN_{\rm jets}/dp_{\rm T}\right)^{\rm med}}{\left(dN_{\rm iets}/dp_{\rm T}\right)^{\rm vac}}.$$
(22)

Table 6 summarizes the parameters used in JEWEL to calculate this quantity in 0–10% centrality Ar–Ar collisions, compared with the parameterization for Pb–Pb and Xe–Xe. The T_i^{v1} values were obtained assuming the same pre-factors as in [400]:

$$T(t) = \left(\epsilon(t)\frac{30}{47.5\,\pi^2}\right)^{1/4}\,,\tag{23}$$

where the energy density $\epsilon(t)$ follows a Bjorken evolution:

$$\epsilon(t) = \frac{1}{\pi R_{nucl}^2 t} \frac{\mathrm{d}E}{\mathrm{d}\eta} \,. \tag{24}$$

The energy per unit of pseudo-rapidity is taken from centrality-dependent measurements in Pb–Pb collisions. Finally the temperature T_i^{v1} is evaluated at τ_i .

Table 6: Energy and medium parameter used in JEWEL simulation of dijets and Z+jet events.

	Pb–Pb	Xe–Xe	Ar–Ar
$\sqrt{s_{\rm NN}}$ (TeV)	5.02	5.80	6.30
$\langle N_{\rm part} \rangle$	353	210	66
$R_{\rm nucl}$ (fm)	6.6	5.4	3.6
$\tau_i (\text{fm/}c)$	0.6	0.57	0.63
T_i^{v1} (MeV)	360	350	318
T_i^{v2} (MeV)	260	250	218

With these medium parameters, JEWEL results lie quite below the ATLAS R_{AA} results for the Pb–Pb 0–10% centrality class [401]. JEWEL was run with medium recoil effects off, although they are known to contribute to increase the jet R_{AA} by ~ 0.1 –0.2 in the most central events [402]. This may explain the discrepancy. Alternatively, the discrepancy can be eliminated by reducing the temperature in JEWEL. Starting from the Pb–Pb temperature changed in order to match the ATLAS results for the



Fig. 42: Jet R_{AA} obtained from a JEWEL simulations using the medium parameters listed in Table 6, with the temperatures listed as T_i^{v2} . On the left, the jet R_{AA} is shown as a function of the jet p_T , and on the right as a function of $\langle N_{part} \rangle$ for a jet $100 < p_T < 126 \text{ GeV/}c$.

Pb–Pb 0–10% centrality class, temperatures for Xe–Xe and Ar–Ar are obtained by assuming that the energy density scales with $A^{1/3}$. Thus, for an arbitrary collision system XX one has:

$$\left(\frac{T_{XX}}{T_{\rm Pb-Pb}}\right)^4 = \left(\frac{A_{XX}}{A_{\rm Pb}}\right)^{1/3}.$$
(25)

This parameterisation is used to obtain the T_i^{v2} values listed in Table 6 which are used to calculate the jet R_{AA} shown in Figure 42. The figure shows the JEWEL calculations for Pb–Pb, Xe–Xe, and Ar–Ar along with ATLAS Pb–Pb measurements in centrality classes chosen to match the $\langle N_{part} \rangle$ values in Table 6.

To further investigate jet energy loss in Ar–Ar collisions, Z boson + jet events are studied within the same JEWEL framework. The importance of boson + jet events for the precision study of energy loss is discussed in Sect. 6.2. Events with a Z boson decaying into $\mu^+\mu^-$ associated with a jet were simulated with JEWEL + PYTHIA Monte Carlo [403]. Events were selected for a reconstructed Z boson with a mass within 70–110 GeV, a minimum $p_{\rm T}$ of 10 GeV/c for its decay muons, and an associated jet with a $p_{\rm T} > 30$ GeV/c and a $|\delta \varphi| > 7\pi/8$ with respect to the boson momentum direction. The resulting energy asymmetry distributions, $x_{\rm jZ} = p_{\rm T}^{\rm jet}/p_{\rm T}^{\rm Z}$, normalized to the number of reconstructed Z bosons are shown in Figure 43.The distribution for central Ar–Ar collisions is similar to those for central and semi-central Pb–Pb collisions. The effect of jet quenching is very significant, as apparent in the comparison with the distribution in pp collisions.



Fig. 43: Boson-Jet energy asymmetry, x_{jZ} obtained from JEWEL simulations using the medium parameters listed in table 6, with the temperatures listed as T_i^{v2} .

Taken together, these studies suggest that JEWEL does somewhat over-predict suppression in smaller systems, e.g. Xe–Xe. However, considering that central collisions are reproduced well by this model and that allowing as an upper limit the suppression measured in the Pb–Pb 40–50% centrality class, a significant suppression is expected in central Ar–Ar collisions. The expected suppression combined with the much larger nucleon–nucleon integrated luminosity (e.g. 8–25 larger with Ar–Ar than with Pb–Pb collisions) makes lighter ion collisions at the LHC an attractive possibility for the study of parton energy loss.

7 Quarkonia

Coordinators: Anton Andronic (Münster University) and Emilien Chapon (CERN)

Contributors: E. G. Ferreiro (Instituto Galego de Fisica de Altas Enerxias (IGFAE) Universidade de Santiago de Compostela), J.-P. Lansberg (Institut de Physique Nucléaire d'Orsay), R. Rapp (Texas A&M University, College Station), J. Castillo Castellanos (IRFU/DPhN, CEA Saclay), C. Cheshkov (IPN Lyon), J. Martin Blanco (Laboratoire Leprince Ringuet), J. Park (Korea University), X. Du (Texas A&M University, College Station), M. Strickland (Kent State University), R. Venugopalan (BNL), I. Vitev (Los Alamos National Laboratory)

7.1 Introduction

A key objective in high-energy heavy-ion physics is to determine the in-medium forces that give rise to the remarkable many-body features of the QGP. In the QCD vacuum, the unravelling of the fundamental force between two static Color charges was made possible by the discovery of the charmonium and bot-tomonium states in the 1970's. Subsequent quantitative analyses of the bound-state spectra established a phenomenological potential of the Cornell type [404],

$$V(r) = -\frac{4}{3}\frac{\alpha_s}{r} + \sigma r , \qquad (26)$$

with a colour-Coulomb term due to gluon exchange dominant at short distances, and a linear term with string tension $\sigma \simeq 0.9$ GeV/fm to account for confinement at large distance. This potential has also been quantitatively confirmed by lattice-QCD (lQCD) calculations [405, 406]. The corresponding effective field theory of QCD, potential non-relativistic QCD (pNRQCD), allows for the definition of the static potential [407] in a $1/m_Q$ expansion for large heavy-quark mass, m_Q [408,409]. The heavy-quark (HQ) potential thus provides a well calibrated starting point to probe the QCD medium, and the in-medium spectroscopy of quarkonia is the natural tool to carry this out in heavy-ion collisions, cf. [410–414] for recent reviews. The string term in the HQ potential, eq. (26), characterises the long-range nonperturbative part of the force and is associated with the confining property of QCD. It is expected to play a critical role in the transition from hadronic to partonic degrees of freedom, and may be responsible for the remarkable transport properties of the QGP, *i.e.*, its strongly coupled nature, up to temperatures of 2-3 times the (pseudo-)critical temperature, T_c [415].

Much like in vacuum, a systematic investigation of the in-medium force must involve the spectroscopy of different states, as they subsequently dissolve with increasing temperature. The complexity in describing the in-medium properties of quarkonia and their implementation into transport calculations in heavy-ion collisions prevents their use as a straightforward thermometer of the medium produced in these reactions. On the contrary, using information on the space-time and temperature evolution in heavy-ion collisions from other sources (e.g.,, hydrodynamics and electromagnetic radiation), on can utilize quarkonium observables to deduce their in-medium properties and infer the fundamental interactions in QCD matter. In the vacuum, only the 1S ground-state bottomonia (Y(1S) and η_b) are small enough in size to be mostly bound by the colour-Coulomb force. All excited bottomonia and all charmonia are predominantly bound by the nonperturbative string term (and/or residual mesonic forces). Thus, charmonia and excited bottomonia are excellent probes of the in-medium confining force, as originally envisioned for the J/ ψ [416]. However, in the cooling of the expanding fireball, quarkonia can also be "(re)generated" through recombination of individual heavy quarks and anti-quarks diffusing through the medium. It is important to emphasise that quarkonium formation occurs also from quarks and antiquarks from different initial pairs. This mechanism [417-419] has turned out to be critical in understanding the rise of J/ ψ production from RHIC to the LHC where (re)generation seems to constitute the major part of the yield observed in central Pb–Pb collisions [420]. The data is also compatible with production of J/ ψ exclusively through statistical hadronisation at the crossover phase boundary [421]. Precise measurements of the $c\bar{c}$ production cross section and the extraction of the charm-quark diffusion coefficient in Runs 3 & 4 will be important for making a more definite statement; these are key objectives discussed

in the chapter 5 on open heavy-flavor production. Information from $p_{\rm T}$ spectra and elliptic flow will help to complete the picture.

Regarding bottomonia, the current understanding suggests that (re)generation is less important for Y(1S), but possibly figures as a major component in the strongly suppressed yield of excited states [422, 423]. It is therefore of great importance to obtain additional information about the typical time at which quarkonia are produced, in particular through p_T spectra and elliptic flow which contain information about the fireball collectivity imprinted on the quarkonia by the time of their decoupling. A schematic illustration of the current knowledge extracted from in-medium quarkonium spectroscopy, *i.e.*, their production systematics in heavy-ion collisions is shown in Fig. 44.



Fig. 44: The vacuum heavy-quark potential as a function of $Q\bar{Q}$ separation. The horizontal lines indicate the approximate locations of the vacuum bound states while the vertical arrows indicate the minimal screening distances of the media produced at the SPS, RHIC and LHC, as deduced from approximate initial temperatures reached in these collisions extracted from data (indicated above the arrows in MeV) and from quarkonium production systematics in Pb–Pb and Au-Au collisions. Figure taken from Ref. [420].

On the theoretical side, the basic objects are the quarkonium spectral functions which encode the information on the quarkonium binding energies, in-medium HQ masses and the (inelastic) reaction rates. Ample constraints on the determination of the quarkonium spectral functions are available from thermal lQCD, *e.g.*, in terms of the heavy-quark free energy, euclidean and spatial quarkonium correlation functions, and HQ susceptibilities, and are being implemented into potential model calculations [424– 430]. In particular, the role of dissociation reactions has received increasing attention. Early calculations of gluo-dissociation [431, 432] or inelastic parton scattering [433] have been revisited and reformulated, e.g., as a singlet-to-octet transition mechanism [427] or in terms of an imaginary part of a two-body potential [434], respectively. In particular, the latter accounts for interference effects which reduce the rate relative to "quasi-free" dissociation [433] in the limit of small binding; interference effects can also be calculated diagrammatically [435]; they ensure that, in the limit of vanishing size, a Color-neutral $Q\bar{Q}$ dipole becomes "invisible" to the Color charges in the QGP.

The information from the spectral functions can then be utilised in heavy-ion phenomenology via transport models. The latter provide the connection between first-principles information from lQCD and experiment that greatly benefits the extraction of robust information on the in-medium QCD force and its emergent transport properties, most notably the (chemical) equilibration rates of quarkonia. Thus far most transport models are based on rate equations and/or semiclassical Boltzmann equations. In recent years quantum transport approaches have been developed using, e.g., a Schrödinger-Langevin [436–439]

or density-matrix [440, 441] formulation. These will enable to test the classical approximation underlying the Boltzmann and rate equation treatments and ultimately quantify the corrections. Quantum effects may be particularly relevant at high $p_{\rm T}$ in connection with the in-medium formation times of quarkonia, augmented by the Lorentz time dilation in the moving frame; schematic treatments of this effect in semiclassical approaches suggest that varying formation times can leave observable differences for high-momentum charmonia and bottomonia [422, 423, 442–444]. Finally, the implementation of phasespace distributions of explicitly diffusing heavy quarks into quarkonium transport is being investigated by various groups (see, e.g., Ref. [445]), which, as mentioned above, will provide valuable constraints on the magnitude and $p_{\rm T}$ dependence of (re)generation processes. In particular, the role of non-perturbative effects in the HQ interactions in the QGP (which are believed to be essential to explain the large elliptic flow observed for D-mesons [246]) needs to be accounted for; the associated large scattering widths are likely to require quarkonium transport implementations beyond semi-classical (or perturbative) approximations, which reiterates the need for a quantum treatment of recombination processes.

The larger experimental data samples in Runs 3 & 4, combined with improved detector performance and measurement techniques, will allow one to significantly improve over the current measurements, with extended kinematic coverage (in p_T) and allowing one to reach also currently-unobserved quarkonium states, like Y(3S). The complementarity (and overlap) of all 4 LHC experiments is crucial in this endeavour and will call for a data combination strategy, for instance for Y azimuthal anisotropy. Quarkonia are measured in the dimuon channel in ATLAS (|y| < 2.0), CMS (|y| < 2.4), LHCb (2.0 < y < 4.5), and ALICE (2.5 < y < 4.0), and in the dielectron channel with ALICE (|y| < 0.9). We present below data projections and simulations for a selection of observables and compare to model predictions (which sometimes constitute the basis for the projections). The model uncertainties shown in this section represent the current knowledge; significant improvements are expected both in what concerns the conceptual aspects discussed above as well for the input parameters, which will be constrained by data and theory (for instance in what concerns nuclear PDFs, see also Section 10.4).

All four LHC experiments will benefit from a large upgrade program, during the Long Shutdown 2 (2019–2020) for ALICE and LHCb, and during Long Shutdown 3 (2024–2025) for ATLAS and CMS [253]. The addition of the Muon Forward Tracker (MFT) will allow ALICE to separate the prompt charmonium from the contribution from B meson decays. In addition, the background will be reduced, yielding to better signal over background ratios. Regarding ATLAS and CMS, the upgraded inner tracker will extend to $|\eta| \leq 4.0$ after LS3, and the muon system coverage to $|\eta| \leq 2.7$ (3.0) for ATLAS (CMS). While the detector improvements will have a smaller impact than the increase in data sample size, this increase in pseudorapidity coverage is appreciable in also giving an overlap with the range of ALICE and LHCb. Better track momentum resolution is also expected from these upgraded inner trackers, with an improvement of about 30% of the mass resolution of quarkonia for CMS [446]. The expected improvement in the relative statistical uncertainty, due to a better signal over background ratio, is in the range 10-25% [291].

7.2 Charmonia in Pb–Pb collisions

A remarkable discovery at the LHC was that the suppression of J/ ψ is significantly reduced in comparison to lower energies [447] and that this reduction is concentrated at lower $p_{\rm T}$ [448, 449], compatible with predictions of (re)generation at the phase boundary of QCD [417] or throughout the deconfined phase [418, 450, 451], via recombination of diagonal (correlated pairs) or off-diagonal cc pairs [419]. No significant difference is however found between measurements at $\sqrt{s_{\rm NN}} = 2.76 \,{\rm TeV}$ and $\sqrt{s_{\rm NN}} =$ $5.02 \,{\rm TeV}$ [449, 452]. Recently, the measurement of a significant elliptic flow coefficient v_2 both for D mesons [453–455] and J/ ψ [248, 456–459], which was shown to be correlated to the flow of the bulk particles [267, 459], can be seen as another indication for the thermalisation of charm quarks in the QGP. Transport model calculations [460, 461] currently underestimate the data for $p_{\rm T} \gtrsim 6 \,{\rm GeV}/c$ [248, 457, 458].



Fig. 45: Projected measurement of elliptic flow coefficient v_2 as a function of p_T for J/ ψ mesons (measured in ALICE, for 2.5 < y < 4), for the centrality class 20–40%, in comparison to model calculations [461].

The projected ALICE measurement of inclusive $J/\psi v_2$ as a function of p_T for the centrality class 20–40%, for 2.5 < y < 4, in comparison to model calculations [461] is shown in Fig. 45. Disentangling the contributions of prompt and non-prompt J/ψ and considering (path-length dependent) energy loss seems mandatory to understand the details of the $J/\psi v_2$ pattern, which will be facilitated with the detector upgrades and higher luminosity of Runs 3 & 4. The measurement of higher harmonics, e.g., v_3 , which are sensitive to initial state energy density fluctuations, will also become available and provide further insight into the charmonium production mechanisms. Precise prompt and non-prompt $J/\psi v_2$ and v_3 measurements at low p_T will be reachable using the ALICE central barrel. Polarisation will be measured too [462], providing further insight in the different production mechanisms involved in Pb-Pb collisions as compared to pp. Under the statistical hadronisation paradigm, the prompt J/ψ yield in Pb-Pb collisions should be unpolarised with the 3 polarisation states equally populated.

At high $p_{\rm T}$, where a raising trend is currently hinted by Run 2 $R_{\rm AA}$ measurements [452, 463], the production mechanisms cannot currently be resolved, given the statistical limitation in the data (see Fig. 46, left). The high $p_{\rm T}$ reach of Runs 3 & 4 data (illustrated in Fig. 46 (right) for CMS) will allow one to conclude on the important question of whether J/ ψ formation at high $p_{\rm T}$ is determined by the Debye screening mechanism [444, 464], or by energy loss of the charm quark or the cc pair [465, 466].

The measurement of $\psi(2S)$ mesons is more difficult than that of J/ψ , because of a much smaller production cross section times branching ratio and even larger suppression in Pb–Pb, yielding a very low signal to background ratio. The projections for the measurement of the $\psi(2S)$ state in ALICE are shown in Fig. 47 as a function of centrality and compared to model predictions in the transport approach [461] and from the statistical hadronisation model [421]. This (p_T -integrated) measurement will significantly contribute to make a distinction between the two models. Projections are also available from the CMS experiment [259]. Other states, for instance χ_c , may be measured too, albeit the measurement down to $p_T = 0$ will remain challenging. B_c^+ mesons can also be measured, either in the $J/\psi\pi$ or $J/\psi\mu\nu_{\mu}$ channel, to further study recombination in the beauty sector.



Fig. 46: Left: R_{AA} vs p_T for prompt J/ ψ in central (0–20%) collisions (ATLAS, |y| < 2, [463]). Right: Prompt J/ ψ high p_T bin boundaries as a function of luminosity with the CMS experiment [291]. The boundaries are chosen in such a way the number of mesons in the bin for the corresponding luminosity equals the number of mesons found in the last p_T bin of the current measurement [452].

7.3 Bottomonia in Pb–Pb collisions

The study of bottomonia with Pb–Pb data from the Runs 3 & 4 of the LHC will bring further information on the physics aspects described above. Although their production is a priori sensitive to the same effects as charmonia, in practice the two quarkonium families feature some fundamental differences. Binding energies differ, which is reflected in the different dissociation temperatures. Experimentally, compared to charmonia, the absence of contribution from B meson decays and the more similar cross section times branching ratio between the ground and excited states make bottomonia measurements easier. At the same time, in pp collisions, up to 30–50% of the measured Y(1S) and Y(2S) yields actually result from the feed-down from other states [244, 468]: a large portion of measured Y(1S) suppression can be due to the stronger suppression of the feed-down states – Y(2S) and Y(3S) mesons also receive a significant contribution from feed-down. The impact of (re)generation from uncorrelated bb is also expected to be much smaller than for charmonia, because of the much smaller number of bb pairs per Pb– Pb event compared to that of cc pairs. The importance of regeneration for bottomonia, from correlated or uncorrelated pairs [439], is however still very model dependent, and no unambiguous experimental signal for it has been found yet. Possible ways of constraining this contribution will be discussed in this section.

Experimentally, the higher mass of bottomonia compared to charmonia implies higher p_T decay leptons, allowing the ATLAS and CMS experiments to measure the production down to zero transverse momentum, as is possible for ALICE for both charmonia and bottomonia [469, 470]. The proximity in mass between the different mass states, especially between the Y(2S) and Y(3S) states, also means that good muon (or electron) momentum resolution is essential to their measurement, especially for excited states.

It is useful to remind quickly the status in 2018, based on results from Run 1 and early Run 2 LHC data as well as RHIC data. Y production is found to be suppressed in Pb–Pb compared to expectations from a scaling of incoherent pp collisions, in all rapidity, p_T and centrality ranges measured [469–472]. Suppression is stronger in central events, as expected from the hotter and longer-lived medium in such events. The results from the most central collisions suggest that a certain amount of suppression of the



Fig. 47: Production ratio $\psi(2S)/J/\psi$ vs. N_{part} for |y| < 0.9 (left) and 2.5 < y < 4 (right) [462, 467]. Model predictions in the transport approach [461] and from statistical hadronisation [421] are included. The values of the ratio used for the projections are quasi-arbitrary.



Fig. 48: Centrality dependence of Y(1S), Y(2S) and Y(3S) R_{AA} , as projected by the CMS [259, 474] (left) and ALICE (centre) experiments, and from a transport model [422]

directly produced Y(1S) might be needed to explain the data in addition to cold nuclear matter effects and melting of the excited Y and χ_b states. The excited states Y(2S) and Y(3S) show higher suppression wrt the ground state, with R_{AA} values which respect the hierarchy expected based on their binding energies. The Y(3S) is still unobserved in Pb–Pb collisions ($R_{AA}(Y(3S)) < 0.094$ at 95% confidence level, for $\sqrt{s_{NN}} = 5.02 \text{ TeV}$ [471,473]). No significant dependence of the suppression of Y states is found at the LHC on collision energy or rapidity.

Differences exist between models in the theoretical treatment of the suppression of the bottomonia in the medium, as summarised earlier in Section. 7.1. Different assumptions are used regarding the production mechanism, the heavy quark potential, or the evolution of the quarkonia with the medium. The understanding of hot medium effects will be also improved thanks to higher precision measurements in pp collisions of the feed-down fractions and to stronger constrains of the cold nuclear matter and initial state effects (including nPDF or coherent energy loss effects [475]) from p–Pb collision measurements.



Fig. 49: Projected R_{AA} for Y(1S) and Y(2S) expected from the ALICE (left) and CMS [291] (center) experiments, as a function of p_T , with 10 nb^{-1} of Pb–Pb data. The expected p_T reach for Y(1S) from the CMS experiment is also shown [291], as the position of the last p_T bin of the measurement, with constant number of observed Y(1S), as a function of integrated luminosity.



Fig. 50: Projected R_{AA} for Y(1S) and Y(2S) expected from the ALICE (left) and CMS [291] (right) experiments, as a function of rapidity, with 10 nb⁻¹ of Pb–Pb data.

Figure 48 shows that the projected uncertainty on the R_{AA} of Y with 10 nb^{-1} will be much smaller than the current model uncertainties. Bottomonia may bring information complementary to other probes, using the sensitivity of the suppression to the medium shear viscosity or to the initial temperature of the fireball.

A precise measurement of the $p_{\rm T}$ dependence of the Y(1S) $R_{\rm AA}$ will be possible using LHC data from Runs 3 & 4. At low and medium $p_{\rm T}$, the measurement is sensitive to the possible regeneration component in Y meson production [422]. Projections for the expected precision of Y measurements from the ALICE and CMS detectors using an integrated luminosity of $10 \, {\rm nb}^{-1}$ after the Runs 3 & 4 are shown as a function of $p_{\rm T}$ in Fig. 49 and y in Fig. 50, and compared to the expectations from two models [422, 423]. In the Kent state model calculations [423] (not shown), where Y mesons are originating only from the primordial production, with no regeneration component, the $R_{\rm AA}$ is rather flat in the low and medium $p_{\rm T}$ range. Only at higher $p_{\rm T}$ (above 10–15 GeV/c) is a small rise predicted, which can be looked for in Runs 3 & 4 data: as can be seen in Fig. 49, it is expected that a measurement up to a $p_{\rm T}$ of about 50 GeV/c can be performed with the ATLAS and CMS detectors with 10 nb⁻¹ of data. In the TAMU model [422] however, a regeneration component is considered, and several assumptions are explored, especially on the degree of thermalisation of the bottom quarks. It predicts a maximum in the Y(1S) $R_{\rm AA}$ at a $p_{\rm T}$ around 10 GeV/c. The current data is not precise enough to confirm or disfavour



Fig. 51: v_2 projections for the CMS [291] (left and centre) and ALICE (right) experiments for the Y(1S) and Y(2) mesons, assuming the predictions from a transport model [422].

such a local maximum in the R_{AA} , but Runs 3 & 4 data will allow to conclude.

Almost no rapidity dependence is expected at the LHC for the nuclear modification factor of Y mesons within the acceptance of ATLAS and CMS ($|\eta| \leq 2.5 - 3$), which can be better tested using Runs 3 & 4 data. This will be further made significant considering the ALICE acceptance (2.5 < y < 4), allowing to confirm or disprove the prediction of the hydrodynamic model, see Fig. 50.

Coming back to the matter of regeneration, much can be learnt about it by a measurement of the elliptic flow of Y(1S) mesons [476], unmeasured to date in any collision system. A parallel can be drawn with that of J/ψ , which is still not properly described by models. This observable requires a more detailed implementation of the dynamics of the interactions between the quarkonium and the medium: thermalisation of the heavy quarks, time dependence of regeneration, path length dependence of energy loss, as well as initial geometry fluctuations and elastic rescattering of the quarkonia in the medium. Thus, collective flow brings complementary information to the R_{AA} , and its measurement can help disentangle some effects. In the case of Y(1S) mesons, a small v_2 (order of 1–2%) is expected [422, 477, 478], as can be seen in Fig. 51. The elliptic flow of Y(2S) could be significantly higher [422, 478], both from the regenerated and primordial components. For both states, projections show that experimental precision may not be enough for a significant v_2 measurement, assuming v_2 values as in Ref. [422]. For this reason, combining results between the different LHC experiments would be beneficial to reach a better sensitivity.

While we have focused on the R_{AA} and v_2 in this section, bottomonium production can be studied using other observables. For instance, fully corrected yields or cross sections in Pb–Pb can be studied, without making the ratio to a pp measurement in a R_{AA} . Such a measurement, already reported in some of the available experimental results [471], can directly be compared to production models.

7.4 Quarkonia in p-Pb and pp collisions

7.4.1 p-Pb collisions

Quarkonium-production studies in high-energy p–Pb collisions are usually carried out to measure how much specific nuclear effects, those which do *not* result from the creation of a deconfined state of matter, can alter the quarkonium yields. They should indeed be accounted for in the interpretation of Pb–Pb results. They are also interesting on their own as they provide means to probe the modification of the gluon densities in the nuclei, the interaction between such pure heavy-quark bound states and light hadrons, or phenomena such as the coherent medium-induced energy loss of these quark-antiquark pairs. The measurements as a function of event activity brought several surprises, hotly discussed presently.

Usually, a separation into initial-state and final-state effects is done (coherent energy loss effects

[479] can be seen as an interplay between the two types of effects). Yet, it is probably more instructive to separate out the effects which are believed to impact *all* the states of the charmonium or the bottomonium family with the *same* magnitude from those which are expected to impact differently the ground and the excited states. In principle, initial-state effects (in particular gluon shadowing [317]) are of the first kind as the nature of the to-be-produced quarkonium state is not yet fixed when the effects are at work. On the contrary, final-state effects (like regeneration [480]) do depend on the properties of the produced quarkonium state and are thus be of the second kind.

However, in p–Pb collisions at LHC energies, final-state interactions between the heavy-quark pair and the nuclear matter likely occur *before* the pair hadronises. This is due to the large boost between the nucleus and the pair – and thus the quarkonium. At rest, a $c\bar{c}$ or $b\bar{b}$ pair takes 0.3–0.4 fm/c to hadronise; seen from the nucleus, at, for instance $y_{\text{pair}}^{\text{lab}} - y_{\text{beam}} \sim 7$, it takes $\gamma = \cosh(7) \simeq 500$ times longer. As such, final-state interactions with the compounds nucleus likely do not discriminate ground and excited quarkonium states, unless rescattering in the nucleus affects the QQ wave function, overlapping with the quarkonium wave function at large distance [481]. Such an argument based on the existence of a large boost is nevertheless not applicable if one considers effects arising from the interactions between the pair and other particles *produced* by the p–Pb collisions, not those contained in the Pb nucleus. The former are indeed not moving at the Pb projectile rapidity. In fact, some of these particles can have similar rapidities as the quarkonium and can thus be considered as comoving with it [482–484].

The simultaneous study of open-heavy flavoured hadrons along with both ground and excited quarkonium states can shed light on all these phenomena. Along the lines exposed above, one expects forward-quarkonium production in p–Pb collisions (namely when the quarkonia flies in the direction of the proton) to be sensitive to low-x phenomena (like the gluon shadowing or saturation in the lead ion) and to the coherent energy loss. On the contrary, the backward production should be sensitive to the gluon antishadowing and to fully coherent energy loss. Moreover, the scatterings of quarkonia with comoving particles occur more often backward than forward, due to the rapidity-asymmetric particle multiplicities, and more often as well with the larger and less tightly bound excited states.

With a wide rapidity coverage spanning from about -5 to 5, the LHC data from the 4 experiments are unique as they allow one to probe much smaller x values than at RHIC and with a larger reach in $p_{\rm T}$. The higher c.m.s. energy, the competitive luminosities and the resolution of the detectors also allow for more extensive studies of the bottomonium family. In fact, an important observation made with Run 1 data was that of a relative suppression in p–Pb collisions of the excited Y(2S),Y(3S) states compared to that of the Y(1S) observed by CMS [485] as a function of the event activity (recently confirmed by ATLAS [486], but also observed in pp collisions by CMS [487]). Not only was it unexpected, but it constitutes a challenge to the conventional interpretation of suppression observed in Pb–Pb collisions [471,473,488], which is of a significantly larger magnitude, but of a similar pattern. Such a relative suppression was also observed in the charmonium sector [489], where it is as well remarkable.

As far as the suppression of the Y(1S) and J/ ψ is concerned, they seem to follow the expectations based on the RHIC results with a strong forward suppression described by shadowing – of a compatible magnitude to that observed with HF data [317], or with the coherent energy loss mechanism [479]. More data, including that on Y(nS) and Drell-Yan production, are clearly needed to disentangle both effects [490] (see also Section 10.3.4). More precision for Y(nS) and non-prompt J/ ψ is in general critically needed as the typical experimental uncertainties are still on the order of the expected effects. As a case in point, backward y data are not yet precise enough to quantify the magnitude of the gluon antishadowing, see Section 10 for the possible relevance of quarkonium p–Pb LHC data on nuclear PDF fits. Direct inclusion of this data in nPDF fits is however not yet possible, pending unambiguous clarification of the different effects impacting quarkonium production in p–Pb collisions.

Recently, the measurement of v_2 of J/ ψ in p–Pb collisions became available [491,492], indicating a large azimuthal anisotropy, $v_2 \leq 0.1$ up to $p_T \leq 8 \text{ GeV}/c$. Recent transport model calculations [480], which are successful in describing the features of the data, including the transverse momentum and



Fig. 52: The $p_{\rm T}$ dependence of the v_2 coefficient of J/ ψ mesons in p–Pb collisions, for 500 nb⁻¹ (AL-ICE). The projections are based on current ALICE data for 0–20% centrality [491] and are shown separately for negative and positive y_{CM}), assuming the same magnitude and are compared with transport model (TAMU) calculations [480] for midrapidity.

centrality dependence of J/ψ and $\psi(2S)$ production in p–Pb, cannot reach the high value of the v_2 coefficient seen in data [491,492] (see Fig. 52), suggesting that the observed v_2 in p–Pb collisions might also originate from initial state effects. A precision measurement in Runs 3 & 4 for a broad rapidity range will clarify this.

In addition to conventional LHC collider data, one should not overlook the discriminating power of data which can be collected in the fixed-target mode [493,494]. Not only they correspond to completely different energy and (c.m.s.) rapidity ranges, but extremely competitive luminosities, up to a few fb^{-1} , are easily reachable, beyond what can be reached in the collider mode during Runs 3 & 4. The LHCb collaboration has paved the way for a full fixed-target program at the LHC with their SMOG luminosity monitor [495] used as an internal (He, Ne, Ar) gas target [496] (see also Section 11.4). It is now clear that corresponding studies to those suggested above are possible [497] with the LHCb and ALICE detectors with minor technical adjustments. They would drastically expand the scope of current proton-nucleus quarkonium studies.

7.4.2 High-multiplicity pp collisions

Systematic studies of the quarkonium production in high-multiplicity pp events can play an important role in understanding hadronisation. In particular, the correlation of the quarkonium yields with the charged-particle multiplicity can provide new insights into the interplay between hard and soft processes in particle production. Hidden and open heavy-flavour production measurements as a function of the event activity were carried out at the LHC during Run 1 [485, 498]. The striking feature of the data is that the production yields of quarkonia in high multiplicity events are significantly enhanced relative to minimum bias events, like for D mesons [499]. Specifically, the measurements of the self-normalised yields (the yield divided by the mean yield in minimum bias collisions) as a function of the self-normalised charged-particle multiplicity show an increase which is stronger than linear at the highest multiplicities. The similarity between the D-meson and J/ ψ results [498, 499] suggests that this behaviour is most likely related to the production processes, and that hadronisation may only play a secondary role. When comparing J/ ψ preliminary results at $\sqrt{s} = 13$ TeV [500] to the ones previously obtained at $\sqrt{s} = 7$ TeV [498], no significant energy dependence is observed, *i.e.* the relative J/ ψ yields for events with identical relative multiplicities give similar results. In addition, a dependence of the

excited-to-ground-state ratio with charged particle multiplicity is observed in the bottomonium sector in pp collisions [485, 487].

The data are described both by initial-state models as well as by a model assuming hydrodynamic evolution [501], considering that the energy density reached in pp collisions at LHC is high enough to apply such evolution. Initial-state (saturation) effects are considered within i) the Color-Glass-Condensate (CGC) framework [502]; ii) the percolation approach [503, 504]; iii) a model with higher Fock states [505], based on parameters derived from p–Pb collisions. The energy dependence of the cross sections is controlled by the saturation momentum $Q_s(x)$ in the case of the CGC or density of colour ropes $\rho_s(y, p_T)$ in the percolation model, which also governs the charged-hadron multiplicity; events at different energies with the same Q_s or ρ_s are therefore identical. For a given event multiplicity, they predict the relative yields to be almost energy independent. It seems that, in any case, multiple interactions at the partonic level need to be taken into account in order to reproduce the data [506–508].

Runs 3 & 4 data, reaching unprecedented high multiplicities because of larger data samples, and allowing for differential studies in $p_{\rm T}$, will certainly help discriminate models. For instance, in the percolation model, where colour interactions produce a reduction of the charged-particle multiplicities, the deviation from the linear behaviour is expected to be steeper for high- $p_{\rm T}$ quarkonia (and D mesons). Moreover, measurements of J/ ψ yields relative to those of D mesons with the same transverse mass could help elucidate the relative contribution of hadronisation and initial-state effects.

Studies of double differential ratios of excited-to-ground quarkonium states versus relative multiplicity could help clarify the presence of final-state effects, either QGP-like or the ones proposed by the comover model [509, 510]. Also, within the CGC+NRQCD framework [502], the relative contributions of the 4 leading J/ ψ Fock states have been calculated as a function of the event activity, showing a different dependence for different Fock states.

8 Electromagnetic radiation

Coordinator: Michael Weber (Stefan Meyer Institute Vienna, Austrian Academy of Sciences)

Contributors: R. Bailhache (Goethe-University Frankfurt), R. Chatterjee (VECC Calcutta), T. Dahms (Excellence Cluster Universe, Technical University Munich), T. Gunji (Center for Nuclear Study, Graduate School of Science, the University of Tokyo), M. He (Nanjing University of Science and Technology), S.R. Klein (Lawrence Berkeley National Laboratory), A. Marin (GSI Helmholtzzentrum für Schwerionenforschung GmbH), D. Peresunko (National Research Centre Kurchatov Institute, Moscow), R. Rapp (Texas A&M University, College Station), K. Reygers (Heidelberg University), T. Song (University of Gießen), A. Uras (Université de Lyon, CNRS/IN2P3, IPN-Lyon), G. Vujanovic (Ohio State University and Wayne State University)

The strongly interacting system formed in ultrarelativistic heavy-ion collisions emits electromagnetic radiation that can be detected using different probes: real *direct* photons or virtual photons measurable via dilepton pairs. Direct photons can be split into *prompt* photons, emitted by the partons of colliding nuclei during their inter-penetration, and *thermal* photons, emitted by the almost thermalized hot system. For dileptons these contributions are called Drell-Yan and thermal, respectively. In contrast to real photons, dileptons carry a mass and thus can be used to study the decay of massive particles, such as the in-medium modified spectral shape of vector mesons, the ρ meson being the most prominent one, and the search for particles beyond the Standard Model, e.g., dark photons. In this section, we outline the measurement of photons via calorimetry and the so-called photon conversion method, as well as dielectron (e⁺e⁻), and dimuon ($\mu^+\mu^-$) pairs in A–A collisions in the ALICE detector at the LHC. Moreover, the photoproduction of dilepton pairs in peripheral collisions and the expected sensitivity for the search of dark photons are discussed in subsections 8.2 and 8.3, respectively. We begin with a short review of previous experimental results together with a summary of the basic theoretical models employed to describe these data.

8.1 Thermal radiation and in-medium spectral function

Electromagnetic radiation from the hot and dense system formed in ultrarelativistic heavy-ion collisions in the form of real photons was measured for the first time at the SPS by WA98 [511]. The direct photon spectrum measured in Pb–Pb collisions at $\sqrt{s_{NN}} = 17.3$ GeV showed an excess above the extrapolated prompt photon signal based on measurements in proton induced reactions. The excess is described by a large variety of hydrodynamic and cascade models (see [512] for review), most of which assume the formation of a QGP phase. Also at the SPS, a modification of low-mass dilepton pairs in S-Au and Pb-Au collisions relative to the expectation of in-vacuum hadron decays was observed by CERES [513–517] and studied with high precision by NA60 in In–In collisions [518–521]. The data are consistent with an in-medium ρ spectral function that, driven by the coupling to baryons, melts and approaches the one from $q\bar{q}$ annihilation in the vicinity of the phase transition [522–524], which is compatible with chiral symmetry restoration [123, 525]. On the other hand, the data cannot be described with a dropping mass scenario, in which the ρ mass drops to zero as chiral symmetry is restored [526]. Beyond the issue of chiral symmetry restoration, NA60 measured an excess of prompt dimuons in the intermediate mass region between the ϕ and the J/ ψ masses [519, 520]. Contrary to transverse-mass spectra of the dimuon continuum at lower masses, this excess shows no increase of the exponential inverse slope with mass, i.e., blue shift, that is typical for radial flow. This suggests that the source of this enhancement is from the earliest phase of the collision, before significant radial flow has built up. This supports the idea that the inverse slope of the invariant mass spectrum is insensitive to the expansion of the medium and therefore a true measure of the average temperature. NA60 measured a value of $T = 205 \pm 12$ MeV [521], which significantly exceeds the temperature of 154 ± 9 MeV, above which the formation of a QGP has been predicted [37, 122].

At RHIC energies, PHENIX and STAR have measured an enhancement of e^+e^- pairs in the low mass region in Au–Au collisions [527–530] that can be described with the same model of collisional broadening as used at the SPS. STAR measured that the enhancement above the hadron decay back-

ground does not change with collision energy between $\sqrt{s_{NN}} = 19$ and 200 GeV [530]. Despite a marked decrease of the net-baryon chemical potential in this energy range, the total baryon plus antibaryon density does not change much, providing further evidence that the ρ coupling to baryons and antibaryons is responsible for the enhancement. Real direct photon production in Au-Au collisions was measured by PHENIX [531-533]. An excess was observed compared to binary scaled direct photon production in pp collisions. The signal was measured via quasi-real virtual photons, i.e., e^+e^- pairs with small invariant mass, as well as real photons converting in detector material. The excess yield at low $p_{\rm T}$ appears to have a universal multiplicity dependence, scaling with the charged-particle multiplicity at midrapidity to the power of about 1.25, independent of collision energy between $\sqrt{s_{_{\rm NN}}} = 39$ and 200 GeV [533]. The transverse momentum spectrum of the excess yield has an exponential inverse slope of $T = 221 \pm 19$ (stat.) ± 19 (syst.) MeV for central collisions and values close to that for other centralities. The spectrum, however, is strongly blue shifted by radial flow in the later stages of the fireball radiation, which is further supported by a sizeable elliptic flow (v_2) of the direct photon signal [534]. Therefore, the inverse slope cannot directly be interpreted as an average temperature, which highlights the importance of thermal dilepton measurements as a function of invariant mass. However, the modelling of the space-time evolution offers the possibility of extracting temperature information from the photon data [535]. The direct photon v_2 is indeed comparable to the v_2 of pions, which suggests late emission of direct photons dominated by the hadronic phase [536]. A simultaneous description of the elliptic flow effect, as well as the large direct photon excess, which implies early production, poses a significant challenge to theoretical models.

The first measurement of direct photon production in Pb-Pb collisions with the ALICE detector [537] at the LHC also show an excess of thermal production at low $p_{\rm T} < 3$ GeV/c with respect to the prompt direct photon expectation [538]. The extracted effective temperatures $T = 297 \pm 12 \text{ (stat.)} \pm 12 \text{ (stat.)}$ 41 (syst.) MeV in central collisions and $T = 410 \pm 84$ (stat.) ± 140 (syst.) MeV in semi-central collisions are higher than those at RHIC energies, as expected. The direct photon elliptic flow was also extracted in central and semi-central collisions [539]. The measured flow is close to the one at RHIC energy and at low $p_{\rm T} < 4$ GeV/c to the one of final hadrons. However, this measurement does not cause the same challenges to models, since the experimental uncertainties are still large at this point. The reduction of systematic uncertainties of the direct photon measurement is the main objective for Run 3 to improve its significance. Moreover, a low magnetic field run will allow one to access the $p_{\rm T} < 1$ GeV/c region where the thermal photon production increases rapidly. Theoretical calculations of thermal and prompt photon productions are available at $\sqrt{s_{\rm NN}} = 5.02$ TeV (Fig. 53) [195, 540–542]. The thermal contribution dominant at low $p_{\rm T}$ is given by the QGP photon emission rates and the hadronic photon production rates, integrated over the space-time evolution of the system. In [195] a (2+1)D hydrodynamic evolution with IP-Glasma initial conditions with initial flow and finite shear and bulk viscosity is followed by a hadronic phase modelled using UrQMD. A longitudinal boost invariant (2+1)D ideal hydrodynamics is used in [541], while in [542] a (2+1)D ideal hydrodynamic model including nonvanishing initial flow is employed. The prompt photon component, dominant at high $p_{\rm T}$, is very similar in all models. It is obtained from NLO pQCD calculations using the BFG-II photon fragmentation function and the CTEQ6.6 [541] or nCTEQ15 parton distribution function [540]. An increase by a factor \sim 1.5 at about $p_{\rm T} \approx 1$ GeV/c and by a factor 1.5 to 2 for the prompt photons is predicted compared to yields at $\sqrt{s_{NN}} = 2.76$ TeV. The predicted thermal photon elliptic flow parameters for central collisions are close to each other at the two LHC energies and are very small. Differences become larger as one goes towards peripheral collisions. Simultaneous measurements of photon yields and photon flow with high accuracy and lower $p_{\rm T}$ reach will provide constraints to theoretical models.

Dilepton measurements by ALICE at the LHC are not yet sensitive to possible low-mass enhancement and thermal signals [545]. A precise measurement of the low-mass dielectron continuum will be one of the main objectives of the ALICE physics programme during the LHC Run 3 and 4. In the meanwhile, the dominant background of dielectrons from correlated semileptonic open heavy-flavour decays



Fig. 53: Direct photon differential invariant yield for central 0–10% Pb–Pb collisions at $\sqrt{s_{NN}} = 5.02$ TeV as predicted by several models [195,541–544].



Fig. 54: Model predictions for the invariant mass spectrum of e^+e^- pairs in central (0–10%) Pb–Pb collisions at $\sqrt{s_{NN}} = 5.5$ TeV. Left panel: in-medium radiation plus decays of the ρ meson at the end of the system evolution by R. Rapp *et al.*. Right panel: Expectations from the PHSD model including the in-medium ρ meson, $q\bar{q} \rightarrow e^+e^-$, $q\bar{q} \rightarrow e^+e^-g$, and $q(\bar{q})g \rightarrow e^+e^-q(\bar{q})$, hadronic sources, and semileptonic decays of $c\bar{c}$ and $b\bar{b}$.

is utilised to learn more about open heavy-flavour production in pp collisions at LHC energies [546,547].

The model by R. Rapp *et al.*, an approach that has been proven to provide a quantitative description of the existing dilepton results [548], is based on two ingredients that are put into a realistic spacetime evolution [549]. The thermal dilepton radiation is modelled by emission rates from the hadronic phase and the Quark–Gluon Plasma [524, 550]. A hadronic many-body approach [523] is used for the medium-modified spectral functions of ρ and ω mesons. In addition, the equation of state is updated to a cross-over transition around $T_c = 170$ MeV extracted from with recent lattice QCD computations, and hadro-chemical freezeout at $T_{chem} = 160$ MeV [551]. Figure 54 (left) shows the calculations performed for central Pb–Pb collisions at $\sqrt{s_{NN}} = 5.5$ TeV for in-medium radiation plus decays of the ρ meson at the end of the system evolution. The pair-yield is estimated for the rapidity range $|y_e| < 0.85$ and transverse momentum of single electrons $p_{\rm T}^{\rm e}>0.2~{\rm GeV\!/\!c}$ and is normalized to the number of events $N_{\rm evt}.$

A complementary approach to study dilepton spectra and thermal radiation is provided by the parton-hadron-string dynamics (PHSD) transport approach, which also successfully describes the existing experimental data [552, 553]. The in-medium modification of the ρ meson is incorporated in PHSD by an off-shell transport of vector mesons with a dynamically changing set of spectral functions [554] evolving towards the vacuum spectral function at the end of the collision history. The electromagnetic radiation of the QGP is modelled by $q\bar{q} \rightarrow e^+e^-$, $q\bar{q} \rightarrow e^+e^-g$, and $q(\bar{q})g \rightarrow e^+e^-q(\bar{q})$ using effective propagators for quarks and gluons from a dynamical quasi-particle model [555]. Figure 54 (right) shows both contributions to the dielectron spectrum in central Pb–Pb collisions at $\sqrt{s_{NN}} = 5.5$ TeV calculated from PHSD together with other sources of dielectrons: decays of long-lived light mesons into e^+e^- (the so-called hadronic cocktail) and the semileptonic decay of hadrons containing heavy quarks, such as D and B mesons.

Important input for models aiming to describe the dilepton yield at LHC energies are the inmedium spectral functions for the vector mesons, most importantly the ρ meson, as well as the photon and dilepton rates from the QGP. For the latter, Lattice QCD calculations, which are currently limited to the quenched approximation, will hopefully be extended (e.g., larger lattices, especially in the time direction, or facilitating extrapolations to the continuum limit) and be available at higher accuracy for realistic systems including light dynamical degrees of freedom in the future. Recent updates on calculations of the photon rate [556], the electrical conductivity [557], and dilepton rates [558] are promising. The photons and dilepton rates from Lattice calculations should in the future be combined with dynamical models like those in Fig. 54, thus improving their results. In addition, the in-medium spectral functions could also use direct input from Lattice QCD [559, 560] or from a functional renormalization group approach [561]. These models can further be refined by including the effects of dissipation, and in that case the electrical conductivity will become of interest to both the dynamical evolution of the medium as well as the electromagnetic rates. In order for that to be achieved self-consistently, the evolution of the medium and the electromagnetic rates need to be modified to account for dissipative effects, which is a currently ongoing effort [195, 562, 563].

More differential information can be used to study the equation of state of the system throughout the full collision history. The measurement of the elliptic flow coefficient v_2 of thermal photons and dileptons, especially if combined with results from hadronic channels, should put tighter constraints on fundamental properties of the medium (e.g., transport coefficients), as well as its "initial conditions" or "pre-equilibrium" dynamics [564]. For example, owing to the penetrating nature of dileptons, the invariant mass dependence of the dilepton v_2 is sensitive to the temperature dependence of both shear [563] and bulk viscosity [562] in a way that is difficult to access using hadronic observables alone.

8.1.1 Real photons

ALICE has measured direct photon spectra in three centrality classes in Pb–Pb collisions at $\sqrt{s_{NN}} = 2.76$ TeV [538]. An excess of direct photons was quantified by the p_T dependent double ratio

$$R_{\gamma} \equiv \frac{\gamma_{\rm incl}}{\pi_{\rm param}^{0}} \bigg/ \frac{\gamma_{\rm decay}}{\pi_{\rm param}^{0}} = \frac{\gamma_{\rm incl}}{\gamma_{\rm decay}}, \tag{27}$$

where $\gamma_{\rm incl}$ is the measured inclusive photon spectrum, $\pi^0_{\rm param}$ a parametrization of the measured π^0 spectrum, and $\gamma_{\rm decay}$ the calculated decay photon spectrum. The double ratio has the advantage that some of the largest systematic uncertainties cancel partially or completely. The measurement combines results of the Photon Conversion Method (PCM) and of the Photon Spectrometer (PHOS), see Fig. 55, left. In central collisions at low $p_{\rm T} < 4$ GeV/c an excess with respect to prompt photon predictions is observed that is attributed to thermal photon emission from the QGP. In the 20% most central collisions



Fig. 55: R_{γ} measured [538] (left) and R_{γ} projected keeping the measured values of R_{γ} and recalculating the uncertainties as explained in the text.

the low $p_{\rm T}$ excess is of the order of 10–15%, while the total uncertainty of the order of 6%. A signal of direct photons is found in central collisions, but on the level of $\sim 2\sigma$, while in mid-central and especially in peripheral the significance is even smaller. On the other hand, peripheral collisions are important since there one can estimate and restrict the contribution from prompt direct photons.

For Run 3 the PCM measurement will be influenced by the ALICE Inner Tracking System (ITS) and Time Projection Chamber (TPC) upgrades, while PHOS and the Electromagnetic Calorimeter (EM-Cal) will be kept unchanged. The new ITS shows an improved low $p_{\rm T}$ tracking efficiency, that will partially compensate the efficiency loss due to the $\sim 30\%$ reduction of its material thickness. Two 1 mm tungsten wires with well known thickness will be installed parallel to the beam direction for precise calibration of the material thickness as described later. The TPC continuous readout mode together with large pile-up may prevent the use of photon conversions beyond a radius of 35 cm. This restriction will translate into a $\sim 35\%$ lower photon efficiency. On the other hand, the PCM measurement will also profit from the dedicated heavy-ion run with reduced magnetic field of the ALICE solenoid, which will considerably increase the low $p_{\rm T}$ reconstruction efficiency. To estimate how one can improve the accuracy of the measurement, the uncertainties are split into three classes: those which can be improved with increase of statistics (statistical uncertainties, uncertainties related to π^0 spectrum extraction, η/π^0 ratio); uncertainties which can be reduced using new techniques and some special methods (material budget estimate - with calibrated material analysis, energy scale in calorimeters with new hybrid π^0 methods); and uncertainties related to the properties of the detector which can not be improved (hadron contamination in calorimeters, electron identification in conversion method etc.). To estimate the improvement of the uncertainties it is assumed that the integrated luminosity will reach $L_{int} = 3.1 \text{ nb}^{-1}$ per Pb–Pb run and in total $L_{\text{int}} = 13 \text{ mb}^{-1}$ at the end of Run 4, which is more than a factor of 100 larger the than integrated luminosity $L_{\text{int}} = 10 \text{ µb}^{-1}$ used in the published analysis [538]. The major improvement foreseen for Run 3 is the use of calibrated tungsten wires inserted into the ITS to determine the product of the photon flux times the γ reconstruction efficiency. This product would then be used to precisely determine the material thickness in the rest of the ITS (assuming φ -independent photon flux and taking the radial dependence of the reconstruction efficiency from simulation). The proposed calibration method is based



Fig. 56: Direct photon flow in mid-central collisions. Left: direct photon collective flow measured in Pb–Pb collisions compared to decay photon flow and several theoretical predictions. Right: expected accuracy in Run 3 keeping the measured values of R_{γ} and v_2^{γ} and recalculating the uncertainties as explained in the text.

on weights calculated as the double ratio:

$$\omega_{i} = \left(\frac{N_{\gamma}^{\text{rec}}(r_{i})}{N_{\gamma}^{\text{rec}}(r_{\text{wire}})}\right)^{\text{data}} / \left(\frac{N_{\gamma}^{\text{rec}}(r_{i})}{N_{\gamma}^{\text{rec}}(r_{\text{wire}})}\right)^{\text{MC}}$$
(28)

where $N_{\gamma}^{\text{rec}}(r_i)$ and $N_{\gamma}^{\text{rec}}(r_{\text{wire}})$ are the number of reconstructed γ 's in data or in MC simulations in a given radial bin and the calibrated wire, respectively. For the Run 3 projections a systematic uncertainty of 1% on the ITS thickness is taken. The uncorrelated systematic uncertainties on the π^0 and η measurements will be reduced by a factor 10 due to the increased luminosity as they are defined mostly by the raw yield extraction uncertainties which scale proportional to statistical uncertainties. The systematic uncertainties on the photon selection and particle identification are expected to be reduced by 50%. Figure 55 (right) shows the projection of the R_{γ} measurement for Run 3 calculated with these assumptions: the measured values of R_{γ} are kept but the uncertainties are recalculated. The total errors are reduced by $\sim 50\%$. In addition to the reduction of the uncertainties, the large data set foreseen for Run 3 will allow exploration of the 0–1% centrality range.

The ALICE Collaboration carried out measurements of the direct photon elliptic flow [539] in Pb–Pb collisions at $\sqrt{s_{\text{NN}}} = 2.76$ TeV for the two centrality classes, 0–20% and 20–40%, see Fig. 56, left plot for 20–40% centrality. The measured direct photon elliptic flow $v_2^{\gamma,\text{dir}}$ is compared to the estimated decay photon elliptic flow $v_2^{\gamma,\text{dec}}$, marked as cocktail, and to the predictions of several theoretical models. Similar to RHIC measurements, the direct and decay photon elliptic flow are very close and systematically higher than theoretical predictions of hydrodynamic [565, 566] and the transport [567] models. However, because of the large uncertainties one can not presently exclude either of theoretical calculations. Using the same assumption concerning photon and neutral measurements in Run 3 as for R_{γ} , the expected accuracy of $v_2^{\gamma,\text{dir}}$ measurements in Run 3 is estimated, see Fig. 56 (right). The mean

values are kept the same but the uncertainties are reduced as expected. Similar to R_{γ} with the current assumptions the total errors will be reduced by factor ~ 2 and one will be able to exclude or confirm available theoretical calculations.

8.1.2 Dileptons

The sensitivity to the expected signal of thermal radiation and an in-medium modification of the ρ spectral function in the dielectron and dimuon channels with the ALICE detector [74,537] was studied already in preparation for ITS upgrade in 2019/20 [55, 56, 462, 467]. The measurement of low-mass dileptons after this upgrade will profit from

- an improved vertex resolution, which leads to a better separation of electrons from prompt sources, like thermal radiation, and electrons from the decays of heavy-flavour hadrons, for which $c\tau$ is about 150 µm (open-charm hadrons) or 400 µm (open-beauty hadrons),
- a reduced material budget and improved tracking efficiency at low transverse momentum $p_{\rm T}$, which leads to a smaller background of electrons and positrons from photon conversion in the detector material,
- the installation of the muon forward tracker, that will lead to an improved mass resolution and reduced background in the dimuon channel,
- and a higher rate capability (50 kHz in Pb–Pb collisions) that will increase the expected number of events in the central barrel detector by a factor of 100. The increased rate capability also enables the possibility to record in a single Pb–Pb run a large data sample with reduced magnetic field value in the ALICE central barrel (B = 0.2 T instead of 0.5 T), which increases the phase-space acceptance and the reconstruction efficiency of low momentum electrons and positrons.

The expected measured spectra discussed in this section closely follow the strategy that is discussed in more detail in [55, 56, 462, 467].

For the dielectron channel an integrated luminosity $L_{int} \approx 3 \text{ mb}^{-1}$ is assumed, which should be collected in the dedicated Pb–Pb run at low field. The corresponding number of events in central (0–10%) collisions is 2.5×10^9 . The input for the signal is composed of:

- contributions from the decays of long-lived light pseudoscalar and vector mesons (hadronic cocktail consisting of $\pi^0, \eta, \eta', \omega$, and ϕ), with particle ratios and spectral shapes extrapolated from existing heavy-ion data at lower energies,
- correlated semileptonic charm decays based on calculations from the PYTHIA event generator [568],
- and the radiation of thermal dileptons and a medium-modified spectral function for the ρ meson in a realistic space-time evolution (see Fig. 54 (left)).

With respect to earlier calculations [55, 56, 462] a fast simulation of central Pb–Pb collisions is used here to estimate the combinatorial background and the statistical significance of the signal. The particles are produced with the event generator HIJING [569] and then propagated through the detector material by GEANT3 [570]. An updated geometry of the ITS is utilised in the detector description and leads to a more realistic treatment of conversion electrons and the subsequent background. Electrons are reconstructed and identified via signals in the ALICE Time Projection Chamber (TPC) and Time-Of-Flight (TOF) detector, a parametrised efficiency from runs at low magnetic field during LHC Run 2 is applied. After pairing electrons and positrons an additional selection on the pair distance of closest approach

$$DCA_{ee}(\sigma) = \sqrt{\left(\text{DCA}_{xy,1}/\sigma_{xy,1}\right)^2 + \left(\text{DCA}_{xy,2}/\sigma_{xy,2}\right)^2}$$
(29)

is applied to reduce the contribution from correlated semileptonic charm decays. The selection is chosen such that 95% of these pairs are rejected, while having an efficiency for prompt pairs of $\sim 17\%$. The



Fig. 57: Inclusive e^+e^- (left) and $\mu^+\mu^-$ (right) invariant mass spectrum for 0–10% most central Pb–Pb collisions at $\sqrt{s_{NN}} = 5.5$ TeV. The green boxes show the systematic uncertainties from the combinatorial background subtraction.

signal distribution S, which includes the remaining charm and beauty hadron decays, is obtained by subtraction of the combinatorial background from all e^+e^- pairs. The combinatorial background B is estimated from like-sign pairs and a correction factor R that takes into account the different acceptance of the apparatus for unlike- and like-sign pairs [545–547]. The significance that is used to project the statistical uncertainty on the measurement is calculated as $S/\sqrt{S+2B}$. The signal S is shown in Fig. 57 (left) together with all input distributions. In order to extract the QGP component and the in-medium modified ρ spectral function, the hadronic cocktail and the contribution from correlated semileptonic charm decays is subtracted and shown in Fig. 58 (left). In addition, the systematic uncertainties from the combinatorial background and signal extraction, as well as physical backgrounds after subtraction are shown. The relative systematic uncertainty of the signal, from tracking, track matching, and particle identification, is assumed to be 10%. For the systematic uncertainty of B a mass independent uncertainty of the R factor of 0.02% is used. Relative systematic uncertainties from the light-hadron cocktail and the total charm cross section of 10% and 15%, respectively, are applied.

In the dimuon channel, the integrated luminosity of Pb–Pb collisions $(L_{int} \approx 10 \text{ nb}^{-1})$ is used. In this channel, the main source of background is represented by the combinatorial pairs of muons coming from uncorrelated semimuonic decays of light-flavoured mesons, mainly pions and kaons, copiously produced in high-energy nuclear collisions. The opposite-sign dimuon mass spectrum obtained after the subtraction of the combinatorial background evaluated by means of an event mixing technique, results from the superposition of several opposite-sign correlated dimuon sources, represented in the right panel of Fig. 57. In order to isolate the thermal dimuon radiation and the in-medium modified line shapes of the ρ meson, the known and well-identifiable sources of the hadronic cocktail — 2-body and Dalitz decays of the η , ω , ϕ mesons, for which no in-medium effect is expected — are subtracted from the total opposite-sign correlated dimuon mass spectrum. A 10% systematic uncertainty in the evaluation of the shape and the normalization of these sources has been considered in the performance studies. The same procedure has been also applied for the subtraction of the dimuons from the open charm and open beauty processes; alternatively, these two sources could be separated from the prompt ones by means of an analysis based on the discrimination of the dimuon offset at the primary vertex.

The spectral function of low-mass dielectrons and dimuons in the mass region of the modified ρ -meson spectral function $M_{\rm ee} \approx 0.5 \text{ GeV/c}^2$ can be extracted with a systematic uncertainty of $\approx 40\%$



Fig. 58: Excess (after subtraction of light hadron decays and from correlated charm semileptonic decays) e^+e^- (left) and $\mu^+\mu^-$ (right) invariant mass spectrum for 0–10% most central Pb–Pb collisions at $\sqrt{s_{NN}} = 5.5$ TeV. The green boxes show the systematic uncertainties from the combinatorial background subtraction, the magenta (left) and blue (right) boxes indicate systematic errors related to the subtraction of the cocktail and charm contribution.

and $\approx 20\%$, respectively (see Fig. 58). The sizeable contribution of thermal dilepton pairs above $M_{\rm ee} > 1.1 \text{ GeV/c}^2$ can be used to extract the temperature of the system. An exponential fit with $dN/dM_{\rm ee} \sim M_{\rm ee}^{3/2} \exp(-M_{\rm ee}/T_{\rm fit})$ to the subtracted e^+e^- spectra in the invariant mass region $1.1 < M_{\rm ee} < 2.0 \text{ GeV/c}^2$ was performed. Comparing the fit parameter $T_{\rm fit}$ to the real temperature $T_{\rm real}$ from the fit to the thermal contribution, a statistical uncertainty of 5% and systematic uncertainty of 10% and 20% for the background and the charm subtraction, respectively, were estimated. The same kind of measurement is also expected to be possible in the dimuon channel, considering a dedicated set of cuts optimized for the analysis of the intermediate mass region (the cuts considered in the right panel of Fig. 58 being optimized for the signal extraction in the mass region below $\sim 1 \text{ GeV/c}^2$).

An alternative method to separate the thermal component from the modified heavy-flavour production in the intermediate mass range, is to fit the measured DCA_{ee} distribution as a function of the dielectron invariant mass and pair transverse momentum with a three component function, including the contributions from prompt dielectron sources, from open-charm hadron decays and from open-beauty hadron decays. Since the shape of the heavy-flavour DCA_{ee} spectra is quasi model independent, the dielectron yield of open heavy-flavour decays in the ALICE acceptance can be determined from the data with small uncertainties, without relying on theoretical calculations. Such fits were performed already with the Run 1 data in pp collisions at $\sqrt{s} = 7$ TeV [546]. Nevertheless, the statistics available did not allow for a differential study.

The measurement of the dielectron elliptic flow coefficient v_2 as a function of $M_{\rm ee}$ in peripheral Pb–Pb collisions (40–60%) was studied already in [462]. It was shown that an absolute statistical uncertainty on v_2 of $\sigma_{v_2} \approx 0.01$ can be achieved. Such uncertainties present a very encouraging prospect for dilepton studies since temperature dependent shear viscosity [563], bulk viscosity [562, 571], and early stages of reaction dynamics [564] have effects on the order of a few up to tens of percent on dilepton v_2 , and thus future constraints on these properties will greatly benefit from a statistically improved dilepton v_2 measurement.

8.2 Two-photon and photonuclear interactions

Heavy ions carry strong electromagnetic fields. Their electric and magnetic fields are perpendicular, so may be treated as a flux of nearly-real photons [572]. These photon fields can give rise to photonuclear (photon on nucleus) and two-photon interactions. Although these interactions are expected to occur in both ultra-peripheral (UPC) and more central collisions, they were not generally expected to be visible in non-UPC collisions. The few final state particles from the photon-mediated interactions were expected to be swamped by the more copious hadronically produced particles. That expectation changed recently, when ALICE [573] and then STAR [574,575] and ATLAS [576] observed excesses of dileptons produced at very small pair $p_{\rm T}$, $p_{\rm T} < 100$ MeV/c. These pairs were prominent in Pb–Pb and Au–Au collisions, but not in pp interactions; the excess corresponded to $R_{\rm AA} > 5$. This is inconsistent with all expectations for hadroproduction, but consistent with photoproduction, where the pair $p_{\rm T}$ scale is set by the nuclear radius $R_{\rm A}$, with $p_{\rm T} \approx \hbar/R_{\rm A}$. The kinematics of these pairs may be affected by the medium in which they are produced or propagate, so they may probe the evolving Quark–Gluon Plasma or associated magnetic fields.

UPC photon-mediated interactions have been studied at both RHIC and the LHC [572, 577–580]. The agreement between data and calculations is quite good. Photoproduction of ρ , ω , ρ' , J/ψ , J/ψ' , Y and direct $\pi^+ \pi^-$ pairs has been observed, along with two-photon production of dilepton pairs and light-by-light scattering. In peripheral collisions, photon-mediated interactions might be used to probe the nuclear medium that they may occur in, including the QGP [574, 576]. The produced leptons may interact with this medium, leading to alterations in their momentum.

Peripheral collisions introduce several new considerations for photon-mediated reactions, particularly evolving coherence conditions for both photon emission and coherent photon-nucleus scattering. Photon emission in both $\gamma\gamma$ and photonuclear interactions is expected to be completely coherent, governed by the nuclear form factor F(q) [581]. The photon emission from a nucleus moving with Lorentz boost γ should occur before the hadronic interaction (which is taken to occur at t = 0), at a retarded time, t - x/c [582], where $x = |b|/\gamma$; |b| is the transverse distance from the photon emission point to where it interacts. For very small impact parameters, some coherence may be lost, and a more detailed calculation is needed. For photon-nucleus collisions, the situation is more complicated, and will be discussed below.

Here, two-photon interactions and coherent photonuclear interactions are discussed.

8.2.1 Two-photon interactions

In two-photon interactions, each nucleus emits a photon, which then interact and form a lepton pair. In UPCs, this process is well described by the Weizsäcker-Williams approach (where each photon is treated as real), except at very low pair $p_{\rm T}$, where a lowest-order QED calculation works better [583]. UPC calculations can be easily extended to include peripheral collisions [584–586]. The kinematic distributions are similar to those in UPCs, and the cross-section depends on the range of impact parameters.

Recently, the ATLAS collaboration [576] presented results showing a dramatic modification to $\gamma\gamma \rightarrow \mu^+\mu^-$ in peripheral collisions. Figure 59 shows the pair acoplanarity α , the azimuthal angular deviation from being perfectly back-to-back, and A, the energy imbalance between the two leptons. For UPCs, they found good agreement with the STARlight [587,588] reference, with the data and calculations peaked at small α and A. More central collisions show dramatic changes with the low- α peak largely disappearing, and the A distributions only minimally changed. ATLAS described this as "Consistent with order of magnitude estimates from kinetic theory for multiple scattering off electric charges in thermal plasma." Multiple scattering would remove the peak at low α , but leave A largely unaffected. A recent calculation finds that the magnitude of the change in α is at least roughly consistent with that expected for leptons propagating through a Quark–Gluon Plasma [589]. If multiple scattering is large, though, one might also expect some bremsstrahlung, which should increase A. To evaluate this further requires a calculation of how many of the produced leptons are produced in the medium, and/or traverse it. An



Fig. 59: Acoplanarity (α , top) and lepton energy imbalance (A, bottom) as a function of centrality, for dimuon pairs with pair mass above 10 GeV/c², observed in the ATLAS detector. From Ref. [576].

alternate explanation could involve the leptons bending in the magnetic field from the QGP. If a QGP is electrically conducting, then it may acquire an induced magnetic field from the short-lived magnetic fields carried by the two nuclei [590]. The QGP field, however, will be longer lived, and could bend the produced leptons in opposite directions, reducing their coplanarity. Symmetry also predicts that it should disappear for the most central collisions [589], except possibly for event-by-event fluctuations.

The STAR Collaboration also has studied two-photon e^+e^- production in peripheral Au–Au collisions; they found a small difference between their pair p_T spectrum and calculations, and suggest that it might be due to medium effects [574]. ALICE has not yet seen these pairs [573], likely because their pair acceptance requires lepton $p_T > 1$ GeV/c, eliminating most pairs from $\gamma\gamma$ reactions.

Coupled with better theoretical calculations, the large Pb–Pb integrated luminosity in Run 3 and 4 can confirm and dramatically expand our understanding of this effect. One important goal is to expand the study to cover a much wider range of masses. Figure 60 shows the expected mass spectrum obtainable by ATLAS for a 13 nb^{-1} integrated luminosity run, assuming no changes in the trigger; masses up to 100 GeV/ c^2 should be accessible. These high mass pairs correspond to two-photon interactions in or very near the two nuclei, so should show increased effects due to interactions with the medium or magnetic fields associated with the Quark–Gluon Plasma.

In contrast, lower masses correspond to larger distances between the dilepton production point and the nuclei, so in-medium effects may be smaller. These lower masses should be accessible with a softer requirement on the muon momentum. It would also be interesting to compare e^+e^- with $\mu^+\mu^-$ (and possibly $\tau^+ \tau^-$), since the lighter leptons should interact more. If the leptons interact with the medium, then the electron A distribution should show more change than that for muons.

8.2.2 Photonuclear interactions

In photonuclear interactions, a photon emitted by one nucleus fluctuates to a quark-antiquark dipole, which then scatters elastically from the other (target) nucleus, emerging as a real vector meson. The scattering occurs via Pomeron exchange, which preserves the photon quantum numbers. In perturbative QCD, Pomerons are made up of gluons, so the process is sensitive to the gluon distribution in the target nucleus. UPC measurements are consistent with moderate gluon shadowing. In coherent scattering, the typical pair $p_{\rm T}$ is $\hbar/R_{\rm A}$. Incoherent scattering is also possible, with a lower cross-section. There the quark-antiquark dipole scatters elastically from a single nucleon (or, at still higher $p_{\rm T}$ inelastically from a single nucleon), producing a vector meson with a typical $p_{\rm T}$ of a few hundred MeV/c.



Fig. 60: Expected dimuon yield in ATLAS acceptance (both muons with $p_{\rm T} > 4$ GeV/c and $|\eta| < 2.4$), for 40–80% centrality Pb–Pb collisions and the expected Run 3/4 integrated luminosity of 13 nb⁻¹. Masses up to 100 GeV/c² are accessible. The effective 8 GeV/c² minimum mass is because of the nearly back-to-back topology and the 4 GeV/c minimum muon $p_{\rm T}$ cut. This was calculated using STARlight [584, 588].

Both ALICE [573] and STAR [575] have observed coherent J/ψ photoproduction in peripheral heavy-ion collisions. There are a number of parallel theoretical calculations [582, 591]. The photon emission process is similar to the two-photon case, but the dipole-nucleons scattering happens at the same time as the hadronic interaction, introducing several complications to the calculations. This immediately raises several questions: What happens to the coherence if a target nucleon is involved in an interaction? Does the dipole-nucleon interaction occur before or after the nuclear collisions? If the hadronic interaction occurs first, the target nucleon will have lost energy, so the photon-nucleon crosssection will be smaller. A detailed calculation should consider both possibilities. There is also destructive interference between photoproduction from the two possible target nuclei [592]; this interference extends to higher $p_{\rm T}$ for more central collisions, and should reduce the cross-section for the region where nuclear collisions occur. At b = 0, we expect complete destructive interference. Ref. [582] makes predictions for a variety of coherence conditions, and as Fig. 61 shows, finds that the ALICE and STAR data likely lie below the region where there is complete coherence for both photon emission and scattering, but probably above that where coherence is limited to only the spectator nucleons. This is not surprising, but there is at least one element missing from this calculation. The lifetime of J/ψ particles is of the order 10^{-20} s, far shorter than that of the expanding Quark–Gluon Plasma. Coherently photoproduced J/ ψ have $p_{\rm T} \sim 100$ MeV/c, so, near mid-rapidity, are moving at a small fraction of the speed of light. Particularly for more central collisions, one would expect many of them to be engulfed by the expanded QGP, before they have a chance to decay.

The ALICE error bars are large, and more data, from the current and future runs are needed to pin down the centrality dependence of the cross section. More data will also allow access to additional observables. A detailed study of the shape of $d\sigma/dp_T$ would shed more light on the possible loss of coherence in more central collisions. There are also expected correlations between the reaction plane, which can be determined from the hadronic part of the collision, with the photonuclear interaction. Because the destructive interference between photoproduction at mid-rapidity on the two nuclei goes as $\sigma \sim |1 - \exp(i\vec{b} \cdot \vec{p_T})|^2$ [593], the azimuthal direction of $\vec{p_T}$ provides information about the azimuthal direction of \vec{b} , i.e., the reaction plane. Thus, it can be used either as an independent measurement of the reaction plane, or as a test of the loss of correlation. Also, the J/ ψ polarization follows that of the photon that produced it, so it also follows \vec{b} , providing another probe of the reaction plane. With a large data sample, one may also be able to probe incoherent J/ ψ photoproduction, at least in very peripheral hadronic collisions, where the signal-to-noise ratio is high.

It will be very interesting to study ψ' and Y photoproduction in peripheral collisions. Since these



Fig. 61: J/ψ coherent photoproduction cross-sections in peripheral collisions, as a function of the number of participants (bottom), and impact parameter (top) with Au–Au collisions at RHIC (left) and Pb–Pb at the LHC (right). The four curves are for different assumptions regarding centrality for the photon emitter (first particle listed) and the target (second particle listed). From Ref. [573, 582].

mesons have different sizes from the J/ψ , they should interact with the medium with different strengths. These studies should be possible at HL-LHC.

8.3 Dark photons

Dark Matter is a hypothetical form of matter that is responsible for accounting for approximately 80% of the matter in the Universe [594]. Dark matter cannot be incorporated into the Standard Model, so the introduction of dark matter requires new interactions between dark matter particles and the ordinary Standard Model particles via unknown dark-sector forces [595]. The dark sector could have a rich structure with a few possible candidates, where one of them is regarded as Dark Photon (A') with $L_{mix} \propto \frac{g}{2}F^{\mu\nu}X_{\mu\nu}$. The dark photon is introduced as an extra-U(1) gauge boson and acts as a messenger particle of a dark sector with the residual interaction (g) to the Standard Model particles. Understanding of possible interactions of dark photons has been motivated by a number of astrophysical anomalies such as antiproton spectrum in the cosmic rays measured by AMS Collaboration, positron excess in the cosmic rays observed earlier by PAMELA [596] and confirmed by FERMI [597] and AMS [598], and the long standing discrepancy between the measured and the calculated anomalous magnetic moment of the muon $(g-2)_{\mu}$, where the difference is more than three standard deviations away from zero [599].

If the dark photon is the lightest state of the dark sector and therefore can decay only into the Standard Model particles, dark photons with mass $m_{A'} \leq 2m_{\mu}$ decay only into electron-positron pairs. For dark photons above 2 muon threshold $(m_{A'} \geq 2m_{\mu})$, dark photons can decay into muon pairs. For $(m_{A'} \geq 2m_{\pi})$, dark photons can decay into hadrons as well. A lot of experimental activities have been seen recently and constraints of mixing parameter (g^2) as a function of dark photon mass $(m_{A'})$ has been done from many experiments. They are, for example, beam-dump experiments (measurement of lepton pairs from dark photons behind a sufficiently long shield. Examples are E141 [600] and E137 [601] at SLAC, E774 [602] at Fermilab), fixed-target experiments (by scattering the electron beam on a nuclear target, the dark photon may be emitted in the initial or final state and coupling to electron-positron pairs is studied by looking for a bump in the electron-positron invariant mass. Examples are A1 [603] at MAMI in Mainz, APEX [604] at JLAB, DarkLight [605] at JLAB) and collider experiments (BABAR [606], NA48/2 [607] at SPS, WASA [608] at COSY, HADES [609] at GSI, PHENIX [610] at



Fig. 62: 90% of confidence level of mixing parameter as a function of dark photon mass. Figure is adapted from Ref. [614]. Red and blue are from ALICE and LHCb [611]. Light grey band contains results from BABAR, KLOE, A1, APEX, NA48/2, E774, E141, E137, KEK, Orsay, BESIII, CHARM, HPS, NA64, NOMAD, NuCAL, and PS191 [614].

RHIC, LHCb [611] and ALICE at LHC). Since any process in which a virtual photon couples to lepton pairs or hadrons can be used to search for dark photons, following processes are used in the collider experiments: measurements of Dalitz decays of the $\pi^0/\eta/\eta' \rightarrow \gamma A'$ mesons and rare meson decays such as $K \rightarrow \pi A'$, $\phi \rightarrow \eta A'$, and $D^* \rightarrow D^0 A'$, Bremsstrahlung process ($e^- Z \rightarrow e^- Z A'$ with A' emitted at very forward direction), radiative decay of vector resonances and initial state radiation (done by BABAR using radiative decays of Y(3S) and done by KLOE [612] using $\phi \rightarrow e^+ e^-$). The 79-string IceCube search for dark matter in the Sun public data is used to test Secluded Dark Matter models [613]. Dark matter particles can be captured by the Sun, annihilate, and produce a neutrino flux that can be observed at Earth and that depends on the dark matter scattering cross section off nuclei and on the dark matter annihilation rate and final states. This analysis constrains a kinetic mixing parameter $g \sim 10^{-9}$ between 0.22 and ~ 1 GeV [613].

ALICE has good capabilities for electron identification in the low transverse momentum region, that enables the measurement of a large sample of the π^0 Dalitz decays [546]. ALICE searches for possible decays of $\pi^0 \rightarrow \gamma A', A' \rightarrow e^+e^-$ by examining the electron-positron invariant mass in a large sample of π^0 Dalitz decay for $20 \le M_{\rm ee} \le 90 \text{ MeV/c}^2$ in pp collisions at 7 TeV ($L_{\rm int} \sim 4 \text{ nb}^{-1}$) and p-Pb collisions at 5.0 TeV ($L_{\rm int} \sim 40 \text{ µb}^{-1}$) as shown in Fig. 62.

LHCb has good capabilities to measure muons and hardware and software triggers enable the accumulation of a large sample of dimuon pairs. LHCb searches for prompt-like and long-lived dark photons produced in pp collisions at 13 TeV, using $A' \rightarrow \mu^+\mu^-$ decays from a large data sample corresponding to $L_{\text{int}} \sim 1.6 \text{ fb}^{-1}$ collected during 2016, where the prompt-like A' search is shown in Fig. 62 [611].

The ALICE upgrade during LS2 will greatly improve the efficiency of electron-positron measurements and data taking capability. Figure 63 shows expected constraints that will be achieved by ALICE and LHCb together with the future experiments. After the major ALICE upgrade, ALICE will accumulate 6 pb^{-1} , 0.3 pb^{-1} , 10 nb^{-1} , 0.3 pb^{-1} , and 3 nb^{-1} of pp, p–Pb, and Pb–Pb collisions at



Fig. 63: 90% of CL constrained by ALICE and LHCb in HL-LHC era. Constraints by ALICE are based on 6 pb^{-1} , 0.3 pb^{-1} , 10 nb^{-1} , 0.3 pb^{-1} , and 3 nb^{-1} of pp, p–Pb, and Pb–Pb collisions at 0.5 T, and p–Pb and Pb–Pb collisions at 0.2 T and by LHCb are based on 15 fb^{-1} . Figures are adopted from Ref. [614].

0.5 T, and p–Pb and Pb–Pb collisions at 0.2 T, respectively. LHCb will improve sensitivity of dark photon searches to large regions of the unexplored space. These new constraints leverage the improved invariant-mass and vertex resolution, as well as the unique capabilities of the particle-identification and real-time data-analysis with triggerless readout, that enables to accumulate $L_{int} \sim 15$ fb⁻¹ [615].

8.4 Limitations and outlook

While the statistical precision for the measurement of low mass dielectrons and dimuons as well as real photons will be sufficient in LHC Run 3 and 4 to study their yield as a function transverse momentum and with respect to the event plane (elliptic flow), more differential measurements might still be limited. The measurement of the photon polarization via the angular distribution of dileptons can not only provide information on the thermalization of the system, but also on the early stages of the collision [616]. Experimentally these distributions have been measured in the NA60 experiment [617], where no polarization was found concluding that the observed excess dimuons are in agreement with the thermal emission from a a randomized system. In order to study the angular distributions, for example in the Collins-Soper reference frame [618–620] in the polar angle θ and the azimuthal angle φ , a large data set is needed (NA60 used ~ 50000 excess $\mu^+\mu^-$ pairs).

Another promising direction is measurement of Bose-Einstein (BE) correlations of direct photons. With this probe one can trace space-time dimensions of the hottest part of the fireball and moreover, varying $k_{\rm T}$ of the photon pair, one can select pairs coming mostly from earlier or later stages of the collision and thus look at evolution of the fireball. On the other hand, from the correlation strength parameter one can extract the direct photon spectrum down to very low $p_{\rm T} \sim 100$ MeV/c. So far there was one successful measurement of direct photon BE correlations by the WA98 Collaboration [621], while at RHIC and LHC energies these measurements are still unavailable. The reason is that the expected strength of these correlations $\lambda_{PGg} = 1/2(N_{\gamma}^{\rm dir}/N_{\gamma}^{\rm tot})^2$ is extremely small. Moreover,

in contrast to massive particles, averaging of full 3D correlation function $C_2(q_{\text{out}}, q_{\text{side}}, q_{\text{long}})$ to 1D $C_2(q_{\text{inv}})$ results in further dramatic decrease of correlation strength [621]. This requires very large statistics in addition to understanding the detector response.

A first step to increase the statistical precision and the available data set for low-mass dileptons could be a further upgrade of the inner barrel of silicon detectors of the ALICE apparatus that is currently under discussion [292]. The planned reduction of the material budget would reduce conversion probability. In addition, an improvement of the tracking efficiency especially at low momentum would increase the conversion rejection efficiency even further. First studies [292] showed that the statistical uncertainty can be reduced by a factor 1.3, while the systematic uncertainty from the subtraction of the rejection of charm background is improved and would lead to a reduced systematic uncertainty from the subtraction of the subtraction of the light-hadron and charm decay backgrounds by a factor of two.

9 Emergence of hot and dense QCD matter in small systems

Coordinators: Jan Fiete Grosse-Oetringhaus (CERN) and Constantin Loizides (Oak Ridge National Laboratory) **Contributors**: R. Bi (Massachusetts Institute of Technology), C. Bierlich (Lund University and Niels Bohr Institute), E. Bruna (University and INFN Torino), Z. Chen (Rice University), C. Cheshkov (IPN Lyon), Z. Citron (Ben-Gurion University of the Negev), A.F. Dobrin (CERN), M. Dumancic (Weizmann Institute of Science), M. Guilbaud (CERN), P.M. Jacobs (Lawrence Berkeley National Laboratory), J. Jia (Stony Brook University and Brookhaven National Lab), A.P. Kalweit (CERN), F. Krizek (Academy of Sciences, Prague), A. Kurkela (CERN and Stavanger University), Y.-L. Lee (Massachusetts Institute of Technology), N. Mohammadi (CERN), D.V. Perepelitsa (University of Colorado Boulder), R. Rapp (Texas A&M University, College Station), B. Schenke (Brookhaven National Lab), K. Tatar (Massachusetts Institute of Technology), M. Weber (Austrian Academy of Sciences), H. Zanoli (Universidade de Sao Paulo), M. Zhou (Stony Brook University)

9.1 Introduction

In the program of proton–proton collisions at the LHC, the main effort is focused on hard processes which are embedded in an underlying event consisting of soft low- p_T particles. The underlying event is described using models, such as Pythia [506] or HERWIG [622], based on essentially free streaming (i.e. no final-state interactions) of the produced particles, supplemented by a non-perturbative cluster or string fragmentation picture [623, 624] to model the non-perturbative soft-particle production. The same models are used to describe minimum-bias events, i.e. events without any signal trigger, primarily consisting of soft QCD interactions. In the past years at LHC, during Run 1 and 2, this picture was challenged by several observations that qualitatively differ from the model expectations and cannot be accommodated by tuning of the existing models used to describe minimum-bias collisions and the underlying event [625].

The first such observation was the unexpected discovery in 2010 of azimuthal correlations of finalstate hadrons in very high multiplicity proton–proton collisions [302], referred to by *the ridge*. These persist at large separation in rapidity on the near side surrounding the jet-like peak. A few years later, a similar observation was made in high multiplicity p–Pb collisions [303]. By subtracting the jet-like contribution in p–Pb collisions, a second long-range rapidity correlation on the away side, back-to-back in azimuth to the first observed correlation, was extracted [304, 305]. Even later, the procedure was adapted to pp collisions, allowing one to identify two long-range contributions also in high-multiplicity pp collisions [626, 627]. Under certain assumptions even lower-multiplicity pp collisions show the same features [626]. With these observations the similarity of small and large collision systems with respect to azimuthal correlations had been clearly demonstrated.

The second observation was that of enhanced production of multi-strange hadrons in highmultiplicity pp collisions extending the puzzle from final-state particle kinematics to include also hadrochemistry. Already after Run 1, several experiments reported that ratios of strange to non-strange particle yields, in minimum-bias collisions, could not be described using model fits obtained from LEP data [628–631]. After systematic studies of this discrepancy, it was found that not only does strangeness increase smoothly with particle density at mid-rapidity in pp collisions, the dependence on this observable continues smoothly to p–Pb and Pb–Pb collisions [632].

Initially, these collision systems were thought as a reference for the effects observed in Pb–Pb collisions. But the discovery of these qualitatively new features has turned the study of small systems into a field on its own, with significant interest in both the heavy-ion and the high-energy physics community.

In ultra-relativistic nucleus–nucleus collisions, ranging from early SPS experiments at CERN through the Relativistic Heavy-Ion Collider (RHIC) at Brookhaven National Laboratory (BNL) to LHC, similar observations have been interpreted as evidence of formation of a droplet of thermalised Quark–Gluon Plasma. The long-range azimuthal correlations, and in particular their lowest harmonic component v_2 , have been used in combination with relativistic fluid-dynamical modeling to constrain the material

properties of the plasma. The striking result from RHIC was that the plasma formed in central nucleus– nucleus collisions flows as a liquid nearly without dissipation such that its specific shear viscosity η/s – quantifying the dissipative properties of the medium – was found to be smaller than that of any other known substance. The inferred value of the specific shear viscosity $\eta/s \sim 0.07 - 0.16$ was found to be significantly smaller than the expectation from perturbative QCD and other quasiparticle models, and closer to the expectation of holographic model calculations of strongly coupled (maximally supersymmetric N = 4) gauge theories in the limit of large number of colors $N_c \to \infty$. These models can be seen as models of fluids with minimal dissipation allowed by basic principles of quantum mechanics thus giving rise to the paradigm of Quark–Gluon Plasma as a perfect liquid. Perfect liquid models do, by definition, not have any quasi-particle structure. This means that they do not have any degrees of freedom which can free stream for an appreciable amount of time, compared to their de Broglie wavelength. In that context it is notable that the observation of fluid–like signatures in small systems can be described as a small modification of the free streaming evolution, challenging the perfect-fluid paradigm.

There are several theoretical pictures that have been suggested to explain the smooth onset of signals of collectivity in small systems. The pp event generators have been supplemented on the one hand with elements describing string or cluster fragmentation in a dense medium [633, 634] to address the hadrochemistry, and on the other hand with final-state interactions between the fragmenting strings to account for the final-state kinematical correlations [635]. The models underlying pp event generators can then in turn be extrapolated to cover p–Pb and Pb–Pb collision systems, which is an approach used since the 1980's [569, 636]. Recent theoretical developments [637, 638] have improved the state of such extrapolations to a degree where also the supplemented hadronisation models can be extrapolated in order to provide a microscopic picture of the QGP even in large systems. The question whether such extrapolations will give even a qualitative description of the observed features is still open.

At the same time, the description of large systems has been employed in regimes initially thought to be not accessible by models implementing the perfect-fluid paradigm. Their application down to proton–proton collisions [197, 639, 640], taken at face value, would imply the formation of a nearly perfect liquid even in the smallest collision systems. Furthermore, pQCD based saturation models can describe the emergence of charged-particle azimuthal anisotropy (v_n) [641]. In these models the finalstate azimuthal correlations can arise either from the intrinsic correlations in the nuclear wave function (initial-state correlations) as correlated anisotropic particle production or as a final-state interaction after the initial particle production. In heavy-ion collisions, statistical models [642] have been very successful in describing the hadrochemistry of particle yields. Their extension to pp collisions shows promise, but similarly with points of tension [643]. Regardless of whether the approach is to extrapolate from pp to Pb–Pb collisions or the other way around, it is crucial to establish that any such model can capture the essential features of intermediate systems. Asymmetric collision systems such as p–Pb, provide challenges and opportunities for both approaches. They may be used as a necessary intermediate stepping stone for pp to A–A extrapolations, and they provide possible discrimination between saturation and fluid approaches as a possible point of tension [644,645].

A question remains to what extent these different models are describing qualitatively different physical phenomena or to what extent they are different representations of the same underlying physics of final-state interactions. For this reason it is important to develop theoretical tools that encompass both the fluid-dynamic limit and the free-streaming limit to theoretically describe how the microscopic physics that leads to fluid-dynamic behavior in A–A collisions should represent itself in small systems. An attempt to do so comes from transport theory, which can describe microscopic interactions but in the limit of large number of final-state interactions allows for a coarse-grained effective description in terms of fluid dynamics. As such, transport theory has the potential to bridge the gap between small systems, where final-state interactions act as small modification to the free-streaming evolution, and central nucleus–nucleus collisions where the final-state interactions bring the matter to the fluid-dynamical limit [202].
The experimental program in the large intermediate region — spanning from mid-central Pb–Pb and Xe–Xe collisions, through p–Pb collisions down to minimum-bias pp collisions — offers a possibility to bridge the difference between the two limits by providing a setup where the microscopic final-state interactions that lead in central Pb–Pb collision to the formation of a QGP may be studied in isolation in the limit of small number of final-state interactions.

The suggested theoretical pictures may have implications for high-energy physics analyses, which depend on reliable models of the underlying event. As an example, it has been recently shown that the discussed long-range correlations are also present in the underlying event of Z-tagged pp collisions [646]. The direct implication is the necessity for questioning the correct description of the underlying event of MC models. As the usual models used to describe the underlying event do not describe such long-range correlations, even qualitatively, the uncertainty introduced by imposing a model dependence, might be larger than expected, as *e.g.* shown for colour reconnection effects on $t\bar{t}$ final states [647]. As such, better descriptions of collective effects in small systems could also probe vital for reducing uncertainties in high-energy physics analyses.

The main experimental task in future years is a detailed examination and characterization of the observed effects in pp, p-Pb and Pb-Pb collisions, in order to understand whether such effects are different or similar in origin in small and large systems. For such a task to be successful, all three types of collision systems, pp, p-Pb and Pb-Pb must be utilized, as they each offer unique features not obtainable from the other systems. The central Pb-Pb collision system is so far the only one where all features of collectivity (including multi-particle correlations, jet quenching, quarkonia supression, thermal photons and hadrochemistry) have been observed. For the study of small collision systems, central Pb–Pb offers the only viable *true collective* reference. Conversely, pp is so far the smallest collision system where collective effects have been observed, and the only system where a smooth transition to the e⁺e⁻ expectation could be reasonably expected. In the intermediate region p-Pb collisions are the only one of the three collision systems which offer, both, a saturation dominated initial state with a well known geometry, and, in a single event the Pb-going and p-going direction allowing the study between both regimes. The potential study of O-O collisions provides an interesting system with smaller fluctuations in the number of participating nucleons. Furthermore, the detailed study of asymmetric collision systems provides valuable input both to models extrapolating pp dynamics to Pb–Pb collisions, and for providing quantitative distinction between initial-state saturation effects and final-state interactions.

This chapter is structured as follows: Section 9.2 gives an overview presenting the observations that have been made and comparing them between pp, p–Pb and Pb–Pb collisions. Subsequently, Sect. 9.3 summarizes the open questions and discusses how these can be addressed at HL-LHC. Sect. 9.4 details the multiplicity distribution which needs to be extrapolated to be used for performance studies, the expected energy densities and the data-taking conditions assumed. A set of performance studies which address the open questions are introduced in Sect. 9.5–9.9 ranging from correlation measures, over hadrochemistry to signatures of energy loss and thermal radiation. Finally, Sect. 9.10 presents the opportunities of a short run of O–O collisions.

9.2 Overview of experimental results and critical assessment

This section will give an overview where the measurements in Pb–Pb, p–Pb and pp collisions at LHC provide a consistent picture and where differences emerge. In addition, it is pointed out where measurements are missing and where additional data is needed which forms the basis for the projections given in the subsequent sections of this chapter. Table 7 lists the different observables and if they have been measured in the Pb–Pb, p–Pb and pp collision systems. In the following a critical assessment of the findings is performed.

Particle spectra In all three systems, the $p_{\rm T}$ spectra of identified particles harden with increasing

Observable or effect	Pb-Pb	p–Pb (high mult.)	pp (high mult.)	Refs.
Low $p_{\rm T}$ spectra ("radial flow")	yes	yes	yes	[34, 59, 308, 309, 648, 651, 657, 658, 661, 662]
Intermediate $p_{\rm T}$ ("recombination")	yes	yes	yes	[308, 651–657]
Particle ratios	GC level	GC level except Ω	GC level except Ω	[309, 632, 658, 659]
Statistical model	$\gamma^{ m GC}_s=1,10 ext{-}30\%$	$\gamma^{ m GC}_s pprox 1, 20{-}40\%$	MB: $\gamma_s^{\mathrm{C}} < 1, 2040\%$	[309, 632, 663]
HBT radii $(R(k_{\rm T}), R(\sqrt[3]{N_{\rm ch}}))$	$R_{ m out}/R_{ m side}pprox 1$	$R_{ m out}/R_{ m side} \lesssim 1$	$R_{ m out}/R_{ m side} \lesssim 1$	[664-671]
Azimuthal anisotropy (v_n)	$v_1 - v_7$	$v_1 - v_5$	$v_2 - v_4$	[35, 303 - 305, 626, 627, 646, 672 - 682]
(from two particle correlations)				
Characteristic mass dependence	$v_2 - v_5$	v_2, v_3	v_2	[35, 306, 318, 677, 680, 683-685]
Directed flow (from spectators)	yes	no	no	[686]
Charge-dependent correlations	yes	yes	yes	[235, 239, 240, 687-690]
Higher-order cumulants	".4 $\approx 6 \approx 8 \approx LYZ$ "	" $4 \approx 6 \approx 8 \approx LYZ$ "	$4 \approx 6$,	[307, 677, 682, 691-702]
(mainly $v_2\{n\}, n \ge 4$)	+higher harmonics	+higher harmonics		
Symmetric cumulants	up to $SC(5,3)$	only $SC(4, 2), SC(3, 2)$	only $SC(4, 2), SC(3, 2)$	[213, 681, 703–706]
Non-linear flow modes	up to v_6	not measured	not measured	[707]
Weak η dependence	yes	yes	not measured	[679, 701, 708 - 713]
Factorization breaking	yes $(n = 2, 3)$	yes $(n = 2, 3)$	not measured	[676, 678, 714 - 716]
Event-by-event v_{n} distributions	n = 2-4	not measured	not measured	[717-719]
Direct photons at low $p_{\rm T}$	yes	not measured	not observed	[538,720]
Jet quenching through dijet asymmetry	yes	not observed	not observed	[340, 352, 367, 721 - 723]
Jet quenching through R_{AA}	yes	not observed	not observed	[314, 336, 338, 339, 344, 724 - 731]
Jet quenching through correlations	yes (Z-jet, γ -jet, h-jet)	not observed (h-jet)	not measured	[346, 349, 368, 369, 373, 381, 727, 732 - 734]
Heavy flavor anisotropy	yes	yes	not measured	[248, 318, 453 - 457, 491, 735 - 739]
Quarkonia production	$suppressed^\dagger$	suppressed	not measured	$[248,447,449,452,472,473,485,486,488,\\489,491,573,740-749]$
		-		

Table 7: Summary of bulk observables or effects in Pb–Pb collisions, as well as in high multiplicity p–Pb and pp collisions at the LHC. References to key measurements for the various observables and systems are given. See text for details. Table adapted from Ref. [660].

 $\dagger J/\psi \uparrow, Y(\downarrow)$ w.r.t. RHIC energies.

multiplicity. If this is interpreted by using a combined blast-wave parametrisation in Pb–Pb collisions⁴ a larger radial flow is observed in pp and p–Pb collisions at the same multiplicity [649] as expected by Ref. [650]. In the intermediate $p_{\rm T}$ region ($2 < p_{\rm T} < 5$ GeV/c), enhancement of baryon-to-meson ratios is observed in all three systems. Recombination models suggest that the number of constituent quarks of the hadrons determine this enhancement [308, 651–657]. Particle ratios and yields are described as in the Grand Canonical ensemble by the statistical model with the strangeness undersaturation factor $\gamma_S \approx 1$ with an accuracy of approximately 10–30% for Pb–Pb collisions and at 20–40% level for p–Pb collisions (except for the Ω meson). The statistical model has been so far applied to minimum-bias pp collisions and when treated as a canonical ensemble, was found to describe the yields with $\gamma_S^C < 1$ and deviations of only about 20–40% from the expected yields [309, 632, 658, 659].

Pressure-driven expansion and anisotropies Assuming that the pressure gradients build up early in the evolution of the created system, initial spatial anisotropies (ε_n) translate into final momentum anisotropies, namely anisotropic flow (v_n) in a system with small viscosity. A large number of detailed studies have been done on different coefficients of anisotropic flow. Higher-order flow harmonics are in particular more sensitive to initial-state fluctuations and therefore can constrain the initial conditions of the system. Anisotropic flow has been measured with two-particle correlation techniques up to v_7 in Pb–Pb, v_5 in p–Pb, and v_4 in pp collisions for charged particles. These v_n coefficients exhibit weaker multiplicity dependence in pp and p–Pb collisions than in Pb–Pb collisions where this is closely related to the shape of the overlap region [35, 303–305, 626, 627, 672–682].

Higher-order cumulants have been measured using Lee-Yang Zeros (LYZ) method and multiparticle correlation techniques with up to 8 particles in both Pb–Pb and p–Pb collisions and up to 6 particles in pp collisions [307,677,682,691–702]. Interestingly, for each collision system, the measurements of the cumulants at different orders ($n \ge 4$) are similar within 10%. The presence of non-zero higherorder cumulants with similar magnitude can be interpreted as evidence for a hydrodynamically evolving system. However, some disfavor this interpretation since models that do not incorporate hydrodynamics have also been able to reproduce these results [750–752]. The $p_{\rm T}$ -differential $v_{\rm n}$ measurements for identified particles show the characteristic mass dependence of anisotropic flow up to v_5 in Pb–Pb, v_3 in p–Pb, and v_2 in pp collisions where heavier particles are depleted at low $p_{\rm T}$ [35, 306, 318, 677, 680, 683–685]. In Pb–Pb collisions this is ascribed to the interplay between radial flow and anisotropic flow harmonics at low $p_{\rm T}$ and recombination at higher $p_{\rm T}$. This characteristic mass dependence has been described by hydrodynamic calculations to a good approximation in all three systems. In the intermediate $p_{\rm T}$ values in all three systems a meson-baryon grouping can be observed which points to a combination of hydrodynamics and quark coalescence (or recombination).

The non-linear hydrodynamic response of the system has been probed using symmetric cumulants which quantify the correlation between different anisotropic flow harmonics. Symmetric cumulants, that are also known as mixed harmonics, have been measured in all three systems up to SC(5,3) in Pb–Pb and SC(4,2) in p–Pb and pp collisions [213,681,703–706]. Different order harmonic correlations have different sensitivities to the transport properties of the system and the initial conditions. Based on the hydrodynamic calculations the data favour a small shear viscosity [753]. In addition, the linear and non-linear hydrodynamic response has been investigated in Pb–Pb collisions up to the sixth order flow harmonic [707]. These new observables, i.e. linear and non-linear flow modes, are very sensitive to details of the hydrodynamic modelling, i.e. initial conditions and the transport properties of the system. Current data–model comparison show this sensitivity which help to constrain the transport properties of the QGP created in Pb–Pb collisions [707]. Linear (v_n^L) and non-linear (v_n^{NL}) flow modes in pp and p–Pb collisions are not yet measured and can constrain the transport properties as well as initial conditions of these small systems. Furthermore, hydrodynamic calculations capture qualitatively "higher-order"

⁴A combined blast-wave parametrisation model is a blast-wave model that fits charged pions, kaons and (anti-)protons simultaneously. In [648], combined blast-wave parametrisation perfectly describes π^{\pm} (0.5 < $p_{\rm T}$ < 1 GeV/c), K^{\pm} (0.2 < $p_{\rm T}$ < 1.5 GeV/c) and p + $\overline{\rm p}$ (0.3 < $p_{\rm T}$ < 3 GeV/c).

details, such as the breaking of factorization due to event-plane angle decorrelations in p_T and η measured in both Pb–Pb and p–Pb collisions [678, 714, 715]. With the existing data such measurements are not yet possible in pp collisions. Similarly, event-by-event v_n measurements have only been done in Pb–Pb collisions [717–719] and it would be interesting to study those in both p–Pb and pp collisions.

Directed flow, for the rapidity-odd as well as the rapidity-even components, of charged particles at mid-rapidity was measured relative to the collision symmetry plane defined by the spectator nucleons, and evidence for dipole-like initial-state density fluctuations in the overlap region was found in Pb–Pb collisions [686]. In small systems, the concept of directed flow is less clear, especially in pp collisions. If there is collectivity in pp collisions, one could also expect a non-zero directed flow measurement. This is technically challenging since the measurement of the spectator plane is not feasible in small systems and, hence, v_1 could only be measured using higher-order ($n \ge 4$) cumulants. The width of the balance functions, $\langle \Delta \eta \rangle$ and $\langle \Delta \varphi \rangle$, have been measured for charged particles in pp, p–Pb and Pb– Pb collisions [687, 689]. The balance function probes the charge creation time and the development of collectivity in the produced system. These measurements are consistent with the picture of a system exhibiting larger radial flow with increasing multiplicity but also whose charges are created at the later stages of the collision. The charge-dependent azimuthal correlations are measured in both Pb-Pb and p-Pb collisions [666–669, 671]. These correlations quantify the influence of the chiral magnetic effect (CME) and the chiral magnetic wave (CMW) on the produced particles. These correlators are also sensitive to strong background contributions, for example from local charge conservation and possibly radial and anisotropic flow.

The freeze-out radii in three orthogonal directions ("out", "side", "long") can be deduced from measurements of quantum-statistic correlations between pairs of same-charge pions and kaons (HBT) at low-momentum transfer. The HBT radii in all collision systems are found to scale with $\sqrt[3]{N_{ch}}$ indicating a constant density at freeze-out, and to decrease with increasing pair momentum $k_{\rm T}$ as expected from hydrodynamics. The size along the emission direction is similar to the geometric size of the system $(R_{out}/R_{side} \approx 1)$ in Pb–Pb collisions [664–666, 669, 670, 670] and $R_{out}/R_{side} \leq 1$ for both p–Pb and pp collisions [666–669, 671].

Direct photons Direct-photon measurements in the low p_T region are so far performed in Pb–Pb and pp collisions. The measurements are reproduced by models assuming the formation of a QGP in Pb–Pb collisions [538]. In this measurement, one cannot discriminate between the available models due to the large systematic uncertainties: models incorporating different initial temperatures, i.e. from 385 to 740 MeV in the most central Pb–Pb collisions, are able to reproduce the measurements. Nevertheless, the comparison among these models suggests that the initial temperature in central Pb–Pb collisions must exceed about 400 MeV [538]. No significant direct-photon signal has been extracted in pp collisions at current available center-of-mass energies [720].

Energy loss The created system in Pb–Pb collisions is opaque for high- p_T colored probes. Due to radiational and collisional energy loss high- p_T colored probes are strongly suppressed (jet quenching) whereas the system is transparent for photons and other colorless probes [314, 336, 338, 339, 344, 724–726, 729]. Jet quenching leads to a large asymmetry in back-to-back jet p_T and slightly modified jet fragmentation functions inside small jet cone sizes (R = 0.4). Most of the radiated energy appears at large angles (R > 0.8) [340, 352, 367, 721, 722].

On the contrary, the picture is different in p–Pb collisions: measurements of inclusive high- $p_{\rm T}$ hadron and inclusive jet yields in minimum-bias p–Pb collisions at the LHC are consistent with $R_{\rm pPb} = 1$ within the current accuracy of approximately 20%; i.e. no evidence of medium-induced modification is observed [314, 727, 728, 730, 731]. For event classes split by event activity, neither medium-induced modification in inclusive hadron production nor dijet transverse momentum imbalance are observed [314, 723, 731]; in contrast, for inclusive jet yields $R_{\rm pPb}$ is strongly suppressed relative to unity in "central" p–Pb collisions, and strongly enhanced in "peripheral" p–Pb collisions [730], attributed to selection biases [732].

The semi-inclusive yield of jets recoiling from a high- p_T trigger hadron has been used to search for jet quenching in p–Pb collisions [732]. This observable is trigger-normalized and semi-inclusive, and it therefore has greater systematic sensitivity to jet quenching effects in small systems than inclusive jet observables. Nevertheless, no significant jet quenching effects within the uncertainties of the measurement have been observed. These uncertainties can be expressed as an upper limit of 400 MeV (at 90% CL) on medium-induced energy transport outside a jet cone with R = 0.4. This value is a factor 20 smaller than the magnitude of out-of-cone energy transport measured by a similar approach in Pb–Pb collisions [346].

Heavy flavour Due to interactions and rescattering with the medium, also heavy-flavour particles exhibit finite anisotropies as shown with non-zero v_2 measurements for heavy flavour particles in both Pb–Pb and p–Pb collisions [248,318,453–457,491,735–739], see also Chapter 5. In addition, J/ ψ suppression in Pb–Pb collisions shows an enhancement w.r.t. RHIC energies [744]. Models incorporating a J/ ψ regeneration component from deconfined charm quarks in the medium can reproduce these measurements [447,744]. The limited understanding of cold nuclear matter effects in the open charm cross section determination, however, restricts the ability of these models to fully describe the experimental data on J/ ψ production in Pb–Pb collisions [749]. The size of these effects can be quantified by measurements in p–Pb collision. In p–Pb collisions, J/ ψ is suppressed relative to pp collisions [749]. The production of the excited charmonium state, $\Psi(2S)$ as well as different bottomonium states (Y(nS)) have been measured in both Pb–Pb and p–Pb collisions [472,486,742,744,748] which shows a suppression w.r.t. the ground state.

9.3 Open questions and new opportunities at HL-LHC

The previous section has extensively reviewed the state-of-the-art experimental knowledge of pp and p–Pb collisions. Certain gaps in knowledge became apparent due to either insufficient available data or shortcomings in the present detectors. The HL era of LHC can make a significant step ahead in many areas. The most relevant ones are discussed with dedicated performance projections in the remainder of this chapter.

Run 3 and 4 will allow the study of unprecedentedly high-multiplicity pp collisions. In order to do estimates in this regime, Sect. 9.4 will establish a firm extrapolation of the multiplicity distribution based on current LHC data together with a review of the data sample to expect. The large multiplicities bring a qualitatively new feature: a wide overlap between pp and Pb–Pb collisions up to about 65% central collisions allowing a unique opportunity to compare observables in a small (pp) and large (Pb–Pb) system at the same multiplicity. Studies in p–Pb collisions amend the picture. Given that multiplicity is not the only driving variable of a system, comparisons of estimates of the energy density in pp, p–Pb and Pb–Pb collisions are made below. The uniqueness of these extreme multiplicity pp collisions warrants that the study of their global-event properties are an interesting subject in itself, see Sect. 9.5.

Subsequently, a set of key observables is presented which require either the large data samples or the upgraded detectors. In particular the measurement of thermal dileptons profits from the new ALICE pixel detector with reduced material budget, and the measurement of higher-order correlations from the extended tracker acceptance in Run 4 in ATLAS and CMS. Correlations at higher orders using the subevent method will provide an essentially non-flow free measurement of v_n coefficients and their inter-correlations measured through symmetric cumulants, see Sect. 9.6. These measurements focus on two interesting regimes: at high multiplicity where the overlap with Pb–Pb collisions will be studied, and at low multiplicity to answer the question on the onset of collective phenomena. This section also shows that the measurement of the probability distribution of event-by-event v_2 becomes for the first time feasible in small systems, a quantity presently completely unknown in pp collisions.

The smooth increase of strange-particle production across system size is one of the key surprising findings from Run 2 pp physics. A projection of the reach at HL-LHC is given in Sect. 9.7 showing that the question if the thermal limit, given by statistical models in Pb–Pb collisions, is reached also in



Fig. 64: Multiplicity distributions measured by ALICE [754] (left panel) and ATLAS [755, 756] (right panel) overlaid by the fit with a negative binomial distribution. For details see text.

pp collisions can be answered. A puzzling finding is the absence of jet quenching in p–Pb collisions with the measurements performed in Run 1 and 2. If final-state interactions are to explain the observed collective phenomena, also energy loss of traversing partons should be measurable. Section 9.8 discusses how jet quenching can be observed in Run 3 and 4 if present, or alternatively how a stringent limit can be set. Performance studies are presented for hadron–jet, γ –jet and Z–jet correlations, both, in p–Pb and pp collisions. Finally, the potential to detect thermal radiation and extract a medium temperature in p–Pb collisions is presented in Sect. 9.9. Such a measurement would constitute a strong indication of the formation of an emitting medium. Finally, the potential of colliding smaller nuclei, in particular oxygen, is assessed in Sect. 9.10.

9.4 Proton-proton collisions at extreme multiplicities

9.4.1 Multiplicity distribution

For the performance estimates at high multiplicity in pp collisions, a multiplicity-distribution extrapolation has been used which is based on existing ALICE ($|\eta| < 1.5$) [754] and ATLAS ($|\eta| < 2.5$) [755,756] data. Data from CMS [757] is compatible with the used distributions and is therefore not explicitly included in the extrapolation. A parameterisation with a single⁵ negative binomial distribution is used to characterize the multiplicity distribution [758,759].

The data is shown in Fig. 64 overlaid with the fit with a single negative binomial distribution of the tail of the distribution (20–40% of the cross-section). The three parameters of this fit are itself fit with a power law to extrapolate to $\sqrt{s} = 14$ TeV.

The resulting extrapolated multiplicity distribution for 14 TeV is shown in Fig. 65 for the ALICE and ATLAS case. In addition, these are compared scaled by their respective average multiplicities. The agreement is rather good, with some discrepancy in the tail of the distribution. The extrapolation based on the smaller phase-space region falls off more quickly with multiplicity, and is therefore used as the conservative estimate for the extrapolations in this chapter.

⁵At LHC energies two NBDs are needed for a good fit to the full distribution, but one is sufficient for the tail of the distribution.



Fig. 65: Extrapolated multiplicity distributions in pp collisions within $|\eta| < 1.5$ (left panel) and $|\eta| < 2.5$ (centre panel). The indicated regions are (from left to right) 5–7, 7–10, 10–12, 12–14, 14–16 times the average multiplicity. In the left panel the multiplicity distribution of Pb–Pb and p–Pb collisions is also plotted. The right panel compares these two distributions scaled by the average multiplicity. The extrapolation for $|\eta| < 2.5$ turns out to be a bit wider at large multiplicities; therefore the one based on $|\eta| < 1.5$ is used as baseline.



Fig. 66: Energy density as a function of dN_{ch}/dy calculated by IP-Glasma (solid lines) and with MC Glauber and the Bjorken formula (dashed lines); for details see text. Compared are pp ($\sqrt{s} = 7$ TeV), p–Pb ($\sqrt{s_{NN}} = 5.02$ TeV) and Pb–Pb ($\sqrt{s_{NN}} = 5.02$ TeV) collisions at $\tau = 0.2$ fm/c. The horizontal line indicates the energy density reached in central Pb–Pb collisions ($dN_{ch}/dy \approx 2000$).

9.4.2 Energy density

While the multiplicity is a convenient and well-defined observable to compare different collision systems, the underlying dynamics may be driven by other properties. In large collision systems, the energy density ϵ is often used to characterize the system and the expected effects. Figure 66 shows an estimate of the energy density for pp, p–Pb and Pb–Pb collisions based on IP-Glasma [760] as well as on the Bjorken estimate:

$$\epsilon = \frac{1}{A\tau} \langle E \rangle \frac{3}{2} \frac{dN_{ch}}{dy}.$$
(30)

Range	${\rm d}N_{\rm ch}/{\rm d}\eta$	Fraction	Events per pb^{-1}	Events in $200 \mathrm{pb}^{-1}$
5–7 $\langle N_{ m ch} \rangle$	35–49	2.4e-03	1.9e+08	3.7e+10
7–10 $\langle N_{ m ch} angle$	49–70	1.3e-04	1.0e+07	2.0e+09
10–12 $\langle N_{ m ch} angle$	70–84	1.1e-06	9.0e+04	1.8e+07
12–14 $\langle N_{\rm ch} \rangle$	84–98	4.7e-08	3.7e+03	7.3e+05
14–16 $\langle N_{\rm ch} \rangle$	98–112	1.8e-09	1.4e+02	2.8e+04

Table 8: Number of pp events at $\sqrt{s} = 14$ TeV in selected high-multiplicity bins.

Table 9: Number of events in pp collisions at $\sqrt{s} = 14$ TeV sliced in equivalent multiplicity bins as in p-Pb and Pb-Pb collisions.

Range	$\mathrm{d}N_{\mathrm{ch}}/\mathrm{d}\eta$	Events per pb^{-1}	Events in 200pb^{-1}
0–5% p–Pb	41–56	4.9e+07	9.8e+09
5–10% p–Pb	34–41	1.9e+08	3.8e+10
1020% p–Pb	27–34	6.6e+08	1.3e+11
60–65% Pb–Pb	98–137	1.5e+02	3.0e+04
65–70% Pb–Pb	68–98	1.6e+05	3.1e+07
7075% Pb-Pb	45–68	2.1e+07	4.2e+09
75–80% Pb–Pb	29–45	5.9e+08	1.2e+11

For the latter, the input is the multiplicity-dependent $\langle p_T \rangle$ [729,761] as well as the multiplicity-dependent transverse overlap from a Glauber MC [762]. The energy density is calculated at fixed $\tau = 0.2 \text{ fm/c}$. It should be noted that these assumptions can be challenged and other ways to calculate ϵ are available which can lead to largely different value in particular in pp and p–Pb collisions. Here the aim is only to show that the energy density depends on the system at a fixed multiplicity, and can reach large values in pp and p–Pb collisions, of the order of central Pb–Pb collisions.

9.4.3 Data-taking conditions and integrated luminosity for pp collisions

For the performance studies in this chapter, a high-multiplicity sample of $L_{int} = 200 \text{ pb}^{-1}$ is assumed per experiment. In order to assure a clean trigger, collisions at low $\mu \approx 1$ are needed which requires special runs or special conditions at the end of fill for ATLAS and CMS. For LHCb, the comparatively low pileup and good vertex resolution should allow for recording high-multiplicity events during normal running conditions at a pile up of about five visible pp interactions. ALICE generally runs at low μ and can collect a similar sample over a longer data-taking period. For Run 4, the upgraded tracking and vertex detectors in ATLAS and CMS may allow to isolate high-multiplicity collisions also in a large μ environment. This option needs to be carefully studied.

Table 8 gives the fraction of cross-section and the number of events in five multiplicity classes: 5–7, 7–10, 10–12, 12–14 and 14–16 times the average multiplicity. Table 9 gives the number of events of bins with equivalent multiplicity than commonly measured multiplicity bins in p–Pb and Pb–Pb collisions. For the calculation of the number of events $\sigma_{\text{inel}} = 78.4 \text{ mb}$ [762] is used. These tables are the key input for the performance figures presented in this chapter. The conversion of the provided $dN_{\text{ch}}/d\eta$ to multiplicity ranges with larger pseudorapidity coverage is done for simplicity assuming a flat pseudorapidity distribution within $|\eta| < 2.5$. For the conversion to a phase space with a p_{T} cut as employed in many current measurements a set of conversation factors is used, listed in Tab. 10.

Table 10: Conversion factors between $N_{\rm ch}$ with a $p_{\rm T}$ threshold, and $N_{\rm ch}$ including particles down to $p_{\rm T} = 0$. The factor shown is $N_{\rm ch}/N_{\rm ch}(p_{\rm T} > X)$, extracted with Pythia 8, tune CUETP8M1 [763]. A potential multiplicity dependence of this factor is neglected for the projections in this chapter.

p_{T}	> 0.1 GeV/c	> 0.2 GeV/c	> 0.3 GeV/c	> 0.4 GeV/c	> 0.5 GeV/c
$ \eta < 1.5$	1.03	1.11	1.22	1.31	1.40
$ \eta < 2.4$	1.04	1.14	1.27	1.42	1.55

9.5 Global-event properties

The measurement of global-event observables in rare high-multiplicity pp collisions are of interest in itself. The shape of the multiplicity distribution, which has been largely extrapolated in the previous section, is today a challenge for models. The dynamics of producing very large multiplicity events is not understood in detail and therefore the shape of the distribution is an important input. Furthermore, studies of $\langle p_{\rm T} \rangle$ as a function of multiplicity [761] have shown a strong increase with multiplicity. However, those measurements exist only up to $dN_{\rm ch}/d\eta \approx 55$, while the measurements at HL-LHC promise a measurement beyond twice that value.

The shape of the multiplicity distribution and the growth of $\langle p_T \rangle$ are closely connected to the physics of multiple parton interactions: high-multiplicity collisions are understood as originating from the collision of multiple partons within the same pp collisions. It has been shown that the number of (low momentum transfer) parton interactions increases linearly with multiplicity with a possible saturation at large multiplicity [764]. The prospect of showing that adding another parton interaction to an already busy event may be strongly suppressed, is an important ingredient to a revised conceptual understanding of particle production in high-energy pp collisions. Together with the studies of symmetric cumulants (see the subsequent Section), HL-LHC will determine not only if there is a saturation limit for multiple parton interactions, but also the parton structure within the proton.

9.6 Particle correlations

The measurements of two-particle correlations and higher-order cumulants have been the initial observations of collective-like effects in small systems. In pp collisions, two distinct regions are of interest at HL-LHC: the high-multiplicity tail to compare to p–Pb and Pb–Pb collisions and the low-multiplicity region to investigate the onset of these effects. In the following, several performance estimates are given as examples for the rich physics which can be addressed.

State-of-the-art measured 4-particle cumulants of v_3 (c_3 {4}) in pp and p–Pb collisions are presented in Fig. 67 overlaid with the projection for HL-LHC. In order to remove non-flow contributions, the 3-subevent method is applied. In pp collisions, with the data collected in Run 2, the statistical uncertainties are large and the c_3 {4} values are consistent with zero in most of the N_{ch} range. On the contrary, in large systems, significant non-zero c_3 {4} up to $-0.4 \cdot 10^{-6}$ depending on centrality has been measured [698], which reflects the nucleonic fluctuations in the initial state. Whether similar behaviour is observed in small systems still needs to be studied. The increase in luminosity in Run 3 and 4 provides a great opportunity to measure c_3 {4} in pp collisions with high precision: the statistics are sufficient to measure a signal as small as v_3 {4} = 1.5% for $N_{ch} \gtrsim 170$, while 2% are accessible with large significance over a wide multiplicity range ($N_{ch} \gtrsim 100$). Similarly, in p–Pb collision, the current result shows that c_3 {4} is consistent with zero, but increased statistics will help to detect a potential non-zero c_3 {4} smaller than 1.5% for 100 $\leq N_{ch} \lesssim 500$. Similarly, the precision of the already measured non-zero c_2 {4} will be greatly improved [260].

The correlations of flow harmonics between different orders, called symmetric cumulants, are very sensitive to the initial state and the hydrodynamic evolution in heavy-ion collisions. In Pb–Pb collisions



Fig. 67: 4-particle cumulants c_3 {4} measured with 3-subevent method for pp (left panel) and p–Pb collisions (right panel) as a function of $N_{\rm ch}$ ($|\eta| < 2.5$ and $0.3 < p_{\rm T} < 3$ GeV/c). Only statistical uncertainties are shown in the figure and the gray band represents the projected statistical uncertainty, with c_3 {4} assumed to be zero. The red and green dash lines represent 1.5% and 2.0% v_3 {4} signal, respectively. The vertical line in the left panel indicates the transition between minimum-bias and high-multiplicity triggered data. Figures from Ref. [260].



Fig. 68: Symmetric cumulants extracted with and without applying subevents for pp (left panel) and p–Pb collisions (right panel) as a function of $N_{\rm ch}$ ($|\eta| < 2.4$ and $0.3 < p_{\rm T} < 3$ GeV/c). The projected reach is shown for the case of 2, 3 and 4 subevents assuming a constant signal as a function of multiplicity indicated by the lower horizontal line. The vertical line indicates the transition between minimum-bias and high-multiplicity triggered data. Figures from Ref. [315].

these are, for instance, used to constrain the shear viscosity over entropy ratio η/s . In addition, they challenge the description of the observed phenomena within initial-state saturation models. Their measurement in small systems can provide important insight in the validity of the hydrodynamic description of the observed phenomena. Here, symmetric cumulants probe in particular the proton substructure [765]



Fig. 69: Demonstration of the influence of the larger tracking acceptances for ATLAS and CMS available in Run 4. Left panel: 4-particle cumulant c_3 {4} (as in the left panel of Fig. 67) for pp collisions with $L_{\text{int}} = 200 \text{ pb}^{-1}$. The data points indicate the reach with the detector in Run 3 ($|\eta| < 2.5$) while the gray band the enlarged acceptance of $|\eta| < 4$ in Run 4. The yellow, red and green dash lines represent 1.0%, 1.5% and 2.0% v_3 {4} signal, respectively. Figure from Ref. [260]. Right panel: Symmetric cumulants with 3 subevents (as in left panel of Fig. 68) for pp collisions with $L_{\text{int}} = 200 \text{ pb}^{-1}$. The blue (gray) area indicates the projected uncertainty for Run 3 (4). Figure from Ref. [315].

which is needed to provide a solid description of the initial state, a necessary ingredient for the hydrodynamic description. The present uncertainties of such measurement in small systems are too large for a definitive conclusion, in particular in pp collisions, due to the dominance of non-flow like jets and resonance decays. Figure 68 shows the performance projection of $SC(2,3) = \langle v_2^2 v_3^2 \rangle - \langle v_2^2 \rangle \langle v_3^2 \rangle$ for HL-LHC for pp and p–Pb collisions. The uncertainties of the measurement without subevents would become practically invisible, however, those stay dominated by non-flow effects. A measurement requiring two, three and even four subevents becomes possible with uncertainties of the order of a few times 10^{-7} depending on multiplicity. Such results can give a definitive answer if a similar hydrodynamic footprint is observed in small and large systems.

Figure 69 illustrates the reduction of the statistical uncertainty due to the larger tracker acceptance in Run 4 for ATLAS and CMS. For this 4-particle correlator a reduction of the uncertainties of about 2.5 is expected, and therefore even the measurement of a 1% v_3 {4} signal comes into reach. The influence of the acceptance increase on the uncertainties of 6- and 8-particle cumulants will be larger, factor 4 and 6.5, respectively. Similarly, the uncertainties on the SC measurement reduce significantly at larger p_T .

Figure 70 (left panel) illustrates the reach which can be obtained for the v_2 measurement of heavyflavoured objects in p–Pb collisions. Shown are projections for heavy-flavour electrons and inclusive J/ψ by ALICE as well as for prompt D and J/ψ by CMS. Minor uncertainties are expected for this observable with the potential to demonstrate for the first time with significance the final-state interaction of charm and beauty in a small collision system.

The v_n fluctuates on an event-by-event basis as no two nuclei have identical parton distribution. The probability density distribution of v_n , $p(v_n)$ is closely related to event-by-event fluctuations of the eccentricities, $p(\varepsilon_n)$ as $v_2 \propto k_2 \epsilon_2$. Therefore its measurement provides crucial information about the initial conditions and the final-state dynamics of the medium. To characterise the initial-state spatial anisotropy these measurements are fitted with Bessel-Gaussian and elliptic power functions. The measurements of



Fig. 70: Left panel: Particle identified v_2 coefficients for p–Pb collisions as a function of $p_{\rm T}$. Two different cases are shown: the ALICE projections are for the 20% highest-multiplicity collision ($L_{\rm int} = 500 \,{\rm nb}^{-1}$) demonstrating the negligible statistical uncertainties for heavy-flavour decay electrons and J/ψ , while the CMS projection is for a bin with $4 - 5\langle N_{\rm ch} \rangle$ ($L_{\rm int} = 2 \,{\rm pb}^{-1}$) demonstrating the wide reach in multiplicity achievable for D mesons and J/ψ . Right panel: Projection of the measurement in pp collisions of the probability distribution of v_2 . To illustrate the reach the same signal as in Pb–Pb [717] is assumed although the mean and the width of the distribution is most likely smaller in pp collisions. The projection is for the equivalent pp multiplicity (circles) as in 60–65% centrality in Pb–Pb collisions (squares).

probability density distributions for v_2 at Pb–Pb collisions are described well by the Bessel-Gaussian function at central collisions and less in peripheral collisions [717–719]. This deviation from the Bessel-Gaussian function is expected in peripheral collisions as k_2 increases slightly at large ϵ_2 values [766]. Measurements are well described by the elliptic power functions in all centrality intervals of Pb–Pb collisions [718]. These measurement have not yet been attempted in small systems due to the insufficient available statistics. Figure 70 (right panel) presents a projection for the measurement of $p(v_n)$ in pp collisions. This extrapolation is based on the $p(v_2)$ measurement in 60–65% centrality Pb–Pb collisions at $\sqrt{s_{NN}} = 2.76$ TeV [717]. In this simple study, the same signal is assumed although the mean and width of the distribution is most likely smaller in pp collisions. Such a measurement would constitute the first measurement of $p(v_n)$ in pp collisions, and can shed important light on the nature of the observed v_2 coefficients.

9.7 Strangeness enhancement

The unexpected increase of the strange-particle yield normalized by the pion yield as a function of $N_{\rm ch}$ is one of the key observations in small systems. In pp collisions these ratios are measured up to $dN_{\rm ch}/d\eta \approx 17$ with some overlap with p–Pb collisions. The most peripheral Pb–Pb collisions measured have a $dN_{\rm ch}/d\eta \approx 96$, nearly 6 times larger. Figure 71 presents the expected reach of the Ω/π ratio in pp collisions which will bridge the present gap between pp and Pb–Pb collisions. In particular, if the measured increasing trend would continue, the Ω/π ratio would grow larger than in peripheral Pb–Pb collisions. Assuming that strangeness enhancement scales with the energy density of the system, Fig. 66 suggests that it should indeed be possible to see that the high-multiplicity pp results exceed the low multiplicity Pb–Pb results (crossover). Whether the signature will be as striking as the



Fig. 71: Ω/π ratio as a function of $dN_{ch}/d\eta$ for pp, p–Pb, and Pb–Pb collisions. The existing data (from Ref. [632]) is shown in open black symbols (pp), grey diamonds (p–Pb) and grey squares (Pb–Pb), while the extrapolation for pp collision is shown in blue filled circles. Two scenarios are shown: a) assuming that the ratio continues increasing following the measured trend, and b) assuming that the value stays the same as at the largest measured $dN_{ch}/d\eta$.

projection in Fig. 71, depends on the details of the assumed scaling law. At this point simulations are not precise enough to provide quantitative predictions of such a crossover, and HL-LHC experimental results on strangeness enhancement will as such be driving the theoretical development. The scenario with a clear crossover will be immediately distinguishable from a scenario where the Ω/π ratio flattens, and connects smoothly with the Pb–Pb result. Such a result will in itself also be groundbreaking, as it will indicate that the thermal limit reached in Pb–Pb collisions will already be realized in high-multiplicity pp collision.

9.8 Energy loss

As discussed in Sect. 9.2, inclusive high- $p_{\rm T}$ hadron and jet yields show at present no evidence of mediuminduced energy loss in p–Pb collisions, and suffer from selection biases if measured in event classes. Inclusive measurements with the large event set expected at HL-LHC therefore do not help to resolve the question of energy loss in small systems. However, coincidence measurements of jets recoiling against a trigger object are not subject to such biases, and have the potential to identify small energy-loss effects or put stringent upper limits. In this section, projections are given for correlations between high- $p_{\rm T}$ hadrons and jets, as well as jets and γ and Z.

Figure 72 shows a projection of the measurement of semi-inclusive hadron-jet correlations in LHC Run 3 and 4, for pp collisions at $\sqrt{s} = 14$ TeV and p-Pb collisions at $\sqrt{s_{NN}} = 5.02$ TeV. The figure shows the ratio of trigger-normalized recoil spectra for events selected on high and low event-activity (EA) classes. This projection is based on Pythia simulations for pp collisions, which gives the expected number of charged-hadron triggers in the interval $15 < p_{T,trig} < 20$ GeV (scaled by A to model p-Pb collisions), and the per-trigger recoil jet spectrum. The measured enhancement in the per-event high- p_T hadron yield for pp collisions in high-multiplicity collisions [649] has been taken into account.

The projection represents the case where no energy loss occurs for high-EA relative to low-EA collisions, and demonstrates the statistically achievable limit. The 90% confidence level for a possible EA-dependent spectrum shift due to large-angle energy transport from jet quenching [732] is 70 MeV/c



Fig. 72: Modification of jet-recoil yields extracted from semi-inclusive hadron-jet correlations for pp collisions (left) and p-Pb collisions (right) within the ALICE acceptance. Shown is the ratio of high event-activity (EA) and low-EA recoil spectra as a function of $p_{T,jet}^{ch}$, with high-EA corresponding to 5–7 $\langle N_{ch} \rangle$ in pp collisions (left panel), and the 0–5% bin for p-Pb collisions (right panel). Since no EA-dependent shift is imposed, the parent distribution of the ratio has the value of unity at all p_T . The red lines show the 90% CL limit for a possible EA-dependent spectrum shift.

for p–Pb (5% highest EA) and 21 MeV/c for pp collisions (5–7 $\langle N_{ch} \rangle$). These values are over 100 times smaller than the spectrum shift measured in Pb–Pb collisions [346]. The high statistics of the HL-LHC dataset enables this approach to be applied to yet more stringent EA selections; for 7–10 $\langle N_{ch} \rangle$ (10–12 $\langle N_{ch} \rangle$) the corresponding 90% CL limit on energy loss is expected to be 69 MeV/c (590 MeV/c).

Projections for the correlation of jets and γ as well as jets and Z are presented in Fig. 73 for pp and p–Pb collisions. Shown are distribution of the momentum fraction $x_{jX} = p_T^{jet}/p_T^X$ where X is the γ or Z. Given that the γ and Z can be considered unmodified by final-state interactions, a potential energy loss acting on the jet would directly alter the x_{jX} distribution. For pp collisions, the left panel of Fig. 73 presents the distribution for different classes in multiplicity based on Pythia, demonstrating the reach. It can be also seen that the distribution shifts significantly as a function of multiplicity without final-state interactions, but purely due to the presence of an underlying event. This shift is caused by selection biases, e.g. the likelihood of multi-jet events is increased by requiring higher event multiplicity. In order to extract a firm conclusion on energy loss, this observable needs to be compared to theoretical calculations or tuned generators which reproduce multiplicity and underlying-event fluctuations in e⁺e⁻ and pp collisions. The right panel of Fig. 73 presents the projection for p–Pb collisions for MB collisions but in different pseudorapidity intervals sensitive to potential differences in the p and Pb hemisphere.

9.9 Thermal Radiation

The measurement of thermal radiation in p–Pb collisions can be considered as a smoking gun for the formation of a system with an energy scale above the phase-transition temperature, see Chapter 8. In order to estimate the sensitivity to the thermal radiation in p–Pb collisions, a similar strategy as in Sec. 8.1.2 was used. The combinatorial background was scaled from Pb–Pb collisions to the expected number of pairs in p–Pb collisions. The pair efficiency (including the efficiency for rejecting e^+e^- pairs from semileptonic charm decays) is assumed to be the same as in Pb–Pb collisions. Subsequently, the temperature of the QGP is extracted in the same way as in Sec. 8.1.2.



Fig. 73: Left panel: CMS projection of the measurement of the jet- γ momentum fraction $x_{j\gamma}$ in pp collisions for selected high-multiplicity bins. A jet with $p_T > 30$ GeV/c is required to be back-to-back $(\Delta \varphi > 7/8\pi)$ with a γ with $p_T > 60$ GeV/c. The shape is based on Pythia and shifts due to selection biases as a function of multiplicity (see text). Figure from Ref. [767]. Right panel: ATLAS projection of the measurement of the Z-jet momentum fraction x_{jZ} in p-Pb collisions in different pseudorapidity intervals. The momentum requirements are also 30 GeV/c for the jet and at least 60 GeV/c for the Z, with the same back-to-back requirement ($\Delta \varphi > 7/8\pi$). The projection is based on Powheg + Pythia 8 Monte Carlo samples with the CT10 PDF set. Figure from Ref. [768].



Fig. 74: Projection of the measurement of the medium temperature extracted from thermal dileptons in p– Pb collisions. As the expected signal is uncertain, the figure presents the required integrated luminosity relative to the prediction based on Ref. [548] denoted by N_{ee} (thermal radiation). It is expressed as the minimum thermal photon N_{ee} (thermal radiation) to π^0 ratio (left axis) and scaling of N_{ee} (thermal radiation) (right axis) for achieving a 10% statistical uncertainty on the extracted temperature.



Fig. 75: Glauber MC calculations are presented for for p–Pb (top panels) and O–O collisions (bottom panels). Shown are N_{part} as a function of impact parameter (left panels) and forward multiplicity as a function of N_{part} (right panels).

 π^0 (both decaying into e^+e^-) ratio that is needed for a fit to the invariant mass spectrum with a statistical uncertainty $\sigma_{T,stat} = 10\%$ as a function of L_{int} up to 2000 nb⁻¹ is shown in Fig. 74. If the considered prediction is accurate, an integrated luminosity of about 50 nb^{-1} is sufficient for the measurement. In case the signal is 50% smaller about 4–5 times the statistics is needed.

9.10 Potential of O–O Collisions

A promising opportunity to study the emergence of collective phenomena further as well as the presence of possible parton energy loss in small collision systems, are collisions of smaller nuclei. In particular, collisions of oxygen are an efficient way of investigating the properties of low-multiplicity heavy-ion collisions, which in large A–A systems only occurs for peripheral geometries. The achieved multiplicities in O–O collisions are similar to p–Pb collisions with the significant advantage that the collision geometry is much better defined. This is demonstrated in Fig. 75, which shows the correlations between number of participants, multiplicities and impact parameter in O–O and p–Pb collisions. The correlations between N_{part} and impact parameter as well as multiplicity and N_{part} is much narrower in O–O collisions are compared to p–Pb collisions. Consequently, highest multiplicities in p–Pb collisions are only accessible in the tail of the distribution while similar multiplicities are already reached in O–O collisions in the plateau region. This is shown in Fig. 76 illustrated for the 20–40% event class.

Scaling the measured nuclear-modification factor in Pb-Pb collisions at 5.02 TeV [314] at sim-



Fig. 76: Estimated multiplicity distributions in the forward region in O–O and p–Pb collisions. The event classes of highest multiplicity in O–O collisions are indicated with 0–20% and 20–40%. The O–O studies have been performed for the 2018 beam configurations at $\sqrt{s_{NN}} = 6.5$ TeV, while the Run 3 configuration would yield $\sqrt{s_{NN}} = 7$ TeV, with the advantage that a large pp reference data set already exists from earlier LHC running (2010–11).

ilar multiplicities as for central O–O collisions while roughly accounting for the artificial suppression caused by the multiplicity bias present in such peripheral Pb–Pb collisions [729], allows to estimate the expected effect on the R_{AA} in O–O collisions to about 20%. Out of this deviation from unity, about half can be attributed to biases due to the multiplicity selection in such small collision systems already in absence of nuclear effects [769]. An observable deviations from unity of about 10% caused by energy loss in the produced medium remains. While this expected suppression may seem small, it should be possible to measure it already with an L_{int} of a few $100 \,\mu b^{-1}$. In case such a suppression was absent, the conclusion can be drawn that small collision systems do not exhibit measurable energy loss, while other collective features are present, challenging the role of significant final-state interactions as underlying mechanism. It should be noted that the absence of suppression can most likely not be taken as a proof against the formation of the QGP, as it may be that the fast partons are emerging without seeing the medium, either because they are emitted from the surface or because their formation time is longer than the time within the medium. On the contrary, as discussed in Sect. 9.8 the first observation of energy loss in small systems would clearly confirm models in which final-state interactions play an important role

The study of bulk particle production in O–O collisions has twofold interest: Firstly, the dependence of the pseudorapidity density of charged particles scaled by participant pairs on \sqrt{s} is stronger in A–A collisions than in pp and p–A collisions [772]. The rise for p–Pb and d–Au collisions is similar to the one of inelastic pp collisions indicating that the stronger rise in A–A collisions might not be solely related to the multiple collisions undergone by the participants since the proton in p–A collisions also encounters multiple nucleons. High-energy O–O collisions promise to study this \sqrt{s} dependence further in a regime of small number of participants. Secondly, a strong correlation between initial state geometry (collision eccentricity) and observed flow has been established since RHIC [773], and pertains at LHC, in Pb–Pb and Xe–Xe collisions [774]. As shown in Fig. 77, the eccentricity profiles of Xe–Xe and Pb–Pb collisions are, however, quite similar, and collision systems exhibiting a different geometry as O–O could therefore provide further insight into the connection between initial-state geometry and multi-particle correlations.



Fig. 77: Second-order eccentricity coefficient ϵ_2 as a function of centrality for O–O, Ar–Ar, Xe–Xe and Pb–Pb collisions from MC-Glauber (left panel) [762, 770] and Trento (right panel) [771] initial conditions.

9.11 Summary

The discoveries made in recent years in small collision systems have challenged two paradigms, the modelling of heavy-ion collisions, as well as the modelling of the underlying event of elementary pp collisions. The experimental observations of strangeness enhancement and multi-particle correlations suggests that energy loss should also be present, as both are a consequence of significant final-state interactions. But up to this point no hint of energy loss in pp or p–Pb collisions has been seen. The increased luminosity will allow both for precision studies of the established signatures of small system collectivity, and to either establish evidence or place exclusion limits on the latter.

This chapter has presented a HL-LHC small system experimental program promising significant progress shown by the set of performance studies, ranging from largely non-flow suppressed high-order correlations, over measurement of strange-particle yields and thermal radiation, to energy-loss signals. In addition, new theoretical insights and phenomenological developments are also needed for developing a unified picture of parton dynamics and particle production valid for both small and large systems. This working group encourages both investment in the necessary theoretical development, and facilitation of collaboration between theorists and experimentalists.

The physics community can look forward to a deepened understanding of hot and dense QCD and a universal description of small to large collision systems.

10 High energy QCD with proton-nucleus collisions and ultra-peripheral collisions

Coordinator: Michael Winn (LAL and IRFU/DPhN, CEA Saclay)

Contributors: A. Angerami (LLNL Livermore), F. Arleo (LLR/École Polytechnique), N. Armesto (Instituto Galego de Fisica de Altas Enerxias (IGFAE) Universidade de Santiago de Compostela), R. Bi (Massachusetts Institute of Technology), É. Chapon (CERN), Z. Citron (Ben-Gurion University of the Negev), J. G. Contreras Nuno (Czech Technical University in Prague), M. Dumancic (Weizmann Institute of Science), D. d'Enterria (CERN), G. Giacalone (IPhT, CNRS/CEA), A. Giammanco (Université Louvain), V. Guzey (Helsinki University, University of Jyvaskyla, Kurchatov institute, Gatchina),I. Helenius (University of Jyvaskyla and Universitä Tübingen), P. Janus (AGH University of Science and Technology), H. Kim (Chonnam National University), M. Klasen (Münster University), S.R. Klein (Lawrence Berkeley National Laboratory), J. Kremer (AGH University of Science and Technology), G. Krintiras (Université Louvain), E. Kryshen (Kurchatov institute, Gatchina), A. Kusina (IFJ PAN, PL-31342 Kraków, Poland), Y.-J. Lee (Massachusetts Institute of Technology), M. van Leeuwen (Utrecht University/Nikhef), C. Marquet (CPHT/École Polytechnique), C. Mayer (IFJ PAN, Kraków, Poland), M. Mulders (CERN), M. Murray (University of Kansas), F. Olness (SMU Dallas), P. Paakkinen (University of Jyvaskyla), H. Paukkunen (University of Jyvaskyla, Helsinki Institute of Physics), I. Schienbein (LPSC/Université Grenoble Alpes), P. Silva (CERN), D. Stocco (Subatech Nantes), M. Strikman (Pennsylvania State University), D. Tapia Takaki (University of Kansas).

10.1 Introduction

Proton-nucleus [4] and ultraperipheral (UPC) [577] collisions offer the opportunity to study the behaviour of QCD at high energies and large partonic densities [775]. As the theory of the strong interaction, QCD is analytically well understood only in a perturbative regime of small coupling constant and where radiation of gluons and quarks is a linear process that can be described with linear evolution equations of the non-perturbative parton densities, i.e., hadrons and nuclei are considered as dilute partonic objects. However, non-linear effects are unavoidable in QCD, and they should in principle dominate at large densities reached at high collision energies and for large nuclei. It was proposed long ago [776,777] that at such large densities a resummation of powers of density scaled by the strong coupling constant is possible, resulting in a non-perturbative but weak coupling regime where parton densities saturate, and whose effective field theory incarnation is the Color Glass Condensate (CGC) [778]. Particle production in p–A collisions in the forward rapidity region is dominated by small–x partons in the nucleus. Therefore, saturation effects are expected to be largest there. Furthermore, UPCs as a source of large fluxes of quasi-real photons, provide the opportunity to study the partonic structure of protons (in pp and p–A) and nuclei (in p–A and Å).

The structure of nucleons and nuclei, and the mechanism of particle production at small x, are also key ingredients for a detailed description of heavy-ion collisions and for the characterisation of the produced hot and dense medium. For hard probes, the nuclear modification of standard collinear parton densities is one of the uncertainties in the extraction of medium characteristics [246, 374]. For soft and semihard observables, both the parton densities and the detailed transverse spatial structure of nucleons and nuclei are crucial ingredients for their description, e.g. for the initial conditions for hydrodynamical evolution [197, 198, 779].

In the collinear framework, parton densities inside nuclei (nPDFs) [780, 781] are poorly constrained due to two primary factors. On the one hand, data come from a large variety of nuclei and the number of data points for any of them is very small compared to the proton analysis. In particular, for the Pb nucleus, there are less than 50 points coming from fixed target Deep Inelastic Scattering (DIS) and Drell-Yan experiments and from p–Pb collisions at the LHC. The fit for a single nucleus is therefore impossible and the modeling of the A-dependence of the parameters in the initial conditions becomes mandatory [782, 783]. On the other hand, the kinematic coverage in Q^2 and x with existing data is very small compared to the requirements of present hadronic colliders, see Fig. 78. In spite of the uncertainties in the applicability of collinear factorisation, scale choices and other theoretical caveats for



Fig. 78: Left: $x-Q^2$ plane to be explored in proton-nucleus at the LHC and the FCC, and in proposed electron-nucleus colliders, compared with the regions where the experimental data presently used in the EPPS16 analysis [782] lie. Right: $x-Q^2$ plane to be explored in UPCs, taken from [577].

nPDFs extraction in hadronic collisions and UPCs, these are the only experimental collision systems where the nPDFs can be constrained before electron-ion colliders become available. The most up to date analyses include between 1000 and 2000 data points for 14 nuclei and are performed at next-to-leading accuracy [782–784], there even exists a first attempt at next-to-next-to-leading [785]. Differences between them mainly arise from the different sets of data included in the analysis and from the different functional forms employed for the initial conditions. All in all, all parton species are very weakly constrained at small $x < 10^{-2}$, gluons at large x > 0.2, and the flavour decomposition is largely unknown - a natural fact for u and d due to the approximate isospin symmetry in nuclei. The impact of presently available LHC data, studied in [782], is quite modest with some constrains on the gluon in the region 0.01 < x < 0.3. On the other hand, theoretical predictions for nuclear shadowing of quark and gluon PDFs based on *s*-channel unitarity and diffractive nucleon PDFs are available down to $x \sim 10^{-4} - 10^{-5}$ [786, 787].

In the context of phenomena beyond collinear factorisation and PDF evolution in $\ln(Q^2)$, there have been recent claims [788, 789] that resummation of logarithms of x may be required for a better description of DIS data from HERA at small x, and searches for long range azimuthal correlations are undergoing [790]. But no conclusive evidence of saturation, i.e., of non-linear dynamics, has been found in hadronic collisions. While the CGC provides a calculational framework for several observables in pp, p–A and Å, see e.g. the reviews [791,792], like the ridge, back-to-back hadron correlations in the forward region, multiplicities and transverse momentum distributions,..., there is no consensus in the field in the interpretation of these results, or they involve non-perturbative modeling, or they are affected by large theoretical uncertainties and, for some of them, higher-order calculations are missing, or the data lie at the border of phase space where extracting clear conclusions is very delicate. Therefore, high-energy p–A collisions and UPCs are two promising systems where data can offer clear evidences of non-linear effects.

In Fig. 78, the kinematic regions covered by proton-nucleus collisions at the LHC and the FCC (the left panel, [400]) and UPCs at the LHC (right) are shown and compared with the regions where data currently used to constrain nPDFs lie. A huge enlargement is evident with respect to the presently existing data at the LHC. The HL-LHC offers new improved detectors and larger statistics for some observables like dijets or photon-jet correlations. The HE-LHC would enlarge the kinematic plane in a region intermediate between the LHC and the FCC. On the other hand, electron-nucleus collisions [793,

794], if they eventually happen in the 2030's, would be complementary. They offer measurements in a cleaner experimental environment (no pileup, full kinematic reconstruction) and under better theoretical control as first-principle calculations are easier in DIS albeit in a more restricted kinematical region. The comparison between the kinematic regions covered by the LHC and Future Circular Collider (FCC) in p–A mode, and the Electron-Ion Collider (EIC) [794] (10-20 GeVelectrons combined with the RHIC nuclear beams or with a new hadron machine at Jefferson Lab) and the LHeC [793] (60 GeVelectrons colliding with the HL-LHC, HE-LHC or FCC nuclear beams) is shown in Fig. 78 (left).

10.2 The physics of ultra-peripheral collisions

Ultra-peripheral collisions are interactions with impact parameter larger than the radial size of the colliding nuclei or protons. In these collisions, one nucleus acts as a source of quasi-real photons that interact with the crossing nucleus or proton. The electromagnetic field intensity and hence the photon flux scales with the charge number squared. At the LHC, these collisions occur abundantly and serve as a laboratory for strong and electromagnetic interactions [577]. After an experimental overview discussing the opportunities, the discussion focuses on vector meson production studies to probe nuclear gluons and dijet production in photonuclear or photon-proton interactions in LHC Run 3 and 4 as a direct access of nPDFs. A dedicated section 11.2 addresses the physics of $\gamma\gamma$ interactions as a probe of QED and beyond the standard model physics. The investigation of γ -proton interactions in pp collision allowed interesting measurements on vector meson production by LHCb [795-798] and will allow for interesting measurements with already recorded and future data. This possibility will not be discussed in detail because the standard running conditions in terms of pile-up and beam optics relevant for forward proton tagging in ATLAS/CMS and the pile-up in LHCb in Run 3 and Run 4 will make comparatively low Q^2 measurements primarily discussed here challenging and likely restricted to short special runs whereas in ALICE, the equivalent luminosity γ -proton luminosity will be smaller or similar than in p-Pb collisions but with the ambiguity which beam emitted the photon.

10.2.1 Experimental overview

An overview of the detector capabilities in Run 3 and Run 4 is given by the four collaborations AL-ICE,ATLAS,CMS and LHCb in the following. Subsequently, the expected statistics for vector meson observables is given in Tab. 12,13,14 followed by an estimate of the available precision for coherent heavy vector meson production and the theoretical assessment of vector mesons and dijet production in UPCs.

- ALICE [462] will take data in both in triggered and in continuous readout mode during Run 3 and Run 4 [58]. Using continuous readout [57] essentially the full delivered luminosity can be integrated without significant trigger and dead-time inefficiencies. Therefore the total acceptance-efficiency factor for UPC events is significantly larger than in LHC Runs 1 and 2; it is determined by the tracking efficiencies and the geometrical acceptances of the Inner Tracking System (ITS) [55], of the Time Projection Chamber (TPC) [56], and of the Muon Spectrometer [799]. The geometrical acceptances of these detectors correspond to the narrow central and forward acceptances defined in Tab. 11. Vector meson yields in the corresponding acceptance are given in Tab. 12,13,14. Final-state neutron emission in UPC events can be detected by the zero-degree calorimeters (ZDC) which will also take data in continuous readout mode, and vetoes can be imposed using the fast interaction trigger detector (FIT).
- ATLAS UPC measurements in Run 3 and Run 4 will continue to be an important part of the experiments heavy-ion programme and will benefit from several detector upgrades in that period. Prior to Run 3, the Zero Degree Calorimeter (ZDC), which is key for UPC event identification by measuring forward neutrons, will be upgraded. At mid-rapidity, the Inner Tracker (ITk) upgrade [251] will increase the acceptance for charged particles to $|\eta| < 4$ for Run 4. The High Granularity

Timing Detector (HGTD) [252] will complement the spatial information of the ITk with timing information in the region $2.4 < \eta < 4.0$, with a resolution of 30 ps. In addition, upgrades to the ATLAS trigger and data-acquisition system will take place prior to both Run 3 and Run 4 [800], and will enable advanced triggering capabilities. Together these will aid the study of low-mass resonances and the continuum as well as jets in UPC events.

- CMS UPC studies will benefit from the upgraded inner tracker for Run 4 which will provide a large acceptance for charged particles up to $|\eta| < 4$ [446]. The improved CMS level-1 trigger and data acquisition rate (up to 60 GB/s) will provide opportunities for more innovative and sophisticated triggers to capture a wide variety of processes. In addition, the proposed MIP Timing Detector [801], which is around 1.16 m away from the beam pipe could provide a time resolution around 30 ps. By combining it with other detectors, MIP will provide proton, pion, and kaon separation for $p_{\rm T}$ values between 0.7 to 2 GeV/cin the midrapidity region ($|\eta| < 1.5$). These detector upgrades and the increased trigger performance summarised in [802] will facilitate the study low-mass UPC resonant states and UPC heavy flavour studies. In addition, the reconstruction algorithms will be improved with the addition of the four-layer pixel system. The CMS-acceptance in Run 3 corresponds to the wide central range defined in Tab. 11. Yields in acceptance for vector mesons are provided in Tab. 12,13,14.
- LHCb is well suited for exclusive production studies in ultra-peripheral collisions. In particular, its optimisation for flavour physics within its acceptance $2 < \eta < 5$ provides an excellent resolution for typical momenta in quarkonium and heavy-flavour exclusive production as demonstrated in Ref. [803]. Its particle identification capabilities allow to measure final states with charged muons, pions, kaons and protons. The upgraded detector will be able to efficiently sample the full delivered luminosity for UPC final states based on a purely software based trigger scheme [804]. Exclusive diphoton analyses as pioneered in ATLAS [805] and CMS [806] are also conceivable with lower $E_{\rm T}$ -thresholds discussed in Section 11.2. The feasibility of inclusive γ -induced measurements in LHCb will require further studies. In Tab. 12,13,14 conceivable vector meson final states are given for LHCb represented by the wide forward acceptance defined in Tab. 11.

10.2.2 Vector meson production

10.2.2.1 Experimental reach

The number of vector mesons expected in the upcoming runs provides an estimate of the expected physics reach. Four toy-model experimental acceptances are shown in Tab. 11. The effects of cuts on the $p_{\rm T}$ of the final state daughter particles are not estimated, because this depends on the analysis and trigger conditions. In general, for two-prong decays, as long as the minimum detectable $p_{\rm T}$ is less than about 1/4 of the final state mass, the effect on the efficiency is limited. The acceptance cuts are applied to both the vector meson rapidity and the daughter particle pseudorapidities. These two selection types have similar effects on the production phase space, and the results would not be very different without the cut on the vector meson rapidity.

Table 12 shows the expected cross sections and rates for five representative vector meson decays, including the decay branching ratios. The rates are calculated using STARlight [588], which has been shown to predict the cross sections for all mesons on proton targets and for ρ photoproduction on heavy targets [807, 808]. Since the ρ vector resonance is broad, the mass range from $2m_{\pi}$ up to $M_{\rho} + 5\Gamma_{\rho}$ is considered. Non-resonant $\pi^{+}\pi^{-}$ production is not included which would lead to a 8% increase of the production rate. The ρ' state represents $\pi^{+}\pi^{-}\pi^{+}\pi^{-}$ states with masses in the range from $4m_{\pi}$ up to 2.5 GeV/ c^{2} . In absence of other guidance and the possibility of more complex resonance structure, the calculation is anchored to the STAR measurement of the $\pi^{+}\pi^{-}\pi^{+}\pi^{-}$ final state in Au–Au collisions at $\sqrt{s_{NN}} = 200$ Gev [809] and uses STARlight to extrapolate in collision energy and collision system.

Since STARlight does not include nuclear shadowing, a rapidity-dependent nuclear shadowing

correction following Ref. [810] for the heavy quarkonium (J/ ψ , ψ (2S) and Y(1S) was applied. The cross sections and rates are hence reduced by factors of 0.42, 0.475 and 0.77 for the J/ ψ , ψ (2S) and Y(1S) respectively.

Condition	Tot.	Central 1	Central 2	Forward 1	Forward 2
		Narrow	Wide	Narrow	Wide
Rapidity	-	y < 0.9	y < 2.4	2.5 < y < 4.0	2 < y < 5
$e/\pi/\mu$ pseudorapidity	-	$ \eta < 0.9$	$ \eta < 2.4$	$2.5 < \eta < 4.0$	$2 < \eta < 5$

Table 11: Table of toy-model acceptance cuts for the different experiments.

Table 12: Table of cross sections and numbers of events in 13 nb⁻¹ integrated luminosity for the different mesons in Pb–Pb collisions. B, M and K denote 10^9 , 10^6 and 10^3 respectively. Both the rates and cross sections include the relevant branching ratios. The cross sections and toy-model acceptances are determined using STARlight [588]. For the J/ ψ , ψ (2S) and Y(1S), rapidity-dependent nuclear shadowing cross sections have been applied following the approach in Ref. [810].

			PbPb			
	σ	All	Central 1	Central 2	Forward 1	Forward 2
Meson		Total	Total	Total	Total 1	Total
$\rho \to \pi^+ \pi^-$	5.2b	68 B	5.5 B	21B	4.9 B	13 B
$\rho' \to \pi^+ \pi^- \pi^+ \pi^-$	730 mb	9.5 B	210 M	2.5 B	190 M	1.2 B
$\phi \to K^+ K^-$	0.22b	2.9 B	82 M	490 M	15 M	330 M
${ m J}/\psi o \mu^+\mu^-$	1.0 mb	14 M	1.1 M	5.7 M	600 K	1.6 M
$\psi(2S) \to \mu^+ \mu^-$	30µb	400 K	35 K	180 K	19 K	47 K
$Y(1S) \rightarrow \mu^+ \mu^-$	2.0 µb	26 K	2.8 K	14 K	880	2.0 K

The rates for light mesons are very large, enough to support billion-event samples of the ρ and ρ' , and hundreds of millions of ϕ , allowing the studies discussed below. Beyond precise cross section measurements detailed below, the rates for J/ ψ , $\psi(2S)$ and Y(1S) should allow tomographic measurements which can be used to infer information on the nuclear wave function and which is outlined in Section 10.2.2.3. In the $\pi^+\pi^-$ channel, the study of pairs with masses above 2 Gev/ c^2 comes in reach.

In the hadronic decay $\phi \to K^+ K^-$, the kaons have a momentum of only 135 MeV/c in the ϕ rest frame. Since the kaon momentum is dominated by the longitudinal momentum acquired from the ϕ -meson, the kaons are produced with large pseudorapidity. Therefore, the acceptance for this channel is low at around midrapidity even without a considering a minimal kaon- $p_{\rm T}$. Consequently, an observation is very difficult with the potential exception of the far forward region where the kaons are significantly Lorentz boosted. Alternately, measurements exploiting the leptonic decay channels $\phi \to \mu^+ \mu^-$ ore⁺e⁻ despite the small branching ratios. The feasibility can be conservatively estimated by scaling the $\phi \to K^+ K^-$ rates down by Br ($\phi \to \mu^+ \mu^-$)/Br($\phi \to K^+ K^-$) $\approx 5.9 \times 10^{-4}$ neglecting the acceptance increase for leptons w.r.t. kaons.

The exploration of double vector mesons photoproduction by a single ion-ion pair by exchange of two independent photons becomes available. The expected ratio of $\rho\rho$ photoproduction to single ρ photoproduction is about 1 : 600 while the predicted ratio of $\rho J/\psi$ to ρ is about 1 : 160 [811]. These events should display significant quantum correlations.

In p–Pb collisions, the per-nucleon centre-of-mass system is boosted by 0.465 units of rapidity from the lab frame, and there are two possible directions for the beams, protons from the +z direction, or from the -z direction influencing strongly the available kinematics for the forward detectors. Furthermore, there are two possibilities for the photon emitter, 'lead-shine' (γ p), the photon from the lead nucleus, or 'proton-shine' (γ Pb), the photon from the proton. These two vector meson production channels can be in principle separated by their different $p_{\rm T}$ scale corresponding to $\hbar/R_{\rm proton}$ for 'lead-shine' (γ p) and $\hbar/R_{\rm Pb}$ for 'proton-shine' (γ Pb). The rates are calculated for these two possibilities in separate tables.

The total p–Pb luminosity of 2000 nb⁻¹ assumed to be equally divided between the two possible proton beam directions. For the asymmetric (around y = 0) detectors, these two Runs are considered separately, listing them as 'forward' (in the proton-going hemisphere) and 'backward' (in the lead-going hemisphere).

Tables 13 and 14 show the cross sections and rates for the lead-shine and proton-shine cases respectively. Lead-shine is dominant, with proton-shine contributing less than 10% to the total rate. The extraction of the proton-shine component by fitting to the different p_T spectra for proton-shine and lead-shine will be challenging and the measurement precision depends strongly of the momentum resolution of the detector. Nuclear shadowing corrections are not applied, the reduction factors are similar for the proton-shine cross sections as for the lead-lead collisions. No attempt to calculate the rates for the ρ' on proton targets is undertaken due to the large rate uncertainties.

Table 13: Table of cross sections and numbers of events for the different mesons in p–Pb collisions for 'lead-shine' (a photon from the lead scattering from the proton). The rates are for the 2000 nb^{-1} integrated luminosity noted above, split evenly between the two possible proton directions. For the central regions, the net luminosity is 2000 nb^{-1} since both directions contribute, but for the forward (FW) and backward (BW) directions, the net luminosity is only 1000 nb^{-1} each. B, M and K denote 10^9 , 10^6 and 10^3 respectively. Both the rates and cross sections include the relevant branching ratios.

pPb - lead shine, γp									
	σ	All	Ctl. 1	Ctl. 2	FW 1	FW 2	BW 1	BW 2	
Meson		Total	Total	Total	Total	Toal	Total	Total	
$\rho \to \pi^+ \pi^-$	35 mb	70 B	3.9 B	15 B	2.0 B	5.5 B	850 M	2.0 B	
$\phi \to \mathrm{K}^+\mathrm{K}^-$	$870 \ \mu b$	1.7 B	65 M	290 M	22 M	120 M	9.7 M	52 M	
$J/\psi \rightarrow \mu^+\mu^-$	6.2 μb	12 M	1.0 M	5.2 M	260 K	800 K	180 K	430 K	
$\psi(2S) \to \mu^+ \mu^-$	134 nb	270 K	22 K	110 K	6.0 K	18 K	3.2 K	7.7 K	
$Y(1S) \rightarrow \mu^+ \mu^-$	5.74 nb	11 K	1.1 K	5.4 K	310	880	41	100	

Table 14: Table of cross sections and rates for the different mesons in p–Pb collisions for 'proton-shine' (a photon from the proton scattering from the lead nucleus). The rates are for the 2000 nb^{-1} integrated luminosity noted above, split evenly between the two possible proton directions. For the central regions, the net luminosity is 2000 nb^{-1} since both directions contribute, but for the forward (FW) and backward (BW) directions, the net luminosity is only 1000 nb^{-1} each. B, M and K denote 10^9 , 10^6 and 10^3 respectively. Both the rates and cross sections include the relevant branching ratios.

pPb - proton shine, γA									
	σ	All	Ctl. 1	Ctl. 2	FW 1	FW 2	BW 1	BW 2	
Meson		Total	Total	Total	Total	Total	Total	Total	
$\rho \to \pi^+ \pi^-$	531µb	1.1 B	83 M	360 M	20 M	44 M	56 M	150 M	
$\phi \to K^+ K^-$	23 µb	46 M	1.3 M	8.0 M	120 K	1.7 M	210 K	3.9 M	
${ m J}/\psi o \mu^+\mu^-$	333 nb	670 K	55 K	290 K	14K	36 K	15 K	41 K	
$\psi(2S) \to \mu^+ \mu^-$	8.9 nb	18 K	1.5 K	7.9 K	380	990	380	1.0 K	
$Y(1S) \rightarrow \mu^+ \mu^-$	0.43 nb	860	93	460	14	34	14	30	

In case of coherent heavy vector meson production in UPC of lead nuclei, the expected experimen-

tal uncertainties are evaluated by the ALICE and CMS collaborations. The vector meson cross section in Pb–Pb UPC can be expressed as a sum of two terms reflecting the fact that either of the colliding ions can serve as a photon source:

$$\sigma(y) = n(+y)\sigma_{\gamma Pb}(+y) + n(-y)\sigma_{\gamma Pb}(-y)$$
(31)

The photoproduction cross sections $\sigma_{\gamma Pb}(y)$ and $\sigma_{\gamma Pb}(-y)$ are coupled and cannot be extracted unambiguously from the measured rapidity differential cross section. However, one can decouple them by measuring vector meson production in UPC with and without additional neutron activity in Zero Degree Calorimeters [812]. Measurement of $\sigma_{0N0N}(y)$ (no neutrons on both sides) and $\sigma_{0NXN}(y)$ (at least one neutron on one of the sides) cross sections provides a system of two equations of two unknown photoproduction cross sections $\sigma_{\gamma Pb}(\pm y)$:

$$\sigma_{0\rm N0N}(y) = n_{0\rm N0N}(+y)\sigma_{\gamma\rm Pb}(+y) + n_{0\rm N0N}(-y)\sigma_{\gamma\rm Pb}(-y), \tag{32}$$

$$\sigma_{0\rm NXN}(y) = n_{0\rm NXN}(+y)\sigma_{\gamma\rm Pb}(+y) + n_{0\rm NXN}(-y)\sigma_{\gamma\rm Pb}(-y), \tag{33}$$

where $n_{0N0N}(\pm y)$ and $n_{0NXN}(\pm y)$ are corresponding photon fluxes, calculable with high accuracy. Solutions of this system of equations can be used to extract photoproduction cross section $\sigma_{\gamma Pb}$.

The expected experimental uncertainties are evaluated in terms of the nuclear suppression factor $R_{\rm Pb}$ which is defined as root square of the ratio of photoproduction cross section $\sigma_{\gamma \rm Pb}$ measured in Pb–Pb UPC and photoproduction cross section in the Impulse Approximation calculated as a reference photoproduction cross section off proton scaled by the integral over squared Pb form factor [813]:

$$R_{\rm Pb}(x) = \left(\frac{\sigma_{\gamma \rm Pb}(x)}{\sigma_{\rm IA}(x)}\right)^{1/2}, \qquad \text{where} \qquad x = \frac{m_V}{\sqrt{s_{\rm NN}}} \exp(-y). \tag{34}$$

Under the assumption that the coherent photoproduction cross section is proportional to the squared gluon density at the scale $Q = m_V/2$, where m_V is the mass of the produced vector meson, this nuclear suppression factor can be used to constrain nuclear shadowing at different scales Q. The theoretical discussion is given in the following Section 10.2.2.2.

The ALICE and CMS collaborations estimate that the uncertainties on luminosity (4%), reference cross section (5%) and photon flux (5%) result in ~ 8% systematic uncertainty on the ratio $\sigma_{\gamma Pb}(x)/\sigma_{IA}(x)$ and ~ 4% uncertainty on the nuclear suppression factor $R_{Pb}(x)$. Detailed information about the uncertainty calculation can be found in existing publications [814–816]. The pseudodata projections for the nuclear suppression factor are shown in Fig. 79 at different scales corresponding to J/ψ , $\psi(2S)$ and Y(1S) photoproduction measurements demonstrating that precision measurements with a range of different scales become available.

10.2.2.2 Coherent vector meson production off nuclei

Extensive data were produced in the previous LHC heavy ion Runs on coherent photoproduction of ρ mesons [807], J/ψ [814,815,817] and to a lesser extent $\psi(2S)$ [816]: $\gamma + A \rightarrow V + A$. Due to the presence of two photon sources, the x-range of these studies is largely limited to $x \ge m_V/\sqrt{s}$. For light mesons these measurements provide information on the pattern of interactions of extended pion-size mesons and such phenomena as shadowing and color fluctuations. For small dipoles like J/ψ and Y(1S), it provided information on the leading twist nuclear shadowing at moderate Q^2 , which is difficult to obtain using other hard probes. The measured factor of ~ 3 reduction of the J/ψ cross section compared to the γp case has significantly constrained the gluon distribution in leading twist approaches [813,818–820]. In the dipole model framework, e.g., [821,822], have had a tendency to predict less supression than seen in the data (in this case, the shape and normalisation of the rapidity distribution rather strongly depends on the form of the dipole cross section and charmonium wave function). The gluon nuclear shadowing in



Fig. 79: Pseudodata projections for the nuclear suppression factor by ALICE and CMS measured with the photoproduction of three heavy vector mesons in Pb–Pb ultra-peripheral collisions are shown. The pseudodata points are derived from EPS09-based photoproduction cross section projections following the method described in Ref. [813].

coherent J/ ψ photoproduction in UPCs was also studied in the k_t -factorization approach [823] in terms of the unintegrated nuclear gluon distribution, which determines the initial condition for the non-linear evolution equation. In the case of ρ meson production, shadowing is a factor of ~ 2 stronger [824] than in the approach based on the Glauber model and the vector meson dominance model.

The higher LHC luminosity and experimental upgrades will allow us to collect vastly improved samples of UPC events. In particular, the planned ALICE continuous readout [825], will eliminate many of the trigger-based constraints that have limited UPC data collection, allowing for high-efficiency collection of large samples of photoproduced light mesons. The increases in sample sizes should be considerably larger than one would expect from merely scaling the luminosity.

In order to conclude this section on the opportunities with vector meson production, we want to give a list of not yet exploited measurements that provide further insight into photonuclear interactions with heavy, light and multiple vector meson production:

- Extend substantially the x range for coherent J/ ψ photoproduction on nuclei using information on the impact parameter distribution in peripheral and ultra-peripheral collisions provided by forward neutron production [812]. The impact parameter distribution can be accessed in the context of UPCs by exploiting the properties of additional photon or hadronic interactions in addition to the photon that produces the vector meson. The rates for the combined processed can be found in [826] and the relationship between impact parameter and additional photon interactions is discussed in [827]. The x-range can be also extended by using p–A collisions to probe the nucleus. In the latter case, one would have to separate coherent J/ ψ production in γA and γp using a much more narrow $p_{\rm T}$ distribution of J/ ψ produced in coherent γA scattering and very good $p_{\rm T}$ resolution for the transverse momentum of the pair (LHCb).
- Measure with high enough statistics coherent Y(1S) production in γp and γA scattering to check the expectation of the 20% reduction of the coherent cross section, which would allow one to probe gluon shadowing at a factor of ~ 10 higher Q^2 than in J/ ψ production.
- Study coherent production of two pions with masses above 1 GeV/ c^2 to study an interplay of soft and hard dynamics as a function of $M_{\pi\pi}$ and $p_T(\pi)$.
- Measure the production of heavier 2π [828], 4π and other resonances on ion targets, and search for the photoproduction of the observed exotic mesons. By using data from both proton targets (at

HERA or the LHC) and ion targets, it is possible to separate the photon-meson coupling constant and the meson-nucleon interaction cross sections.

- Study the photoproduction of multiple vector mesons by a single ion pair [811]. Double photoproduction introduces many quantum correlations, including the possibility of observing stimulated decays of vector mesons. Since the two photons share the same linear polarisation, it will be possible to study photoproduction with polarised photons. In the discussed process, the charged particles from the vector meson decays are aligned (as $\cos^2(\theta)$) with the plane of the linear polarization. If the two vector mesons are aligned along the same plane, then the planes formed by the decay particles will be correlated [827].

10.2.2.3 Nuclear imaging with coherent photoproduction

In coherent photoproduction, production amplitudes from each individual scattering site add with a phase factor $\exp i(\vec{x} \cdot \vec{k})$, where \vec{x} is the location within the nucleus and \vec{k} is the momentum transfer from the nucleus to the vector meson. So, one can Fourier transform the coherent photoproduction cross section, $d\sigma_{coherent}/dt$ to find the location of the scattering sites within the nucleus. This can provide information on the spatial dependence of nuclear shadowing, allowing us to compare shadowing in the centre of nuclei vs. shadowing in the periphery through the transverse profile F(b). Assuming azimuthal symmetry [829, 830]

$$F(b) \propto \frac{1}{2\pi} \int_0^\infty p_{\rm T} dp_{\rm T} J_0(bp_T) \sqrt{\frac{{\rm d}\sigma_{\rm coherent}}{{\rm d}t}}.$$
(35)

One complication is that it is necessary to flip the sign of $\sqrt{d\sigma_{coherent}/dt}$ when crossing each diffractive minimum in the sample.

This calculation is data-hungry, and is subject to a number of theoretical uncertainties. It is also necessary to separate the cross section into its coherent and incoherent components. Nevertheless, the STAR Collaboration applied it to ρ^0 photoproduction in UPCs [831], finding a relatively boxy shape, inconsistent with a Woods-Saxon distribution, but seemingly consistent with expectations from nuclear shadowing. A follow-on study explored the Q^2 evolution of this transverse distribution by dividing the sample into three bins with different dipion mass [832]. That study also considered some of the uncertainties inherent in the analysis, including the finite experimental reach in p_T and the effects of the photon p_T , and the impact of the vector meson wave function.

Studies at the LHC could avoid many of these issues, by triggering on exclusive vector mesons (STAR required that the vector mesons be accompanied by neutrons from mutual Coulomb excitation). This would expand the $p_{\rm T}$ range where a meaningful $d\sigma_{\rm coherent}/dt$ can be extracted, increasing the accuracy of the transform. Also, because of the higher beam energies, the photon $p_{\rm T}$ is less important than it is at RHIC. More importantly, LHC Run 3 and 4 could also extend this study to a wider variety of mesons, including the ρ' (generically, 4π final states with a mass in the 1-2 GeV/ c^2 range) and the J/ ψ . The latter is of particular interest because it is heavy enough to probe gluon shadowing, rather than just nuclear shadowing.

10.2.2.4 Incoherent vector meson production off nuclei

Incoherent diffractive processes provide information on nuclear dynamics, which is complementary to the information one can obtain from coherent scattering. In particular, incoherent J/ψ photoproduction probes quantum fluctuations of the target gluon density [833–838]. The corresponding cross section can be measured in a much larger range of $W_{\gamma N}$ than in the coherent case. This is possible since the activity in the nucleus fragmentation region, for example, neutrons in a ZDC can be used to determine which of two nuclei was a source of photons almost in each event.

One can distinguish two contributions to incoherent diffraction: quasielastic, when the nucleus involved in the strong interaction breaks down into nucleons and nuclear fragments, and inelastic, when

hadrons are produced in the nucleus fragmentation region. At small t the second mechanism gives a $\sim 20\%$ contribution to the incoherent cross section [833, 839]. However, since the t-dependence of the inelastic mechanism is weaker, it dominates the nuclear incoherent cross section for $|t| \ge 0.5 \text{ GeV}/c^2$.

While it is generally understood and accepted that J/ψ photoproduction with target dissociation is sensitive to fluctuations of the gluon density of the target, practical realisations of this notion require modeling. Notably, proton size fluctuations at t = 0 [833] should be contrasted with proton shape fluctuations [834–836], which in turn can be refined by including fluctuations of number of subnucleonic degrees of freedoms representing regions of high-gluon density, so-called hot spots [837,838]. The latter two approaches are assumed to be valid in an entire range of |t|. Hence, studies of incoherent diffraction in γp scattering via p–A UPCs and γA scattering via ÅUPCs would allow one to map variations of the gluon fluctuations with energy (x) including a possible approach to the black disk regime, where the fluctuations are strongly suppressed [837].

For very large $|t| \ge 1 - 2$ GeV/ c^2 one enters the regime of pQCD, which corresponds to elastic scattering of small-size dipoles off individual small-x partons of the nuclear target [840,841].

Note also that for the proton target, one can use the process $\gamma + p \rightarrow VM(J/\psi) + gap + Y$ at $-t \geq \text{few GeV}/c^2$ to study the perturbative Pomeron. In the kinematics, where Δy is fixed, $d\sigma/dy_{VM} \propto (y_{VM} - \Delta y)^{2\alpha_{Pomeron}-2}$. In BFKL one expects $2\alpha_{Pomeron} - 2 \sim 0.4$ and, hence, a strong rapidity dependence of the corresponding cross section. A larger acceptance of the ATLAS and CMS detectors should allow for a more effective study of these processes than at HERA.

By studying the *t*-dependence and activity in the nucleus fragmentation region it would be possible to separate the two mechanisms of incoherent nuclear scattering as a function of *t*. For small $|t| \le 0.3 - 0.5$ GeV², one can calculate nuclear shadowing for both mechanisms within the leading twist shadowing framework [812]. The shadowing turns out to be large and sensitive to the details of the leading twist shadowing dynamics. At large |t|, one can study the *A* dependence of the discussed reaction for different rapidity gap intervals to track propagation of a small dipole through the nuclear medium. By changing Δy it would be possible to vary strongly the relative role of the initial and final state interaction.

10.2.3 Inclusive and diffractive dijet production in UPC

Ultra-peripheral heavy-ion collisions provide an opportunity to study nuclear modifications of the PDFs in clean photon-nucleus interactions. One possible observable is dijet production as suggested in Ref. [842]. Compared to the dijet production in p–Pb collisions the photo-nuclear events have less underlying event activity since multiparton interactions are significantly suppressed. This enables jet reconstruction at lower transverse momenta allowing to study nPDFs at smaller scales Q^2 and x where the current PDF uncertainties are more pronounced. As the virtuality of the photons emitted by the nucleus is negligible, there are two components that need to be taken into account: the photons may interact as unresolved particles or the quasi-real photons may fluctuate into a hadronic state described with photon PDFs. The relative contribution of the direct and resolved components depends on the kinematics of the final state jets. Hence, the uncertainty related to weakly-constrained photon PDFs can be reduced by focusing on the region where direct processes dominate the dijet production.

Here, the photoproduction framework is applied which has been recently implemented into the PYTHIA 8 Monte-Carlo event generator [506] and validated against HERA data [843], to study the potential of the Run 3 and Run 4 program to constrain nPDFs using photo-nuclear dijets. The relevant part of the photon flux is obtained by integrating the impact-parameter dependent flux from $b_{\rm min} = 2R_{\rm Pb} \approx 13.27$ fm. Two different jet kinematics are considered, one corresponding to the preliminary ATLAS measurement [844] with $p_{\rm T}^{\rm lead} > 20$ GeV/c and $m_{\rm jets} > 35$ GeV/c² and one similar to HERA dijet photoproduction data [845, 846] with $p_{\rm T}^{\rm lead} > 8$ GeV/c and $m_{\rm jets} > 14$ GeV/c². In both cases the jets were reconstructed from the generated events using the anti- $k_{\rm T}$ algorithm with R = 0.4 implemented in the FASTJET package [398]. The differential cross sections are shown as a function of x_A in Fig. 80

using NNPDF2.3LO proton PDFs [847] with and without EPPS16 nuclear modifications [782]. The kinematic variables used in the ATLAS study [844] are defined as

$$x_{A} = \frac{m_{\text{jets}}}{\sqrt{s_{\text{NN}}}} e^{-y_{\text{jets}}}, \qquad z_{\gamma} = \frac{m_{\text{jets}}}{\sqrt{s_{\text{NN}}}} e^{y_{\text{jets}}},$$
$$m_{\text{jets}} = \left[\left(\sum_{i} E_{i} \right)^{2} - \left| \sum_{i} \vec{p_{i}} \right|^{2} \right]^{1/2}, \qquad y_{\text{jets}} = \frac{1}{2} \ln \left(\frac{\sum_{i} E_{i} + p_{i,z}}{\sum_{i} E_{i} - p_{i,z}} \right). \tag{36}$$

where the index *i* runs over all accepted jets; E_i and \vec{p}_i denote the jet energy and momentum, respectively. Note that in a leading-order (LO) parton-level calculation, the definitions of x_A and z_γ would exactly match the momentum fractions probed in the PDFs of the nucleus and the photon. The parton-shower emissions and MPIs, and NLO corrections considered below, smear this connection but x_A and z_γ do serve as rather precise hadron-level estimators for the momentum fractions [848].

The uncertainty bands are derived from the EPPS16 error sets and reflect the uncertainties in the current nPDF analyses which are compared to the expected statistical uncertainties of the data in the ratio. Also the contributions from direct and resolved processes are separately plotted. Furthermore, results with the default CJKL photon PDFs [849] are compared to GRV [850] and SASGAM [851] analyses to study the underlying photon PDF uncertainty.

As shown in Figure 80(left), the contribution from resolved processes becomes dominant around $x_A > 0.02$ for $p_{\rm T}^{\rm lead} > 20$ GeV/c. This leads to a more pronounced dependence on the photon PDFs in this region, partly hindering the use of the data from this region in a global nPDF analysis. However, at small-x region, where the nPDF uncertainties are currently large and the dijets in p–Pb do not provide additional constraints, the direct processes dominate the dijet production and the dependence on the photon PDFs is negligible. The dijet cross sections fall off rapidly at small- x_A region which increases the expected statistical uncertainty limiting the small- x_A reach of the observable. With an integrated luminosity of $L_{\text{int}} = 2 \text{ nb}^{-1}$ in Pb-Pb collisions and jet kinematics of the ATLAS preliminary study the expected statistical uncertainties become significant at $x_A \lesssim 2 \cdot 10^{-3}$. The increased luminosity of the LHC Run 3 and 4 increases the potential small- x_A reach only slightly but in the region where nPDF constraints are currently sparse. An effective way to extend the small- x_A reach is to consider jets with lower p_T as demonstrated in Figure 80(right). With a cut of $p_T^{\text{lead}} > 8 \text{ GeV}/c$ and an integrated luminosity $L_{\text{int}} = \text{nb}^{-1}13$ in Runs 3 and 4 it is possible to obtain nPDF constraints down to $x_A \approx 10^{-4}$. Also, the small-x nPDF uncertainties are more pronounced with a lower p_T^{lead} -cut since the nuclei are probed at smaller scales. The theoretical uncertainty related to the limited precision of the photon PDFs could be reduced by performing a similar measurement in p-Pb collisions where the photon flux would be dominantly provided by the Pb ion and the jets produced by γ -p system without any nuclear modifications. This measurement would constrain also the uncertainty related to the impact-parameter rejection that removes the events with hadronic interactions. However, as the minimum allowed impact parameter is smaller in case of p-Pb compared to Pb-Pb and the spectrum of photons is correlated with the impact parameter cut, the kinematics would not be fully comparable preventing a full calibration of the photon flux. As the nPDFs mainly vary the shape of the x_A distributions, part of the theoretical and experimental uncertainties could also be reduced by considering x_A distributions normalized with the integrated cross section.

Inclusive dijet photoproduction in UPCs can also be evaluated to the next-to-leading order (NLO) accuracy of perturbative QCD. The corresponding cross section can be written in the following form [853]:

$$d\sigma(AA \to A + 2jets + X) =$$
(37)

$$\sum_{a,b} \int \mathrm{d}y \int \mathrm{d}x_{\gamma} \int \mathrm{d}x_{b} f_{\gamma/A}(y) f_{a/\gamma}(x_{\gamma},\mu^{2}) f_{b/A}(x_{A},\mu^{2}) \mathrm{d}\hat{\sigma}(ab \to \text{jets}), \qquad (38)$$

where a, b are parton flavours; $f_{\gamma/A}(y)$ is the flux of equivalent photons emitted by one ion, which depends on the photon light-cone momentum fraction y; $f_{a/\gamma}(x_{\gamma}, \mu^2)$ is the PDF of the photon, which



Fig. 80: Photo-nuclear dijet cross sections in ultra-peripheral Pb–Pb collisions at $\sqrt{s_{\text{NN}}} = 5.5$ TeV with leading jet p_{T} cut of 20 GeV/c (left) and 8 GeV/c (right). Results based on PYTHIA simulations are calculated with EPPS16 nuclear modification (blue) and the contributions from resolved (green) and direct (orange) photons are separately shown. Ratio plots show also results with different photon PDF sets and the expected statistical uncertainties corresponding to the LHC (brown) and the Run 3 and and Run 4 (dark blue) luminosities. Corresponding results based on NLO calculations for Pb–Pb collisions at $\sqrt{s_{\text{NN}}} = 5.02$ TeV with nCTEQ15 nPDFs [852] (red) are shown in case leading jet p_{T} cut of 20 GeV/c.

depends on the momentum fraction x_{γ} and the factorisation scale μ ; $f_{b/A}(x_A, \mu^2)$ is the nPDF with x_A being the corresponding parton momentum fraction; $d\hat{\sigma}(ab \rightarrow \text{jets})$ is the elementary cross section for production of two- and three-parton final states emerging as jets in the interaction of partons a and b. The sum over a involves quarks and gluons for the resolved photon contribution and the photon for the direct photon contribution dominating at $x_{\gamma} \approx 1$.

Figure 80 (left) presents predictions of Eq. (38) for the cross section of dijet photoproduction in Pb–Pb UPCs at $\sqrt{s_{\rm NN}} = 5.02$ TeV in the ATLAS kinematics as a function of x_A . The red solid lines and the associated shaded band correspond to the central fit of nCTEQ15 nPDFs and their uncertainty, respectively. The top panel of this figure demonstrates that NLO pQCD correctly reproduces the shape and, at least semi-quantitatively, the normalisation of the preliminary ATLAS data. The lower panel of Fig. 80 shows the ratio of the curves from the upper panel to the result of the calculation, where nCTEQ15 nPDFs are substituted by free proton and neutron PDFs. One can see from this panel that the central value of the ratio of the two cross sections reveals the expected trend of nuclear modifications of nPDFs: $\sim 10\%$ shadowing for small $x_A < 0.01$, which is followed by $\sim 20\%$ antishadowing (enhancement) around x = 0.1 and then $\sim 10\%$ suppression for $x_A > 0.3$. Note that since the uncertainties of nCTEQ15 nPDFs are of the same magnitude as the effect of nuclear modifications, inclusion of this dijet data if global QCD fits of nPDFs should in principle reduce the existing uncertainty.

It is also important to study diffractive dijet photoproduction in UPCs in the reaction $A + A \rightarrow A + jet1 + jet2 + X + A$. NLO pQCD predictions for the cross section of this process in pp, p–A, and ÅUPCs in the LHC kinematics were made in [854]. It was shown that studies of this process on nuclei may shed some light on the mechanism of QCD factorisation breaking in diffractive photoproduction

and, for the first time, give access to nuclear diffractive PDFs and test their models.

10.2.4 Photoproduction of heavy quarks

Photoproduction of open charm and bottom is a direct probe of the gluon content of the target nucleus [855, 856]. The lowest order process, $\gamma + gluon \rightarrow c\overline{c}$ (or $b\overline{b}$) dominates over higher order, resolved processes, in which the photon radiates before it interacts with the target gluon [855]. This process type is a subset of flavour untagged dijets, discussed above. However, open charm and bottom offer some advantages, including very high rates. Based on the leading order calculations in Ref. [855] with the EKS nPDF [857], a total number of 22 billion (10 million) $c\overline{c}$ ($b\overline{b}$) pairs are produced in γ Pb interactions in a 13 nb⁻¹ PbPb data set at $\sqrt{s_{NN}}$ =5.5 TeV. In pPb collisions at $\sqrt{s_{NN}}$ =8.8 TeV with a luminosity of 2 pb⁻¹, 18 billion (100 million) $c\overline{c}$ ($b\overline{b}$) pairs are produced. The measurement of both of these processes should be well feasible despite the small experimental acceptances and branching ratios.

Vertex detectors can detect separated vertices from charm production if the charm is moderately relativistic [858], so it should be possible to study pairs with an invariant mass of $M_{c\overline{c}} \approx 4m_c \approx 6 \text{ GeV}/c^2$. This will enable to study gluon distributions down to Bjorken-x values of around: $x \approx M_{Q\overline{Q}}/(4\gamma m_p) \exp(y)$, where the relativistic boost of the nucleus γ , the rapidity y of the $Q\overline{Q}$ and the proton mass m_p appear. The corresponding scale is $Q^2 \approx M_{Q\overline{Q}}$. Assuming scales down to $M_{c\overline{c}} \approx 6 \text{ GeV}/c^2$, a Bjorken-x lower by a factor 6 can be reached compared to the kinematic selection in the preliminary inclusive dijet ATLAS analysis [844] for which the Bjorken-x reach is depicted in Fig. 80(left). By comparing results from proton targets (in p–A collisions) and heavy-ion targets (in Åcollisions), it is possible to make a direct measurement of nuclear shadowing.

The clear experimental signature for this process consists of a pair of separated vertices each corresponding to a D-meson decay in an event with one large rapidity gap. The ion moving in the same direction as the rapidity gap should remain intact. The full reconstruction of both D-mesons, the cleanest channel, the in specific decay channels requires large large event statistics and might be complemented by an analysis with one D-meson decaying semileptonically. However, the charm production rates are high enough that the low efficiency should be affordable for a precise measurement.

In addition to charm and bottom, it may also be possible to study the photoproduction of top [859, 860]. The rates are small for lead-lead collisions at the LHC (7 pairs in 13 nb⁻¹), but for pPb collisions at $\sqrt{s_{NN}}$ =8.8 TeV, the calculation of Ref. [859] finds that an integrated luminosity of 2 pb⁻¹ corresponds to a production of 110 pairs. These pairs are mostly at fairly central rapidities and the large top semileptonic branching ratio should provide a clear experimental tag. This process could provide a separate probe of gluon distributions at larger x, and very high Q^2 . It would also allow a direct measurement of the electric charge of the top quark.

10.3 The physics of inelastic p-Pb collisions

10.3.1 Experimental overview

Proton-lead collisions are an integral part of the LHC program since the 2012 pilot run. Within collinear factorisation, constraints on our knowledge of the nuclear wave functions could be extended at high Q^2 by dijets and heavy gauge boson available for the first time in nuclear collisions [782]. Insights have been gained at lower Q^2 with heavy-flavour production based on the assumption that their nuclear production modification is dominated by nPDFs [317]. In Run 3 and 4, the increased luminosities and detector upgrades will allow to improve the statistical precision, to extent the kinematic reach and the available processes. The detector capabilities and observables are outlined in view of high-energy QCD studies in p–Pb collisions and experimental measurement projections on selected observables are shown. One upgrade dedicated to low-x physics is separately discussed. In all cases, the statistical and point-by-point uncorrelated systematic uncertainties are not taking into account to place the central point. After

this introduction, two theory contributions zoom in on specific aspects of Run 3-4 data.

- ALICE will measure heavy flavour and charged particle production at mid-rapidity in p–Pb collisions during Run 3 and Run 4. These measurements will constrain the parton densities in the nucleus at moderate x. At forward and backward rapidity, measurements of W and Z-boson production, as well as a range of quarkonia (J/ ψ , $\psi(2S)$, and the Y(nS) family) will be performed. In particular the W and Z boson measurements will constrain the nuclear PDFs, while the quarkonia are also sensitive to final state effects. An example is given for the Z boson in Fig. 83 on the right.
- ATLAS will measure heavy electro-weak bosons, W and Z, with the larger p-Pb dataset which will be available in Run 3-4. The much larger luminosity will allow precision in these measurements significantly surpassing the one currently available. Projected W and Z boson cross sections published in Ref. [768] are shown in Figure 81. Previous results have suggested that the modification of EW boson cross sections that is described with nPDFs, appears to be stronger in more central collisions [861, 862]. Figure 82 (left) shows the projected boson yield ratio of central to peripheral collisions, R_{CP}, for different centrality bins from Ref. [768]. Figure 82(right) shows the yield of Z bosons scaled by the nuclear overlap function, T_{AB}, as a function of centrality. It indicates that the Z boson yield uncertainties will be considerably smaller than the T_{AB} uncertainty.
- CMS will exploit the larger p–Pb dataset available during Run 3–4 and it will hence further improve the precision on differential measurements of W and Z bosons (constraining quark and antiquark nPDFs) [863, 864], as well as dijet and top quark pair ($t\bar{t}$) production (gluon nPDFs) [865, 866]. While first Run 2 results on W boson production already feature experimental uncertainties smaller than the nPDF ones [867], the larger luminosity by a factor 5 to 10 expected in Runs 3–4 will allow for another large jump in the constraints on nPDFs from data. The CMS dijet capabilities with Run 3–4 data [868] are shown in Fig. 84. In addition, novel studies will become possible, such as the measurement of differential cross sections for $t\bar{t}$ production down to the Y meson mass region. The projection for $t\bar{t}$ production [868] is shown together with the precise W boson asymmetry measurement in Run 3–4 [259] in Fig. 85. Heavy flavour meson cross sections will also be measured, which are sensitive to low-x gluon nPDFs: D mesons ($p_{\rm T} > 0.5$ GeV/c), B mesons ($p_{\rm T} > 5$ GeV/c), prompt and non-prompt J/ ψ ($p_{\rm T} > 3$ GeV/c), and Y(nS) down to $p_{\rm T} = 0$. These measurements will benefit from the improved detector and trigger performance in Run 4 [802].
- LHCb will operate during Run 3 and Run 4 with the average charged particle multiplicity in p–Pb and Pb–p collisions that are smaller than the nominal conditions in pp running with an average pile-up of 5 interactions. All p–Pb collisions will be processed in the software trigger. Hence, LHCb can fully profit from the luminosity increase including $p_T = 0$ heavy-flavour production. A natural focus is the study of open and hidden beauty and charm production with improved precision. A novel measurement in the p–Pb collision system is discussed in detail in view of saturation physics in section 10.3.3 and is shown in Fig. 87. A new focus of LHCb, profitting from the increased ion in one of the considered rapidity range is shown in Fig. 83 down to 5 GeV/ c^2 . Direct photon studies in the conversion channel will strongly profit from the increased luminosity. The projections of the LHCb collaboration for p–Pb collisions are discussed in detail in Ref. [300].

10.3.2 Forward calorimeter upgrade of ALICE

The ALICE collaboration is considering to add a high-granularity forward calorimeter (FOCAL) to the experiment to measure direct photon production in the rapidity range 3.0–5.0 and at low $p_{\rm T}$, to probe the gluon density in protons and nuclei at $x \sim 10^{-5}$ where gluon saturation and non-linear effects in the gluon density may become apparent. The FOCAL design is based on the Si-W calorimeter technology, with two or three high-granularity layers with silicon pixel sensors that allow to separate electromagnetic



Fig. 81: Top row: Fiducial cross sections for W^+ (left) and W^- (right) boson production in p–Pb collisions at $\sqrt{s_{NN}} = 8.8$ TeV differential in the charged lepton pseudorapidity measured in the laboratory frame η_{lab} [768]. The cross sections are projected with nuclear effects described by the EPPS16 nPDF set and without any nuclear effects. The boxes represent the projected total uncertainties (quadratic sum of statistical and systematic uncertainties), while vertical bars represent statistical uncertainties (smaller than the marker size). Botton row: Z boson rapidity (left) and transverse momentum (right) differential cross sections [768].



Fig. 82: Left: Uncertainties on measurments of the W^+ (left) and W^- (right) boson yield ratio of central to peripheral collisions $R_{\rm CP}$. The boxes represent the projected total uncertainties (quadratic sum of statistical and systematic uncertainties), while vertical bars represent statistical uncertainties [768]. Right: Projected Z boson yields scaled by $T_{\rm AB}$ as a function of the number of nucleon participants [768]. Statistical uncertainties are everywhere smaller than the marker size, and projected systematic uncertainties (green) are described in the text. The projections are compared with a previous run 1 analysis [861].



Fig. 83: Left: Expected statistical significance of the Z boson rapidity distribution for p–Pb collisions at $\sqrt{s_{_{\rm NN}}} = 8.8$ TeV in Run 3-4 assuming an integrated luminosity of 0.25/pb (0.25/pb) obtained with the proton (Pb ion) going toward the ALICE muon spectrometer. The results are compared with the statistical significance of the published data collected in p–Pb collisions in 2013. Right: $R_{\rm pPb}$ of inclusive Drell-Yan production in $2.5 < y_{cms} < 3.0$, one of five rapidity bins accessible with LHCb. The projection is detailed in Ref. [300].



Fig. 84: Projected dijet η measurements in p–Pb collisions with combined Run 3 and Run 4 data [868].

showers with only a few mm distance between them. This unique high granularity makes it possible to reconstruct neutral pions in the forward direction and to reject the decay photon background for the direct photon measurement.

Figure 86 shows the expected performance of the FOCAL detector for the direct photon measurement in p–Pb collisions. The left panel shows the projected uncertainties, which are 7-8% at high p_T and increase at lower momentum due to the combinatorial background. The right panel shows the impact of the measurement on our knowledge of the nuclear modification of the gluon distribution; the red lines show the current uncertainty, based on the EPS09 nPDFs, but using a broader set of parametrisations for the nuclear PDFs at the initial scale Q_0 , similar to [870]. The grey band shows the uncertainty after including the FOCAL pseudo-data. The improvement in the uncertainty is about a factor 2, but it should be noted that the direct photon measurements probes the gluon distribution at low x directly, while the most of the existing data only probe $x \gtrsim 10^{-3}$ (see Fig. 78). At present, the only measurements that probe the very low x region for nuclear PDFs are forward heavy flavour measurements from LHCb [749, 871] and ALICE [872]; theoretical developments are under way to use these data to constrain the PDFs [317, 873].



Fig. 85: Left: W boson asymmetry measurement with Run 3 and Run 4 with CMS [259]. Middle and right: CMS projected differential $t\bar{t}$ production cross section as a function of the reconstructed lepton $p_{\rm T}$ (middle) and rapidity (right) [868]. The total and statistical uncertainties are represented by the outer and inner error bars, respectively. The bottom panels represent the relative uncertainties (68 confidence-level error bands) in the projected data and theory predictions employing the CT14NLO [869] free proton and EPPS16NLO [782] bound nucleon PDF sets.



Fig. 86: Left: Projected uncertainties on the nuclear modification factor for the forward direct photon measurement in p–Pb collisions with the ALICE FOCAL upgrade. Right: Uncertainty on the modification ratio of the gluon density in the Pb nucleus from a fit to FOCAL pseudo-data. The red lines show the current uncertainty and the grey band shows the uncertainty after including the FOCAL pseudo-data.

The FOCAL program will probe small x in different production channels (quark-gluon Compton scattering vs gluon fusion) and therefore also further test universality and factorisation in this regime [874]. Future measurements of Drell-Yan production in LHCb, as shown in Fig. 91, will also probe this region, but will have a weaker impact for gluons at small Q^2 according to current experimental uncertainty estimates.

In addition to the inclusive direct photon measurement, the FOCAL program will measure forward π^0 production in pp, p–Pb and Pb–Pb collisions, which also provides important constraints for the nuclear PDFs and parton energy loss in Pb–Pb collisions at large rapidity. Correlation measurements of neutral pions and photons will be used to further probe the gluon density and to search for evidence of multiple-gluon interactions which are expected to be important in the high gluon density of the Color Glass Condensate [875, 876].



Fig. 87: LHCb projections [300] for $D^0 \overline{D^0}$ in LHC Run 3 and Run 4 compared with theoretical calculations decribed in section 10.3.3 in 10 % largest multiplicity p–Pb collisions (left panel) and minimum bias, i.e., inclusive, pp collisions (right panel). The D-mesons are both reconstructed in the πK -decay channel.

10.3.3 Transverse momentum dependent and low-x phenomena sensitive observables in $c\bar{c}$ and $b\bar{b}$ production

In recent years, many efforts have been devoted to elucidate the properties of transverse-momentumdependent gluon distributions in the high-energy or small-x limit, and in particular to determine how non-linear saturation effects impact the various distributions. A process which is particularly interesting in this regard is the forward production of a heavy quark-antiquark pair, in high-energy proton-proton or proton-nucleus collisions. For kinematical reasons, in the proton-nucleus case, the proton side of the collision involves large-x partons, while on the nucleus side, small-x gluons participate. Hence, this process can be described in a hybrid approach in which the proton content is described by regular, integrated PDFs, while the small-x dynamics in the nuclear wave function is dealt with using the Color Glass Condensate (CGC) effective theory, from which the gluon transverse momentum dependent distributions (TMDs) naturally emerge [877, 878].

In terms of sensitivity to the QCD saturation regime, the same manifestations are expected than with light quarks [879], although smaller values of x can be reached in the latter case. From the point of view of the TMD content however, heavy quarks are interesting because their production involves linearly-polarized gluons TMDs [880], in addition to the usual unpolarized gluon TMDs, and because, due to non-linear effects, the linearly-polarized TMDs generally differ from their unpolarized partners [881]. The LHCb detector is well-suited to measure heavy hadrons in the forward rapidity region, and given large-enough statistics, one could potentially try and extract information about the linearly-polarized gluons in protons and lead nuclei, from data obtained with LHC unpolarized beams. Before giving cross section estimates, let us briefly give a bit more details about the process and the physics involved.

Forward quark-antiquark pair production in dilute-dense collisions is characterized by three momentum scales: P_t , the typical transverse momentum of a single quark, and always one of the largest scales; k_t , the total transverse momentum of the pair, which is a measure of the transverse momentum of the small-x gluons coming from the target; and Q_s , the saturation scale of the target, which is always one of the softest scales. The value of k_t with respect to Q_s and P_t governs which factorisation scheme is relevant. Indeed, when $k_t \sim Q_s \ll P_t$ (the quark and the antiquark are almost back-to-back), there are effectively two strongly ordered scales k_t and P_t in the problem and TMD factorisation applies [877], implying the involvement of several gluon TMDs that differ significantly from each other, especially in the saturation regime, when $k_t \leq Q_s$ [878]. In the other regime which shall not be discussed here (away
from back-to-back production), k_t and P_t are of the same order and far above the saturation scale, only the linear small-x dynamics is important and the various TMDs differ no more.

In [877,882], the cross section for forward di-jet production in proton-nucleus collisions was calculated within the CGC. It was then shown that, in the back-to-back limit, a TMD factorisation formula could be extracted, the result being the same as in a direct TMD approach (i.e., without resorting to the CGC). However, in contrast to the direct TMD approach, the calculation in the CGC yields explicit expressions for the TMDs in terms of Wilson lines, which can be evolved in rapidity through the nonlinear Jalilian-Marian-Iancu-McLerran-Weigert-Leonidov-Kovner (JIMWLK) equation. In [881], those results were extended to the case of a forward heavy quark-antiquark pair. As already observed earlier (see for instance [880, 883–887]), by keeping a non-zero quark mass, the cross section becomes sensitive to additional TMDs, which describe the linearly-polarized gluon content of the unpolarized target.

The three unpolarized gluon TMDs which describe the quark-antiquark pair production are the adjoint-dipole TMD, the Weizsäcker-Williams TMD, and a third one which is roughly a convolution of two fundamental-dipole TMDs. They are accompanied by three 'polarized' partners, which couple through the quark mass and via azimuthal-angle modulation. This is analogous to what happens in the $\gamma^* A \rightarrow q\bar{q}$ process (in that case not only a non-zero quark mass but also a non-zero photon virtuality brings sensitivity to linearly-polarized gluons) described in Section 10.2.4,10.2.3, although there, only one unpolarized gluon TMD is involved (the Weizsäcker-Williams distribution), along with its polarized partner [886, 887].

Predictions for D mesons are presented in Figure 87 as a function of their relative azimuthal angle near π , along with projections from the LHCb experiment. In the p–Pb kinematical range that is investigated here, roughly 2.800 raw counts are expected over the full azimuthal range for the specified decay channel that can be complemented with other channels as in pp collisions in Ref. [888].

10.3.4 Fully coherent energy loss effects on different final states

The multiple scattering of quarks and gluons traveling in a QCD medium induces the radiation of gluons which carry away some energy of the propagating parton, leading for instance to the jet quenching phenomenon. Therefore, a key ingredient of any parton energy loss calculation is the medium-induced gluon spectrum radiated by the fast propagating color charge. It is of course of crucial importance to know the correct parametric dependence of the induced spectrum, which in general depends on the parton properties (in particular its energy and mass) and those of the medium, since it has a direct impact on the phenomenology of particle production in p–A and Åcollisions. The emission of a gluon radiated by an energetic parton experiencing multiple scattering in a medium takes a typical formation time, t_f , which needs to be compared to the length of the medium, L.

Over the last few years, it has been realized that in a hard process involving incoming and outgoing energetic color charges (which do not have to be identical) being quasi-collinear in the rest frame of the medium, the associated medium-induced gluon spectrum is dominated by *large* gluon formation times, $t_f \gg L$ [479, 889, 890]. In this so-called fully coherent (FC) region, the medium-induced radiated energy is similar to the energy loss of an asymptotic charge. In particular it scales as the energy, and thus exceeds at high energy the average parton energy loss in the Landau–Pomeranchuk–Migdal (LPM) regime, $t_f \lesssim L$, for which the energy dependence is at most logarithmic. This different parametric behavior has important consequences on the phenomenology of hadron production in p–A collisions. In particular, a model based on the fully coherent induced gluon spectrum was shown to describe accurately the quarkonium suppression observed in p–A collisions at all centre-of-mass energies, from the SPS fixed-target experiments ($\sqrt{s} \simeq 20$ GeV) to the LHC ($\sqrt{s} = 5.02$ TeVand $\sqrt{s} = 8.16$ TeV) [891–893]. It is therefore necessary to investigate further the role of fully coherent energy loss on other processes. Because the fully coherent induced gluon spectrum arises from the interference between the emission amplitudes off the initial charge and that off the final state, the effects of FC energy loss are process dependent. Let us review, at a qualitative level, the expected nuclear dependence of different processes which could be measured at the LHC with a high luminosity.

In the absence of color charge in the partonic final state, the energy loss in Drell-Yan (DY) production at leading order is expected to be that of a suddenly decelerated parton, that is, independent of its energy (LPM regime). The effects of parton energy loss in nuclei should therefore play almost no role on DY production in high-energy p–A collisions, since $\Delta E/E \rightarrow 0$ in the high-energy limit. The inclusive production of DY lepton pairs in p–A collisions at the LHC should therefore be free of any parton energy loss effect. As a result, this process might be used advantageously in order to probe possible nuclear PDF effects at small values of x, see e.g., Fig. 83 and Ref. [490] for more details. Assuming a luminosity of $L_{\rm int} = 250 \text{ nb}^{-1}$ and taking the conservative value for the cross section in p–Pb collisions at $\sqrt{s} = 5 \text{ TeV}$, $d\sigma/dy = 40 \text{ nb}$ [490], the number of forward DY lepton pairs, which could be measured by the LHCb experiment is typically 10^4 per rapidity unit. Also interesting, and accessible with a high luminosity, would be the production of diphotons in p–A collisions. This process would be free of energy loss effects, for the same reasons as the DY process, while being sensitive to nPDF in the quark sector, $q\bar{q} \rightarrow \gamma\gamma$, and in the gluon sector through the 'box diagram', $gg \rightarrow \gamma\gamma$.

Note that the energy loss scaling as E (FC regime) would come into play only if *another* energetic charged particle is produced in the final-state in association with the virtual photon (or the diphoton), such that the final state carries a global color charge. Such a situation typically occurs in DY+jet production in p–A collisions. Consider for instance the Compton scattering process, $qg \rightarrow q\gamma^*$. At forward rapidity, an incoming quark from the proton projectile scatters in the medium to produce the final state in a color triplet representation. In this peculiar case of quark to a color triplet final state, one would expect a negative *medium-induced* gluon spectrum (i.e., with stronger gluon radiation in pp than in p–A collisions), leading to an energy *gain* (with respect to pp collisions), $\Delta E \propto (-1/2N_c) \times E$ [890]. Such an unusual dependence would manifest by a slight *enhancement* of DY+jet production in p–A collisions with respect to pp collisions [490], although a quantitative study would be needed to answer whether this effect could be visible in the experiment. Similarly, should this enhancement be small or negligible, the associate production of a prompt photon with a heavy-quark jet might be sensitive to the nPDF of heavy quarks and gluons, as emphasized in Ref. [894]. Using $\sigma_{\gamma c} = 1.2 \times 10^5$ pb in the acceptance of the AL-ICE calorimeter [894] leads to $2.4 \times 10^5 \gamma + c$ -jet events in p–Pb collisions at $\sqrt{s} = 8.8$ TeV assuming $\mathcal{L} = 2 \text{ pb}^{-1}$. This observable should be also accessible in ATLAS, CMS and LHCb acceptances as well.

More pronounced effects of fully coherent energy loss are expected when the final state is in a color octet state, or possibly in higher color representations. An example is the production of a jet pair with not too large transverse momenta (ideally only a few times the saturation scale of the target nucleus). In the case of di-gluon production, the final state can be produced in the 27-plet color representation (with Casimir $C_{27} = 8$) that would lead to an average coherent energy loss proportional to $(N_c + C_{27})/2 = 11/2$, that is almost twice larger that expected if the final state is in a color octet state. Such higher color representations could also be probed in the production of B_c mesons (or in the associate production of a D and a B meson), with a complex final state c c b b. From the number of fitted signal candidates $\mathcal{N} = 10^4$ B_c mesons extracted at LHCb in the semileptonic decay channel in pp collisions at $\sqrt{s} = 8$ TeV with $\mathcal{L} = 2 \text{ fb}^{-1}$ [895], the expected B_c rate in the LHCb acceptance using $\mathcal{L} = 0.5 \text{ pb}^{-1}$ in p–Pb collisions at the same energy is $\mathcal{N} = 5 \times 10^2 \text{ B}_c$ mesons.

10.4 Constraints on nuclear PDFs

10.4.1 Overview

As previously discussed in Sec. 10.1, the nuclear Parton Distribution Functions (nPDFs) are poorly constrained as compared to the proton PDFs. This is mainly due to the lack of high statistics data across the very large nuclear mass number (A) range. In fact, even the precision of the proton PDFs rely crucially on nuclear target data [869, 896, 897]; for example, the neutrino-nucleon deep-inelastic-scattering (DIS) structure functions are essential for decomposing the flavour components of the proton [898–902]. Consequently, improved determinations of the nuclear PDFs and nuclear correction factors could improve the proton PDF precision. Thus, the future LHC Runs 3 and 4 could provide the opportunity to precisely constrain the nPDFs for the Pb-nucleus and, in Run 5 and later, one or more lighter nuclei, and thereby disentangle the nuclear effects from the individual flavour components.

In Fig. 88 selected nPDFs from the literature are displayed.

10.4.2 W and Z boson production



Fig. 88: Comparison of the gluon (left) and strange (right) quark nPDF for a lead nucleus as a function of x for a selection of nuclear PDF sets: HKN07 [903], DSSZ [784], EPPS16 [782], and nCTEQ15 [783].

Inclusive W and Z boson production: Inclusive production of W and Z bosons in pp collisions at the LHC can provide new information on the strange, charm and beauty quark PDFs. Additionally, heavyion W and Z production data from p–Pb and Pb–Pb can provide insight on the nuclear corrections, and this complements other data on nuclear targets as it is at large A (lead) and high energy (and thus, smaller x) [861–864, 867, 907, 908]. For example, ATLAS used inclusive W/Z production data to extract the strange quark component of the proton as displayed in Fig. 89 (left). This yielded a larger strange quark PDF than commonly expected [904, 909]. A recent analysis of the ATLAS and CMS inclusive W and Z differential cross section data at 7 and 8 TeV [910] and the combined HERA inclusive data indicates that while there is no tension between the data sets, the LHC data does support unsuppressed strangeness in the proton at low x at both low and high scales. Certainly this is an area that warrants further study.

To highlight the sensitivity of the heavy ion W/Z production to both the heavy flavour components and the nuclear corrections, in Fig. 89 (right), the correlations between W^+ and W^- cross sections for proton-lead interactions calculated with different input PDFs and assumptions [905] are shown. By comparing the results with and without the strange, charm, beauty quark flavours, it can be observed that these quarks do have a large impact on this observable; hence, this process can provide incisive information about the corresponding PDFs. To see the effect of the nuclear corrections, the CT10 proton result is compared with the other calculations. For the case that only two flavours are considered, the separation of the proton result (CT10) and the nuclear results are quite distinct. In this case, the effect of the specific nuclear correction (nCTEQ15 or EPS09) or the effect of the underling base PDF (CTEQ6.1 or CT10) is minimal. In contrast, when this picture is compared to the five flavour results, the division between the proton and nuclear result is not as simple as the different nuclear corrections and proton baseline PDFs yield a broader range of results. In particular, the strange quark PDFs in the CTEQ6.1 and CT10 proton baseline PDFs are quite different, and this will contribute to the spread of results. Thus, proton-lead production of W/Z is an ideal "laboratory" as this process is sensitive to i) the heavy flavour components, ii) the nuclear corrections, and iii) the underlying "baseline" proton PDF. Thus, high-statistics heavy-ion run data during Run 3 and Run 4 has the potential to reduce the current uncertainties and improve the nuclear PDF determination as illustrated by the projections from ATLAS in Fig. 81 and CMS 85.



Fig. 89: Left: The relative strange-to-down sea quark fractions $r_s = 0.5(s + \bar{s})/\bar{d}$ as compared with predictions from different NNLO PDF sets; figure from Ref. [904]. Right: correlations between W^+ and W^- p–Pb cross sections calculated with different input PDFs and assumptions to illustrate the separate impact of the i) nuclear corrections, ii) heavy flavour components, and iii) base PDFs [905,906].

Fig. 90: Left: The CMS projections [259] for the forward-backward asymmetry in W^{\pm} production (Fig. 85) compared to the original EPPS16 90% confidence-level error bands and those after reweighting with these W^{\pm} data. Right: The change in EPPS16 nuclear PDF modifications for sea quarks and gluons at $Q^2 = 100 \text{ GeV}^2$ upon reweighting with the data shown in the left-hand panel.



 W^{\pm} asymmetries: A way to disentangle the nuclear effects from proton PDF and other theory uncertainties like higher-order corrections even in the absence of a pp reference, is the forward-to-backward ratio [911]. To showcase the potential in the case of W^{\pm} projections (shown in Fig. 85), Fig. 90 presents the effect that PDF reweighting [912] analysis with these data has on EPPS16 nuclear PDFs and the corresponding theory predictions for the asymmetry. The most notable effect is the dramatic reduction in the uncertainties of the gluon PDF. Indeed, W^{\pm} production takes place at a high scale $Q^2 \sim M_W^2$ and even though it is mostly $q_i \bar{q}_j$ processes that make the W bosons, the \bar{q}_j PDFs probed at $\eta_{CM} \gtrsim 0$ are, in practice, dominated by the evolution effects at small x. Thus, it is not that surprising that it is predominantly the gluon component that gets tightly constrained by the W data. The improvement for the light sea quarks (d,u,s) is merely a consequence of better constrained gluons, through QCD dynamics. The large-x ($x \gtrsim 0.1$) part is not really affected by the W[±] data.



Fig. 91: Left: The LHCb projections [300] on $R_{\rm pPb}$ in Drell-Yan process compared to the original EPPS16 error bands and those after reweighting with these data. Right: The improvement in EPPS16 nuclear PDFs for sea quarks and gluons at $Q^2 = 100 \text{ GeV}^2$ upon reweighting with the data shown in the left-hand panel.

Another W^{\pm} observable that would benefit from a high luminosity is an asymmetry

$$[\sigma^{W^+}(y_{\rm CM}) - \sigma^{W^+}(-y_{\rm CM})] / [\sigma^{W^-}(y_{\rm CM}) - \sigma^{W^-}(-y_{\rm CM})]$$

proposed in Ref. [911]. In order to measure such a quantity involving four cross sections and subtractions among them (particularly in the denominator which may fluctuate between positive and negative), an excellent statistical precision — like that achievable during Run 3 and 4 at the LHC would be advantageous. With the present Run 2 luminosity it appears that the data [867] are still not accurate enough to measure this across the full rapidity acceptance.

Low-mass Drell-Yan: The Drell-Yan process at low invariant masses is a luminosity-hungry process where LHC Run 3 and Run 4 can make a difference. It would be very much advantageous to reach this low-mass region experimentally as it offers a possibility to constrain the nuclear PDFs at lower factorisation scale where the nuclear effects are larger. The estimated impact is shown in Fig. 91 where the LHCb estimates for $R_{\rm pPb}$ are compared to the reduction of EPPS16 uncertainties upon performing a PDF reweighting [912] with these data described in detail in Ref. [300]. In the calculation, a full decay-lepton phase space has been assumed, but this has a feeble effect on $R_{\rm pPb}$. Although the Drell-Yan process occurs predominantly via $q\bar{q}$ annihilation, the scale-evolution effects are large and these data mostly constrain the gluon PDFs. Here, it should be noted that the probed x values are already so small, that the parametrisation bias which is prominent at small x [870,913] is probably significant and understates the "true" effect of these measurements. This kind of scan over a wide range of invariant masses with the LHC Run 3 and 4 precision would offer a possibility to test the Q^2 evolution of nuclear PDFs – whether there are corrections to standard DGLAP – much more systematically than only on-shell W[±] - and Z-production measurements do (involving only one scale).

10.4.3 Heavy quark production

Inclusive Heavy Quark Production: A recent study of inclusive heavy quark production in protonlead collisions at the LHC demonstrates this can also help constrain the gluon distribution in nuclei. Specifically, Ref. [914] makes use of LHC p–Pb data on D^0 , J/ψ , $B \rightarrow J/\psi$, Y(1S) meson production [486, 496, 749, 871, 915–926]. They obtain a consistent description of these data assuming nPDF modifications are the dominant source of nuclear modifications in p–Pb collisions. Under this assumption, a clear confirmation of gluon shadowing at small x is found. Additionally, they demonstrate that the inclusion of such heavy-flavour data in a global fit can significantly reduce the uncertainty on the gluon density down to $x \simeq 7 \times 10^{-6}$ while keeping an agreement with the other data of the global fits. A reweighting of the current nPDFs sets with the LHC heavy-flavour data was performed for the nCTEQ15 and EPPS16 sets. In a recent analysis of beauty hadrons by the LHCb collaboration [749, 919, 920] excellent agreement between the reweighted predictions and the measured data for the nuclear modification factor $R_{\rm pPb}$ was found. Furthermore, the precision of these data will allow to further reduce the uncertainty of the low-x nuclear gluon distribution. Due to the lower production rates for beauty hadrons, this kind of observables would clearly profit from a higher luminosity as shown in Fig. 25 in the heavy-flavour chapter. Note that heavy flavour measurements from LHCb extending to larger transverse momenta have also the potential to constrain the intrinsic heavy quark component [927–929].

Prompt Diphoton Production: As discussed in Sec. 10.3.4, other nuclear effects such as coherent small angle gluon radiation may explain the heavy-flavour data without a strong shadowing of the small-x gluon. A way to contrast parton shadowing against effects from coherent energy loss discussed in section 10.3.4, will be to study hard processes with color neutral final states in the context of a global analysis in order to see whether a coherent description of all the data remains possible. The case of vector boson production has already been discussed above which is rather sensitive to the quark distributions. Interesting processes (with more or less color neutral final states) which are sensitive to the gluon distribution are inclusive prompt photon production and diphoton production. The prompt photon observable has been discussed in Section 10.3.2 in the context of the ALICE upgrade. The diphoton production is rather clean with an essentially color neutral final state assuming that the contribution from the fragmentation of quarks and gluons into photons can be strongly suppressed by photon isolation criteria [720, 930]. Due to the small diphoton cross section, this measurement requires high luminosities. Based on the available calculations at next-to-leading order [931, 932], about 6000 events within the fiducial experimental acceptance used in pp collisions by ATLAS can be expected with 2 pb⁻¹ p–Pb collisions.

Heavy Quark Associated Production: The associated production of a heavy quark Q and a vector boson $\gamma/Z/W^{\pm}$ also provides incisive information about the PDFs. For all these processes, the LO contribution comes from the gluon-heavy-quark (gQ) initiated subprocess, making this process very sensitive to the gluon and the heavy quark nuclear parton densities. For the neutral current processes, a prompt photon γ or Z together with a c or b quark can be considered to obtain information of the c and b PDFs, respectively; for the charged current process, the Wc and Wb final states are sensitive to the strange and charm quark, respectively. These channels have been analysed for the LHC pp data [933– 938] and the Wc channel is a key input for the ATLAS/CMS comparison of the strange sea quark content of the proton [910]. The event statistics in 2 pb⁻¹ p-Pb collisions can be expected to be a factor 10 smaller than in the Run 1 CMS [937] and ATLAS [938] allowing for a first measurement in p-Pb collisions.

Additionally, the γ/Z transverse momentum can be used to gauge the initial energy of the massive parton propagating through the dense QCD medium produced in those collisions, making $\gamma/Z+Q$ production a powerful process in order to probe energy loss dynamics in the heavy-quark sector. Furthermore, the comparison of the photon-jet pair momentum, from pp to Pb–Pb collisions, is sensitive to the amount of energy loss by the heavy-quarks and could therefore be used in order to better understand parton energy loss processes in the heavy quark sector.

10.4.4 Top production

The $t\bar{t}$ production provides a complementary observable to the dijets for probing high-*x* gluons [866, 939]. For the very short t-quark life time, they decay, in practice exclusively, to W boson and b quarks. Although, the cleanest channel is the one where both W bosons decay leptonically, it has been recently demonstrated [866] that it is possible to get a clear signal when one of the two W bosons decays leptonically and the other one goes to light-quark jets. This is the preferred observable as the yields are 3 times higher than when both W bosons decay leptonically. The cross sections using the MCFM code [940],



Fig. 92: Left: Projected data for $t\bar{t}$ production during Run 3 and 4 compared with uncertanties of the EPPS16 pdf-set. Right: Number of expected $t\bar{t}$ events in Ar–Ar collisions and in Pb–Pb as a function of the top-quark transverse momentum. the used luminosities are discussed in the text.

process 146 (where the leptonically decaying W comes from top quark) have been estimated at NLO level. The following fiducial cuts have been applied:

$$p_{\rm T}^{\ell^{\pm}} > 30 \text{ GeV/c}, |y^{\ell^{\pm}}| < 2.5, p_{\rm T}^{\rm missing} > 30 \text{ GeV/c},$$

 $p_{\rm T}^{\rm jets} > 30 \text{ GeV/c}, |y^{\rm jets}| < 2.5, R_{\rm isolation}^{\rm jets} = 0.5.$

In this setup, the NLO calculation with the factorisation and renormalisation scales set to the topquark mass yields, per-nucleon, $\sigma_{n-n} \approx 4200$ fb in p–Pb collisions at $\sqrt{s} = 8.8$ TeV, with CT14NLO [869] proton PDFs and EPPS16 [782] nuclear corrections. This is to be multiplied by a factor two to account for the electron and muon final states and by a factor of two to account for the process where the antitop is the one from which the leptonically decaying W originates from. Since this is the nucleon-nucleon cross section, a factor of 208 has to be still multiplied to get the p-Pb cross section. This yields $\sigma_{pPb} \approx 3.5 \text{ nb}$. Considering the 2000 nb^{-1} scenario, and efficiency of 0.75 [939] in b-jet tagging, around 5200 events are expected. Based on this number, the expected nuclear modification factor $R_{\rm pPb}$ following the steps of Ref. [939] has been estimated, assigning each data point a 5% uncorrelated systematic uncertainty (in the current p-Pb measurement of CMS [866] the systematic uncertainty is ~20%). Dividing the < 2.5 interval to 20 bins, the statistical uncertainty varies from 5% ($|y^{\ell^{\pm}}|$ < 1) to 10% (most $|v^{\ell^{\pm}}|$ forward/backward). The resulting $R_{\rm pPb}$ is compared in Fig. 92 with the EPPS16 uncertainty band. In the considered kinematic configuration the expected $R_{\rm pPb}$ is typically a bit above unity for the gluon antishadowing in EPPS16. As can be seen from the left-hand panel of Fig. 92, the expected precision of the measurement is not enough to give significant constraints on nuclear PDFs. In particular, the dijets considered in Section 11.4.3 will probe the same kinematic configuration with a clearly higher precision. However, here much depends on the expected systematic error (taken here to be 5% for each data point separately) and how large are the bin-to-bin correlations. In the p-Ar mode, the higher c.m. energy of $\sqrt{s} \approx 9.4 \,\mathrm{TeV}$ increases the yields around 50% and would also benefit from the higher luminosities, see the next subsection and Sec. 2.4.

This, plus a higher achievable nucleon-nucleon luminosity would render the above case almost completely systematics dominated.

Using the same framework and assumptions as above, the fiducial $t\bar{t}$ yields in A-A collisions have been estimated. Here, the Pb–Pb and Ar–Ar cases have been considered. In Pb–Pb collisions



Fig. 93: The EPPS16 and nCTEQ15 nuclear PDF modifications for gluons at $Q^2 = 10 \text{ GeV}^2$. Left-hand panel corresponds to the Argon nucleus and the right-hand panel to the Lead nucleus. From Ref. [942]

at $\sqrt{s_{\rm NN}} = 5.5 \,{\rm TeV}$, the per-nucleon cross section is $\sigma_{\rm n-n} \approx 1200 \,{\rm fb}$ which translates to around 2000 reconstructed events for $13 \,{\rm nb}^{-1}$ ion-ion luminosity. In the Ar–Ar option, the c.m. energy is slightly higher, $\sqrt{s} = 6.3 \,{\rm TeV}$, which increases the cross section by some 50%. In addition, the achievable ion-ion luminosity is much higher, $3000 \,{\rm nb}^{-1}$ -8800 ${\rm nb}^{-1}$ within 2.75 months of running, see Tables in Chapter 2.4. Thus, the estimated amount of $t\bar{t}$ events is clearly larger, around 30000. The right-hand panel of Fig. 92 shows the event distributions as a function of top-quark transverse momentum $p_{\rm T}(t)$. This shows that, the HL-LHC may allow to probe the space-time picture of heavy-ion collisions using top quarks [941] up to $p_{\rm T}(t) \approx 400 \,{\rm GeV/c}$ in the Pb–Pb case, and up to $p_{\rm T}(t) \approx 700 \,{\rm GeV/c}$ in the Ar–Ar alternative.

10.5 Perspectives with lighter ions

Lighter ions, with the possibility to achieve large integrated luminosities in modest running times, see Sect. 2.4 in the accelerator chapter, offer several interesting opportunities for the study of the initial stage of ion collisions, for small-x physics and for the determination of nuclear parton densities, see Section 10.1.

First, concerning nPDFs, it should be noted that due to the scarcity of nuclear data, a PDF fit or a single nucleus is impossible as discussed in Section 10.1. The different groups [782–784] have adopted different strategies but, generically, they give the parameters in the initial condition to be fitted a dependence on the nuclear mass number. Such dependence acquires different functional forms and, therefore, it constitutes part of the parametrisation bias in the nPDF set. Data on lighter nuclei may help to constrain such parametrisations, see the discussions for UPCs and p–A collisions in Section 10.4. To highlight the current unknowns, Fig. 93 compares the nuclear modifications for Argon and Lead, as given by the EPPS16 and nCTEQ15 global analyses. In particular, the nCTEQ15 prediction varies, even qualitatively, quite significantly between Argon and Lead. This underscores the usefulness of e.g. a p-Ar run at the LHC.

On the other hand, the impact parameter dependence of nPDFs is linked to their dependence on nuclear size. Several models exist (see e.g. [943, 944]), and even less model-dependent approaches like the EPS09s analysis [945] where the dependence on nuclear size was used to constrain the impact parameter dependence. First-principle calculations combining the Gribov theory of inelastic shadowing and factorisation theorems for hard diffraction and DIS relate diffraction in electron-proton collisions with nuclear shadowing. This has been used to predict nuclear shadowing [786,787], including its nuclear size and impact parameter dependence. While such relation is exact for the deuteron, its extension

to larger nuclei has some degree of model dependence. Lighter ions are the ideal place to test the nuclear size dependence without resorting to centrality selection, whose relation with impact parameter is doomed to be as problematic - at least - as found in p–Pb collisions at the LHC.

Lighter ions also offer large luminosities that are important for several aspects:

- Data on beauty mesons and bottomonium in p–A collisions can be used to constrain nPDFs [317] better than their charm counterparts, because of the larger scale given by the mass and by the opportunity that they are less affected by collective effects. But they demand large statistics that can be achieved with lighter ions.
- Larger luminosities will benefit measurements in p-A collisions for observables with large scales, like high-mass DY or dijets, for precise determination of nPDFs.
- UPCs and photon-photon studies [577] in Å collisions will greatly benefit in spite of the $Z^{2(4)}$ factor for the photon luminosities that can be overcompensated by the larger nucleon-nucleon luminosity.
- Larger luminosities can also benefit small-x forward observables that, like dijets [400], aim to reach quite large transverse momenta in p-A collisions.

Lighter ions offer a bridge between small systems and Pb–Pb without requiring centrality selection that is problematic both in pp, p–A and peripheral Å. In the framework of saturation models [778] that aim to describe collective effects in small systems without requiring final state interactions (see e.g. [946] and references therein), the extension from the proton to the nuclear case in some of the phenomenological realisations is done by a simple rescaling of the squared saturation momentum $Q_s^2 \propto A^{1/3}$. And the centrality dependence is assumed to be proportional to the nuclear profile, which leads to strong problems in the nuclear periphery where a dilute situation is restored. Lighter ions offer a check of our ideas on the nuclear size versus energy leading the density that determines saturation, and the use of minimum bias observables instead of centrality-sliced ones that would greatly simplify the phenomenology.

To conclude, lighter ions offer several advantages and disadvantages for initial stage studies. The main disadvantage is the fact that theory calculations usually assume the limit of scattering of a dilute projectile (proton) on a dense target (nucleus). Lighter ion-ion collisions are further from this limit. On the other hand, they offer: (i) a bridge between small and large systems without resorting to centrality selection that would be useful for constraining the nuclear wave function, the collision dynamics and the interpretation of collectivity; (ii) the possibility of larger luminosities for UPCs and forward observables for nPDFs determination and small-x studies; and (iii) a more affordable setup for microscopic calculations of nuclear corrections.

11 Other opportunities with ion and proton beams at the LHC

11.1 Physics motivation for collisions of light ions

Coordinator: Zvi Citron (Ben-Gurion University of the Negev)

Contributors: L. Apolinario (LIP and IST Lisbon), A. Dainese (INFN Padova), J.F. Grosse-Oetringhaus (CERN), J.M. Jowett (CERN), Y.-J. Lee (Massachusetts Institute of Technology), C. Loizides (Oak Ridge National Laboratory), G. Milhano (LIP and IST Lisbon, CERN), A. Milov (Weizmann Institute of Science), J. Pires (CFTP and IST Lisbon), A.M. Sickles (U. Illinois, Urbana-Champaign), U. Wiedemann (CERN), M. Winn (LAL, Orsay and IRFU/DPhN, Saclay)

The collision of ion species with $A \ll A_{Pb}$ is an appealing opportunity to expand the physics programme presented in this document. The recent Xe–Xe run of only eight hours has provided valuable input for the physics performance of ion collisions lighter than Pb at the LHC. Broadly, the advantages of using $A \ll A_{Pb}$ collisions are twofold: smaller collision systems sample key physical parameters beyond what can be probed with Pb–Pb and p–Pb collisions, and they allow higher luminosity running to maximize the accumulation of rare events in heavy-ion collisions. This higher luminosity would enable high-precision measurements for currently rare observables in Pb–Pb collisions as well as the study of observables totally inaccessible in Pb–Pb collisions.

A scenario is envisioned in which the programme is extended in two directions: a) a short run of O–O (A = 16) to study system-size dependence and b) longer running of a species of intermediate A to achieve a large luminosity increase. The choice of the intermediate species will be dictated by the competition of increased luminosity with lower A against the goal of studying the properties of an extended QGP system. Optimizing the choice of species will require further study from the accelerator, experimental, and theoretical communities; in this document Ar–Ar (A = 40) collisions are considered as a test-case for the choice of intermediate ion. It is understood that any choice of collision species will likely require a pilot run prior to beginning dedicated collisions.

Section 2.4 describes the technical capabilities of the LHC to provide lighter-ion collisions, as well as the expected performance for several ion species. For example, for Ar–Ar the expectation for one month of collisions is 1080 nb^{-1} (p = 1.5). In order to compare across different collisions species we consider the nucleon–nucleon integrated luminosity per month of running, which could be larger by a factor 8–25 (for p = 1.5–1.9) with respect to Pb–Pb collisions, i.e. one month of Ar–Ar collisions would be equivalent to ~ 25–80 nb⁻¹ of Pb–Pb collisions. This gain would be the same in all centrality classes (defined in terms of percentiles of the hadronic cross section). Jet quenching estimates for light ion (Ar–Ar) collisions are presented in Sect. 6.5. Section 4.3.2 discusses flow measurements with light ions. A discussion of the role that light ion collisions can play for small-x and nPDF studies is reported in Sect. 10.5. Finally, the implications of Ar–Ar collisions for the study of light-by-light scattering are discussed in Sect. 11.2.

Even a short O–O run can help clarify the uncertainty concerning the onset of QGP or QGP-like phenomena in high-multiplicity pA and pp collisions, as discussed in section 9.10. The search for signals associated with the QGP in O–O collisions should complement the searches in pp and p–Pb collisions. In particular, the O–O system has well-understood collision geometry as described in detail in Sect. 9.10, enabling the study of collisions with low values of $\langle N_{part} \rangle$ that are difficult to select and study in Pb–Pb collisions and that are similar to those associated to high-multiplicity p–Pb events. Colliding O–O at the LHC naturally dovetails with p–O collisions whose significance for cosmic-ray physics is detailed in Sect. 11.3.

Complementing Pb–Pb collisions, the possibility of high-luminosity extended LHC runs with intermediate-A nuclei (e.g. Ar–Ar or Kr–Kr) is an appealing long-term option. The chief promise of these collisions is the possibility of reaching much higher luminosity than Pb–Pb collisions while still producing a QGP over an extended volume of ~ 1000 fm³ in central events. Based on a Glauber Monte Carlo simulation [762], the mean number of participants for Ar–Ar ranges from $\langle N_{part} \rangle \sim 7$ for 60–



Fig. 94: Maximum medium lifetime that can be distinguished from a full quenching baseline with a statistical significance of two standard deviations (2σ) , as a function of luminosity (shown in terms of equivalent Pb–Pb luminosity) for different species and collider energies. A single Ar–Ar run is expected to provide ~ 25–80 nb⁻¹ of Pb–Pb equivalent luminosity. Adapted from [941].

80% centrality to \sim 70 for 0–5% centrality collisions. QGP effects are observed in Pb–Pb collisions with similar number of participants (see e.g. [947,948]). In addition, the much lower underlying event multiplicity in Ar–Ar relative to central Pb–Pb collisions is expected to lead to reduced systematic uncertainties for several observables, from reconstructed jets to all signal affected by large combinatorial backgrounds. These features, combined with the possibility to increase the nucleon–nucleon luminosity by more than one order of magnitude, make Ar–Ar collisions an extremely attractive option for hard-probe measurements that in Pb–Pb collisions are limited or impossible, such as boosted top-quark decay chains for QGP studies [941]. Figure 94 extends the analysis to lighter nuclei and shows that one month of Ar–Ar collisions, with nucleon–nucleon luminosity equivalent to 25–80 nb⁻¹ for Pb–Pb, allows a similar physics reach as the entire Pb–Pb future programme (13 nb⁻¹), namely to probe the QGP density evolution up to a time of about 1.5–2 fm/c. Top quark studies in Ar–Ar collisions in the context of constraints on nPDFs are discussed in Sect. 10.4.4.

Studies with Z bosons are representative examples of the types of measurements that may be undertaken in a lighter ion rare-probes programme. In Fig. 95 the expected number of Z boson candidates (assuming a selection similar to that used by ATLAS and CMS in previous studies) for one month of heavy-ion running as a function of $\langle N_{\rm part} \rangle$ is shown for several colliding ion species, and compared with the expectation for the full Pb-Pb and p-Pb programmes of Runs 3 and 4. The figure demonstrates that the overall yield of Z bosons would be considerably higher for one Ar-Ar run than for several years of Pb–Pb running including both a sufficient number of candidates to study low $\langle N_{\text{part}} \rangle$ collisions unreachable with Pb–Pb collisions as well as moderate $\langle N_{\text{part}} \rangle$ values in which QGP formation is expected. Z bosons are a powerful tool to probe the properties of the QGP in particular in Z+jet events. In these studies the energy of the Z is a direct measurement of the energy of parton that initiated the recoil jet. Therefore, the coverage of a broad range in Z momentum gives access to a jet-energy differential study of jet quenching. the The expected number of Z+jet events from the 0-10% centrality class above a given $p_{\rm T}$ of the Z boson is calculated at NLO (and without including jet suppression) and shown in Fig. 96 as a function of Pb–Pb equivalent luminosity. The Pb–Pb programme in Runs 3 and 4 (13 nb^{-1}) gives 1000 events with $Z p_T > 120$ GeV/c, while a single Ar–Ar one-month run extends the coverage with the same number of events to 140-180 GeV/c. A three-months Ar-Ar programme extends well above 200 GeV/c.

For further study, Z+jet events were simulated in the 10% most central events in Ar-Ar collisions



Fig. 95: The number of Z bosons as a function of $\langle N_{\text{part}} \rangle$ expected for one month of O–O,Ar–Ar, and Kr–Kr collisions at the LHC and for the full expected Pb–Pb and p–Pb programmes. The Z bosons are reconstructed via the di-lepton decay channel with leptonic $p_{\text{T}} > 20 \text{ GeV}/c$ and $|\eta| < 2.5$, and a mass selection of $66 < M_{\ell\ell} < 116 \text{ GeV}$. The bands shown indicate the range of the expected luminosity ranging from p = 1.5 to p = 1.9, as discussed in section 2.4.



Fig. 96: The kinematic reach of Z+jet events as a function of Pb–Pb equivalent luminosity. The curves indicate the maximum $p_{\rm T}$ of the Z boson for which 100 (or 1000) events are expected from 0–10% centrality collisions for a given Pb–Pb equivalent luminosity. 25 and $80 \,{\rm nb}^{-1}$, corresponding to the upper and lower expectations for one month of Ar–Ar collisions, are indicated by the vertical lines.



Fig. 97: The x_{jZ} distribution for pp, Pb–Pb, and Ar–Ar collisions calculated with JEWEL. The 10% most central events are shown for Pb–Pb and Ar–Ar. Ar–Ar collisions are calculated as $N_{\text{part}} = 60$, T = 318 MeV at thermalization time $\tau = 0.63 \text{ fm/}c$ for $\sqrt{s} = 6.3 \text{ TeV}$. The Z boson must have $p_{\rm T} > 60 \text{ GeV}/c$ and be back-to-back ($|\Delta \varphi| > 7/8\pi$) to a jet with $p_{\rm T} > 30 \text{ GeV}/c$.

using Jewel [949] to estimate the expected jet-quenching effects. Details of the use of and limitations of JEWEL for this purpose are discussed in Sect. 6.5. Figure 97 shows the distribution of x_{jZ} , the ratio of the jet transverse momentum to that of the Z boson for 0–10% centrality Ar–Ar events, as well as pp collisions and 0–10% centrality Pb–Pb events ($N_{part} = 356$, T = 260 MeV at thermalization time $\tau = 0.6$ fm/c for $\sqrt{s} = 5.02$ TeV). The Z boson must have $p_T > 60$ GeV/c and be back-to-back ($|\Delta \varphi| > 7/8\pi$) to a jet with $p_T > 30$ GeV/c. The Figure clearly shows that for this observable the jet-quenching phenomena observed in Pb–Pb collisions as modelled by JEWEL are present also in Ar–Ar collisions. More studies are needed to refine modelling of more dilute systems and optimize the choice of colliding species, but taken together the available information suggests the potential of light ion collision systems like Ar–Ar for a heavy-ion rare-probes programme.

11.2 Physics of $\gamma\gamma$ interactions in heavy-ion collisions

Coordinators: Iwona Grabowska-Bold (AGH University of Science and Technology)

Contributors: M. Dyndal (DESY), S. Hassani (Université Paris-Saclay), M. Klusek-Gawenda (IFJ PAN, PL-31342 Kraków, Poland), L. Schoeffel (Université Paris-Saclay), Peter Steinberg (BNL)

Heavy-ion beams are composed of nuclei which carry electric charge Ze (*e* is the electron charge and Z is the atomic number). They are accelerated to nearly the speed of light, thus they generate large electromagnetic (EM) fields. The EM fields generated by the relativistic ion can interact with the other nucleus or its EM fields. Therefore, besides nuclear hadronic interactions, EM interactions also occur in ultra-relativistic heavy-ion collisions. These EM interactions can be studied in so-called ultra-peripheral collisions (UPC) which occur when the distance between two nuclei in the transverse plane is larger than two times the nuclear radius, and hadronic interactions are thus suppressed [572].

A broad range of processes can be studied with $\gamma\gamma$ interactions in UPC. In the following, a few examples of photon-induced processes are considered at the HL-LHC: exclusive production of $\mu^+\mu^-$ or $p\bar{p}$ pairs, a rare process of light-by-light (LbyL) scattering and a potential of searches for axion-like particles (ALP).



Fig. 98: (Upper) Differential cross section for exclusive production of the di-muon pairs as a function of the di-muon mass for $10 < m_{\mu\mu} < 200$ GeV extracted from STARLight. Two scenarios are considered for the nuclear geometry: a realistic skin depth of the nucleus (solid line) or a hard sphere (dashed line). (Bottom) Ratio to nominal as a function of the di-muon mass, where "nominal" stands for the realistic skin depth of the nucleus. Shaded bands represent expected statistical uncertainties associated with a number of signal events in each bin for integrated luminosity of 0.5 nb⁻¹ (yellow), and 10 nb⁻¹ (cyan).

Exclusive production of di-muon pairs $(\gamma \gamma \rightarrow \mu^+ \mu^-)$ in UPC can offer a precision measurement of photon fluxes associated with ion beams, and as such can be used to constrain predictions for the other processes covered in this section. The cross section at high pair mass is also sensitive to the nuclear



Fig. 99: Differential cross sections as a function of $p\bar{p}$ invariant mass (left) and rapidity distance between proton and anti-proton (right) in Pb–Pb collisions at $\sqrt{s_{NN}} = 5.52$ TeV for four experimental acceptance requirements. For ATLAS and CMS experiments two requirements for proton $p_T > 0.2$ GeV/c or $p_T > 0.5$ GeV/c are considered.

geometry assumed in the calculations. Figure 98 presents a differential cross section as a function of the invariant mass of the di-muon system in the range of 10–200 GeV with expected statistical uncertainties represented by two bands corresponding to integrated luminosities of 0.5 nb^{-1} and 10 nb^{-1} . Two scenarios are considered for the nuclear geometry: a realistic skin depth of the nucleus or a hard sphere [950]. For the 10 nb^{-1} scenario, a significant reduction of the statistical uncertainty is expected. This will help in reducing uncertainties from the modelling of the nuclear geometry. The expected upgrades of the ATLAS Zero Degree Calorimeters (ZDC) in the LHC Run 3 will also be important for isolating the contributions to the cross section stemming from dissociative processes.

Exclusive production of $p\bar{p}$ pairs $(\gamma\gamma \rightarrow p\bar{p})$ in heavy-ion collisions is considered as a process which can help verify the existing theoretical approaches. It has been demonstrated that the $\gamma\gamma \rightarrow p\bar{p}$ experimental data [951] from the Belle Collaboration can be successfully described by implementation of several components [952]: the non-resonant proton exchange, *s*-channel tensor meson exchange and the hand-bag model [953]. Figure 99 shows the calculated distributions of invariant mass of the $p\bar{p}$ system, $W_{\gamma\gamma} = M_{p\bar{p}}$ (left panel) and of the difference of rapidities for protons and anti-protons, $y_{\text{diff}} = y_p - y_{\bar{p}}$ (right panel). The ALICE Collaboration can measure $p\bar{p}$ pairs in Pb–Pb collisions at mid-rapidity (|y| < 0.9). The LHCb Collaboration could also provide a complementary measurement of $p\bar{p}$ production in the forward region ($2 < \eta < 4.5$). The upgraded charged particle tracking capabilities of ATLAS and CMS experiments for Run 4 will measure in |y| < 4.0. Corresponding kinematic requirements on transverse momenta and rapidity or pseudorapidity specific for each experiment are presented in the figure legend. The calculations are made for Pb–Pb collisions with $\sqrt{s_{NN}} = 5.52$ TeV. The total cross section predicted for the ATLAS and CMS acceptances for $p_{\rm T} > 0.2$ GeV/*c* ($p_{\rm T} > 0.5$ GeV/*c*) is $\sigma = 793 \,\mu$ b (248 μ b), while LHCb and ALICE requirements lead to $\sigma = 125$ and 105 μ b, respectively.

From the left panel of Fig. 99 one can deduce that the dependence on invariant mass of the $p\bar{p}$ pair is sensitive to the rapidity/pseudorapidity of the outgoing particle. The cut-off at the minimal value of $W_{\gamma\gamma}$ is determined by the minimum $p_{\rm T}$ requirement. The $y_{\rm diff}$ distribution shown in the right panel of Fig. 99 is of particular interest. The broad maximum at $y_{\rm diff} = 0$ corresponds to the region with $|\cos \theta| < 0.6$, where θ denotes the angle of the outgoing nucleon relative to the beam direction in the centre-of-mass frame. An observation of peaks at $y_{\rm diff} = \pm 1$ could be a good test to constraint the theoretical models which predict the elementary cross section. The proposed model has a few parameters (i.e. vertex form factors for the proton exchange, tensor meson s-channel exchanges and a form factor in the hand-bag contribution) which could be constrained with the help of the y_{diff} distributions. The fiducial acceptance requirements imposed on p_{T} do not distort the maxima. If the structures in the y_{diff} distributions indeed exist, the study of $p\bar{p}$ production in UPC can provide an important complimentary information to the existing $\gamma\gamma \rightarrow \ell^+\ell^-$ and $J/\psi \rightarrow \ell^+\ell^-$ data with $\ell = e, \mu$ [815,954].

Evidence of the rare process of LbyL scattering has been established by the ATLAS and CMS Collaborations using Pb–Pb data obtained in 2015 [805, 955] with an integrated luminosity of about 0.4 nb^{-1} . That process can be studied with higher precision using heavy-ion data collected at the HL-LHC. The left panel of Fig. 100 presents a differential cross section as a function of the di-photon rapidity for LbyL scattering for photons with $|\eta^{\gamma}| < 4$ with two photon p_T^{γ} thresholds: 2.0 and 2.5 GeV/c. The LbyL scattering occurs in the central region: 91% of the integrated cross section resides within $|\eta^{\gamma}| < 2.37$. A strong dependence on the p_T^{γ} requirement is observed. The cross section increases by a factor of two when the single photon p_T^{γ} threshold is lowered by half a GeV/c from 2.5 to 2.0 GeV/c. The corresponding integrated cross sections in the fiducial region are 112 nb for $p_T^{\gamma} > 2.5$ GeV/c and 221 nb for $p_T^{\gamma} > 2.0$ GeV/c.

The right panel of Fig. 100 shows a detector-level acoplanarity $(=1 - |\varphi_1^{\gamma} - \varphi_2^{\gamma}|/\pi)$ distribution for the di-photon system from LbyL signal and two background processes originating from exclusive production of di-electron pairs $(\gamma\gamma \rightarrow e^+e^-)$ and di-photons produced in central exclusive production $(gg \rightarrow \gamma\gamma)$. The distributions depict simulated events which passed a full simulation of the ATLAS detector with the extended acceptance in pseudorapidity. About 640 LbyL events pass the selection requirements for acoplanarity below 0.01 and $p_T^{\gamma} > 2.5 \text{ GeV}/c$ in 5.02 TeV Pb–Pb collisions with an integrated luminosity of 10 nb^{-1} , in comparison to about 13 events observed in the 2015 data set with the $p_T^{\gamma} > 3.0 \text{ GeV}/c$ requirement. The signal events are peaked at acoplanarities close to zero, while the background processes are distributed either uniformly (di-photons from central exclusive production) or even grow with acoplanarity (e⁺e⁻ pairs from exclusive di-electron production). The latter originates from e⁺e⁻ pairs which trajectories have been bent in the magnetic field before emitting hardbremsstrahlung photons. A limitation of the current analysis is lack of simulation of the trigger response. Based on experience from the analyses of 2015 Pb–Pb data, triggering on photons with $p_T^{\gamma} < 3.0 \text{ GeV}/c$ is challenging, and therefore a dedicated trigger system [956,957].

The LbyL process can also be studied at lower di-photon masses. The differential cross sections as a function of the di-photon mass can be evaluated taking into account acceptance of the AL-ICE experiment, i.e. pseudorapidity limited to $|\eta^{\gamma}| < 0.9$ or in the forward region defined by 2 < $\eta^{\gamma} < 4.5$ in the LHCb experiment, and relatively low energies of outgoing photons [959]. At lower energies ($W_{\gamma\gamma} < 4$ GeV) meson resonances [960] may play an important role in addition to the Standard Model box diagrams [961, 962] or double photon fluctuations into light vector mesons [962] or two-gluon exchanges [963]. Figure 101 shows predictions for LbyL and background processes in the ALICE and LHCb experiments with photon acceptance in $|\eta^{\gamma}| < 0.9$ and $E_{\gamma} > 200$ MeV (top panel) or $2 < \eta^{\gamma} < 4.5$ and $E_{\mathrm{T},\gamma}$ > 200 MeV (bottom panel), respectively, for two systems: Pb–Pb collisions at 5.52 TeV (left panel) and Ar-Ar collisions at 6.3 TeV (right panel). Presented results include the effect of the experimental energy resolution [964, 965]. The black-solid lines depict the LO QED fermionic box mechanism with leptons and quarks. Presented results for the $\gamma\gamma \rightarrow \gamma\gamma$ process are in agreement with calculations from Refs. [966-968]. The green-solid lines show results for the s-channel $\gamma\gamma \rightarrow$ pseudoscalar/scalar/tensor resonances that contribute to the LbyL process. In the present analysis, $\eta, \eta'(958), \eta_c(1S), \eta_c(2S), \chi_{c0}(1P)$ mesons are considered. Their masses, total widths and branching ratios are taken from the PDG [969]. The contributions of pseudoscalar mesons from radiative decays of a coherently-produced vector meson could be sizeable [970] and should be quantified in future studies. The dominant background from the $\gamma\gamma \to \pi^0\pi^0$ process is shown by the blue lines. It becomes



Fig. 100: (Left) Predicted differential cross section as a function of the di-photon rapidity for LbyL scattering for photons with $p_T^{\gamma} > 2.5$ GeV/c (dashed) or $p_T^{\gamma} > 2.0$ GeV/c (solid), and $|\eta^{\gamma}| < 4$ extracted from SuperChic [958]. (Right) Detector-level acoplanarity distribution of the di-photon system for photons from the LbyL signal and background processes in 5.02 TeV Pb–Pb collisions with an integrated luminosity of 10 nb⁻¹. The shaded band in cyan represents expected statistical uncertainties.

non-negligible only when one photon from each $\pi^0 \to \gamma\gamma$ decay is reconstructed in the detector. Two scenarios with and without the acoplanarity requirement of 0.01 are considered. The acoplanarity requirement reduces this background contribution by a factor of 5 in the full $W_{\gamma\gamma}$ region. The experimental data for the $\gamma\gamma \to \pi\pi$ elementary cross section were very well described in Ref. [971]. There simultaneously the total cross section and angular distributions for both charged and neutral pions are shown. Following Ref. [971], here nine resonances, $\gamma\gamma \to \rho^{\pm} \to \pi^0\pi^0$ continuum, Brodsky-Lepage and hand-bag mechanism are included. Figure 101 shows that pionic background dominates at low invariant di-photon mass (below 2 GeV). In the same energy region, one can observe a very clear dominance of η , $\eta'(958)$ mesons over other processes. The inclusion of energy resolution introduces mainly smearing of the contribution from $\gamma\gamma \to \eta$, $\eta' \to \gamma\gamma$ resonance scattering. This contribution is supposed to be measured with good precision. These results suggest that both ALICE and LHCb Collaborations could measure LbyL scattering for $W_{\gamma\gamma} > 2$ GeV in Pb–Pb collisions.

In the case of Ar–Ar collisions, the cross sections are about two orders of magnitude lower, because of the smaller electric charge of Ar nulcei with respect to Pb nuclei. Assuming integrated luminosities of $3.0 - 8.8 \text{ pb}^{-1}$ in a dedicated Ar–Ar run, the LbyL production cross section leads to 1460–4280 signal events for ALICE and 11–34 events for LHCb in a range of $W_{\gamma\gamma} > 2$ GeV. A background contribution from $\gamma\gamma \rightarrow \pi^0\pi^0$ is at the level of 20% for ALICE and 134% for LHCb in this region.

Axions and axion-like particles (ALP) are fundamental components of extensions of the Standard Model, occurring in most solutions of the strong CP problem [972, 973]. Recently an increasing interest has been paid to ALP masses above 1 GeV [974–978]. In particular the Higgs discovery has set spin zero particles in the spotlight of searches for new physics, with scalar and pseudo-scalar particles (elementary or not) as heralds of new phenomena. An interesting feature is that ALP (generically labelled as a in the following) in this mass range would induce an anomalous contribution to the LbyL, via the reaction: $\gamma \gamma \rightarrow a \rightarrow \gamma \gamma$, under the condition that the magnitudes of the EM fields associated with the incident photon are large enough, typically $|\vec{E}| > 10^{18}$ V/m. This has triggered the study presented in Ref. [977], and then in Ref. [979] using the recent observation of LbyL scattering published by the ATLAS experiment in Pb–Pb collisions [805], where the electric field produced by the ultra-relativistic Pb is of the order of 10^{25} V/m (thus satisfying the above condition).

The potential of ALP searches in UPC Pb–Pb collisions is studied using detector-level quantities after the LbyL selection requirements are imposed. The overall selection efficiency (times acceptance)



Fig. 101: Di-photon invariant mass distributions for Pb–Pb collisions at $\sqrt{s_{\text{NN}}} = 5.52 \text{ TeV}$ (left) and Ar–Ar collisions at $\sqrt{s_{\text{NN}}} = 6.3 \text{ TeV}$ (right) for ALICE at mid-rapidity (top) and LHCb at forward pseudorapidity (bottom). The $\pi^0 \pi^0$ background is shown with the acoplanarity requirement of 0.01 (dotted line) and also without it (dashed line).

relative to generated events increases from about 40% to 65% for ALP masses ranging from 7 GeV to 80 GeV. Also, the mass resolution varies from 0.5 GeV at low masses (below 15 GeV) up to 1 GeV for larger masses. In the left panel of Fig. 102 the expected mass distributions for three ALP signal mass values, and the main background from LbyL normalised to integrated luminosity of 10 nb⁻¹ are shown. In this study, other sources of backgrounds are neglected, since they have been found to be small in the LbyL measurement [805]. The invariant mass distribution is used as the discriminating variable, with bin widths comparable to the expected resolution of a narrow resonant signal. Upper limits are set on the product of the production cross section of new resonances and their decay branching ratio into $\gamma\gamma$. Exclusion intervals are derived using the CLs method [980] in the asymptotic approximation. The limit set on the signal strength μ is then translated into a limit on the signal cross section times branching ratio as presented in the right panel of Fig. 102.

In Fig. 103 exclusion limits on the coupling, $1/\Lambda$, as a function of m_a are presented along with the existing results from the compilation discussed in Ref. [981]. The ATLAS 20 nb⁻¹ limit is derived using



Fig. 102: (Left) Mass distribution for the ALP signal shown for three values of the ALP mass: $m_a = 10, 30$ and 80 GeV (in red). Also shown (in blue) the LbyL background (see text). All ALP mass points are generated with $\Lambda = 1$ TeV (1/ Λ is the coupling of the interaction) which follows a convention defined in Ref. [977]. (Right) Expected 95% CLs upper limits on $\sigma_{a \to \gamma\gamma}$.

Pb–Pb collisions at 5.52 TeV. These results demonstrate that heavy-ion collisions have unique sensitivity to ALP searches in the range of $m_a = 7 - 140$ GeV, where the previous results based on available Pb–Pb data by ATLAS and CMS [955, 977] are also shown (labelled as ATLAS $\gamma\gamma \rightarrow \gamma\gamma$ and CMS $\gamma\gamma \rightarrow \gamma\gamma$ in the figure).



Fig. 103: Compilation of exclusion limits obtained by different experiments (see text). In light grey, the ATLAS 20 nb⁻¹ limit at $\sqrt{s_{NN}} = 5.52$ TeV is presented. The ATLAS $\gamma\gamma \rightarrow \gamma\gamma$ represents the exclusion limit derived from the LbyL cross section measured in Pb—Pb collisions by ATLAS [805], while the CMS $\gamma\gamma \rightarrow \gamma\gamma$ limit comes from the recent analysis described in Ref. [955]. A more complete version of the existing constraints on ALPs masses versus coupling, including the constraints in the sub meV range from astrophysical observations and from dedicated experiments such as CAST can be found in Ref. [974].



Fig. 104: Left: Mass composition of cosmic rays quantified by $\langle \ln A \rangle$ as a function of cosmic ray energy *E*. See Ref. [982] for references to data (bands) and model predictions (markers and lines), and the text for a discussion. Right: Impact of changes of the hadron multiplicity N_{mult} (dashed lines) and the energy fraction α (dotted lines) which goes into neutral pions in collisions at the LHC energy scale on EPOS-LHC predictions for X_{max} and $\ln N_{\mu}$ in 10¹⁹ eV air showers, compared to Auger data [983]. The model lines represent all values that can be obtained for any mixture of cosmic nuclei from proton (bottom right) to iron (top left). The dashed and dotted lines represent modifications of N_{mult} and α in steps of $\pm 10\%$ from their nominal values.

11.3 Proton-oxygen collisions for cosmic ray research

Coordinators: Hans Dembinski (MPI for Nuclear Physics, Heidelberg)

Contributors: T. Pierog (Karlsruhe Institute of Technology), R. Ulrich (Karlsruhe Institute of Technology)

The recent coincident observations of gamma rays and neutrinos from the flaring blazar TXS 0506+056 confirmed that active galactic nuclei produce high-energy cosmic rays [984]. This long awaited finding demonstrates that sources of cosmic rays are linked to the most violent places in our universe. Measurements of cosmic rays contribute to the understanding of the high-energy universe. Since cosmic rays are charged and bent by magnetic fields in space onto chaotic paths, their arrival directions at Earth are highly isotropic, but their mass composition carries an imprint of the source physics. Precision measurements of minimum-bias events in proton-oxygen collisions have the unique power to resolve current ambiguities in the mass composition measured with atmospheric air-shower techniques.

Cosmic rays are nuclei from protons to iron (heavier elements are negligible). The energydependent mass composition of cosmic rays is characteristic for different source scenarios, as shown in Fig. 104, left-hand-side, which displays predictions (lines and markers) of the mean-logarithmic-mass $\langle \ln A \rangle$ of cosmic rays. Above particle energies of 10^{15} eV, $\langle \ln A \rangle$ can only be indirectly inferred from extensive air showers, huge secondary particle cascades produced by collisions between cosmic rays and nuclei in the atmosphere. The two leading observables to infer $\langle \ln A \rangle$ are the depth X_{max} of the shower maximum in the atmosphere (yellow band in Fig. 104), and the number N_{μ} of muons produced in the shower (green band in Fig. 104). The width of those bands has two main contributions: the experimental uncertainties, and the hadronic model uncertainties inherent in converting the air shower observables into $\langle \ln A \rangle$.

Leading experiments achieve an instrumental accuracy of 10% of the proton-iron difference, which would strongly discriminate between source scenarios, but air shower simulations are required to convert N_{μ} and X_{max} to $\langle \ln A \rangle$ and this adds a large model uncertainty. The simulations use the multi-



Fig. 105: Comparison of charged particle multiplicity measurements at different center-of-mass energies and in different colliding systems with the EPOS-LHC model [995]. Shown in both plots is $dN/d\eta$.

purpose heavy-ion event generator EPOS-LHC [985], or specialized hadronic interaction models such as QGSJet-II.04 [986] and SIBYLL-2.3c [987]. All are designed to describe nucleus-nucleus and soft-QCD interactions by extrapolating combinations of Regge field theory tuned to available data and perturbative QCD. Uncertainties in these models arise from a lack of data on multiparticle production in the very forward phase-space in hadron-nucleus interactions at the TeV scale.

LHC measurements have already reduced the spread of model predictions for X_{max} in the latest generation of models. This big improvement was due to high-precision measurements of the inelastic cross-section (see e.g. [988] and references therein). Further measurements now have the potential to make the spread negligible. The model spread for N_{μ} is still large and predictions are not consistent with X_{max} for cosmic rays with the same mass. There is overwhelming evidence from air shower experiments [983, 989–992] that the muon number N_{μ} is underestimated in simulations starting at about 10^{16} eV. This corresponds to a cms energy of 4.3 TeV, well accessible by the LHC. Shown in Fig. 104, right-hand-side, is a representative data point from the Pierre Auger Observatory, which is well above EPOS-LHC predictions – and EPOS-LHC and SIBYLL-2.3c are already models which produce the highest muon number of all hadronic interaction models models. This is called the *Muon Puzzle*.

Two aspects of multi-particle production with a strong effect on N_{μ} have been identified [993], the hadron multiplicity N_{mult} and the energy fraction α that goes into neutral pions. The impact of changing these variables in EPOS-LHC at 13 TeV cms energy and extrapolating upward in energy is also shown in Fig.104, right-hand-side. A combined measurement to 5% accuracy of both variables at the LHC would reduce the model uncertainty for the conversion of X_{max} to $\langle \ln A \rangle$ well below the experimental uncertainty of 10%, and has the clear potential to resolve the discrepancy in the muon number N_{μ} . To reach the accuracy goal, the following minimum-bias measurements are desired:

- Double-differential production cross-section for charged pions, kaons, and protons: ALICE $|\eta| < 0.9$, LHCb $2 < \eta < 5$
- Production cross-section for neutral pions and neutrons: LHCf $\eta > 8.4$.
- Energy flow over pseudo-rapidity, separated for hadrons and gammas: CMS+CASTOR $-6.6 < \eta < 5.2$, ATLAS $-4.5 < \eta < 4.5$.

Energy flow measurements separated by hadronic and electromagnetic energy deposit constrain both N_{mult} and α , and can be done further forward than direct measurements of charged tracks. The particle identification provided by the ALICE and LHCb experiments provides important additional information, needed to tune and test internal parameters of hadronic interaction models. In particular, the number of produced baryons was found to strongly affect the number of muons in air showers at ground, despite their small number compared to pions [994].

To meet the accuracy goal, N_{mult} and α need to be measured in proton-oxygen collisions at the LHC, which directly mimic interactions of cosmic rays with the atmosphere. Constraining α and N_{mult}

to 5% with existing and future pp and p–Pb data is very challenging [996], since forward-produced hadrons experience strong nuclear modification [314,749,997]. A sufficiently accurate theory to predict nuclear modification in the p–O system based on pp and p–Pb data is not yet available, and a simple interpolation is not reliable since both systems are far away in ln *A*. The difficulty of predicting hadron production in ion collisions is demonstrated in Fig. 105. EPOS-LHC predictions for Xe–Xe collisions significantly underestimate the observed yields in the central region, despite a satisfactory description of pp, p–Pb, and Pb–Pb collisions. The deviations in Xe–Xe are much larger than what is expected from a simple interpolation [995]. The dominant nuclear effects are expected to be different for light and heavy collision partners. Light nuclei are described by the shell model and nucleon correlations are important. Lead nuclei can be described by a simpler model, essentially a Wood-Saxon potential with reduced nucleon correlations that cannot be probed well in experiments.

Selecting peripheral p–Pb collisions to mimic p-air collisions with the same number of binary collisions was considered as an alternative to direct p–O measurements, but this option also increases the uncertainty too much. Centrality in p–Pb collisions is extracted from the data using various centrality estimators with different selection biases. These biases would increase the uncertainty of the proposed measurements well beyond the target of 5 % [769]. However, p–O measurements could provide a sensitive test of centrality estimators since the thickness of the oxygen nucleus and hence the average number of wounded nucleons is about a factor of two smaller. The advantages of estimating centrality in a small ion system are discussed in Sect. 9.10.

In conclusion, collisions of p-O at the LHC are highly desirable to solve the outlined questions. The corresponding measurements would be a crucial input to cosmic ray physics and have the potential to result in significant advances in the century-long ridde of the origin of cosmic rays. The luminosity requirements to reach the physics goals are moderate. A statistical accuracy better than 5% can be achieved with 100 M minimum-bias events. Luminosity calculations for light ion systems are given in Sect. 2.4. The setup of p-O collisions would follow the successful rapid set-up procedure previously used in the 2012 p-Pb run and the 2017 Xe–Xe run, as described in Sect. 2.5.

It is worthwhile noting that a period of oxygen acceleration in the SPS would also provide the opportunity to complement cosmic-ray related measurements of nuclear fragmentation at NA61/SHINE [998,999] at beam momenta of 150 A GeV/c. These measurements aim at improving our understanding of the cosmic-ray propagation in the Galaxy and to evaluate the cosmic-ray background for signatures of astrophysical dark matter [1000]. Another opportunity is the study of very forward production of hadrons in the p–O system at $\sqrt{s_{NN}} \sim 100 \text{ GeV}$ at the LHCb experiment, by colliding the oxygen beam with proton gas provided by an upgraded SMOG system, as described in Sect. 11.4.

11.4 Fixed-target prospects with LHC beams

Contributors: F. Fleuret (LLLR, Palaiseau), G. Graziani (INFN, Firenze), C. Hadjidakis (IPNO, Orsay), E. Maurice (LLR, Palaiseau), L. Massacrier (IPNO, Orsay), P. Di Nezza (INFN, Frascati), L. Pappalardo (University and INFN Ferrara), P. Robbe (LAL, Orsay), B. Trzeciak (Institute for Subatomic Physics, Utrecht)

Fixed target experiments present many advantages having the versatility of polarised and nuclear targets and allowing one to reach high luminosities with dense and long targets. The 7 TeV proton and 2.76 A.TeV lead beams allow one to reach a centre-of-mass energy per nucleon pair of $\sqrt{s_{\rm NN}} = 115$ GeV and $\sqrt{s_{\rm NN}} = 72$ GeV with a centre-of-mass rapidity boost of 4.8 and 4.2 units, respectively. These energies correspond to an energy domain between the SPS and nominal RHIC energies. The large rapidity boost implies that the backward rapidity region ($y_{\rm cms} \leq 0$) is easily accessible by using standard experimental techniques or existing LHC experiments such as ALICE or LHCb.

The physics opportunities offered by a fixed-target programme at the LHC have been developed in several publications of the AFTER@LHC study group [493, 497, 1001] and can be summarised as follows:

- advance our understanding of the large-x gluon, sea quark and heavy-quark content in the nucleon and nucleus,
- advance our understanding of the dynamics and spin of gluons inside polarised nucleons (if a polarised target were used),
- advance our understanding of the properties of the Quark-Gluon Plasma formed in heavy-ion collisions between SPS and RHIC energies.

11.4.1 Status and future plans in LHCb

The LHCb experiment has pioneered fixed target physics at the LHC since Run 2, using noble-gas targets (helium, neon and argon) obtained by injecting the gas directly in the LHC vacuum pipe in the proximity of the LHCb collision point through the SMOG device [1002]. The nominal target gas pressure of 2×10^{-7} mbar corresponds, for a typical LHC beam of 10^{14} protons, to a luminosity of 6×10^{29} cm⁻²s⁻¹ for collisions occurring in one meter of gas along the beam direction, which is roughly the acceptance of the LHCb vertex detector.

The forward geometry of the detector is particularly well suited for this configuration. It provides three units of pseudorapidity corresponding to mid and backward rapidities ($-2.8 < y_{cms} < 0.2$ for a beam energy of 6.5 TeV), fully equipped with tracking and particle identification. Proton-nucleus and Pb-nucleus collisions using fixed targets of different nuclear size can be studied at the energy scale of $\sqrt{s_{NN}} \sim 100$ GeV with unique coverage of the high-x regime in the target nucleon.

The samples collected during Run 2, corresponding to integrated luminosities up to about 100 nb^{-1} , allowed to perform studies of particle production which are of particular relevance to cosmic ray physics [1003], and to collect unprecedented samples of charmed hadrons in fixed-target collisions at this energy scale [496]. These data can provide unique inputs to discriminate cold nuclear matter effects in heavy-flavour production from the effect of deconfinement, and to study nuclear PDFs at large x. The physics reach of heavy-flavour studies is presently limited by the size of these samples. Measurements of absolute crosssections are also limited in accuracy by the determination of the luminosity, since the gas pressure can be controlled only within $\pm 50\%$ with the SMOG device. For the first fixed-target physics results, the integrated luminosity has been determined from the rate of elastically scattered atomic electrons with a precision of 6% [1003].

An upgraded gas target device, named SMOG2, is currently being developed, and expected to be operational already during Run 3. In the new setup, the gas is contained in a storage cell, consisting of a 20-cm-long open-ended tube with a diameter of 1 cm, fed by a capillary. It allows to increase the gas density in the target by at least one order of magnitude with respect to SMOG, reaching luminosities of

Table 15: Expected yields of reconstructed events for some benchmark channels using the largest fixedtarget data sample acquired with SMOG during the LHC Run 2, and possible with SMOG2, using as example a p–Ar sample of 0.1 fb^{-1} .

	SMOG	SMOG2
	largest sample	example
	p–Ne@68 GeV	p–Ar@115 GeV
Integrated luminosity	$\sim 100~{ m nb}^{-1}$	100 pb^{-1}
syst. error on J/ ψ x-sec.	6–7%	2-3 %
J/ψ yield	15k	35M
D^0 yield	100k	350M
$\Lambda_{ m c}$ yield	1k	3.5M
$\psi(2S)$ yield	150	400k
Y(1S) yield	4	15k
Low-mass (5 $< M_{\mu\mu} < 9 \text{ GeV/}c^2$) Drell-Yan yield	5	20k

order 10^{31} cm⁻²s⁻¹ with proton beams. The target is placed upstream, from -50 to -30 cm, the nominal LHCb collision point and is thus not overlapping the luminous pp region. This opens the possibility to acquire fixed target events simultaneously with collision events with negligible impact to the pp physics program. The new setup would also allow other gases to be injected, notably hydrogen and deuterium, providing pp collisions in fixed-target mode as a reference for all pA collision samples, and extending the physics case to the study of the three-dimensional structure functions of the nucleon through spin-independent observables [1004]. Heavy noble gases as Kr and Xe would also be usable. The device will be equipped with a gas feed system, allowing to know the target gas density at 1% level.

Assuming that about 10% of the beam intensity can be exploited for fixed-target physics, either in synergy with pp data taking or through dedicated runs, samples corresponding to integrated luminosities of order 0.1 fb⁻¹ (using proton beams) and 0.1 pb⁻¹ (using Pb beams) can be collected per year, also profiting from the increased beam intensity provided by the HL-LHC.

Samples of this size would allow copious production of Drell-Yan and heavy flavour states, including $b\bar{b}$ mesons. As an example, rough estimates are provided in Table 15 for the yields of reconstructed events in an assumed sample of p–Ar collisions corresponding to 0.1 fb⁻¹. Substantial advancements in the understanding of parton distributions for gluons, antiquark and heavy-quarks at large x, where PDFs are now poorly constrained, are foreseeable [497, 1001]. The precise determination of heavy hadron production at large x is expected to clarify the extent of the intrinsic heavy quark content in the nucleon [928, 1005], and to constrain modifications of the nuclear PDFs due to initial state effects (anti-shadowing and EMC effect [1006], saturation effects [1007]). Sequential quarkonia suppression is a main signature for deconfinement [416], but is also affected by final state effects as break-up of the heavy quark pair [1008] and statistical recombination [417]. The rich samples of different quarkonia states reconstructed in fixed target data will allow to investigate sequential suppression at an energy scale between the SPS and RHIC/LHC, for collision systems ranging from pp to Pb–Xe. The study of collisions of Pb beams on heavy nuclei has been limited in Run 2 by the detector tracking capabilities and would greatly profit from the higher detector granularity offered by the LHCb upgrade 1 and upgrade 2 detectors.

The fixed-target program also presents a very good testbed for the hydrodynamic description of the QCD medium produced in heavy-ion collisions down to the energy of $\sqrt{s_{NN}} \sim 100$ GeV, thanks to the considerable pseudorapidity coverage, with particle identification capability for pions, kaons and protons as well as neutral particles ϕ , K_S^0 and Λ^0 . Measurements of flow observables and correlations can contribute in particular to shed light on the extension of the hydrodynamic description [1009, 1010] successful at top RHIC energy and at the LHC towards lower beam energies requiring an appropriate treatment of baryon density [1011], and a fully 3 dimensional initial state [1012]. This can provide important inputs to the search for the critical point in the QCD phase diagram [142, 1013]. In addition, the extension of the fluid dynamical paradigm towards smaller collision systems as an explanation of the experimental findings in proton(deuteron)-nucleus and pp collisions at the LHC [302–305, 712] and at RHIC [1014] can be probed in a novel kinematic regime.

The study of ultra-peripheral collisions in the fixed target configuration also offers a complementary kinematic regime to the similar studies in beam-beam collisions discussed in Section 10. An intriguing possibility would be the observation of η_c photoproduction in the PbAr sample, where the cross section is estimated to be of order 1 nb [1015] provided that the signal can be cleanly separated from the background due to radiative J/ ψ decays. This would constitute a confirmation for the existence of the odderon, since the η_c state cannot be produced in the γ -pomeron process and would be under threshold for the γ - γ process in fixed-target PbA collisions. Large samples of exclusively produced ρ^0 and ω are also expected [1016].

Studies of proton collisions on light nuclei provide crucial inputs to the understanding of cosmic ray propagation in the interstellar medium (using H and He targets) and in the atmosphere (using N and O targets). The interpretation of the precise measurements of the antiproton component in cosmic rays, performed in the last decade by the PAMELA [1017] and AMS-02 [1018] space-based missions, is presently limited by the knowledge of the antiproton production in the interstellar medium. The first measurement of antiproton production in pHe collisions has been performed by LHCb with Run 2 data [1003] and has been used to improve the sensitivity to a possible dark matter contribution to the cosmic antiproton flux [1019, 1020]. During Run 3, it is planned to extend these studies with a hydrogen target. The production of antineutrons can also be inferred indirectly by evaluating isospin violation in antiproton production from proton-hydrogen and proton-deuterium collisions.

A better understanding of the bulk of particle production in high-energy hadronic collisions is also crucial to the modeling of atmospheric showers induced by ultra-high-energy cosmic rays. Despite the moderate energy scale, fixed target data offer the unique possibility to use a wide range of nuclear targets, including nitrogen and oxygen, and to study production at large x. The main background for the observation of the astrophysical high-energy neutrino flux, recently established by the IceCube collaboration [1021], originates from neutrinos of PeV energy produced from decays of charmed hadrons in atmospheric showers. A relevant uncertainty on this background is related to the possible intrisic charm contribution to the charm PDF at large x [1022]. Knowledge of light hadron production, notably baryons, at large x will allow to improve the modeling of the number of muons produced in the lateral development of the showers, which is a key observable to infer the mass composition of ultra-high-energy cosmic rays, as discussed in Sect. 11.3.

More ambitious projects for future target upgrades have also been proposed for a possible installation in LHCb on the time scale of HL-LHC. These are beyond the baseline LHCb Upgrade 2. A polarised gas target similar to that used in HERMES [1023], installed upstream of the LHCb vertex detector, would make LHCb a key contributor to spin physics. With this option, LHCb would have access to single spin asymmetries in an unique kinematic range for a variety of final states, including unique measurements with quarkonium and Drell-Yan lepton pairs [1024]. Measurements of the three-dimensional structure of nucleons from hadron collisions would be highly complementary to the program at the electron-ion collider, which is proposed on a comparable time scale.

11.4.2 Opportunities with ALICE

The ALICE detector provides many physics opportunities if running in the fixed-target mode with the LHC proton and lead beams. One of the main strengths of ALICE in the fixed-target mode would be its large rapidity coverage. Assuming a target location at z = 0, the ALICE muon spectrometer would

access the mid- to backward rapidity in the centre-of-mass frame $(-2.3 < y_{cms})^6 < -0.8)$ considering an incident proton beam on the target. For an incident lead beam, the muon spectrometer rapidity coverage is $-1.8 < y_{cms} < -0.3$. In addition, the absorber in front of the muon tracking stations is an asset for background rejection and Drell-Yan studies. The ALICE central barrel offers a complementary coverage to the muon arm by accessing the very backward rapidity region $(-5.7 < y_{cms} < -3.9)$ with the proton beam and $-5.2 < y_{cms} < -3.4$ with the lead beam), reaching the end of the phase space for several probes. Thanks to its excellent particle identification capabilities, particle detection and identification down to low $p_{\rm T}$, unique measurements of soft probes and open heavy flavours can be pursued. Another asset of the ALICE apparatus is the capability to operate with good performance in a high particle density environment. Access to the most central AA collisions at $\sqrt{s_{\rm NN}} = 72$ GeV should be possible if the detector occupancy does not exceed the one expected in Pb–Pb collisions at 50 kHz. In addition, the ALICE Collaboration could potentially devote a significant data taking time to a fixed-target programme (especially with the proton beam), allowing the collection of large integrated luminosities and the investigation of several target species.

Two main solutions are being investigated to deliver fixed-target collisions to ALICE: an internal gaseous target or an internal solid target (coupled to a bent crystal to deflect the beam halo). On the one hand, a gas-jet or a storage cell with levelled gas pressure would allow to deliver about 45 pb⁻¹ of proton-polarised hydrogen collisions to ALICE (260 pb⁻¹ in case of unpolarised H₂ collisions), and 8 nb⁻¹ of Pb–Xe collisions per year. For the luminosity calculation, a time duration of 10^7 s and 10^6 s, corresponding to one LHC year, is considered for the proton and lead beams, respectively. With a gas system the target can be polarised, but requires large space to be installed, most likely outside the ALICE barrel magnet, 7 metres from the nominal Interaction Point (IP). In that case, additional detectors for vertexing are needed as well as studies of the tracking performances of the TPC in such conditions. A simple unpolarised storage cell might potentially be used closer to the current ALICE IP. On the other hand, the usage of an internal solid target coupled to a bent crystal has the advantage of more portability, allowing one to install the target closer to the IP, from 2.75 to 4.7 m to the IP, and thus benefiting of better performances from the current ALICE apparatus. With such a device 37 pb⁻¹ (6 pb⁻¹) of p–C (p–W) collisions, and 5 nb⁻¹ (3 nb⁻¹) of Pb–C (Pb–W) collisions could be registered in ALICE per year. One of these solutions could be installed during LS3.

The performance for several key probes has been investigated using fast simulations [497] and it is summarised in the following.

- Quarkonia: Very large yields are expected for charmonia (up to $\sim 10^6$ J/ ψ) in the ALICE muon spectrometer both in pp and Pb-A collisions⁷ considering one LHC year of data taking. The Y(1S) will also be within reach. Looking at quarkonium suppression as a function of rapidity and the system size would allow one to search for the onset of the QGP formation, and to determine the in-medium modification of the QCD forces for centre-of-mass energies between the ones probed at RHIC and the SPS. In pA collisions, Y(1S) could be used to probe large-x gluons in the target (0.1 < x < 1), in order to constrain the anti-shadowing and EMC effects.
- Drell-Yan: A precise measurement of the Drell-Yan process with the muon spectrometer can probe initial state effects on quarks of momentum fraction 0.05 < x < 0.8 at a mass scale of $M_{\mu\mu} >$ $4 \text{ GeV}/c^2$ from pA to A- collisions. The correlated background from bb and cc pairs in the dimuon decay channel is largely reduced at the lower centre-of-mass energy of the fixed-target mode, with respect to the TeV energy range.
- Open heavy flavour: At $y_{lab} \sim 1$ about 3000 (up to 100) D⁰ are expected to be produced per 0.1 rapidity unit per year in pp (Pb-A) collisions, respectively. The ALICE central barrel probes the end of the D meson phase-space, in particular if the vertex is located at z < 4 m from the IP. This

⁶The rapidity is calculated assuming massless particles.

⁷The largest atomic mass number considered here is W for the solid target and Xe for the gas target.

would allow one to reach very large-x gluon close to 1 at low mass scale, where the contribution of the intrinsic charm component in the proton could largely increase the D meson yield. Hence the ALICE central barrel is well suited to study the large-x intrinsic charm component in the proton.

- Identified charged hadrons: In few hours of Pb–A data taking, it would be already possible to collect up to 10^6 minimum bias events with the ALICE central barrel which would allow one to reach an absolute statistical uncertainty of 0.01 on the elliptic flow coefficient v_2 for pions and protons, 0.02 for kaons, and 0.05 for antiprotons in semi-central events. Measurements of identified particles up to very large rapidities would complement the limiting fragmentation studies carried out by the BRAHMS and PHOBOS experiments at RHIC. In addition, particle yields and flow coefficients measured at large rapidities are powerful tools to constrain the temperature dependence of the medium shear viscosity [1025]. Thermal model calculations also indicate that the temperature and baryonic chemical potential depend on the rapidity [1026–1028] suggesting that one can perform a rapidity scan of the QGP phase diagram in a complementary approach to the Beam Energy Scan programme of RHIC.
- Antiproton production: Measurements of antiproton production cross sections in p–H, p–He, p–C, p–N and p–O collisions are important inputs for theoretical calculations of the secondary cosmic antiproton spectrum [1029–1031], where secondary antiprotons originate from the high energy scattering between the interstellar matter and the primary cosmic rays. The measurement of a cosmic antiproton excess with respect to expectations from secondary antiproton production would open new perspectives on the indirect detection of dark matter or unknown astrophysical mechanisms of cosmic ray acceleration. Complementarily to LHCb [1003], the ALICE central barrel can measure very slow antiprotons down to few hundred MeV momentum. Measuring slow antiprotons produced with the LHC proton beam on a nuclear target is equivalent to the case where the nuclear target travels at TeV energies, hit an interstellar proton at rest and produces an antiproton with high energy. Thanks to the large antiproton yields expected in p–H₂ collisions (larger than 10⁸ per 0.1 rapidity unit per year), the ALICE central barrel is well placed to help constraining the uncertainty on the cosmic antiproton spectrum.
- Strangeness: Thanks to the large yields (~ 10^8 per 0.1 rapidity unit per year at $y_{\rm lab} \sim 1$) of Λ hyperons expected to be produced in the ALICE central barrel by using a longitudinally polarised target with the proton beam, a precise measurement of the longitudinal spin transfer D_{LL} of the Λ hyperon could be carried out. So far only limited experimental results exist with poor precision [1032, 1033]. Measurements with ALICE would give a unique opportunity to study the spin-dependent strange quark (antiquark) densities at 0.35 < x < 0.7.
- Charmonium and pentaquark photoproduction: Exclusive J/ ψ photoproduction is known to be sensitive to gluon Generalised Parton Distributions (GPD) at leading order [1034]. ALICE in fixed target mode would have a unique opportunity to study the yet unknown GPD E_g thanks to the measurement of single transverse spin asymmetries of photoproduced J/ ψ by using a transversally polarised H target with the proton beam [1035, 1036]. About 200 photoproduced J/ ψ per year are expected to be produced in the muon spectrometer acceptance. Moreover, the photoproduction of hidden charm pentaquark states [1037] might be possible in the central barrel acceptance which allows for the access to low photon-proton centre-of-mass energies ($W_{\gamma p} \sim 5$ GeV). About 2 to 20 pentaquarks are expected to be produced in the ALICE central barrel per year in p-H₂ collisions.

Note that these studies were performed assuming a vertex position at the ALICE IP for the quarkonia, identified hadron production, as well as for charmonium and pentaquark photoproduction, and a vertex position at 4.7 m from the IP for the antiproton and strangeness production.

Studies are ongoing to address the technical feasibility of the target system integration in the experiment, as well as simulation studies to evaluate the ALICE apparatus tracking performance for target positions displaced by a few metres from the IP. Moreover, investigations are ongoing to extend the ALICE rapidity coverage for several observables thanks to combined measurements of muons detected

both in the ALICE central and the muon spectrometer (see as an example the work in [1038] in collider mode).

Summary of luminosity requirements and proposed run schedule 12

The physics programme presented in this report requires data-taking campaigns with various colliding systems with centre-of-mass energies and integrated luminosities L_{int} as outlined in the following. In some cases the requirements are updated or new with respect to the present baseline LHC programme (see Sec. 2.2 and Ref. [462]). The main variations are: a much larger L_{int} target for p-Pb collisions, motivated by high-precision studies of both initial and final-state effects, following the surprising discoveries of collective-like effects in small collision systems; a large sample of pp collisions at top LHC energy to reach the highest possible multiplicities with the smallest hadronic colliding system; moderate-statistics samples of O–O and p–O collisions, to study the onset of hot-medium effects and to tune cosmic-ray particle production models, respectively. Finally, as discussed in Sec. 11.1, extended LHC running with colliding intermediate-mass nuclei (as, for example, Ar–Ar or Kr–Kr), offers the unique opportunity of a large increase in nucleon-nucleon luminosity to access novel probes of the QGP and to open a precision era for probes which are still rare with the Pb-Pb system. The working group considers the high-luminosity Pb–Pb and p–Pb programmes to be the priorities that should be pursued in Run 3 and Run 4. High-luminosity runs with intermediate-mass nuclei are regarded as an appealing case for extending the heavy-ion programme at the LHC after LS4. This case, including the choice of the optimal nuclear species, should be studied further from the theoretical and operational points of view, both of which could be informed with one or two pilot runs with different species.

- Pb–Pb at $\sqrt{s_{\text{NN}}} = 5.5 \text{ TeV}$, $L_{\text{int}} = 13 \text{ nb}^{-1}$ (ALICE, ATLAS, CMS), 2 nb^{-1} (LHCb)
- pp at $\sqrt{s} = 5.5$ TeV, $L_{int} = 600 \text{ pb}^{-1}$ (ATLAS, CMS), 6 pb⁻¹ (ALICE), 50 pb⁻¹ (LHCb)
- pp at $\sqrt{s} = 14$ TeV, $L_{\text{int}} = 200 \text{ pb}^{-1}$ with low pileup (ALICE, ATLAS, CMS)
- **p-Pb at** $\sqrt{s_{NN}} = 8.8$ TeV, $L_{int} = 1.2$ pb⁻¹ (ATLAS, CMS), 0.6 pb⁻¹ (ALICE, LHCb)
- pp at $\sqrt{s} = 8.8$ TeV, $L_{\text{int}} = 200 \text{ pb}^{-1}$ (ATLAS, CMS, LHCb), 3 pb^{-1} (ALICE) O-O at $\sqrt{s_{\text{NN}}} = 7$ TeV, $L_{\text{int}} = 500 \ \mu \text{b}^{-1}$ (ALICE, ATLAS, CMS, LHCb)
- p-O at $\sqrt{s_{\text{NN}}}$ = 9.9 TeV, L_{int} = 200 μb^{-1} (ALICE, ATLAS, CMS, LHCb)
- Intermediate AA, e.g. $L_{\text{int}}^{\text{Ar}-\text{Ar}} = 3-9 \text{ pb}^{-1}$ (about 3 months) gives NN luminosity equivalent to Pb-Pb with $L_{\text{int}} = 75-250 \text{ nb}^{-1}$

Based on this requirements, the proposed updated running schedule is reported in the following table. It can be seen that the physics programme for Run 3 and Run 4 discussed in this report is achievable by a modest increase of the "heavy-ion running" time from 12 to 14 weeks per run.

Year	Systems, $\sqrt{s_{_{\rm NN}}}$	Time	$L_{\rm int}$
2021	Pb–Pb 5.5 TeV	3 weeks	$2.3 \mathrm{~nb}^{-1}$
	pp 5.5 TeV	1 week	3 pb^{-1} (ALICE), 300 pb^{-1} (ATLAS, CMS), 25 pb^{-1} (LHCb)
2022	Pb–Pb 5.5 TeV	5 weeks	3.9 nb^{-1}
	O–O, p–O	1 week	$500 \ \mu { m b}^{-1}$ and $200 \ \mu { m b}^{-1}$
2023	p–Pb 8.8 TeV	3 weeks	0.6 pb^{-1} (ATLAS, CMS), 0.3 pb^{-1} (ALICE, LHCb)
	pp 8.8 TeV	few days	1.5 pb^{-1} (ALICE), 100 pb^{-1} (ATLAS, CMS, LHCb)
2027	Pb–Pb 5.5 TeV	5 weeks	3.8 nb^{-1}
	pp 5.5 TeV	1 week	3 pb^{-1} (ALICE), 300 pb ⁻¹ (ATLAS, CMS), 25 pb ⁻¹ (LHCb)
2028	p–Pb 8.8 TeV	3 weeks	0.6 pb^{-1} (ATLAS, CMS), 0.3 pb^{-1} (ALICE, LHCb)
	pp 8.8 TeV	few days	1.5 pb^{-1} (ALICE), 100 pb^{-1} (ATLAS, CMS, LHCb)
2029	Pb–Pb 5.5 TeV	4 weeks	3 nb^{-1}
Run-5	Intermediate AA	11 weeks	e.g. Ar–Ar 3–9 pb^{-1} (optimal species to be defined)
	pp reference	1 week	

13 First considerations on a heavy-ion programme at a High Energy LHC (HE-LHC)

Coordinators: Andrea Dainese (INFN Padova), David d'Enterria (CERN) and Carlos A. Salgado (Instituto Galego de Fisica de Altas Enerxias (IGFAE) Universidade de Santiago de Compostela)

Contributors: L. Apolinario (LIP and IST Lisbon), N. Armesto (Instituto Galego de Fisica de Altas Enerxias (IGFAE) Universidade de Santiago de Compostela), J. Jowett (CERN), G. Milhano (LIP and IST Lisbon, CERN), U.A. Wiedemann (CERN)

13.1 Introduction

In this section the physics opportunities associated with the operation of the HE-LHC with heavy-ion beams are discussed. These first considerations are based on studies carried out in the scope of the Future Circular Collider (FCC) Study group [400, 1039]. For a centre-of-mass energy $\sqrt{s} = 27$ TeV for pp collisions, the relation $\sqrt{s_{\rm NN}} = \sqrt{s}\sqrt{Z_1Z_2/A_1A_2}$ gives the energy per nucleon–nucleon collision of $\sqrt{s_{\rm NN}} = 10.6$ TeV for Pb–Pb (Z = 82, A = 208) and 17 TeV for p–Pb collisions. The present estimate of the integrated luminosity per month of running is larger by a factor two with respect to the current projection for the future LHC runs, i.e. $L_{\rm int} \approx 6 \, {\rm nb}^{-1}$ per experiment, see Section 2.2. The possibility of using nuclei smaller than Pb, like e.g. ⁴⁰Ar or ¹²⁹Xe, to achieve larger instantaneous luminosity is also under consideration. For example, the integrated nucleon–nucleon (NN) luminosity per run for Xe–Xe collisions at $\sqrt{s_{\rm NN}} = 11.5$ TeV could be 2–3 times larger than for Pb–Pb collisions (see integrated $L_{\rm NN}$ values in Tables 3 and 4).

The increase in the centre-of-mass energy and integrated luminosity at the FCC with respect to the LHC opens up novel opportunities for physics studies of the Quark-Gluon Plasma (QGP) described in a recent CERN Yellow Report [400]. Most of these opportunities also apply to the HE-LHC scenario, although with more moderate reach in terms of available probes and kinematics coverage. The main scientific motivations for a heavy-ion programme at the HE-LHC can be summarized as follows.

Novel access to QCD thermodynamics and QCD equilibration processes. Substantially increasing the centre-of-mass energy leads to the creation of initially denser and hotter strongly-interacting systems that expand for a longer duration and over a larger volume, thereby developing stronger collective phenomena. Extrapolations of LHC measurements indicate that the initial energy density increases by a factor about 1.4 from $\sqrt{s_{\rm NN}} = 5.5$ TeV to 10.6 TeV, up to values of about 22–24 GeV/fm³ (at $\tau = 1$ fm/c). These estimates are presented in Section 13.2. The QGP formed at the HE-LHC collision energies reaches closer to a range of temperatures ($T \sim 1$ GeV) where charm quarks start to contribute as active thermal degrees of freedom in the QGP equation of state, thus playing a novel role in QCD equilibration processes.

Characterisation of dense QCD matter through hard-scattering processes. As detailed in Section 13.3, the HE-LHC would provide a much larger abundance of hard-scattering processes than the LHC, as well as novel probes like the top quark and, potentially, the Higgs boson [939, 941, 1040]. A notable example is provided by high-momentum (thus, highly boosted) $t \rightarrow W \rightarrow q\bar{q}$ decay chains, which are promising probes of the QGP time evolution and of the role of colour coherence [941]. The secondary production of charm quarks in scatterings between quark and gluon constituents of the hot QCD medium could reach a substantial fraction of the initial production in partonic hard scatterings and be observed for the first time.

Exploration of saturated parton densities in a previously-uncharted, ultra-dense kinematic domain. As discussed in Section 13.4, the higher centre-of-mass energy of the HE-LHC allows one to explore a wide previously-uncharted kinematic range at low x and Q^2 , where parton saturation is expected to set in. Proton-nucleus collisions would have a coverage down to $x \sim 5 \times 10^{-6}$ in the Pb nucleus at a rapidity of $y \approx 5$.

Table 16: Global properties measured in central Pb–Pb collisions (0–5% centrality class) at $\sqrt{s_{NN}} = 2.76$ TeV and extrapolated to 5.5, 10.6 and 39 TeV. The values for Pb–Pb collisions at the LHC and FCC are from Ref. [400]. The values for Pb–Pb collisions at the HE-LHC are estimated using the same parametrisations as used for the FCC. The values for Xe–Xe collisions at the HE-LHC are all estimated on the basis of the multiplicity extrapolation from the measurement by the ALICE Collaboration [1042] (it is assumed that the transverse energy density scales only with the multiplicity, neglecting possible differences of the average energy per particle between Pb–Pb and Xe–Xe and between the LHC and the HE-LHC).

System, $\sqrt{s_{\rm NN}}$ (Tev)	Pb–Pb, 2.76	Pb–Pb, 5.5	Pb–Pb, 10.6	Xe–Xe, 11.5	Pb–Pb, 39.4
$\mathrm{d}N_{\mathrm{ch}}/\mathrm{d}\eta$ at $\eta=0$	1600	2000	2400	1500	3600
$\mathrm{d}E_{\mathrm{T}}/\mathrm{d}\eta$ at $\eta=0$ (TeV)	1.7 - 2.0	2.3-2.6	3.1-3.4	≈ 1.5	5.2-5.8
Homogeneity volume fm ³	5000	6200	7400	4500	11000
Decoupling time (fm/c)	10	11	11.5	10	13
ε at $\tau = 1 \text{ fm/}c \text{ (GeV/fm}^3)$	12–13	16–17	22-24	≈ 15	35–40

13.2 Global characteristics of nucleus–nucleus collisions at the HE-LHC

Extrapolating measurements of charged particle multiplicity, transverse energy and femtoscopic correlations at lower energies, one can obtain estimates for the growth of global event characteristics from the LHC to the HE-HC and the FCC. In particular, up to the top LHC energy, the growth of chargedparticle multiplicity per participant pair per unit rapidity in nucleus–nucleus collisions is consistent with a slowly-rising power-law: $dN_{ch}/d\eta$ ($\eta = 0$) $\propto (\sqrt{s_{NN}})^{0.3}$ (see e.g. [1041]). As shown in Table 16, for Pb–Pb this amounts to an increase of a factor ~ 1.2 from the LHC to the HE-LHC. The multiplicity in central Xe–Xe collisions is expected to be lower by 35% with respect to Pb–Pb collisions at the HE-LHC, and similar to that of Pb–Pb collisions at 2.76 TeV.

In general, the global event characteristics listed in Table 16 determine the spatio-temporal extent QGP system, and they constrain the thermodynamic conditions that apply after thermalization. The measured transverse energy per unit rapidity $dE_T/d\eta$ (see Table 16) is of particular importance since it constrains the initial energy density. The energy density is expected to increase by a factor 1.4 from the LHC to the HE-LHC, reaching a value of 22–24 GeV/fm³ at the time of 1 fm/c [400]. Using the arguments presented in Ref. [400], an initial temperature as large as $T_0 \approx 600-800$ MeV is expected at the time $\mathcal{O}(0.05 \text{ fm/c})$ after which both nuclei traverse each other at HE-LHC energies. In the case of Xe–Xe collisions the energy density is estimated to be significantly lower than that for Pb–Pb and similar to that of Pb–Pb at LHC energies.

13.3 QGP studies with hard probes

13.3.1 Hard processes in nucleus–nucleus collisions at the HE-LHC

The increase in energy and luminosity (in the case of Xe–Xe) from the LHC to the HE-LHC provides new tools to study the matter created in the collisions of heavy ions. In Fig. 106 (left), cross sections for different processes and different energies are computed with MCFM [1043] at the highest available order. The increases amount to a factor ~ 2 for charm, beauty, W and Z production, ~ 4 for jets with $p_T > 100 \text{ GeV}/c$ and for Higgs, and ~ 6 for top-pair production.

The motivations for measurements of top quarks in heavy-ion collisions are multifold. In p–Pb collisions the cross sections efficiently probe the nuclear gluon PDFs in a wide range in momentum fraction x at high scale $Q \sim m_t$ [939] (see Section 10.4.4). In Pb–Pb collisions, the top-quark observables are sensitive to the energy-loss of heavy quarks [1044] and by selecting boosted (very high p_T) top quarks one could also probe the QGP medium at later times as the decays of boosted top quarks get



Fig. 106: Left: \sqrt{s} -dependence of the cross sections for hard processes of interest for a heavy-ion programme, calculated with MCFM [1043] at the highest available order. The yields for a one month Pb–Pb run ($L_{\text{int}} = 6 \text{ nb}^{-1}$) are also shown. Right: invariant mass distribution for Higgs boson search in the $\gamma\gamma$ decay channel in Pb–Pb collisions at the HE-LHC with $L_{\text{int}} = 70 \text{ nb}^{-1}$ using the selections described in Ref. [1040].

Lorentz time dilated (see next section). For example, the estimated measurable yields for $t\bar{t} \rightarrow b\bar{b} \ell\ell \nu\nu$ (using the per-month luminosities discussed in Section 13.1) with realistic analysis cuts and conservative 50% efficiency for b-jet tagging are about 10⁴ in Pb–Pb collisions and 3 × 10⁴ in Xe–Xe collisions (for the case of three-fold increase of NN integrated luminosity with respect to Pb–Pb).

Another potential novel probe of the QGP medium at HE-LHC and FCC energies is the Higgs boson. The Higgs boson has a lifetime of $\tau \approx 50$ fm/c, which is much larger than the time extent of the QGP phase [1045, 1046]. In Ref. [1045] it has been argued that the Higgs boson interacts strongly with the quarks and gluons of the OGP and the interactions induce its decay in the gluon-gluon or quark-antiquark channels, thus depleting the branching ratio to the most common "observation" channels $\gamma\gamma$ or ZZ^{*}. This "suppression" could potentially be used to accurately determine the final-state density of the produced QGP. Taking a different perspective, in Ref. [1046] it has been argued that nucleus-nucleus collisions could offer a more favourable environment than pp collisions for the study of the Higgs hadronic decays, because of a larger signal-to-background ratio. The cross section for Higgs boson production in Pb–Pb collisions is expected to increase by a factor about 4 when going from $\sqrt{s_{\rm NN}} = 5.5 \text{ TeV}$ to $\sqrt{s_{\rm NN}} = 10.6 \text{ TeV}$ [1040]. A statistically-significant Higgs boson observation in the $\gamma\gamma$ decay channel in Pb–Pb collisions at the HE-LHC requires an integrated luminosity of 70 nb⁻¹ (estimated as in Ref. [1040]), which corresponds to about 12 months with the present machine performance projections. The analysis used similar photon selections as used by ATLAS and CMS in pp collisions: $p_{\rm T}$ > 30,40 GeV/c, $|\eta|$ < 4, $R_{\rm isol}$ = 0.3. The backgrounds included the irreducible QCD diphoton continuum plus 30% of events coming from misidentified γ -jet and jet-jet processes. The corresponding invariant mass distribution is shown in the right-hand panel of Fig. 106. With Xe-Xe collisions the same statistical significance could be reached in 4 months.

13.3.2 Boosted tops and the time evolution of QGP opacity

The HE-LHC would provide large rates of highly-boosted heavy particles, such as tops, Z and W bosons. It is expected that when these particles decay the density profile of the QGP has already evolved. It has been argued that the hadronically-decaying W bosons in events with a $t\bar{t}$ pair can provide unique insights into the time structure of the QGP [941]. This is because the time decays of the top and the W bosons are



Fig. 107: Upper panel: expected top-quark $p_{\rm T}$ distribution in Pb–Pb in the decay channels of interest for the boosted top analysis at $\sqrt{s_{\rm NN}} = 10.6$ TeV after acceptance and efficiency cuts with the statistical uncertainties for $L_{\rm int} = 18$ nb⁻¹, corresponding to three months of data taking (adapted from [939]). Lower-left panel: reconstructed W boson mass at HE-LHC and FCC energies, as a function of the top $p_{\rm T}$. The upper axis refers to the average total time delay of the corresponding top $p_{\rm T}$ bin. Lower-right panel: maximum medium quenching end-time, τ_m , that can be distinguished from full quenching at two standard deviations, as a function of luminosity for various species at LHC and HE-LHC energies. The luminosity for nuclei lighter than Pb is scaled to the Pb–Pb equivalent nucleon–nucleon luminosity. Figures adapted from Ref. [941].

followed by a time-delay in the interaction of the decay products of the W boson with the surrounding medium due to a colour coherence effect. The sum of these three times, several fm/c for boosted tops, would be the time at which the interaction with the QGP begins, providing a unique way to directly measure the time structure of the QGP evolution. In addition, due to colour coherence effects, energy loss would be initially absent for the colour-singlet $q\bar{q}$ decay products of a highly-boosted W boson: the two quarks would start to be quenched only when their distance becomes larger than the colour correlation length of the medium, which depends on the transport coefficient \hat{q} (the average transverse momentum squared that particles exchange with the medium per unit mean-free path) [1047]. The effect on the reconstructed masses of the top and W was studied with different energy loss scenarios as a proof of concept of the potential of these observables to access completely novel quantities in heavy-ion collisions [941].

For this study, the $p_{\rm T}$ reach of top quarks in Pb–Pb collisions is of special importance, because

it determines the maximum time delay for probing the QGP. The upper panel of Fig. 107 shows the estimated $p_{\rm T}$ distribution of the reconstructed top yields in Pb–Pb collisions at the HE-LHC with an integrated luminosity of 18 nb^{-1} , corresponding to three months of data taking. The figure indicates that events with top $p_{\rm T}$ up to about 500 GeV/c can be studied. The reconstructed W-boson mass as a function of top transverse momentum is shown in Fig. 107 (lower-left panel), together with the FCC case. For details on the simulation and reconstruction procedure see [941]. The scale on the upper horizontal axis shows that a $p_{\rm T}$ reach of 500 GeV/c corresponds to an average total time delay $\langle \tau_{\rm tot} \rangle \sim 1.5$ fm/c. The shaded region corresponds to the statistical uncertainty estimated in central Pb-Pb collisions for $L_{\text{int}} = 30 \text{ nb}^{-1}$ (corresponding to 5 Pb–Pb months or 1.5 Xe–Xe months with the present luminosity estimates) and $L_{\text{int}} = 2 \text{ fb}^{-1}$ for the pp reference. Energy loss was simulated by assuming that all particles, except the W-boson decay products, lose 15% of their initial momentum. Average time delays $\tau_m = 1; 2.5; 5 \text{ and } 10 \text{ fm}/c \text{ were considered as effective QGP time evolution profiles. Fig. 107 (right)}$ shows the maximum medium quenching end-time, τ_m , that can be distinguished from full quenching with two standard deviations, as a function of luminosity for various collider energies and species. For Pb–Pb with $L_{\text{int}} = 30 \text{ nb}^{-1}$ (5 months) at the HE-LHC a maximum time of 5–6 fm/c can be accessed, which is much larger than the time up to 1.5 fm/c that can be probed at the LHC with the nominal programme of 10 nb^{-1} . For Xe–Xe collisions, with equivalent NN luminosity larger by a factor 2–3 with respect to Pb-Pb, a time range longer by $\sim 1-2$ fm/c can be covered in the same running period.

13.3.3 Heavy flavour and quarkonia

Heavy quarks (charm and bottom) are among the hard probes that have provided important insights on the formation and the characterics of the QGP, see Sections 5 and 7, and Ref. [244]. In this section, a few selected aspects that could represent novel or particularly remarkable observations at HE-LHC energy are discussed, namely: i) large production of thermal charm from interactions of light quarks and gluons within the QGP; ii) observation of an enhancement of charmonium production with respect to the binary scaling of the yields in pp collisions, as consequence of (re)generation; iii) observation of a colour screening and (re)generation for the most tightly-bound quarkonium state, the $\Upsilon(1S)$.

Interactions between gluons or light quarks of the QGP can lead to the production of $c\bar{c}$ pairs if the energy in the centre of mass of the interaction is of the order of twice the charm quark mass $\sqrt{\hat{s}} \sim 2 m_c \sim 3$ GeV. In Section 13.2 we have estimated that an initial temperature T_0 of 600–800 MeV could be reached at the HE-LHC. With these QGP temperatures a sizeable fraction of the gluons and light quarks have energies larger than the charm quark mass and $c\bar{c}$ pairs can be produced in their interactions. Figure 108 shows the prediction [1048] for the time-dependence of the $c\bar{c}$ rapidity density at mid-rapidity in central Pb–Pb collisons at the HE-LHC. The value at the initial time τ_0 corresponds to the initial hardscattering cross section. Both calculations show a rapid increase after τ_0 with a final value that is larger by up to 20% than the initial production. This enhancement could be observed for the first time at the HE-LHC and provide a handle on the initial temperature of the QGP. The abundance of charm quarks also has an effect on the QGP equation of the state: the inclusion of the charm quark in lattice-QCD calculations results in a sizeable increase of $P/T^4 \propto n_{d.o.f.}$ for T > 400 MeV, as discussed in the context of the FCC [400].

The measurements of the nuclear modification factor of J/ψ at the LHC [740, 744, 1049] are described by models that include dissociation caused by colour-charge screening and a contribution of recombination (usually denoted (re)generation) from deconfined c and \bar{c} quarks in the QGP [450, 1050, 1051]. The (re)generation contribution the charmonium yield is expected to be proportional to the square of the rapidity density of $c\bar{c}$ pairs in the QGP. Therefore, it is predicted to be much larger at HE-LHC than LHC energies, as a consequence of the larger hard-scattering production cross section of $c\bar{c}$ pairs and the possible sizeable thermal production. This could lead to the observation of an enhancement of J/ψ production with respect to binary scaling of the yield in pp collisions, i.e. $R_{AA} > 1$, which would be striking evidence of $c\bar{c}$ recombination from a deconfined QGP.



Fig. 108: Time-evolution of the $c\bar{c}$ yield (per unit of rapidity at midrapidity) for central Pb–Pb collisions at $\sqrt{s_{\text{NN}}} = 10.6$ TeV, obtained as described in Ref. [1048].

The measurement of Y production would be particularly interesting at the high energies and temperatures reached at the HE-LHC. The LHC data are consistent with a scenario in which the excited states 2S and 3S are partially or totally suppressed by colour screening, while the 1S, which is the most tightly bound state, has no or little direct melting. Its suppression by about 60% could be explained by the lack of feed-down from the (melted) higher states and the effect of nuclear PDF suppression (see e.g. Ref. [244] for a recent review). At HE-LHC energies, on the one hand, the temperature could be large enough to determine a full melting even of the tightly-bound 1S state, on the other hand the large abundance of bb pairs in the QGP could induce substantial Y (re)generation. The role of the two effects —degree of survival of initial bottomonia and contribution of (re)generation— could be separated by means of precise measurements of the bb cross section and of the B meson and Y R_{AA} and elliptic flow v_2 (the regenerated Y states could exhibit a v_2 such that $0 < v_2^Y < v_2^B$).

13.4 Nuclear PDF measurements and search for parton saturation

Parton saturation [776, 777] is based on the idea that standard linear parton branching leads, at small values of momentum fraction x, to parton densities so high that non-linear dynamics (like gluon recombination) becomes important and parton densities are tamed to grow from power-like to logarithmically. Non-linear effects are expected to become important when the density of gluons per unit transverse area exceeds a certain limit, the *saturation density*.

In the framework of QCD collinear factorization, Parton Distribution Functions of nucleons inside nuclei (nuclear PDFs) can be obtained in standard global fit analysis with usual linear evolution equations. The differences with respect to free nuclon PDFs are parametrized in a nuclear modification factor $R_i^A(x,Q^2)$ with i = g, $q_{valence}$, q_{sea} (see e.g. Ref. [1052]). Collinear factorization is expected to break down when the gluon phase-space becomes saturated. The onset of saturation is usually discussed in terms of the saturation momentum Q_S^2 , defined as the scale at which the transverse area of the nucleus is completely saturated and gluons start to overlap. It can be shown that $Q_S^2 \sim A^{1/3} (\sqrt{s_{NN}})^{\lambda} e^{\lambda y}$, with $\lambda \approx 0.3$ [400]. Therefore, the regime of high gluon density is best accessed at a high- $\sqrt{s_{NN}}$ hadron collider with measurements at low p_T and forward rapidity, which probe small x and small Q^2 . In order to firmly establish the existence of this new high-energy regime of QCD and clarify the validity of the different approaches to factorisation and evolution, new kinematic regions must be explored using higher collision energies in order to have a large lever arm in Q^2 in a region that, while perturbative, lies inside the saturation domain. The HE-LHC extends the small-x coverage by a factor of two with respect to the LHC, as shown in Fig. 109.



Fig. 109: Regions of the $x-Q^2$ plane covered by the data used for present nuclear PDF fits (nuclear DIS and Drell-Yan, p–Pb and d–Au at the LHC and RHIC), shown in both panels, by p–Pb collisions at the LHC and the HE-LHC (left) and by e–A collisions at the EIC, LHeC and HE-LHC (right). In the left panel, the thin lines correspond to different rapidities in the laboratory frame $y_{lab} = 0.1, 2, 3, 4, 5, 6$ from right to left, with the left edge defined by $y_{lab} = 6.6$. Values of $Q_S^2(x)$ for Pb are shown for illustration in both panels.

There is a strong complementarity between the physics programmes at hadron colliders and at the proposed electron–hadron colliders (Electron-Ion Collider in the USA [794], Large Hadron Electron Collider LHeC [793]). With kinematic reach at the TeV scale in the c.m.s. (Fig. 109, left), the electron–nucleus option at the HE-LHC would be well-positioned to reach conclusive evidence for the existence of a new non-linear regime of QCD. It would be clearly complementary with the p–Pb case, providing a precise knowledge on the partonic structure of nucleons and nuclei and on the small-*x* dynamics. A specific discussion can be found in the electron–nucleus part of the FCC Conceptual Design Report.

13.5 Photon-photon collisions

Photon-photon collisions in UPCs of proton [1053] and lead (Pb) beams [577] have been experimentally observed at the LHC [815, 1054–1056]. The future prospects at the LHC are extensively discussed in Sect. 11.2. Although the γ spectrum is harder for smaller charges –which favours proton over nuclear beams in the production of heavy diphoton systems– each photon flux scales with the squared charge of the hadron, Z^2 , and thus $\gamma\gamma$ luminosities are extremely enhanced for ion beams ($Z^4 = 5 \cdot 10^7$ in the case of Pb–Pb). The Pb beam Lorentz factor at HE-LHC ($\gamma = 5\,650$) determines the "maximum" quasireal photon energy $\omega_{\text{max}} = \gamma/\text{R}_{\text{Pb}} \approx 160$ GeV, leading to photon–photon collisions up to centre-of-mass energies of $\sqrt{s_{\gamma\gamma}} \approx 320$ GeV, twice larger than those reachable at the LHC.

The very rare elastic scattering of two photons in vacuum $\gamma\gamma \rightarrow \gamma\gamma$ was recently observed for the first time in UPCs at the LHC [805, 806]. At the HE-LHC, due to the higher diphoton masses reached, this process may be sensitive to physics beyond the SM through new heavy charged particles contributing to the virtual loop such as, e.g., from SUSY particles [1057]. Light-by-light (LbyL) scattering has also been proposed as a tool to search for monopoles [1058], axions [1059], unparticles [1060], low-scale gravity effects [1061], and non-commutative interactions [1062].
Acknowledgements

We would like to thank the LHC experimental Collaborations and the WLCG for their essential support. We are especially grateful for the efforts by the computing, generator and validation groups who were instrumental for the creation of large simulation samples. We thank the detector upgrade groups as well as the physics and performance groups for their input. Not least, we thank the many colleagues who have provided useful comments on the analyses.

Specific authors acknowledge the following forms of support:

- Nestor Armesto, Elena G. Ferreiro and Carlos Salgado acknowledge the support of the Ministerio de Ciencia e Innovación of Spain under projects FPA2014-58293-C2-1-P, FPA2017-83814-P and Unidad de Excelencia María de Maetzu under project MDM-2016-0692, by Xunta de Galicia (Consellería de Educación) within the Strategic Unit AGRUP2015/11, and by FEDER.
- Raphaelle Bailhache acknowledges the support of the German Federal Ministry of Education and Research (BMBF).
- Zvi Citron acknowledges the support of the ISRAEL SCIENCE FOUNDATION (grant No. 91/6491).
- Torsten Dahms acknowledges the support of the DFG cluster of excellence "Origin and Structure of the Universe".
- Stefan Floerchinger acknowledges the support of the DFG Collaborative Research Centre SFB 1225 (ISOQUANT).
- Iwona Grabowska-Bold acknowledges the support of the Polish National Science Centre grant DEC-2016/23/B/ST2/01409, of the AGH UST statutory tasks No. 11.11.220.01/4 within subsidy of the Ministry of Science and Higher Education, and of the PL-Grid Infrastructure.
- Ilkka Helenius acknowledges the support of the Carl Zeiss Foundation and the Academy of Finland, Project 308301.
- Jiangyong Jia, Peng Huo and Mingliang Zhou acknowledges the support of the U.S. National Science Foundation under grant number PHY-1613294.
- Spencer Klein acknowledges the support of the U.S. DOE under contract number DE-AC02-05-CH11231.
- Filip Krizek acknowledges the support of the Ministry of Education of the Czech Republic (grant InterExcellence LT17018)
- Mariola Klusek-Gawenda acknowledges the support of the Polish National Science Center Grant No. DEC-2014/15/B/ST2/02528.
- Nirbhay Kumar Behera acknowledges the support of National Research Foundation of Korea (NRF), Basic Science Research Program, funded by the Ministry of Education, Science and Technology (Grant No. NRF-2014R1A1A1008246).
- Constantin Loizides acknowledges the support of the U.S. Department of Energy, Office of Science, Office of Nuclear Physics, under contract number DE-AC05-00OR22725.
- Laure Massacrier acknowledges the support of CNRS under the grant RFBR/CNRS 18-52-15007.
- Christoph Mayer acknowledges the support of the Polish Ministry of Science and Higher Education and from the Polish National Science Center.
- Soumya Mohapatra acknowledges the support of the Division of Nuclear Physics of the US Department of Energy under grant DE-FG02-86ER40281.
- Petja Paakkinen acknowledges the support of the Magnus Ehrnrooth Foundation.
- Hannu Paukkunen acknowledges the support by his Academy-of-Finland project 308301.
- Dmitri Peresunko acknowledges the support of the Russian Science Foundation grant 17-72-20234.
- Mateusz Płoskoń acknowledges the support of the U.S. Department of Energy, Office of Science,

Office of Nuclear Physics, under contract DE-AC02-05CH11231.

- Ralf Rapp acknowledges the support of the U.S. National Science Foundation under grant number PHY-1614484, and in part by the ExtreMe Matter Institute EMMI at the GSI Helmholtzzentrum für Schwerionenforschung (Darmstadt,Germany).
- Kristof Redlich acknowledges the support of the Polish National Science Center NCN under Maestro grant DEC-2013/10/A/ST2/00106.
- Mark Strikman acknowledges the support of the U.S. Department of Energy, Office of Science, Office of Nuclear Physics, under Award No. DE-FG02-93ER40771.
- Adam Trzupek acknowledges the support of the National Science Centre, Poland under the grant no 2016/23/B/ST2/00702.
- Michael Weber acknowledges the support of the Austrian Academy of Sciences and the Nationalstiftung für Forschung, Technologie und Entwicklung, Austria.
- Michael Winn acknowledges the support of the European Research Council (ERC) through the project EXPLORINGMATTER, funded by the ERC through a ERC-Consolidator-Grant.

References

- [1] O. S. Brüning et al. CERN-2004-003-V-1, CERN-2004-003 (2004). https://cds.cern.ch/record/782076.
- [2] J.M. Jowett in Proc. 9th International Particle Accelerator Conference (IPAC'18), no. 9 in International Particle Accelerator Conference, pp. 584–589. JACoW Publishing, Geneva, Switzerland, June, 2018. https://doi.org/10.18429/JACoW-IPAC2018-TUXGBD2.
- [3] M. Schaumann et al. in Proc. 9th International Particle Accelerator Conference (IPAC'18), Vancouver, BC, Canada, April 29-May 4, 2018, no. 9 in International Particle Accelerator Conference, pp. 180–183. JACoW Publishing, Geneva, Switzerland, June, 2018. http://jacow.org/ipac2018/papers/mopmf039.pdf. https://doi.org/10.18429/JACoW-IPAC2018-MOPMF039.
- [4] C. A. Salgado et al. J. Phys. G39 (2012) 015010, arXiv:1105.3919 [hep-ph].
- [5] J. Jowett CERN EDMS Note 134202 (2015).
- [6] J. M. Jowett et al. LHC Performance Workshop (Chamonix 2017). https://indico.cern.ch/event/580313/.
- [7] S. R. Klein Nucl.Instrum.Meth. A459 (2001) 51-57, arXiv:physics/0005032 [physics].
- [8] J. M. Jowett in LHC performance. Proceedings, 12th Workshop, Chamonix, France, March 3-8, 2003, pp. 84-90.
 2003. http://ab-div.web.cern.ch/ab-div/Conferences/Chamonix/chamx2003/PAPERS/2_6_JJ.pdf.
- [9] R. Bruce, D. Bocian, S. Gilardoni, and J. M. JowettPhys. Rev. ST Accel. Beams 12 (Jul, 2009) 071002. http://link.aps.org/doi/10.1103/PhysRevSTAB.12.071002.
- [10] C. Bahamonde Castro et al. (2016) TUPMW006. http://cds.cern.ch/record/2207361.
- [11] R. W. Assmann in LHC Project Workshop 14th Chamonix Workshop Chamonix, Switzerland, January 17-21, 2005, p. 54. 2005. http://ab-div.web.cern.ch/ab-div/Conferences/Chamonix/chamx2005/PAPERS/8_02.pdf.
- [12] G. Robert-Démolaize and R. W. Assmann, *Design and Performance Optimization of the LHC Collimation System*, 2006. https://cds.cern.ch/record/1004869. Presented on 20 Nov 2006.
- [13] R. Assmann et al. Conf. Proc. C060626 (2006) 986–988. [,986(2006)].
- [14] C. Bracco, R. Assmann, and L. Rivkin, *Commissioning Scenarios and Tests for the LHC Collimation System*, 2009. https://cds.cern.ch/record/1174254. Presented on 29 Jan 2009.
- [15] R. Bruce, R. W. Assmann, V. Boccone, et al.Phys. Rev. ST Accel. Beams 17 (Aug, 2014) 081004. http://link.aps.org/doi/10.1103/PhysRevSTAB.17.081004.
- [16] G. Valentino, R. Bruce, M. Fiascaris, et al. CERN-ACC-2015-376, Proceedings of the 6th Evian Workshop, Evian, France (2015). http://cds.cern.ch/record/2156966?ln=en.
- [17] D. Mirarchi, R.Bruce, M. Fiascaris, et al. CERN-ACC-2017-094, Proceedings of the 7th Evian Workshop, Evian, France (2016). http://cds.cern.ch/record/2289585?ln=en.
- [18] R. Bruce, C. Bracco, R. D. Maria, et al.Nucl. Instr. Meth. Phys. Res. A 848 (Jan, 2017) 19 30. http://www.sciencedirect.com/science/article/pii/S0168900216313092.
- [19] P. D. Hermes, J. P. Wessels, and R. Bruce, *Heavy-Ion Collimation at the Large Hadron Collider: Simulations and Measurements*, Sep, 2016. https://cds.cern.ch/record/2241364. Presented 19 Dec 2016.
- [20] P. Hermes, R. Bruce, J. Jowett, et al.Nucl. Instr. Meth. Phys. Res. A 819 (Feb, 2016) 73 83. https://www.sciencedirect.com/science/article/pii/S0168900216002175?via%3Dihub.
- [21] H.-H. Braun, R. W. Assmann, A. Ferrari, et al. in 9th European Particle Accelerator Conference (EPAC 2004) Lucerne, Switzerland, July 5-9, 2004. 2004. http://accelconf.web.cern.ch/AccelConf/e04/PAPERS/MOPLT010.PDF.
- [22] P. D. Hermes, B. Auchmann, R. Bruce, et al. CERN-ACC-NOTE-2016-0031, Mar, 2016. http://cds.cern.ch/record/2136828.
- [23] R. Bruce, A. Marsili, and S. Redaelli Proceedings of the International Particle Accelerator Conference 2014, Dresden, Germany (2014) 170. http://accelconf.web.cern.ch/AccelConf/IPAC2014/papers/mopro042.pdf.
- [24] A. Lechner, B. Auchmann, R. Bruce, et al. Proceedings of the International Particle Accelerator Conference 2014, Dresden, Germany (2014) 112.
 - http://accelconf.web.cern.ch/AccelConf/IPAC2014/papers/mopro021.pdf.
- [25] Apollinari G. and Béjar Alonso I. and Brüning O. and Fessia P. and Lamont M. and Rossi L. and Tavian L., *High-Luminosity Large Hadron Collider (HL-LHC): Technical Design Report V. 0.1.* CERN Yellow Reports: Monographs. CERN, Geneva, 2017. https://cds.cern.ch/record/2284929.
- [26] D. Mirarchi, S. Redaelli, W. Scandale, and G. Hall, Crystal Collimation for LHC, Aug, 2015. http://cds.cern.ch/record/2036210.
- [27] D. Mirarchi, G. Hall, S. Redaelli, and W. ScandaleThe European Physical Journal C 77 (Jun, 2017) 424. https://doi.org/10.1140/epjc/s10052-017-4985-4.
- [28] W. Scandale, G. Arduini, M. Butcher, et al. Physics Letters B 758 (2016) 129 133. http://www.sciencedirect.com/science/article/pii/S0370269316301514.

- [29] J. Jowett, R. Alemany-Fernández, G. Baud, et al. TUPVA014 (2017) no. CERN-ACC-2017-143, . https://cds.cern.ch/record/2289686.
- [30] D. Manglunki et al. in Proceedings, 7th International Particle Accelerator Conference (IPAC 2016): Busan, Korea, May 8-13, 2016, p. TUPMR027. 2016.
- [31] J.M. Jowett in FCC Week, Amsterdam. 2018. https://indico.cern.ch/event/656491/contributions/2923429/.
- [32] A. Andronic, P. Braun-Munzinger, K. Redlich, and J. Stachel arXiv:1710.09425 [nucl-th].
- [33] ALICE Collaboration, S. Acharya et al. Nucl. Phys. A971 (2018) 1–20, arXiv:1710.07531 [nucl-ex].
- [34] ALICE Collaboration, B. Abelev et al. Phys. Rev. C88 (2013) 044910, arXiv:1303.0737 [hep-ex].
- [35] ALICE Collaboration, S. Acharya et al. arXiv:1805.04390 [nucl-ex].
- [36] A. Andronic, P. Braun-Munzinger, J. Stachel, and H. Stöcker Phys. Lett. B697 (2011) 203-207, arXiv:1010.2995 [nucl-th].
- [37] HotQCD Collaboration, A. Bazavov et al. Phys. Rev. D 90 (2014) 094503, arXiv:1407.6387 [hep-lat].
- [38] R. Bellwied, S. Borsanyi, Z. Fodor, S. D. Katz, and C. Ratti Phys. Rev. Lett. 111 (2013) 202302, arXiv:1305.6297 [hep-lat].
- [39] S. T. Butler and C. A. Pearson Phys. Rev. 129 (1963) 836–842.
- [40] J. I. Kapusta Phys. Rev. C21 (1980) 1301–1310.
- [41] J. C. Bergstrom Nucl. Phys. A327 (1979) 458-476.
- [42] H. Sato and K. Yazaki Phys. Lett. B98 (1981) 153-157.
- [43] J. L. Nagle, B. S. Kumar, D. Kusnezov, H. Sorge, and R. Mattiello Phys. Rev. C53 (1996) 367–376.
- [44] R. Scheibl and U. W. Heinz Phys. Rev. C59 (1999) 1585-1602, arXiv:nucl-th/9809092 [nucl-th].
- [45] U. A. Wiedemann and U. W. Heinz Phys. Rept. 319 (1999) 145-230, arXiv:nucl-th/9901094 [nucl-th].
- [46] F. Bellini and A. P. Kalweit arXiv:1807.05894 [hep-ph].
- [47] ExHIC Collaboration, S. Cho et al. Prog. Part. Nucl. Phys. 95 (2017) 279-322, arXiv:1702.00486 [nucl-th].
- [48] Z. Zhang and C. M. Ko Phys. Lett. B780 (2018) 191–195.
- [49] S. Bazak and S. Mrowczynski arXiv:1802.08212 [nucl-th].
- [50] W. Zhao, L. Zhu, H. Zheng, C. M. Ko, and H. Song arXiv: 1807.02813 [nucl-th].
- [51] S. Wheaton and J. Cleymans Comput. Phys. Commun. 180 (2009) 84–106, arXiv:hep-ph/0407174 [hep-ph].
- [52] M. Petran, J. Letessier, J. Rafelski, and G. Torrieri Comput. Phys. Commun. 185 (2014) 2056–2079, arXiv:1310.5108 [hep-ph].
- [53] ALICE Collaboration. https://cds.cern.ch/record/2272148.
- [54] ALICE Collaboration, S. Trogolo PoS EPS-HEP2017 (2017) 200.
- [55] ALICE Collaboration, B. Abelev et al. J. Phys. G41 (2014) 087002.
- [56] ALICE Collaboration CERN-LHCC-2013-020. ALICE-TDR-016, Oct, 2013. http://cds.cern.ch/record/1622286.
- [57] ALICE Collaboration, P. Antonioli, A. Kluge, and W. Riegler CERN-LHCC-2013-019. ALICE-TDR-015, Sep, 2013. https://cds.cern.ch/record/1603472.
- [58] P. Buncic, M. Krzewicki, and P. Vande Vyvre CERN-LHCC-2015-006. ALICE-TDR-019, Apr, 2015. https://cds.cern.ch/record/2011297.
- [59] ALICE Collaboration, S. Acharya et al. Eur. Phys. J. C77 (2017) no. 10, 658, arXiv:1707.07304 [nucl-ex].
- [60] ALICE Collaboration, J. Adam et al. Nature Phys. 11 (2015) no. 10, 811–814, arXiv:1508.03986 [nucl-ex].
- [61] G. A. Miller Nucl. Phys. A518 (1990) 345-357.
- [62] G. Audi, A. H. Wapstra, and C. Thibault Nucl. Phys. A729 (2002) 337-676.
- [63] H. Kamada, J. Golak, K. Miyagawa, H. Witala, and W. Glockle Phys. Rev. C57 (1998) 1595.
- [64] D. H. Davis Nucl. Phys. A754 (2005) 3 13.
- [65] Particle Data Group Collaboration, M. Tanabashi et al. Phys. Rev. D 98 (2018) 030001.
- [66] M. Agnello et al. Nucl. Phys. A954 (2016) 176.
- [67] ALICE Collaboration, J. Adam et al. Phys. Lett. B754 (2016) 360.
- [68] STAR Collaboration, L. Adamczyk et al. Phys. Rev. C97 (2018) 054909.
- [69] R. Dalitz and M. Rayet Il Nuovo Cim. A46 (1966) 786.
- [70] J. Congleton J. Phys. G 18 (1992) 339.
- [71] F.-C. T. Garcilazo H., Valcarce A. Phys. Rev. C 76 (2007) 034001.
- [72] T. Nagae et al. Phys. Rev. Lett. 80 (1998) 1605–1609.
- [73] ALICE Collaboration, A. Borissov EPJ Web Conf. 97 (2015) 00005.

- [74] ALICE Collaboration, B. B. Abelev et al. Int. J. Mod. Phys. A29 (2014) 1430044, arXiv:1402.4476 [nucl-ex].
- [75] ALICE Collaboration, N. Jacazio Nucl. Phys. A967 (2017) 421–424, arXiv:1704.06030 [nucl-ex].
- [76] ALICE Collaboration, S. Acharya et al. Submitted to: Phys. Lett. (2018), arXiv:1809.10922 [nucl-ex].
- [77] Particle Data Group Collaboration, C. Patrignani et al. Chin. Phys. C40 (2016 and 2017 update) 100001.
- [78] J. He Phys. Rev. D95 (2017) no. 7, 074031, arXiv:1701.03738 [hep-ph].
- [79] F. Donato, N. Fornengo, and P. Salati Phys. Rev. D62 (2000) 043003, arXiv:hep-ph/9904481 [hep-ph].
- [80] H. Baer and S. Profumo JCAP 0512 (2005) 008, arXiv:astro-ph/0510722 [astro-ph].
- [81] F. Donato, N. Fornengo, and D. Maurin Phys. Rev. D78 (2008) 043506, arXiv:0803.2640 [hep-ph].
- [82] C. B. Brauninger and M. Cirelli Phys. Lett. B678 (2009) 20-31, arXiv:0904.1165 [hep-ph].
- [83] M. Kadastik, M. Raidal, and A. Strumia Phys. Lett. B683 (2010) 248–254, arXiv:0908.1578 [hep-ph].
- [84] Y. Cui, J. D. Mason, and L. Randall JHEP 11 (2010) 017, arXiv:1006.0983 [hep-ph].
- [85] L. A. Dal and M. Kachelriess Phys. Rev. D86 (2012) 103536, arXiv:1207.4560 [hep-ph].
- [86] A. Ibarra and S. Wild JCAP 1302 (2013) 021, arXiv:1209.5539 [hep-ph].
- [87] N. Fornengo, L. Maccione, and A. Vittino JCAP 1309 (2013) 031, arXiv:1306.4171 [hep-ph].
- [88] E. Carlson, A. Coogan, T. Linden, et al. Phys. Rev. D89 (2014) no. 7, 076005, arXiv:1401.2461 [hep-ph].
- [89] T. Aramaki et al. Phys. Rept. 618 (2016) 1-37, arXiv:1505.07785 [hep-ph].
- [90] M. Korsmeier, F. Donato, and N. Fornengo Phys. Rev. D97 (2018) no. 10, 103011, arXiv:1711.08465 [astro-ph.HE].
- [91] AMS Collaboration, F. Giovacchini and V. Choutko in Proceedings, 30th International Cosmic Ray Conference (ICRC 2007): Merida, Yucatan, Mexico, July 3-11, 2007, vol. 4, pp. 765-768. 2007. http://indico.nucleares.unam.mx/contributionDisplay.py?contribId=1112&confId=4.
- [92] AMS Collaboration, A. Kounine in Proceedings, 32nd International Cosmic Ray Conference (ICRC 2011): Beijing, China, August 11-18, 2011, vol. c, p. 5. 2011. https://inspirehep.net/record/1352202/files/vc_102.pdf.
- [93] P. von Doetinchem, T. Aramaki, S. Boggs, et al. PoS ICRC2015 (2016) 1219, arXiv: 1507.02717
 [astro-ph.IM]. [34,1218(2015)].
- [94] T. Aramaki, C. J. Hailey, S. E. Boggs, et al. Astropart. Phys. 74 (2016) 6-13, arXiv:1506.02513 [astro-ph.HE].
- [95] K. Abe et al. Phys. Rev. Lett. 108 (2012) 051102, arXiv:1107.6000 [astro-ph.HE].
- [96] Berezinsky, V. S. and Bulanov, S. V. and Dogiel, V. A. and Ptuskin, V. S., Astrophysics of cosmic rays. 1990.
- [97] B. Katz, K. Blum, J. Morag, and E. Waxman Mon.Not.Roy.Astron.Soc. 405 (2010) 1458, arXiv:0907.1686 [astro-ph.HE].
- [98] K. Blum, K. C. Y. Ng, R. Sato, and M. Takimoto Phys. Rev. D96 (2017) no. 10, 103021, arXiv:1704.05431 [astro-ph.HE].
- [99] AMS Collaboration, M. Aguilar et al. Phys. Rev. Lett. 117 (2016) no. 9, .
- [100] AMS Collaboration, M. Aguilar et al. Phys. Rev. Lett. **114** (2015) 171103.
- [101] R. P. Duperray, K. V. Protasov, and A. Yu. Voronin Eur. Phys. J. A16 (2003) 27-34, arXiv:nucl-th/0209078 [nucl-th].
- [102] R. P. Duperray, K. V. Protasov, L. Derome, and M. Buenerd Eur. Phys. J. A18 (2003) 597–604, arXiv:nucl-th/0301103 [nucl-th].
- [103] ALICE Collaboration, S. Acharya et al. Phys. Rev. C97 (2018) no. 2, 024615, arXiv:1709.08522 [nucl-ex].
- [104] F. Donato, M. Korsmeier, and M. Di Mauro Phys. Rev. D96 (2017) no. 4, 043007, arXiv:1704.03663 [astro-ph.HE].
- [105] A. Gal, E. V. Hungerford, and D. J. Millener Rev. Mod. Phys. 88 (2016) 035004, arXiv:1605.00557 [nucl-th].
- [106] J. M. Lattimer Annu. Rev. Nucl. Part. Sci. 62 (2012) 485, arXiv: 1305.3510 [nucl-th].
- [107] W. Weis Hyperfine Interact. 233 (2015) no. 1-3, 131, arXiv:1412.7838 [nucl-th].
- [108] P. B. Demorest, T. Pennucci, S. M. Ransom, M. S. E. Roberts, and J. W. T. Hessels Nature 467 (2010) 1081.
- [109] E. Fonseca, T. T. Pennucci, J. A. Ellis, et al. Astrophys. J. 832 (2016) no. 2, 167, arXiv:1603.00545 [astro-ph.HE].
- [110] D. Lonardoni, A. Lovato, S. Gandolfi, and F. Pederiva Phys. Rev. Lett. 114 (2015) no. 9, 092301, arXiv:1407.4448 [nucl-th].
- [111] H. Togashi, E. Hiyama, Y. Yamamoto, and M. Takano Phys. Rev. C 93 (2016) no. 3, 035808, arXiv:1602.08106 [nucl-th].
- [112] ALICE Collaboration, S. Acharya et al. arXiv:1805.12455 [nucl-ex].
- [113] K. Sasaki et al. PoS LATTICE2016 (2017) 116, arXiv:1702.06241 [hep-lat].
- [114] T. Hatsuda, K. Morita, A. Ohnishi, and K. Sasaki Nucl. Phys. A967 (2017) 856-859, arXiv: 1704.05225

[nucl-th].

- [115] J. Haidenbauer Nucl. Phys. A981 (2019) 1-16, arXiv:1808.05049 [hep-ph].
- [116] T. A. Rijken and Y. Yamamoto arXiv:nucl-th/0608074 [nucl-th].
- [117] R. D. Pisarski and F. Wilczek Phys. Rev. D29 (1984) 338-341.
- [118] S. Ejiri, F. Karsch, E. Laermann, et al. Phys. Rev. D80 (2009) 094505, arXiv:0909.5122 [hep-lat].
- [119] H. T. Ding, P. Hegde, F. Karsch, et al. in 27th International Conference on Ultrarelativistic Nucleus-Nucleus Collisions (Quark Matter 2018) Venice, Italy, May 14-19, 2018. 2018. arXiv:1807.05727 [hep-lat].
- [120] Y. Aoki, G. Endrodi, Z. Fodor, S. D. Katz, and K. K. Szabo Nature 443 (2006) 675-678, arXiv:hep-lat/0611014 [hep-lat].
- [121] Y. Aoki, S. Borsanyi, S. Durr, et al. JHEP 06 (2009) 088, arXiv:0903.4155 [hep-lat].
- [122] Wuppertal-Budapest Collaboration, S. Borsanyi, Z. Fodor, C. Hoelbling, et al. JHEP 09 (2010) 073, arXiv:1005.3508 [hep-lat].
- [123] A. Bazavov et al. Phys. Rev. D 85 (2012) 054503, arXiv:1111.1710 [hep-lat].
- [124] T. Bhattacharya et al. Phys. Rev. Lett. 113 (2014) no. 8, 082001, arXiv:1402.5175 [hep-lat].
- [125] H. T. Ding, A. Bazavov, F. Karsch, et al. PoS LATTICE2013 (2014) 157, arXiv:1312.0119 [hep-lat].
- [126] S. Ejiri, F. Karsch, and K. Redlich Phys. Lett. B633 (2006) 275–282, arXiv:hep-ph/0509051 [hep-ph].
- [127] A. Bazavov et al. Phys. Rev. D95 (2017) no. 5, 054504, arXiv:1701.04325 [hep-lat].
- [128] B. Friman, F. Karsch, K. Redlich, and V. Skokov Eur. Phys. J. C71 (2011) 1694, arXiv:1103.3511 [hep-ph].
- [129] HotQCD Collaboration, A. Bazavov et al. Phys. Rev. D86 (2012) 034509, arXiv:1203.0784 [hep-lat].
- [130] R. Bellwied, S. Borsanyi, Z. Fodor, et al. Phys. Rev. D92 (2015) no. 11, 114505, arXiv:1507.04627 [hep-lat].
- [131] F. Karsch and K. Redlich Phys. Lett. B695 (2011) 136–142, arXiv:1007.2581 [hep-ph].
- [132] V. Skokov, B. Friman, and K. Redlich Phys. Rev. C88 (2013) 034911, arXiv:1205.4756 [hep-ph].
- [133] F. Karsch Central Eur. J. Phys. 10 (2012) 1234–1237, arXiv:1202.4173 [hep-lat].
- [134] F. Karsch Acta Phys. Polon. Supp. 10 (2017) 615, arXiv:1703.06702 [hep-lat].
- [135] S. Borsanyi, Z. Fodor, S. D. Katz, et al. Phys. Rev. Lett. 111 (2013) 062005, arXiv:1305.5161 [hep-lat].
- [136] S. Borsanyi, Z. Fodor, S. D. Katz, et al. Phys. Rev. Lett. 113 (2014) 052301, arXiv:1403.4576 [hep-lat].
- [137] G. A. Almasi, B. Friman, and K. Redlich Phys. Rev. D96 (2017) no. 1, 014027, arXiv:1703.05947 [hep-ph].
- [138] P. Parotto, M. Bluhm, D. Mroczek, et al. arXiv:1805.05249 [hep-ph].
- [139] S. Borsanyi, Z. Fodor, J. N. Guenther, et al. arXiv:1805.04445 [hep-lat].
- [140] C. R. Allton, M. Doring, S. Ejiri, et al. Phys. Rev. D71 (2005) 054508, arXiv:hep-lat/0501030 [hep-lat].
- [141] P. Braun-Munzinger, B. Friman, F. Karsch, K. Redlich, and V. Skokov Nucl. Phys. A880 (2012) 48–64, arXiv:1111.5063 [hep-ph].
- [142] X. Luo and N. Xu Nucl. Sci. Tech. 28 (2017) no. 8, 112, arXiv:1701.02105 [nucl-ex].
- [143] P. Braun-Munzinger, B. Friman, F. Karsch, K. Redlich, and V. Skokov Phys. Rev. C84 (2011) 064911, arXiv:1107.4267 [hep-ph].
- [144] P. Braun-Munzinger, A. Rustamov, and J. Stachel arXiv:1807.08927 [nucl-th].
- [145] P. Braun-Munzinger, A. Rustamov, and J. Stachel Nucl. Phys. A960 (2017) 114–130, arXiv:1612.00702 [nucl-th].
- [146] F. Karsch, K. Morita, and K. Redlich Phys. Rev. C93 (2016) no. 3, 034907, arXiv:1508.02614 [hep-ph].
- [147] P. Alba, R. Bellwied, M. Bluhm, et al. Phys. Rev. C92 (2015) no. 6, 064910, arXiv:1504.03262 [hep-ph].
- [148] S. Floerchinger and M. Martinez Phys. Rev. C92 (2015) no. 6, 064906, arXiv:1507.05569 [nucl-th].
- [149] STAR Collaboration, L. Adamczyk et al. Phys. Rev. Lett. 112 (2014) 032302, arXiv:1309.5681 [nucl-ex].
- [150] ALICE Collaboration, A. Rustamov Nucl. Phys. A967 (2017) 453-456, arXiv:1704.05329 [nucl-ex].
- [151] A. Bzdak and V. Koch arXiv:1811.04456 [nucl-th].
- [152] N. K. Behera arXiv:1706.06558 [nucl-ex].
- [153] ALICE Collaboration, N. K. Behera in 27th International Conference on Ultrarelativistic Nucleus-Nucleus Collisions (Quark Matter 2018) Venice, Italy, May 14-19, 2018. 2018. arXiv:1807.06780 [hep-ex].
- [154] P. Steinbrecher arXiv:1807.05607 [hep-lat].
- [155] O. Kaczmarek, F. Karsch, E. Laermann, et al. Phys. Rev. D83 (2011) 014504, arXiv:1011.3130 [hep-lat].
- [156] G. Endrodi, Z. Fodor, S. D. Katz, and K. K. Szabo JHEP 04 (2011) 001, arXiv:1102.1356 [hep-lat].
- [157] A. Bazavov et al. Phys. Rev. D93 (2016) no. 1, 014512, arXiv:1509.05786 [hep-lat].
- [158] C. Bonati, M. D'Elia, F. Negro, F. Sanfilippo, and K. Zambello arXiv:1805.02960 [hep-lat].
- [159] P. Cea, L. Cosmai, and A. Papa Phys. Rev. D89 (2014) no. 7, 074512, arXiv:1403.0821 [hep-lat].

- [160] C. Bonati, P. de Forcrand, M. D'Elia, O. Philipsen, and F. Sanfilippo Phys. Rev. D90 (2014) no. 7, 074030, arXiv:1408.5086 [hep-lat].
- [161] C. Bonati, M. D'Elia, M. Mariti, et al. Phys. Rev. D92 (2015) no. 5, 054503, arXiv:1507.03571 [hep-lat].
- [162] R. Bellwied, S. Borsanyi, Z. Fodor, et al. Phys. Lett. **B751** (2015) 559–564, arXiv:1507.07510 [hep-lat].
- [163] P. Cea, L. Cosmai, and A. Papa Phys. Rev. D93 (2016) no. 1, 014507, arXiv:1508.07599 [hep-lat].
- [164] P. Huovinen and P. Petreczky Nucl. Phys. A837 (2010) 26–53, arXiv:0912.2541 [hep-ph].
- [165] S. Borsanyi, Z. Fodor, C. Hoelbling, et al. Phys. Lett. B730 (2014) 99-104, arXiv:1309.5258 [hep-lat].
- [166] J. S. Moreland and R. A. Soltz Phys. Rev. C93 (2016) no. 4, 044913, arXiv:1512.02189 [nucl-th].
- [167] A. Monnai and J.-Y. Ollitrault Phys. Rev. C96 (2017) no. 4, 044902, arXiv:1707.08466 [nucl-th].
- [168] J. S. Moreland, J. E. Bernhard, and S. A. Bass arXiv:1808.02106 [nucl-th].
- [169] P. F. Kolb and U. W. Heinz arXiv:nucl-th/0305084 [nucl-th].
- [170] P. Huovinen arXiv:nucl-th/0305064 [nucl-th].
- [171] T. Hirano and K. Tsuda Phys. Rev. C66 (2002) 054905, arXiv:nucl-th/0205043 [nucl-th].
- [172] G. Policastro, D. T. Son, and A. O. StarinetsPhys. Rev. Lett. 87 (Aug, 2001) 081601. https://link.aps.org/doi/10.1103/PhysRevLett.87.081601.
- [173] P. K. Kovtun, D. T. Son, and A. O. StarinetsPhys. Rev. Lett. 94 (Mar, 2005) 111601. https://link.aps.org/doi/10.1103/PhysRevLett.94.111601.
- [174] P. B. Arnold, G. D. Moore, and L. G. Yaffe JHEP 05 (2003) 051, arXiv:hep-ph/0302165 [hep-ph].
- [175] J. Ghiglieri, G. D. Moore, and D. Teaney Phys. Rev. Lett. 121 (2018) no. 5, 052302, arXiv:1805.02663 [hep-ph].
- [176] A. Nakamura and S. Sakai Phys. Rev. Lett. 94 (2005) 072305, arXiv:hep-lat/0406009 [hep-lat].
- [177] H. B. Meyer Phys. Rev. D76 (2007) 101701, arXiv:0704.1801 [hep-lat].
- [178] S. Borsányi, Z. Fodor, M. Giordano, et al. Phys. Rev. D98 (2018) no. 1, 014512, arXiv:1802.07718 [hep-lat].
- [179] S. Ryu, J.-F. Paquet, C. Shen, et al. Phys. Rev. C97 (2018) no. 3, 034910, arXiv:1704.04216 [nucl-th].
- [180] R. Horsley and W. Schoenmaker Nucl. Phys. B280 (1987) 716–734.
- [181] P. B. Arnold, C. Dogan, and G. D. Moore Phys. Rev. D74 (2006) 085021, arXiv:hep-ph/0608012 [hep-ph].
- [182] F. Karsch, D. Kharzeev, and K. Tuchin Phys. Lett. B663 (2008) 217-221, arXiv:0711.0914 [hep-ph].
- [183] H. B. Meyer Phys. Rev. Lett. 100 (2008) 162001, arXiv:0710.3717 [hep-lat].
- [184] D. Kharzeev and K. Tuchin JHEP 09 (2008) 093, arXiv:0705.4280 [hep-ph].
- [185] G. S. Denicol, T. Kodama, T. Koide, and P. Mota Phys. Rev. C80 (2009) 064901, arXiv:0903.3595 [hep-ph].
- [186] J. Noronha-Hostler, J. Noronha, and C. Greiner Phys. Rev. Lett. 103 (2009) 172302, arXiv:0811.1571 [nucl-th].
- [187] A. Buchel Phys. Lett. B663 (2008) 286-289, arXiv:0708.3459 [hep-th].
- [188] S. I. Finazzo, R. Rougemont, H. Marrochio, and J. Noronha JHEP 02 (2015) 051, arXiv:1412.2968 [hep-ph].
- [189] C. Gale, S. Jeon, and B. Schenke Int. J. Mod. Phys. A28 (2013) 1340011, arXiv:1301.5893 [nucl-th].
- [190] U. Heinz and R. Snellings Ann. Rev. Nucl. Part. Sci. 63 (2013) 123-151, arXiv:1301.2826 [nucl-th].
- [191] B. Schenke, P. Tribedy, and R. Venugopalan Phys. Rev. Lett. 108 (2012) 252301, arXiv:1202.6646 [nucl-th].
- [192] B. Schenke, P. Tribedy, and R. Venugopalan Phys. Rev. C86 (2012) 034908, arXiv:1206.6805 [hep-ph].
- [193] H. Niemi, K. J. Eskola, and R. Paatelainen Phys. Rev. C93 (2016) no. 2, 024907, arXiv:1505.02677 [hep-ph].
- [194] P. Bozek Phys. Rev. C81 (2010) 034909, arXiv:0911.2397 [nucl-th].
- [195] J.-F. Paquet, C. Shen, G. S. Denicol, et al. Phys. Rev. C93 (2016) no. 4, 044906, arXiv:1509.06738 [hep-ph].
- [196] C. Gale, S. Jeon, B. Schenke, P. Tribedy, and R. Venugopalan Phys. Rev. Lett. 110 (2013) no. 1, 012302, arXiv:1209.6330 [nucl-th].
- [197] R. D. Weller and P. Romatschke Phys. Lett. B774 (2017) 351-356, arXiv:1701.07145 [nucl-th].
- [198] H. Mäntysaari, B. Schenke, C. Shen, and P. Tribedy Phys. Lett. B772 (2017) 681-686, arXiv: 1705.03177 [nucl-th].
- [199] B. Schenke, C. Shen, and P. Tribedy 2018. arXiv:1807.05205 [nucl-th].
- [200] A. Kurkela, A. Mazeliauskas, J.-F. Paquet, S. Schlichting, and D. Teaney arXiv:1805.01604 [hep-ph].
- [201] A. Kurkela, A. Mazeliauskas, J.-F. Paquet, S. Schlichting, and D. Teaney arXiv:1805.00961 [hep-ph].
- [202] A. Kurkela, U. A. Wiedemann, and B. Wu arXiv:1805.04081 [hep-ph].
- [203] CMS Collaboration Phys. Rev. C 92 (2015) 034911, arXiv:1503.01692 [hep-ex].
- [204] CMS Collaboration arXiv:1710.07864 [hep-ex].
- [205] ATLAS Collaboration arXiv:1709.02301 [hep-ex].
- [206] D. Teaney and L. Yan Phys. Rev. C83 (2011) 064904, arXiv:1010.1876 [nucl-th].
- [207] S. Floerchinger and U. A. Wiedemann Phys. Lett. B728 (2014) 407-411, arXiv:1307.3453 [hep-ph].

- [208] S. Floerchinger and U. A. Wiedemann JHEP 08 (2014) 005, arXiv:1405.4393 [hep-ph].
- [209] D. Teaney and L. Yan Phys. Rev. C86 (2012) 044908, arXiv:1206.1905 [nucl-th].
- [210] S. Floerchinger, U. A. Wiedemann, A. Beraudo, et al. Phys. Lett. B735 (2014) 305–310, arXiv:1312.5482 [hep-ph].
- [211] ATLAS Collaboration Phys. Rev. C 90 (2014) 024905, arXiv:1403.0489 [hep-ex].
- [212] ATLAS Collaboration Phys. Rev. C 92 (2015) 034903, arXiv:1504.01289 [hep-ex].
- [213] ALICE Collaboration, J. Adam et al. Phys. Rev. Lett. 117 (2016) 182301, arXiv:1604.07663 [nucl-ex].
- [214] G. Giacalone, J. Noronha-Hostler, and J.-Y. Ollitrault Phys. Rev. C95 (2017) no. 5, 054910, arXiv:1702.01730 [nucl-th].
- [215] G. Giacalone, J. Noronha-Hostler, M. Luzum, and J.-Y. Ollitrault Phys. Rev. C97 (2018) no. 3, 034904, arXiv:1711.08499 [nucl-th]. Predictions for O+O and Ar+Ar obtained by private communications.
- [216] ATLAS Collaboration ATLAS-CONF-2018-011, CERN, Geneva, May, 2018. https://cds.cern.ch/record/2318870.
- [217] ATLAS Collaboration Phys. Rev. C 95 (2017) 064914, arXiv:1606.08170 [hep-ex].
- [218] F. Becattini, V. Chandra, L. Del Zanna, and E. Grossi Annals Phys. 338 (2013) 32-49, arXiv:1303.3431 [nucl-th].
- [219] STAR Collaboration, L. Adamczyk et al. Nature 548 (2017) 62, arXiv:1701.06657 [nucl-ex].
- [220] STAR Collaboration, B. I. Abelev et al. Phys. Rev. C76 (2007) 024915, arXiv:0705.1691 [nucl-ex]. [Erratum: Phys. Rev.C95,no.3,039906(2017)].
- [221] S. Voloshin. https://indico.particle.mephi.ru/event/4/.
- [222] STAR Collaboration, T. Niida 2018. arXiv:1808.10482 [nucl-ex].
- [223] I. Karpenko and F. Becattini arXiv:1710.09726 [nucl-th]. [EPJ Web Conf.171, 17001(2018)].
- [224] T. D. Lee Phys. Rev. D8 (1973) 1226–1239.
- [225] T. D. Lee and G. C. Wick Phys. Rev. D9 (1974) 2291-2316.
- [226] P. D. Morley and I. A. Schmidt Z. Phys. C26 (1985) 627.
- [227] D. Kharzeev, R. D. Pisarski, and M. H. G. Tytgat Phys. Rev. Lett. 81 (1998) 512-515, arXiv:hep-ph/9804221 [hep-ph].
- [228] S.-S. Chern and J. Simons Annals Math. 99 (1974) 48-69.
- [229] W.-T. Deng and X.-G. Huang Phys. Rev. C85 (2012) 044907, arXiv:1201.5108 [nucl-th].
- [230] U. Gursoy, D. Kharzeev, and K. Rajagopal Phys. Rev. C89 (2014) no. 5, 054905, arXiv:1401.3805 [hep-ph].
- [231] D. Kharzeev Phys. Lett. B633 (2006) 260-264, arXiv:hep-ph/0406125 [hep-ph].
- [232] D. Kharzeev and A. Zhitnitsky Nucl. Phys. A797 (2007) 67-79, arXiv:0706.1026 [hep-ph].
- [233] D. E. Kharzeev, L. D. McLerran, and H. J. Warringa Nucl. Phys. A803 (2008) 227–253, arXiv:0711.0950 [hep-ph].
- [234] K. Fukushima, D. E. Kharzeev, and H. J. Warringa Phys. Rev. D78 (2008) 074033, arXiv:0808.3382 [hep-ph].
- [235] ALICE Collaboration, S. Acharya et al. Phys. Lett. B777 (2018) 151-162, arXiv:1709.04723 [nucl-ex].
- [236] S. A. Voloshin Phys. Rev. C70 (2004) 057901, arXiv:hep-ph/0406311 [hep-ph].
- [237] S. Schlichting and S. Pratt Phys. Rev. C83 (2011) 014913, arXiv:1009.4283 [nucl-th].
- [238] S. Pratt, S. Schlichting, and S. Gavin Phys. Rev. C84 (2011) 024909, arXiv:1011.6053 [nucl-th].
- [239] CMS Collaboration, V. Khachatryan et al. Phys. Rev. Lett. 118 (2017) 122301, arXiv:1610.00263 [nucl-ex].
- [240] CMS Collaboration, A. M. Sirunyan et al. Phys. Rev. C97 (2018) no. 4, 044912, arXiv:1708.01602 [nucl-ex].
- [241] J. Schukraft, A. Timmins, and S. A. Voloshin Phys. Lett. B719 (2013) 394–398, arXiv:1208.4563 [nucl-ex].
- [242] ALICE Collaboration, J. Margutti 2017. arXiv:1709.05618 [nucl-ex].
- [243] S. K. Das, S. Plumari, S. Chatterjee, et al. Phys. Lett. B768 (2017) 260-264, arXiv:1608.02231 [nucl-th].
- [244] A. Andronic et al. Eur. Phys. J. C76 (2016) no. 3, 107, arXiv:1506.03981 [nucl-ex].
- [245] F. Prino and R. Rapp J. Phys. G43 (2016) no. 9, 093002, arXiv:1603.00529 [nucl-ex].
- [246] A. Beraudo et al. arXiv:1803.03824 [nucl-th].
- [247] ALICE Collaboration, J. Adam et al. JHEP **11** (2015) 205, arXiv:1506.06604 [nucl-ex]. [Addendum: JHEP06,032(2017)].
- [248] CMS Collaboration, V. Khachatryan et al. Eur. Phys. J. C77 (2017) no. 4, 252, arXiv:1610.00613 [nucl-ex].
- [249] ALICE Collaboration, B. Abelev et al. CERN-LHCC-2013-024. ALICE-TDR-017, Nov, 2013. https://cds.cern.ch/record/1625842.
- [250] ALICE Collaboration, B. Abelev et al. CERN-LHCC-2015-001. ALICE-TDR-018, Jan, 2015. https://cds.cern.ch/record/1981898.

- [251] ATLAS Collaboration ATL-PHYS-PUB-2016-025, CERN, Geneva, Oct, 2016. https://cds.cern.ch/record/2222304.
- [252] ATLAS Collaboration CERN-LHCC-2018-023. LHCC-P-012, CERN, Geneva, Jun, 2018. https://cds.cern.ch/record/2623663.
- [253] CMS Collaboration CMS Physics Analysis Summary CMS-PAS-FTR-18-012, CERN, Geneva, 2018.
- [254] CMS Collaboration, D. Contardo, M. Klute, J. Mans, L. Silvestris, and J. Butler CERN-LHCC-2015-010. LHCC-P-008. CMS-TDR-15-02, Geneva, Jun, 2015. https://cds.cern.ch/record/2020886.
- [255] CMS Collaboration CERN-LHCC-2017-009. CMS-TDR-014, CERN, Geneva, Jun, 2017. https://cds.cern.ch/record/2272264.
- [256] LHCb Collaboration CERN-LHCC-2013-021. LHCB-TDR-013, Nov, 2013. http://cds.cern.ch/record/1624070.
- [257] LHCb Collaboration CERN-LHCC-2014-001. LHCB-TDR-015, Feb, 2014. https://cds.cern.ch/record/1647400.
- [258] Y. Dokshitzer and D. Kharzeev Physics Letters B 519 (2001) no. 3, 199 206. http://www.sciencedirect.com/science/article/pii/S0370269301011303.
- [259] CMS Collaboration CMS-PAS-FTR-17-002, CERN, Geneva, 2017. https://cds.cern.ch/record/2291105.
- [260] ATLAS Collaboration ATL-PHYS-PUB-2018-020, CERN, Geneva, Oct, 2018. http://cds.cern.ch/record/2644407.
- [261] G. D. Moore and D. Teaney Phys. Rev. C71 (2005) 064904, arXiv:hep-ph/0412346 [hep-ph].
- [262] S. K. Das, F. Scardina, S. Plumari, and V. Greco Phys. Lett. B747 (2015) 260–264, arXiv:1502.03757 [nucl-th].
- [263] S. K. Das, F. Scardina, S. Plumari, and V. Greco Phys. Rev. C90 (2014) 044901, arXiv:1312.6857 [nucl-th].
- [264] Y. Xu, J. E. Bernhard, S. A. Bass, M. Nahrgang, and S. Cao Phys. Rev. C97 (2018) no. 1, 014907, arXiv:1710.00807 [nucl-th].
- [265] ALICE Collaboration, J. Adam et al. Phys. Rev. C93 (2016) no. 3, 034916, arXiv:1507.06194 [nucl-ex].
- [266] A. Beraudo, A. De Pace, M. Monteno, M. Nardi, and F. Prino 2018. arXiv:1807.03180 [hep-ph].
- [267] ALICE Collaboration, S. Acharya et al. arXiv:1809.09371 [nucl-ex].
- [268] X. Dong and V. Greco Prog. Part. Nucl. Phys 104 (2018) no. 5, 1-47.
- [269] F. Scardina, S. K. Das, V. Minissale, S. Plumari, and V. Greco Phys. Rev. C96 (2017) no. 4, 044905, arXiv:1707.05452 [nucl-th].
- [270] Z.-w. Lin and D. Molnar Phys. Rev. C68 (2003) 044901, arXiv:nucl-th/0304045 [nucl-th].
- [271] V. Greco, C. Ko, and R. Rapp Physics Letters B 595 (2004) no. 1, 202 208. http://www.sciencedirect.com/science/article/pii/S0370269304009037.
- [272] R. J. Fries, V. Greco, and P. Sorensen Ann.Rev.Nucl.Part.Sci. 58 (2008) 177-205, arXiv:0807.4939 [nucl-th].
- [273] ALICE Collaboration, S. Acharya et al. JHEP 04 (2018) 108, arXiv:1712.09581 [nucl-ex].
- [274] ALICE Collaboration, S. Acharya et al. Phys. Lett. B781 (2018) 8-19, arXiv:1712.04242 [hep-ex].
- [275] V. Greco, C. M. Ko, and P. Levai Phys. Rev. C68 (2003) 034904, arXiv:nucl-th/0305024 [nucl-th].
- [276] V. Greco, C. M. Ko, and R. Rapp Phys. Lett. B595 (2004) 202-208, arXiv:nucl-th/0312100 [nucl-th].
- [277] R. J. Fries, B. Muller, C. Nonaka, and S. A. Bass Phys. Rev. C68 (2003) 044902, arXiv:nucl-th/0306027 [nucl-th].
- [278] R. J. Fries, B. Muller, C. Nonaka, and S. A. Bass Phys. Rev. Lett. 90 (2003) 202303, arXiv:nucl-th/0301087 [nucl-th].
- [279] Y. Oh, C. M. Ko, S. H. Lee, and S. Yasui Phys. Rev. C79 (2009) 044905, arXiv:0901.1382 [nucl-th].
- [280] V. Minissale, F. Scardina, and V. Greco Phys. Rev. C92 (2015) no. 5, 054904, arXiv:1502.06213 [nucl-th].
- [281] S. Plumari, V. Minissale, S. K. Das, G. Coci, and V. Greco Eur. Phys. J. C78 (2018) no. 4, 348, arXiv:1712.00730 [hep-ph].
- [282] L. Ravagli and R. Rapp Physics Letters B 655 (2007) no. 3, 126 131. http://www.sciencedirect.com/science/article/pii/S0370269307008854.
- [283] M. He, R. J. Fries, and R. Rapp Phys. Rev. C86 (2012) 014903, arXiv:1106.6006 [nucl-th].
- [284] A. Andronic, P. Braun-Munzinger, K. Redlich, and J. Stachel Phys. Lett. B659 (2008) 149–155, arXiv:0708.1488 [nucl-th].
- [285] I. Kuznetsova and J. Rafelski J. Phys. G32 (2006) S499–S504, arXiv:hep-ph/0605307 [hep-ph].
- [286] M. He, R. J. Fries, and R. Rapp Phys. Lett. B735 (2014) 445-450, arXiv:1401.3817 [nucl-th].
- [287] T. Song, H. Berrehrah, D. Cabrera, W. Cassing, and E. Bratkovskaya Phys. Rev. C93 (2016) no. 3, 034906, arXiv:1512.00891 [nucl-th].
- [288] ALICE Collaboration, S. Acharya et al. Submitted to: JHEP (2018), arXiv:1804.09083 [nucl-ex].

- [289] ALICE Collaboration, X. Peng 2018. arXiv:1807.07286 [nucl-ex].
- [290] STAR Collaboration, S. K. Radhakrishnan 2018. https://indico.cern.ch/event/656452/contributions/2869736/.
- [291] CMS Collaboration CMS-PAS-FTR-18-024, CERN, Geneva, 2018. http://cds.cern.ch/record/2650897.
- [292] ALICE Collaboration, S. Acharya et al. ALICE-PUBLIC-2018-013, CERN, 2018.
- [293] H. van Hees, V. Greco, and R. Rapp Phys. Rev. C73 (2006) 034913, arXiv:nucl-th/0508055 [nucl-th].
- [294] P. B. Gossiaux and J. Aichelin Phys. Rev. C78 (2008) 014904, arXiv:0802.2525 [hep-ph].
- [295] M. He, R. J. Fries, and R. Rapp Phys. Rev. C86 (2012) 014903, arXiv:1106.6006 [nucl-th].
- [296] S. Plumari, G. L. Guardo, F. Scardina, and V. Greco Phys. Rev. C92 (2015) no. 5, 054902, arXiv:1507.05540 [hep-ph].
- [297] M. Ruggieri, F. Scardina, S. Plumari, and V. Greco Phys. Rev. C89 (2014) no. 5, 054914, arXiv:1312.6060 [nucl-th].
- [298] LHCb collaboration, R. Aaij et al. JHEP 06 (2012) 141 CERN-PH-2012-109, LHCb-PAPER-2012-003, arXiv:1205.0975 [hep-ex].
- [299] G. Aarts et al. Eur. Phys. J. A 53 (2017) no. 5, 93. https://doi.org/10.1140/epja/i2017-12282-9.
- [300] LHCb Collaboration LHCb-CONF-2018-005. CERN-LHCb-CONF-2018-005, CERN, Geneva, Nov, 2018. http://cds.cern.ch/record/2648625.
- [301] S. Chatterjee and P. Bozek arXiv:1804.04893 [nucl-th].
- [302] CMS Collaboration, V. Khachatryan et al. JHEP 09 (2010) 091, arXiv:1009.4122 [hep-ex].
- [303] CMS Collaboration, S. Chatrchyan et al. Phys. Lett. B718 (2013) 795-814, arXiv:1210.5482 [nucl-ex].
- [304] ALICE Collaboration, B. Abelev et al. Phys. Lett. B719 (2013) 29-41, arXiv:1212.2001 [nucl-ex].
- [305] ATLAS Collaboration, G. Aad et al. Phys. Rev. Lett. 110 (2013) no. 18, 182302, arXiv:1212.5198 [hep-ex].
- [306] ALICE Collaboration, B. B. Abelev et al. Phys. Lett. B726 (2013) 164–177, arXiv:1307.3237 [nucl-ex].
- [307] CMS Collaboration, V. Khachatryan et al. Phys. Rev. Lett. **115** (2015) no. 1, 012301, arXiv:1502.05382 [nucl-ex].
- [308] ALICE Collaboration, B. B. Abelev et al. Phys. Lett. **B728** (2014) 25–38, arXiv:1307.6796 [nucl-ex].
- [309] ALICE Collaboration, J. Adam et al. Phys. Lett. B758 (2016) 389-401, arXiv:1512.07227 [nucl-ex].
- [310] K. Dusling and R. Venugopalan Phys. Rev. D87 (2013) no. 9, 094034, arXiv:1302.7018 [hep-ph].
- [311] A. Bzdak and G.-L. Ma Phys. Rev. Lett. 113 (2014) no. 25, 252301, arXiv: 1406.2804 [hep-ph].
- [312] L. He, T. Edmonds, Z.-W. Lin, et al. Physics Letters B 753 (2016) 506 510. http://www.sciencedirect.com/science/article/pii/S0370269315009995.
- [313] B. Blok, C. D. Jäkel, M. Strikman, and U. A. Wiedemann JHEP 12 (2017) 074, arXiv:1708.08241 [hep-ph].
- [314] ALICE Collaboration, S. Acharya et al. arXiv:1802.09145 [nucl-ex].
- [315] CMS Collaboration CMS-PAS-FTR-18-026, CERN, Geneva, 2018. http://cds.cern.ch/record/2650773.
- [316] A. Beraudo, A. De Pace, M. Monteno, M. Nardi, and F. Prino JHEP 03 (2016) 123, arXiv:1512.05186 [hep-ph].
- [317] A. Kusina, J.-P. Lansberg, I. Schienbein, and H.-S. Shao Phys. Rev. Lett. 121 (2018) 052004, arXiv:1712.07024 [hep-ph].
- [318] CMS Collaboration, A. M. Sirunyan et al. Phys. Rev. Lett. (2018), arXiv:1804.09767 [hep-ex].
- [319] J. D. Bjorken FERMILAB-PUB-82-059-THY, FERMILAB-PUB-82-059-T (1982).
- [320] M. Gyulassy and M. Plumer Phys. Lett. B 243 (1990) 432.
- [321] R. Baier, Y. L. Dokshitzer, S. Peigne, and D. Schiff Phys. Lett. B 345 (1995) 277, arXiv:hep-ph/9411409 [hep-ph].
- [322] B. G. Zakharov arXiv:1807.09742 [hep-ph].
- [323] M. Gyulassy, P. Levai, and I. Vitev Nucl. Phys. B571 (2000) 197-233, arXiv:hep-ph/9907461 [hep-ph].
- [324] U. A. Wiedemann arXiv:0908.2306 [hep-ph]. [Landolt-Bornstein23,521(2010)].
- [325] PHENIX Collaboration, K. Adcox et al. Phys. Rev. Lett. 88 (2002) 022301, arXiv:nucl-ex/0109003 [nucl-ex].
- [326] PHENIX Collaboration, S. S. Adler et al. Phys. Rev. Lett. 91 (2003) 072301, arXiv:nucl-ex/0304022 [nucl-ex].
- [327] STAR Collaboration, C. Adler et al. Phys. Rev. Lett. 89 (2002) 202301, arXiv:nucl-ex/0206011 [nucl-ex].
- [328] STAR Collaboration, C. Adler et al. Phys. Rev. Lett. 90 (2003) 082302, arXiv:nucl-ex/0210033 [nucl-ex].
- [329] STAR Collaboration, J. Adams et al. Phys. Rev. Lett. 91 (2003) 172302, arXiv:nucl-ex/0305015 [nucl-ex].
- [330] STAR Collaboration, J. Adams et al. Phys. Rev. Lett. 91 (2003) 072304, arXiv:nucl-ex/0306024 [nucl-ex].
- [331] STAR Collaboration, J. Adams et al. Phys. Rev. Lett. 97 (2006) 162301, arXiv:nucl-ex/0604018 [nucl-ex].
- [332] BRAHMS Collaboration, I. Arsene et al. Phys. Rev. Lett. 91 (2003) 072305, arXiv:nucl-ex/0307003 [nucl-ex].
- [333] PHOBOS Collaboration, B. B. Back et al. Phys. Lett. B 578 (2004) 297, arXiv:nucl-ex/0302015 [nucl-ex].

- [334] STAR Collaboration, L. Adamczyk et al. Phys. Rev. Lett. 119 (2017) 062301, arXiv:1609.03878 [nucl-ex].
- [335] STAR Collaboration, L. Adamczyk et al. Phys. Rev. C 96 (2017) 024905, arXiv:1702.01108 [nucl-ex].
- [336] ALICE Collaboration, K. Aamodt et al. Phys. Lett. B696 (2011) 30-39, arXiv:1012.1004 [nucl-ex].
- [337] ALICE Collaboration, K. Aamodt et al. Phys. Rev. Lett. 108 (2012) 092301, arXiv:1110.0121 [nucl-ex].
- [338] ATLAS Collaboration, G. Aad et al. JHEP 09 (2015) 050, arXiv:1504.04337 [hep-ex].
- [339] CMS Collaboration, S. Chatrchyan et al. Eur. Phys. J. C72 (2012) 1945, arXiv:1202.2554 [nucl-ex].
- [340] ATLAS Collaboration, G. Aad et al. Phys. Rev. Lett. 105 (2010) 252303, arXiv:1011.6182 [hep-ex].
- [341] CMS Collaboration, S. Chatrchyan et al. Phys. Lett. B 712 (2012) 176, arXiv:1202.5022 [nucl-ex].
- [342] ATLAS Collaboration, G. Aad et al. Phys. Lett. B 719 (2013) 220, arXiv:1208.1967 [hep-ex].
- [343] ALICE Collaboration, B. Abelev et al. JHEP 03 (2014) 013, arXiv:1311.0633 [nucl-ex].
- [344] ALICE Collaboration, J. Adam et al. Phys. Lett. B746 (2015) 1-14, arXiv:1502.01689 [nucl-ex].
- [345] CMS Collaboration, V. Khachatryan et al. JHEP 04 (2017) 039, arXiv:1611.01664 [nucl-ex].
- [346] ALICE Collaboration, J. Adam et al. JHEP 09 (2015) 170, arXiv:1506.03984 [nucl-ex].
- [347] CMS Collaboration, S. Chatrchyan et al. Phys. Lett. B 730 (2014) 243, arXiv:1310.0878 [nucl-ex].
- [348] CMS Collaboration, S. Chatrchyan et al. Phys. Rev. C 90 (2014) 024908, arXiv:1406.0932 [nucl-ex].
- [349] ATLAS Collaboration, M. Aaboud et al. Eur. Phys. J. C77 (2017) no. 6, 379, arXiv:1702.00674 [hep-ex].
- [350] ALICE Collaboration, S. Acharya et al., *First measurement of jet mass in Pb-Pb and p-Pb collisions at the LHC*, Submitted to *Phys. Lett. B*, 2017.
- [351] ALICE Collaboration, S. Acharya et al. JHEP 10 (2018) 139, arXiv:1807.06854 [nucl-ex].
- [352] CMS Collaboration, A. M. Sirunyan et al. Phys. Rev. Lett. 120 (2018) no. 14, 142302, arXiv: 1708.09429 [nucl-ex].
- [353] CMS Collaboration, A. M. Sirunyan et al. arXiv:1805.05145 [hep-ex].
- [354] JET Collaboration, K. M. Burke et al. Phys. Rev. C90 (2014) no. 1, 014909, arXiv:1312.5003 [nucl-th].
- [355] CMS Collaboration, V. Khachatryan et al. Phys. Rev. C 96 (2017) no. 1, 015202, arXiv:1609.05383 [nucl-ex].
- [356] Y. He, T. Luo, X.-N. Wang, and Y. Zhu Phys. Rev. C91 (2015) 054908, arXiv:1503.03313 [nucl-th]. [Erratum: Phys. Rev.C97,no.1,019902(2018)].
- [357] Y.-T. Chien, A. Emerman, Z.-B. Kang, G. Ovanesyan, and I. Vitev Phys. Rev. D93 (2016) no. 7, 074030, arXiv:1509.02936 [hep-ph].
- [358] Y.-T. Chien and I. Vitev JHEP 05 (2016) 023, arXiv:1509.07257 [hep-ph].
- [359] I. Vitev, S. Wicks, and B.-W. Zhang JHEP 11 (2008) 093, arXiv:0810.2807 [hep-ph].
- [360] Z.-B. Kang, F. Ringer, and I. Vitev Phys. Lett. B769 (2017) 242-248, arXiv:1701.05839 [hep-ph].
- [361] M. Spousta and B. Cole Eur. Phys. J. C76 (2016) no. 2, 50, arXiv:1504.05169 [hep-ph].
- [362] ATLAS Collaboration ATL-PHYS-PUB-2018-019, CERN, Geneva, Oct, 2018. http://cds.cern.ch/record/2644406.
- [363] ATLAS Collaboration, G. Aad et al. Phys. Rev. Lett. 110 (2013) no. 2, 022301, arXiv:1210.6486 [hep-ex].
- [364] ATLAS Collaboration, G. Aad et al. Phys. Rev. C93 (2016) no. 3, 034914, arXiv:1506.08552 [hep-ex].
- [365] CMS Collaboration, S. Chatrchyan et al. Phys. Lett. B710 (2012) 256–277, arXiv:1201.3093 [nucl-ex].
- [366] CMS Collaboration, S. Chatrchyan et al. JHEP 03 (2015) 022, arXiv:1410.4825 [nucl-ex].
- [367] CMS Collaboration, S. Chatrchyan et al. Phys. Rev. C84 (2011) 024906, arXiv:1102.1957 [nucl-ex].
- [368] CMS Collaboration, A. M. Sirunyan et al. Phys. Rev. Lett. 119 (2017) no. 8, 082301, arXiv:1702.01060 [nucl-ex].
- [369] CMS Collaboration, A. M. Sirunyan et al. arXiv:1711.09738 [nucl-ex].
- [370] R. Haake and C. Loizides arXiv:1810.06324 [nucl-ex].
- [371] F. D'Eramo, M. Lekaveckas, H. Liu, and K. Rajagopal JHEP 05 (2013) 031, arXiv:1211.1922 [hep-ph].
- [372] L. Chen, G.-Y. Qin, S.-Y. Wei, B.-W. Xiao, and H.-Z. Zhang Phys. Lett. B773 (2017) 672–676, arXiv:1607.01932 [hep-ph].
- [373] ATLAS Collaboration, M. Aaboud et al. Phys. Lett. B774 (2017) 379-402, arXiv:1706.09363 [hep-ex].
- [374] H. A. Andrews et al. arXiv:1808.03689 [hep-ph].
- [375] A. H. Mueller, B. Wu, B.-W. Xiao, and F. Yuan Phys. Rev. D95 (2017) no. 3, 034007, arXiv:1608.07339 [hep-ph].
- [376] F. D'Eramo, K. Rajagopal, and Y. Yin arXiv:1808.03250 [hep-ph].
- [377] M. Gyulassy, P. Levai, J. Liao, et al. 2018. arXiv:1808.03238 [hep-ph].
- [378] J. Ghiglieri and H. Kim arXiv:1809.01349 [hep-ph].

- [379] K. C. Zapp Eur. Phys. J. C74 (2014) no. 2, 2762, arXiv:1311.0048 [hep-ph].
- [380] ATLAS Collaboration, M. Aaboud et al. Phys. Rev. C98 (2018) no. 2, 024908, arXiv:1805.05424 [nucl-ex].
- [381] CMS Collaboration, A. M. Sirunyan et al. JHEP 05 (2018) 006, arXiv:1803.00042 [nucl-ex].
- [382] J. Casalderrey-Solana, D. C. Gulhan, J. G. Milhano, D. Pablos, and K. Rajagopal JHEP 10 (2014) 019, arXiv:1405.3864 [hep-ph]. [Erratum: JHEP09,175(2015)].
- [383] Z. Hulcher, D. Pablos, and K. Rajagopal JHEP 03 (2018) 010, arXiv:1707.05245 [hep-ph].
- [384] CMS Collaboration, A. M. Sirunyan et al. Submitted to: Phys. Rev. Lett. (2018), arXiv:1809.08602 [hep-ex].
- [385] ATLAS Collaboration ATLAS-CONF-2017-074, CERN, Geneva, Sep, 2017. https://cds.cern.ch/record/2285812.
- [386] S. D. Ellis, C. K. Vermilion, and J. R. Walsh Phys. Rev. D 81 (2010) 094023, arXiv:0912.0033 [hep-ph].
- [387] J. M. Butterworth, A. R. Davison, M. Rubin, and G. P. Salam Phys. Rev. Lett. 100 (2008) 242001, arXiv:0802.2470 [hep-ph].
- [388] D. Krohn, J. Thaler, and L.-T. Wang JHEP 02 (2010) 084, arXiv:0912.1342 [hep-ph].
- [389] M. Dasgupta, A. Fregoso, S. Marzani, and G. P. Salam JHEP 09 (2013) 029, arXiv:1307.0007 [hep-ph].
- [390] A. J. Larkoski, S. Marzani, G. Soyez, and J. Thaler JHEP 05 (2014) 146, arXiv:1402.2657 [hep-ph].
- [391] G. Altarelli and G. Parisi Nucl. Phys. B 126 (1977) 298.
- [392] A. J. Larkoski, S. Marzani, and J. Thaler Phys. Rev. D 91 (2015) 111501, arXiv:1502.01719 [hep-ph].
- [393] Y. Mehtar-Tani and K. Tywoniuk JHEP 04 (2017) 125, arXiv:1610.08930 [hep-ph].
- [394] Y.-T. Chien and I. Vitev Phys. Rev. Lett. 119 (2017) 112301, arXiv:1608.07283 [hep-ph].
- [395] N.-B. Chang, S. Cao, and G.-Y. Qin Phys. Lett. B781 (2018) 423-432, arXiv:1707.03767 [hep-ph].
- [396] B. Andersson, G. Gustafson, L. Lonnblad, and U. Pettersson Z. Phys. C43 (1989) 625.
- [397] F. A. Dreyer, G. P. Salam, and G. Soyez arXiv: 1807.04758 [hep-ph].
- [398] M. Cacciari, G. P. Salam, and G. Soyez Eur. Phys. J. C72 (2012) 1896, arXiv:1111.6097 [hep-ph].
- [399] M. Cacciari and G. P. Salam Phys. Lett. B641 (2006) 57-61, arXiv:hep-ph/0512210 [hep-ph].
- [400] A. Dainese et al. CERN Yellow Report (2017) no. 3, 635-692, arXiv:1605.01389 [hep-ph].
- [401] ATLAS Collaboration, M. Aaboud et al. arXiv:1805.05635 [nucl-ex].
- [402] R. Kunnawalkam Elayavalli and K. C. Zapp JHEP 07 (2017) 141, arXiv:1707.01539 [hep-ph].
- [403] R. Kunnawalkam Elayavalli and K. C. Zapp Eur. Phys. J. C76 (2016) no. 12, 695, arXiv:1608.03099 [hep-ph].
- [404] E. Eichten, K. Gottfried, T. Kinoshita, K. D. Lane, and T.-M. Yan Phys. Rev. D21 (1980) 203.
- [405] G. S. Bali Phys. Rept. 343 (2001) 1-136, arXiv:hep-ph/0001312 [hep-ph].
- [406] N. Brambilla, A. Pineda, J. Soto, and A. Vairo Rev. Mod. Phys. 77 (2005) 1423, arXiv:hep-ph/0410047 [hep-ph].
- [407] Y. Schroder Phys. Lett. B447 (1999) 321-326, arXiv:hep-ph/9812205 [hep-ph].
- [408] N. Brambilla, A. Pineda, J. Soto, and A. Vairo Nucl. Phys. B566 (2000) 275, arXiv:hep-ph/9907240 [hep-ph].
- [409] Quarkonium Working Group Collaboration, N. Brambilla et al. FERMILAB-FN-0779, CERN-2005-005 (2004), arXiv:hep-ph/0412158 [hep-ph].
- [410] R. Rapp, D. Blaschke, and P. Crochet Prog. Part. Nucl. Phys. 65 (2010) 209–266, arXiv:0807.2470 [hep-ph].
- [411] P. Braun-Munzinger and J. Stachel Landolt-Bornstein 23 (2010) 424, arXiv:0901.2500 [nucl-th].
- [412] L. Kluberg and H. Satz arXiv:0901.3831 [hep-ph].
- [413] A. Mocsy, P. Petreczky, and M. Strickland Int. J. Mod. Phys. A28 (2013) 1340012, arXiv:1302.2180 [hep-ph].
- [414] Y. Liu, K. Zhou, and P. Zhuang Int. J. Mod. Phys. E24 (2015) no. 11, 1530015.
- [415] S. Y. F. Liu and R. Rapp arXiv:1612.09138 [nucl-th].
- [416] T. Matsui and H. Satz Phys. Lett. B178 (1986) 416-422.
- [417] P. Braun-Munzinger and J. Stachel Phys. Lett. B490 (2000) 196–202, arXiv:nucl-th/0007059 [nucl-th].
- [418] R. L. Thews, M. Schroedter, and J. Rafelski Phys. Rev. C63 (2001) 054905, arXiv:hep-ph/0007323 [hep-ph].
- [419] C. Young and E. Shuryak Phys. Rev. C79 (2009) 034907, arXiv:0803.2866 [nucl-th].
- [420] R. Rapp and X. Du Nucl. Phys. A967 (2017) 216-224, arXiv:1704.07923 [hep-ph].
- [421] A. Andronic, P. Braun-Munzinger, K. Redlich, and J. Stachel Nature 561 (2018) 321-330, arXiv:1710.09425 [nucl-th].
- [422] X. Du, R. Rapp, and M. He Phys. Rev. C96 (2017) no. 5, 054901, arXiv:1706.08670 [hep-ph].
- [423] B. Krouppa, A. Rothkopf, and M. Strickland Phys. Rev. D97 (2018) no. 1, 016017, arXiv:1710.02319 [hep-ph].
- [424] C.-Y. Wong Phys. Rev. C72 (2005) 034906, arXiv:hep-ph/0408020 [hep-ph].
- [425] A. Mocsy and P. Petreczky Phys. Rev. D73 (2006) 074007, arXiv:hep-ph/0512156 [hep-ph].

- [426] W. M. Alberico, A. Beraudo, A. De Pace, and A. Molinari Phys. Rev. D75 (2007) 074009, arXiv:hep-ph/0612062 [hep-ph].
- [427] N. Brambilla, J. Ghiglieri, A. Vairo, and P. Petreczky Phys. Rev. D78 (2008) 014017, arXiv:0804.0993 [hep-ph].
- [428] F. Riek and R. Rapp New J. Phys. 13 (2011) 045007, arXiv:1012.0019 [nucl-th].
- [429] Y. Burnier, O. Kaczmarek, and A. Rothkopf JHEP 12 (2015) 101, arXiv:1509.07366 [hep-ph].
- [430] S. Y. F. Liu and R. Rapp Phys. Rev. C97 (2018) no. 3, 034918, arXiv:1711.03282 [nucl-th].
- [431] G. Bhanot and M. E. Peskin Nucl. Phys. B156 (1979) 391-416.
- [432] D. Kharzeev and H. Satz Phys. Lett. B334 (1994) 155-162, arXiv:hep-ph/9405414 [hep-ph].
- [433] L. Grandchamp and R. Rapp Phys. Lett. B523 (2001) 60-66, arXiv:hep-ph/0103124 [hep-ph].
- [434] M. Laine, O. Philipsen, P. Romatschke, and M. Tassler JHEP 03 (2007) 054, arXiv:hep-ph/0611300 [hep-ph].
- [435] Y. Park, K.-I. Kim, T. Song, S. H. Lee, and C.-Y. Wong Phys. Rev. C76 (2007) 044907, arXiv:0704.3770 [hep-ph].
- [436] J.-P. Blaizot, D. De Boni, P. Faccioli, and G. Garberoglio Nucl. Phys. A946 (2016) 49-88, arXiv:1503.03857 [nucl-th].
- [437] R. Katz and P. B. Gossiaux Annals Phys. 368 (2016) 267–295, arXiv:1504.08087 [quant-ph].
- [438] S. Kajimoto, Y. Akamatsu, M. Asakawa, and A. Rothkopf Phys. Rev. D97 (2018) no. 1, 014003, arXiv:1705.03365 [nucl-th].
- [439] P. Petreczky and C. Young Few Body Syst. 58 (2017) no. 2, 61, arXiv:1606.08421 [nucl-th].
- [440] Y. Akamatsu Phys. Rev. D91 (2015) no. 5, 056002, arXiv:1403.5783 [hep-ph].
- [441] N. Brambilla, M. A. Escobedo, J. Soto, and A. Vairo Phys. Rev. D96 (2017) no. 3, 034021, arXiv:1612.07248 [hep-ph].
- [442] T. Song, C. M. Ko, and S. H. Lee Phys. Rev. C91 (2015) no. 4, 044909, arXiv:1502.05734 [nucl-th].
- [443] J. Hoelck, F. Nendzig, and G. Wolschin Phys. Rev. C95 (2017) no. 2, 024905, arXiv:1602.00019 [hep-ph].
- [444] S. Aronson, E. Borras, B. Odegard, R. Sharma, and I. Vitev Phys. Lett. B778 (2018) 384–391, arXiv:1709.02372 [hep-ph].
- [445] X. Yao and B. Müller Phys. Rev. C97 (2018) no. 1, 014908, arXiv:1709.03529 [hep-ph]. [Erratum: Phys. Rev.C97,no.4,049903(2018)].
- [446] D. Contardo, M. Klute, J. Mans, L. Silvestris, and J. Butler CERN-LHCC-2015-010, 2015. https://cds.cern.ch/record/2020886.
- [447] ALICE Collaboration, B. Abelev et al. Phys. Rev. Lett. 109 (2012) 072301, arXiv:1202.1383 [hep-ex].
- [448] ALICE Collaboration, B. B. Abelev et al. Phys. Lett. B734 (2014) 314-327, arXiv:1311.0214 [nucl-ex].
- [449] ALICE Collaboration, J. Adam et al. Phys. Lett. B766 (2017) 212–224, arXiv:1606.08197 [nucl-ex].
- [450] X. Zhao and R. Rapp Nucl. Phys. A859 (2011) 114-125, arXiv:1102.2194 [hep-ph].
- [451] L. Yan, P. Zhuang, and N. Xu Phys. Rev. Lett. 97 (2006) 232301, arXiv:nucl-th/0608010 [nucl-th].
- [452] CMS Collaboration, A. M. Sirunyan et al. Eur. Phys. J. C78 (2018) no. 6, 509, arXiv:1712.08959 [nucl-ex].
- [453] ALICE Collaboration, B. B. Abelev et al. Phys. Rev. C90 (2014) no. 3, 034904, arXiv:1405.2001 [nucl-ex].
- [454] ALICE Collaboration, S. Acharya et al. Phys. Rev. Lett. 120 (2018) no. 10, 102301, arXiv:1707.01005 [nucl-ex].
- [455] CMS Collaboration, A. M. Sirunyan et al. Phys. Rev. Lett. 120 (2018) no. 20, 202301, arXiv: 1708.03497 [nucl-ex].
- [456] ALICE Collaboration, E. Abbas et al. Phys. Rev. Lett. 111 (2013) 162301, arXiv:1303.5880 [nucl-ex].
- [457] ALICE Collaboration, S. Acharya et al. Phys. Rev. Lett. 119 (2017) no. 24, 242301, arXiv: 1709.05260 [nucl-ex].
- [458] ATLAS Collaboration, M. Aaboud et al. Eur. Phys. J. C78 (2018) no. 9, 784, arXiv:1807.05198 [nucl-ex].
- [459] ALICE Collaboration, S. Acharya et al. arXiv:1811.12727 [nucl-ex].
- [460] K. Zhou, N. Xu, Z. Xu, and P. Zhuang Phys. Rev. C89 (2014) no. 5, 054911, arXiv:1401.5845 [nucl-th].
- [461] X. Du and R. Rapp Nucl. Phys. A943 (2015) 147-158, arXiv:1504.00670 [hep-ph].
- [462] ALICE Collaboration, B. Abelev et al. J. Phys. G41 (2014) 087001.
- [463] ATLAS Collaboration, M. Aaboud et al. Eur. Phys. J. C78 (2018) no. 9, 762, arXiv:1805.04077 [nucl-ex].
- [464] B. Z. Kopeliovich, I. K. Potashnikova, I. Schmidt, and M. Siddikov Phys. Rev. C91 (2015) no. 2, 024911, arXiv:1409.5147 [hep-ph].
- [465] M. Spousta Phys. Lett. B767 (2017) 10-15, arXiv:1606.00903 [hep-ph].
- [466] F. Arleo Phys. Rev. Lett. 119 (2017) no. 6, 062302, arXiv:1703.10852 [hep-ph].
- [467] ALICE Collaboration CERN-LHCC-2013-014. LHCC-I-022-ADD-1, CERN, Geneva.

https://cds.cern.ch/record/1592659.

- [468] LHCb Collaboration, R. Aaij et al. Eur. Phys. J. C74 (2014) no. 10, 3092, arXiv:1407.7734 [hep-ex].
- [469] ALICE Collaboration, B. B. Abelev et al. Phys. Lett. B738 (2014) 361-372, arXiv:1405.4493 [nucl-ex].
- [470] ALICE Collaboration, S. Acharya et al. arXiv:1805.04387 [nucl-ex].
- [471] CMS Collaboration, A. M. Sirunyan et al. arXiv:1805.09215 [hep-ex].
- [472] CMS Collaboration, V. Khachatryan et al. Phys. Lett. B770 (2017) 357–379, arXiv:1611.01510 [nucl-ex].
- [473] CMS Collaboration, A. M. Sirunyan et al. Phys. Rev. Lett. 120 (2018) no. 14, 142301, arXiv: 1706.05984 [hep-ex].
- [474] B. Krouppa and M. Strickland Universe 2 (2016) no. 3, 16, arXiv:1605.03561 [hep-ph].
- [475] F. Arleo and S. Peigné JHEP 10 (2014) 73, arXiv:1407.5054 [hep-ph].
- [476] D. Das and N. Dutta Int. J. Mod. Phys. A33 (2018) no. 16, 1850092, arXiv:1802.00414 [nucl-ex].
- [477] X. Yao, W. Ke, Y. Xu, S. Bass, and B. Müller in 27th International Conference on Ultrarelativistic Nucleus-Nucleus Collisions (Quark Matter 2018) Venice, Italy, May 14-19, 2018. 2018. arXiv:1807.06199 [nucl-th].
- [478] P. P. Bhaduri, N. Borghini, A. Jaiswal, and M. Strickland arXiv:1809.06235 [hep-ph].
- [479] F. Arleo, S. Peigné, and T. Sami Phys. Rev. D83 (2011) 114036, arXiv:1006.0818 [hep-ph].
- [480] X. Du and R. Rapp arXiv:1808.10014 [nucl-th].
- [481] C. J. Benesh, J. Qiu, and J. P. Vary Phys. Rev. C 50 (1994) 1015–1023, arXiv:hep-ph/9403265 [hep-ph].
- [482] S. Gavin and R. Vogt Phys. Rev. Lett. 78 (1997) 1006-1009, arXiv:hep-ph/9606460 [hep-ph].
- [483] S. J. Brodsky and A. H. Mueller Phys. Lett. B206 (1988) 685.
- [484] A. Capella, E. G. Ferreiro, and A. B. Kaidalov Phys. Rev. Lett. 85 (2000) 2080-2083, arXiv:hep-ph/0002300 [hep-ph].
- [485] CMS Collaboration, S. Chatrchyan et al. JHEP 04 (2014) 103, arXiv:1312.6300 [nucl-ex].
- [486] ATLAS Collaboration, M. Aaboud et al. Eur. Phys. J. C78 (2018) no. 3, 171, arXiv:1709.03089 [nucl-ex].
- [487] CMS Collaboration CMS-PAS-BPH-14-009, CERN, 2016. https://cds.cern.ch/record/2223879.
- [488] CMS Collaboration, S. Chatrchyan et al. Phys. Rev. Lett. 109 (2012) 222301, arXiv:1208.2826 [nucl-ex].
- [489] ALICE Collaboration, B. B. Abelev et al. JHEP 12 (2014) 073, arXiv:1405.3796 [nucl-ex].
- [490] F. Arleo and S. Peigné Phys. Rev. D 95 (2017) no. 1, 011502, arXiv:1512.01794 [hep-ph].
- [491] ALICE Collaboration, S. Acharya et al. Phys. Lett. B780 (2018) 7-20, arXiv:1709.06807 [nucl-ex].
- [492] CMS Collaboration, A. M. Sirunyan et al. Submitted to: Phys. Lett. (2018), arXiv:1810.01473 [hep-ex].
- [493] S. J. Brodsky, F. Fleuret, C. Hadjidakis, and J. P. Lansberg Phys. Rept. 522 (2013) 239–255, arXiv:1202.6585 [hep-ph].
- [494] J. P. Lansberg, S. J. Brodsky, F. Fleuret, and C. Hadjidakis Few Body Syst. 53 (2012) 11-25, arXiv:1204.5793 [hep-ph].
- [495] M. Ferro-Luzzi Nucl. Instrum. Meth. A553 (2005) 388–399.
- [496] LHCb Collaboration, R. Aaij et al. arXiv:1810.07907 [hep-ex].
- [497] C. Hadjidakis et al. arXiv:1807.00603 [hep-ex].
- [498] ALICE Collaboration, B. Abelev et al. Phys. Lett. B712 (2012) 165–175, arXiv:1202.2816 [hep-ex].
- [499] ALICE Collaboration, J. Adam et al. JHEP 09 (2015) 148, arXiv:1505.00664 [nucl-ex].
- [500] ALICE Collaboration, S. G. Weber Nucl. Phys. A967 (2017) 333-336, arXiv:1704.04735 [hep-ex].
- [501] K. Werner, B. Guiot, I. Karpenko, and T. Pierog Phys. Rev. C89 (2014) no. 6, 064903, arXiv:1312.1233 [nucl-th].
- [502] Y.-Q. Ma, P. Tribedy, R. Venugopalan, and K. Watanabe arXiv:1803.11093 [hep-ph].
- [503] E. G. Ferreiro and C. Pajares Phys. Rev. C86 (2012) 034903, arXiv:1203.5936 [hep-ph].
- [504] E. G. Ferreiro and C. Pajares arXiv:1501.03381 [hep-ph].
- [505] B. Z. Kopeliovich, H. J. Pirner, I. K. Potashnikova, K. Reygers, and I. Schmidt Phys. Rev. D88 (2013) no. 11, 116002, arXiv:1308.3638 [hep-ph].
- [506] T. Sjöstrand, S. Ask, J. R. Christiansen, et al. Comput. Phys. Commun. **191** (2015) 159–177, arXiv:1410.3012 [hep-ph].
- [507] P. Skands, S. Carrazza, and J. Rojo Eur. Phys. J. C74 (2014) no. 8, 3024, arXiv:1404.5630 [hep-ph].
- [508] T. Sjöstrand arXiv:1706.02166 [hep-ph].
- [509] E. G. Ferreiro Phys. Lett. B749 (2015) 98-103, arXiv:1411.0549 [hep-ph].
- [510] E. G. Ferreiro and J.-P. Lansberg JHEP 10 (2018) 094, arXiv:1804.04474 [hep-ph].
- [511] WA98 Collaboration, M. M. Aggarwal et al. Phys. Rev. Lett. 85 (2000) 3595, arXiv:nucl-ex/0006008 [nucl-ex].

- [512] T. Peitzmann and M. H. Thoma Phys. Rept. 364 (2002) 175–246, arXiv:hep-ph/0111114 [hep-ph].
- [513] CERES Collaboration, G. Agakichiev et al. Phys. Rev. Lett. 75 (1995) 1272–1275.
- [514] CERES Collaboration, G. Agakishiev et al. Phys. Lett. B **422** (1998) 405–412, arXiv:nucl-ex/9712008 [nucl-ex].
- [515] CERES/NA45 Collaboration, D. Adamova et al. Phys. Rev. Lett. 91 (2003) 042301, arXiv:nucl-ex/0209024 [nucl-ex].
- [516] CERES Collaboration, G. Agakichiev et al. Eur. Phys. J. C 41 (2005) 475-513, arXiv:nucl-ex/0506002 [nucl-ex].
- [517] CERES Collaboration, D. Adamova et al. Phys. Lett. B 666 (2008) 425, arXiv:nucl-ex/0611022 [nucl-ex].
- [518] NA60 Collaboration, R. Arnaldi et al. Phys. Rev. Lett. 96 (2006) 162302, arXiv:nucl-ex/0605007 [nucl-ex].
- [519] NA60 Collaboration, R. Arnaldi et al. Phys. Rev. Lett. 100 (2008) 022302, arXiv:0711.1816 [nucl-ex].
- [520] NA60 Collaboration, R. Arnaldi et al. Eur. Phys. J. C 61 (2009) 711, arXiv:0812.3053 [nucl-ex].
- [521] NA60 Collaboration, H. J. Specht AIP Conf. Proc. 1322 (2010) 1, arXiv:1011.0615 [nucl-ex].
- [522] R. Rapp, G. Chanfray, and J. Wambach Phys. Rev. Lett. 76 (1996) 368, arXiv:hep-ph/9508353 [hep-ph].
- [523] R. Rapp and J. Wambach Eur. Phys. J. A6 (1999) 415-420, arXiv:hep-ph/9907502 [hep-ph].
- [524] R. Rapp, J. Wambach, and H. van Hees Landolt-Bornstein 23 (2010) 134, arXiv:0901.3289 [hep-ph].
- [525] P. M. Hohler and R. Rapp Phys. Lett. B731 (2014) 103-109, arXiv:1311.2921 [hep-ph].
- [526] G. Brown and M. Rho Phys. Rev. Lett. 66 (1991) 2720.
- [527] PHENIX Collaboration, A. Adare et al. Phys. Rev. C 81 (2010) 034911, arXiv:0912.0244 [nucl-ex].
- [528] PHENIX Collaboration, A. Adare et al. Phys. Rev. C 93 (2016) 014904, arXiv:1509.04667 [nucl-ex].
- [529] STAR Collaboration, L. Adamczyk et al. Phys. Rev. Lett. 113 (2014) 022301, arXiv:1312.7397 [hep-ex].
- [530] STAR Collaboration, L. Adamczyk et al. Phys. Lett. B 750 (2015) 64, arXiv:1501.05341 [hep-ex].
- [531] PHENIX Collaboration, A. Adare et al. Phys. Rev. Lett. 104 (2010) 132301, arXiv:0804.4168 [nucl-ex].
- [532] PHENIX Collaboration, A. Adare et al. Phys. Rev. C 91 (2015) 064904, arXiv:1405.3940 [nucl-ex].
- [533] PHENIX Collaboration, A. Adare et al. arXiv:1805.04084 [hep-ex].
- [534] PHENIX Collaboration, A. Adare et al. Phys. Rev. Lett. 109 (2012) 122302, arXiv:1105.4126 [nucl-ex].
- [535] C. Shen, U. W. Heinz, J.-F. Paquet, and C. Gale Phys. Rev. C89 (2014) no. 4, 044910, arXiv:1308.2440 [nucl-th].
- [536] H. van Hees, C. Gale, and R. Rapp Phys. Rev. C84 (2011) 054906, arXiv:1108.2131 [hep-ph].
- [537] ALICE Collaboration, K. Aamodt et al. JINST 3 (2008) S08002.
- [538] ALICE Collaboration, J. Adam et al. Phys. Lett. B 754 (2016) 235, arXiv:1509.07324 [nucl-ex].
- [539] ALICE Collaboration, S. Acharya et al. arXiv:1805.04403 [nucl-ex].
- [540] C. Shen, J.-F. Paquet, G. S. Denicol, S. Jeon, and C. Gale Phys. Rev. C95 (2017) no. 1, 014906, arXiv:1609.02590 [nucl-th].
- [541] P. Dasgupta, S. De, R. Chatterjee, and D. K. Srivastava arXiv:1804.02828 [nucl-th].
- [542] H. van Hees, M. He, and R. Rapp Nucl. Phys. A933 (2015) 256–271, arXiv:1404.2846 [nucl-th].
- [543] J.-F. Paquet Nucl. Part. Phys. Proc. 289-290 (2017) 89–94, arXiv:1612.07359 [nucl-th].
- [544] J.-F. Paquet Nucl. Phys. A967 (2017) 184–191, arXiv:1704.07842 [nucl-th].
- [545] ALICE Collaboration, S. Acharya et al. arXiv:1807.00923 [nucl-ex].
- [546] ALICE Collaboration, S. Acharya et al. arXiv:1805.04391 [hep-ex].
- [547] ALICE Collaboration, S. Acharya et al. arXiv:1805.04407 [hep-ex].
- [548] R. Rapp Acta Phys. Polon. B42 (2011) 2823-2852, arXiv:1110.4345 [nucl-th].
- [549] R. Rapp Phys. Rev. C63 (2001) 054907, arXiv:hep-ph/0010101 [hep-ph].
- [550] H. van Hees and R. Rapp Nucl. Phys. A806 (2008) 339–387, arXiv:0711.3444 [hep-ph].
- [551] M. He, R. J. Fries, and R. Rapp Phys. Rev. C85 (2012) 044911, arXiv:1112.5894 [nucl-th].
- [552] O. Linnyk, E. L. Bratkovskaya, and W. Cassing Prog. Part. Nucl. Phys. 87 (2016) 50-115, arXiv:1512.08126 [nucl-th].
- [553] W. Cassing and E. L. Bratkovskaya Nucl. Phys. A831 (2009) 215-242, arXiv:0907.5331 [nucl-th].
- [554] E. L. Bratkovskaya and W. Cassing Nucl. Phys. A807 (2008) 214-250, arXiv:0712.0635 [nucl-th].
- [555] O. Linnyk J. Phys. G38 (2011) 025105, arXiv:1004.2591 [hep-ph].
- [556] J. Ghiglieri, O. Kaczmarek, M. Laine, and F. Meyer Phys. Rev. D94 (2016) no. 1, 016005, arXiv:1604.07544 [hep-lat].
- [557] G. Aarts, C. Allton, A. Amato, et al. JHEP 02 (2015) 186, arXiv:1412.6411 [hep-lat].

- [558] H.-T. Ding, O. Kaczmarek, and F. Meyer Phys. Rev. D94 (2016) no. 3, 034504, arXiv:1604.06712 [hep-lat].
- [559] G. Aarts and J. M. Martinez Resco Nucl. Phys. B726 (2005) 93-108, arXiv:hep-lat/0507004 [hep-lat].
- [560] B. B. Brandt, A. Francis, B. Jäger, and H. B. Meyer Phys. Rev. D93 (2016) no. 5, 054510, arXiv:1512.07249 [hep-lat].
- [561] C. Jung, F. Rennecke, R.-A. Tripolt, L. von Smekal, and J. Wambach Phys. Rev. D95 (2017) no. 3, 036020, arXiv:1610.08754 [hep-ph].
- [562] G. Vujanovic, J.-F. Paquet, S. Ryu, et al. Nucl. Phys. A967 (2017) 692-695, arXiv:1704.04687 [nucl-th].
- [563] G. Vujanovic, G. S. Denicol, M. Luzum, S. Jeon, and C. Gale Phys. Rev. C98 (2018) no. 1, 014902, arXiv:1702.02941 [nucl-th].
- [564] G. Vujanovic, J.-F. Paquet, G. S. Denicol, et al. Phys. Rev. C94 (2016) no. 1, 014904, arXiv:1602.01455 [nucl-th].
- [565] C. Gale, Y. Hidaka, S. Jeon, et al. Phys. Rev. Lett. 114 (2015) 072301, arXiv:1409.4778 [hep-ph].
- [566] R. Chatterjee, P. Dasgupta, and D. K. Srivastava Phys. Rev. C96 (2017) no. 1, 014911, arXiv:1702.02378 [nucl-th].
- [567] O. Linnyk, V. Konchakovski, T. Steinert, W. Cassing, and E. L. Bratkovskaya Phys. Rev. C92 (2015) no. 5, 054914, arXiv:1504.05699 [nucl-th].
- [568] T. Sjostrand, S. Mrenna, and P. Z. Skands JHEP 05 (2006) 026, arXiv:hep-ph/0603175 [hep-ph].
- [569] X.-N. Wang and M. Gyulassy Phys. Rev. D44 (1991) 3501–3516.
- [570] R. Brun, F. Carminati, and S. Giani CERN-W5013, CERN-W-5013, CERN, 1994.
- [571] G. Vujanovic, J.-F. Paquet, C. Shen, et al. Nucl. Part. Phys. Proc. 289-290 (2017) 165-168, arXiv:1703.06164 [nucl-th].
- [572] C. A. Bertulani, S. R. Klein, and J. Nystrand Ann. Rev. Nucl. Part. Sci. 55 (2005) 271–310, arXiv:nucl-ex/0502005 [nucl-ex].
- [573] ALICE Collaboration, J. Adam et al. Phys. Rev. Lett. 116 (2016) no. 22, 222301, arXiv:1509.08802 [nucl-ex].
- [574] STAR Collaboration, J. Adam et al. arXiv:1806.02295 [hep-ex].
- [575] W. Zha Int. J. Mod. Phys. Conf. Ser. 46 (2018) 1860012.
- [576] ATLAS Collaboration, M. Aaboud et al. arXiv:1806.08708 [nucl-ex].
- [577] A. J. Baltz Phys. Rept. 458 (2008) 1-171, arXiv:0706.3356 [nucl-ex].
- [578] S. Klein and J. Nystrand Phys. Today 70 (2017) no. 10, 40–47.
- [579] C. A. Bertulani and G. Baur Phys. Rept. 163 (1988) 299.
- [580] G. Baur, K. Hencken, D. Trautmann, S. Sadovsky, and Y. Kharlov Phys. Rept. 364 (2002) 359–450, arXiv:hep-ph/0112211 [hep-ph].
- [581] M. Vidovic, M. Greiner, C. Best, and G. Soff Phys. Rev. C47 (1993) 2308–2319.
- [582] W. Zha, S. R. Klein, R. Ma, et al. Phys. Rev. C97 (2018) no. 4, 044910, arXiv:1705.01460 [nucl-th].
- [583] STAR Collaboration, J. Adams et al. Phys. Rev. C70 (2004) 031902, arXiv:nucl-ex/0404012 [nucl-ex].
- [584] S. R. Klein Phys. Rev. C97 (2018) no. 5, 054903, arXiv:1801.04320 [nucl-th].
- [585] W. Zha, L. Ruan, Z. Tang, Z. Xu, and S. Yang Phys. Lett. B781 (2018) 182-186, arXiv:1804.01813 [hep-ph].
- [586] M. Klusek-Gawenda, R. Rapp, W. Schäfer, and A. Szczurek arXiv:1809.07049 [nucl-th].
- [587] A. J. Baltz, Y. Gorbunov, S. R. Klein, and J. Nystrand Phys. Rev. C80 (2009) 044902, arXiv:0907.1214 [nucl-ex].
- [588] S. R. Klein, J. Nystrand, J. Seger, Y. Gorbunov, and J. Butterworth Comput. Phys. Commun. 212 (2017) 258–268, arXiv:1607.03838 [hep-ph].
- [589] S. Klein, A. H. Mueller, B.-W. Xiao, and F. Yuan arXiv:1811.05519 [hep-ph].
- [590] D. E. Kharzeev, K. Landsteiner, A. Schmitt, and H.-U. Yee Lect. Notes Phys. 871 (2013) 1-11, arXiv:1211.6245 [hep-ph].
- [591] M. Klusek-Gawenda and A. Szczurek Phys. Rev. C93 (2016) no. 4, 044912, arXiv:1509.03173 [nucl-th].
- [592] STAR Collaboration, B. I. Abelev et al. Phys. Rev. Lett. 102 (2009) 112301, arXiv:0812.1063 [nucl-ex].
- [593] S. R. Klein and J. Nystrand Phys. Rev. Lett. 84 (2000) 2330–2333, arXiv:hep-ph/9909237 [hep-ph].
- [594] Planck Collaboration, P. A. R. Ade et al. Astron. Astrophys. 571 (2014) A22, arXiv:1303.5082 [astro-ph.CO].
- [595] J. Alexander et al. 2016. arXiv:1608.08632 [hep-ph]. http://lss.fnal.gov/archive/2016/conf/fermilab-conf-16-421.pdf.
- [596] PAMELA Collaboration, O. Adriani et al. Nature 458 (2009) 607-609, arXiv:0810.4995 [astro-ph].
- [597] Fermi-LAT Collaboration, M. Ackermann et al. Phys. Rev. Lett. 108 (2012) 011103, arXiv:1109.0521 [astro-ph.HE].

- [598] AMS Collaboration, M. Aguilar et al. Phys. Rev. Lett. 110 (2013) 141102.
- [599] Muon g-2 Collaboration, G. W. Bennett et al. Phys. Rev. D73 (2006) 072003, arXiv:hep-ex/0602035 [hep-ex].
- [600] E. M. Riordan et al. Phys. Rev. Lett. 59 (1987) 755.
- [601] J. D. Bjorken, S. Ecklund, W. R. Nelson, et al. Phys. Rev. D38 (1988) 3375.
- [602] A. Bross, M. Crisler, S. H. Pordes, et al. Phys. Rev. Lett. 67 (1991) 2942-2945.
- [603] H. Merkel et al. Phys. Rev. Lett. 112 (2014) no. 22, 221802, arXiv:1404.5502 [hep-ex].
- [604] APEX Collaboration, S. Abrahamyan et al. Phys. Rev. Lett. 107 (2011) 191804, arXiv:1108.2750 [hep-ex].
- [605] J. Balewski et al. in Proceedings, 2013 Community Summer Study on the Future of U.S. Particle Physics: Snowmass on the Mississippi (CSS2013): Minneapolis, MN, USA, July 29-August 6, 2013. 2013. arXiv:1307.4432 [physics.ins-det]. https://misportal.jlab.org/ul/publications/view_pub.cfm?pub_id=12467.
- [606] BaBar Collaboration, J. P. Lees et al. Phys. Rev. Lett. 113 (2014) no. 20, 201801, arXiv:1406.2980 [hep-ex].
- [607] NA48/2 Collaboration, J. R. Batley et al. Phys. Lett. B746 (2015) 178-185, arXiv:1504.00607 [hep-ex].
- [608] WASA-at-COSY Collaboration, P. Moskal in *Proceedings*, 49th Rencontres de Moriond on QCD and High Energy Interactions: La Thuile, Italy, March 22-29, 2014, pp. 237–242. 2014. arXiv:1406.5738 [nucl-ex].
- [609] HADES Collaboration, G. Agakishiev et al. Phys. Lett. B731 (2014) 265-271, arXiv:1311.0216 [hep-ex].
- [610] PHENIX Collaboration, A. Adare et al. Phys. Rev. C91 (2015) no. 3, 031901, arXiv:1409.0851 [nucl-ex].
- [611] LHCb Collaboration, R. Aaij et al. Phys. Rev. Lett. 120 (2018) no. 6, 061801, arXiv:1710.02867 [hep-ex].
- [612] KLOE-2 Collaboration, F. Archilli et al. Phys. Lett. B706 (2012) 251-255, arXiv:1110.0411 [hep-ex].
- [613] M. Ardid, I. Felis, A. Herrero, and J. A. Martínez-Mora JCAP 1704 (2017) no. 04, 010, arXiv: 1701.08863 [astro-ph.HE].
- [614] P. Ilten, Y. Soreq, M. Williams, and W. Xue JHEP 06 (2018) 004, arXiv:1801.04847 [hep-ph].
- [615] P. Ilten, Y. Soreq, J. Thaler, M. Williams, and W. Xue Phys. Rev. Lett. 116 (2016) no. 25, 251803, arXiv:1603.08926 [hep-ph].
- [616] G. Baym, T. Hatsuda, and M. Strickland Phys. Rev. C95 (2017) no. 4, 044907, arXiv:1702.05906 [nucl-th].
- [617] NA60 Collaboration, R. Arnaldi et al. Phys. Rev. Lett. 102 (2009) 222301, arXiv:0812.3100 [nucl-ex].
- [618] J. C. Collins and D. E. Soper Phys. Rev. D16 (1977) 2219.
- [619] C. S. Lam and W.-K. Tung Phys. Rev. D18 (1978) 2447.
- [620] C. S. Lam and W.-K. Tung Phys. Rev. D21 (1980) 2712.
- [621] WA98 Collaboration, M. M. Aggarwal et al. Phys. Rev. Lett. 93 (2004) 022301, arXiv:nucl-ex/0310022 [nucl-ex].
- [622] J. Bellm et al. Eur. Phys. J. C76 (2016) no. 4, 196, arXiv:1512.01178 [hep-ph].
- [623] B. Andersson, G. Gustafson, G. Ingelman, and T. Sjostrand Phys. Rept. 97 (1983) 31-145.
- [624] B. R. Webber Nucl. Phys. **B238** (1984) 492–528.
- [625] N. Fischer and T. Sjöstrand JHEP 01 (2017) 140, arXiv:1610.09818 [hep-ph].
- [626] ATLAS Collaboration, G. Aad et al. arXiv:1509.04776 [hep-ex].
- [627] CMS Collaboration, V. Khachatryan et al. Phys. Rev. Lett. **116** (2016) no. 17, 172302, arXiv:1510.03068 [nucl-ex].
- [628] CMS Collaboration, V. Khachatryan et al. JHEP 05 (2011) 064, arXiv:1102.4282 [hep-ex].
- [629] LHCb Collaboration, R. Aaij et al. Eur. Phys. J. C72 (2012) 2168, arXiv:1206.5160 [hep-ex].
- [630] ALICE Collaboration, K. Aamodt et al. Eur. Phys. J. C71 (2011) 1594, arXiv:1012.3257 [hep-ex].
- [631] ALICE Collaboration, B. Abelev et al. Phys. Lett. B712 (2012) 309-318, arXiv:1204.0282 [nucl-ex].
- [632] ALICE Collaboration, J. Adam et al. Nature Phys. 13 (2017) 535-539, arXiv:1606.07424 [nucl-ex].
- [633] C. Bierlich, G. Gustafson, L. Lönnblad, and A. Tarasov JHEP 03 (2015) 148, arXiv:1412.6259 [hep-ph].
- [634] S. Gieseke, P. Kirchgaeßer, and S. Plätzer Eur. Phys. J. C78 (2018) no. 2, 99, arXiv:1710.10906 [hep-ph].
- [635] C. Bierlich, G. Gustafson, and L. Lönnblad Phys. Lett. B779 (2018) 58-63, arXiv:1710.09725 [hep-ph].
- [636] B. Andersson, G. Gustafson, and B. Nilsson-Almqvist Nucl. Phys. B281 (1987) 289-309.
- [637] C. Bierlich, G. Gustafson, L. Lönnblad, and H. Shah JHEP 10 (2018) 134, arXiv:1806.10820 [hep-ph]. [JHEP18,134(2020)].
- [638] J. Bellm and C. Bierlich arXiv:1807.01291 [hep-ph].
- [639] K. Werner Phys. Rev. Lett. 98 (2007) 152301, arXiv:0704.1270 [nucl-th].
- [640] PHENIX Collaboration, C. Aidala et al. arXiv:1805.02973 [nucl-ex].
- [641] B. Schenke, S. Schlichting, P. Tribedy, and R. Venugopalan Phys. Rev. Lett. 117 (2016) no. 16, 162301, arXiv:1607.02496 [hep-ph].
- [642] P. Braun-Munzinger, K. Redlich, and J. Stachel arXiv:nucl-th/0304013 [nucl-th].

- [643] V. Vislavicius and A. Kalweit arXiv:1610.03001 [nucl-ex].
- [644] M. Mace, V. V. Skokov, P. Tribedy, and R. Venugopalan Phys. Rev. Lett. 121 (2018) no. 5, 052301, arXiv:1805.09342 [hep-ph].
- [645] J. L. Nagle and W. A. Zajc arXiv:1808.01276 [nucl-th].
- [646] ATLAS Collaboration ATLAS-CONF-2017-068, CERN, Geneva, Sep, 2017. http://cds.cern.ch/record/2285806.
- [647] S. Argyropoulos and T. Sjöstrand JHEP 11 (2014) 043, arXiv:1407.6653 [hep-ph].
- [648] ALICE Collaboration, B. Abelev et al. Phys. Rev. Lett. 109 (2012) 252301, arXiv:1208.1974 [hep-ex].
- [649] ALICE Collaboration, S. Acharya et al. Submitted to: Phys. Rev. (2018), arXiv:1807.11321 [nucl-ex].
- [650] E. Shuryak and I. Zahed Phys. Rev. C88 (2013) no. 4, 044915, arXiv:1301.4470 [hep-ph].
- [651] ALICE Collaboration, C. Andrei Nucl. Phys. A931 (2014) 888-892, arXiv:1408.0093 [hep-ex].
- [652] ALICE Collaboration, B. Abelev et al. Phys. Rev. Lett. 111 (2013) 222301, arXiv:1307.5530 [nucl-ex].
- [653] ALICE Collaboration, B. B. Abelev et al. Phys. Rev. C91 (2015) 024609, arXiv:1404.0495 [nucl-ex].
- [654] CMS Collaboration, V. Khachatryan et al. Phys. Lett. **B768** (2017) 103–129, arXiv:1605.06699 [nucl-ex].
- [655] ALICE Collaboration, J. Adam et al. Phys. Lett. **B768** (2017) 203–217, arXiv:1506.09206 [nucl-ex].
- [656] ALICE Collaboration, J. Adam et al. Phys. Lett. B760 (2016) 720-735, arXiv:1601.03658 [nucl-ex].
- [657] ALICE Collaboration, J. Adam et al. Phys. Rev. C95 (2017) no. 6, 064606, arXiv:1702.00555 [nucl-ex].
- [658] ALICE Collaboration, J. Adam et al. Eur. Phys. J. C76 (2016) no. 5, 245, arXiv:1601.07868 [nucl-ex].
- [659] ALICE Collaboration, B. B. Abelev et al. Phys. Lett. **B728** (2014) 216–227, arXiv:1307.5543 [nucl-ex].
- [660] C. Loizides Nucl. Phys. A956 (2016) 200-207, arXiv:1602.09138 [nucl-ex].
- [661] CMS Collaboration, S. Chatrchyan et al. Eur. Phys. J. C74 (2014) no. 6, 2847, arXiv:1307.3442 [hep-ex].
- [662] CMS Collaboration, S. Chatrchyan et al. Eur. Phys. J. C72 (2012) 2164, arXiv:1207.4724 [hep-ex].
- [663] M. Floris Nucl. Phys. A931 (2014) 103-112, arXiv:1408.6403 [nucl-ex].
- [664] ALICE Collaboration, J. Adam et al. Phys. Rev. C93 (2016) no. 2, 024905, arXiv:1507.06842 [nucl-ex].
- [665] ALICE Collaboration, J. Adam et al. Phys. Rev. C92 (2015) no. 5, 054908, arXiv:1506.07884 [nucl-ex].
- [666] ALICE Collaboration, B. Abelev et al. Phys. Lett. B739 (2014) 139–151, arXiv:1404.1194 [nucl-ex].
- [667] ALICE Collaboration, J. Adam et al. Phys. Rev. C91 (2015) 034906, arXiv:1502.00559 [nucl-ex].
- [668] ALICE Collaboration, K. Aamodt et al. Phys. Rev. D84 (2011) 112004, arXiv:1101.3665 [hep-ex].
- [669] CMS Collaboration. https://cds.cern.ch/record/1703272.
- [670] ALICE Collaboration, S. Acharya et al. Phys. Rev. C96 (2017) no. 6, 064613, arXiv:1709.01731 [nucl-ex].
- [671] ATLAS Collaboration, M. Aaboud et al. Phys. Rev. C96 (2017) no. 6, 064908, arXiv:1704.01621 [hep-ex].
- [672] ALICE Collaboration, K. Aamodt et al. Phys. Lett. B708 (2012) 249-264, arXiv:1109.2501 [nucl-ex].
- [673] CMS Collaboration, S. Chatrchyan et al. JHEP 07 (2011) 076, arXiv:1105.2438 [nucl-ex].
- [674] CMS Collaboration, S. Chatrchyan et al. Eur. Phys. J. C72 (2012) 2012, arXiv:1201.3158 [nucl-ex].
- [675] ATLAS Collaboration, G. Aad et al. Phys. Rev. C86 (2012) 014907, arXiv:1203.3087 [hep-ex].
- [676] ATLAS Collaboration, G. Aad et al. Phys. Rev. C90 (2014) no. 4, 044906, arXiv:1409.1792 [hep-ex].
- [677] CMS Collaboration, V. Khachatryan et al. Phys. Lett. B765 (2017) 193-220, arXiv:1606.06198 [nucl-ex].
- [678] ALICE Collaboration, S. Acharya et al. JHEP 09 (2017) 032, arXiv:1707.05690 [nucl-ex].
- [679] ALICE Collaboration, J. Adam et al. Phys. Lett. B762 (2016) 376-388, arXiv:1605.02035 [nucl-ex].
- [680] ALICE Collaboration, J. Adam et al. JHEP 09 (2016) 164, arXiv:1606.06057 [nucl-ex].
- [681] CMS Collaboration, A. M. Sirunyan et al. Phys. Rev. Lett. 120 (2018) no. 9, 092301, arXiv:1709.09189 [nucl-ex].
- [682] ATLAS Collaboration, M. Aaboud et al. Eur. Phys. J. C77 (2017) no. 6, 428, arXiv:1705.04176 [hep-ex].
- [683] ALICE Collaboration, B. Abelev et al. JHEP 06 (2015) 190, arXiv:1405.4632 [nucl-ex].
- [684] ALICE Collaboration, B. Abelev et al. Phys. Lett. **B719** (2013) 18–28, arXiv:1205.5761 [nucl-ex].
- [685] CMS Collaboration, V. Khachatryan et al. Phys. Lett. B742 (2015) 200-224, arXiv:1409.3392 [nucl-ex].
- [686] ALICE Collaboration, B. Abelev et al. Phys. Rev. Lett. 111 (2013) no. 23, 232302, arXiv:1306.4145 [nucl-ex].
- [687] ALICE Collaboration, B. Abelev et al. Phys. Lett. B723 (2013) 267-279, arXiv:1301.3756 [nucl-ex].
- [688] ALICE Collaboration, J. Adam et al. Phys. Rev. C93 (2016) no. 4, 044903, arXiv:1512.05739 [nucl-ex].
- [689] ALICE Collaboration, J. Adam et al. Eur. Phys. J. C76 (2016) no. 2, 86, arXiv:1509.07255 [nucl-ex].
- [690] CMS Collaboration, A. M. Sirunyan et al. arXiv:1708.08901 [nucl-ex].
- [691] ATLAS Collaboration, G. Aad et al. Phys. Lett. B725 (2013) 60-78, arXiv:1303.2084 [hep-ex].
- [692] CMS Collaboration, S. Chatrchyan et al. Phys. Lett. B724 (2013) 213-240, arXiv:1305.0609 [nucl-ex].

[693] ALICE Collaboration, K. Aamodt et al. Phys. Rev. Lett. 105 (2010) 252302, arXiv:1011.3914 [nucl-ex]. [694] ALICE Collaboration, K. Aamodt et al. Phys. Rev. Lett. 107 (2011) 032301, arXiv:1105.3865 [nucl-ex]. [695] CMS Collaboration, S. Chatrchyan et al. Phys. Rev. C87 (2013) no. 1, 014902, arXiv:1204.1409 [nucl-ex]. [696] ALICE Collaboration, B. Abelev et al. Phys. Rev. C90 (2014) no. 5, 054901, arXiv:1406.2474 [nucl-ex]. [697] CMS Collaboration, S. Chatrchyan et al. Phys. Rev. C89 (2014) no. 4, 044906, arXiv:1310.8651 [nucl-ex]. [698] ATLAS Collaboration, G. Aad et al. Eur. Phys. J. C74 (2014) no. 11, 3157, arXiv:1408.4342 [hep-ex]. [699] ALICE Collaboration, J. Adam et al. Phys. Rev. Lett. 116 (2016) no. 13, 132302, arXiv:1602.01119 [nucl-ex]. [700] CMS Collaboration, A. M. Sirunyan et al. Phys. Lett. B776 (2018) 195-216, arXiv:1702.00630 [hep-ex]. [701] CMS Collaboration, A. M. Sirunyan et al. Phys. Rev. C98 (2018) no. 4, 044902, arXiv:1710.07864 [nucl-ex]. [702] ATLAS Collaboration, M. Aaboud et al. Phys. Rev. C97 (2018) no. 2, 024904, arXiv:1708.03559 [hep-ex]. [703] ATLAS Collaboration, G. Aad et al. Phys. Rev. C90 (2014) no. 2, 024905, arXiv:1403.0489 [hep-ex]. [704] ATLAS Collaboration, G. Aad et al. Phys. Rev. C92 (2015) no. 3, 034903, arXiv:1504.01289 [hep-ex]. [705] ALICE Collaboration, S. Acharya et al. Phys. Rev. C97 (2018) no. 2, 024906, arXiv:1709.01127 [nucl-ex]. [706] ATLAS Collaboration, M. Aaboud et al. arXiv:1807.02012 [nucl-ex]. [707] ALICE Collaboration, S. Acharya et al. Phys. Lett. B773 (2017) 68-80, arXiv:1705.04377 [nucl-ex]. [708] ATLAS Collaboration, G. Aad et al. Eur. Phys. J. C74 (2014) no. 8, 2982, arXiv:1405.3936 [hep-ex]. [709] ATLAS Collaboration, G. Aad et al. Phys. Lett. B707 (2012) 330-348, arXiv:1108.6018 [hep-ex]. [710] CMS Collaboration, V. Khachatryan et al. Phys. Rev. C96 (2017) no. 1, 014915, arXiv:1604.05347 [nucl-ex]. [711] ALICE Collaboration, J. Adam et al. Phys. Lett. B753 (2016) 126-139, arXiv:1506.08032 [nucl-ex]. [712] LHCb Collaboration, R. Aaij et al. Phys. Lett. B762 (2016) 473-483, arXiv:1512.00439 [nucl-ex]. [713] ATLAS Collaboration, M. Aaboud et al. Phys. Rev. C95 (2017) no. 6, 064914, arXiv:1606.08170 [hep-ex]. [714] CMS Collaboration, V. Khachatryan et al. Phys. Rev. C92 (2015) no. 3, 034911, arXiv:1503.01692 [nucl-ex]. [715] CMS Collaboration, A. M. Sirunyan et al. Phys. Rev. C96 (2017) no. 6, 064902, arXiv:1708.07113 [nucl-ex]. [716] ATLAS Collaboration, M. Aaboud et al. Eur. Phys. J. C78 (2018) no. 2, 142, arXiv:1709.02301 [nucl-ex]. [717] ATLAS Collaboration, G. Aad et al. JHEP 11 (2013) 183, arXiv:1305.2942 [hep-ex]. [718] CMS Collaboration, A. M. Sirunyan et al. arXiv:1711.05594 [nucl-ex]. [719] ALICE Collaboration, S. Acharya et al. JHEP 07 (2018) 103, arXiv:1804.02944 [nucl-ex]. [720] ALICE Collaboration, S. Acharya et al. arXiv:1803.09857 [nucl-ex]. [721] CMS Collaboration, A. M. Sirunyan et al. JHEP 03 (2018) 181, arXiv:1802.00707 [hep-ex]. [722] CMS Collaboration, V. Khachatryan et al. JHEP 11 (2016) 055, arXiv:1609.02466 [nucl-ex]. [723] CMS Collaboration, S. Chatrchyan et al. Eur. Phys. J. C74 (2014) no. 7, 2951, arXiv:1401.4433 [nucl-ex]. [724] ALICE Collaboration, B. Abelev et al. Phys. Lett. B720 (2013) 52-62, arXiv:1208.2711 [hep-ex]. [725] ALICE Collaboration, B. Abelev et al. JHEP 09 (2012) 112, arXiv:1203.2160 [nucl-ex]. [726] ATLAS Collaboration, G. Aad et al. Phys. Rev. Lett. 114 (2015) no. 7, 072302, arXiv:1411.2357 [hep-ex]. [727] ALICE Collaboration, J. Adam et al. Eur. Phys. J. C76 (2016) no. 5, 271, arXiv:1603.03402 [nucl-ex]. [728] CMS Collaboration, A. M. Sirunyan et al. Phys. Lett. B772 (2017) 306–329, arXiv:1612.08972 [nucl-ex]. [729] ALICE Collaboration, S. Acharya et al. arXiv:1805.05212 [nucl-ex]. [730] ATLAS Collaboration, G. Aad et al. Phys. Lett. B748 (2015) 392-413, arXiv:1412.4092 [hep-ex]. [731] CMS Collaboration, V. Khachatryan et al. Eur. Phys. J. C75 (2015) no. 5, 237, arXiv:1502.05387 [nucl-ex]. [732] ALICE Collaboration, S. Acharya et al. Phys. Lett. B783 (2018) 95-113, arXiv:1712.05603 [nucl-ex]. [733] ALICE Collaboration, J. Adam et al. Phys. Lett. B763 (2016) 238-250, arXiv:1608.07201 [nucl-ex]. [734] CMS Collaboration, A. M. Sirunyan et al. Phys. Rev. Lett. 121 (2018) 242301, arXiv:1801.04895 [hep-ex]. [735] ALICE Collaboration, B. Abelev et al. Phys. Rev. Lett. 111 (2013) 102301, arXiv:1305.2707 [nucl-ex]. [736] ALICE Collaboration, J. Adam et al. Phys. Lett. B753 (2016) 41-56, arXiv:1507.03134 [nucl-ex]. [737] ALICE Collaboration, J. Adam et al. JHEP 09 (2016) 028, arXiv:1606.00321 [nucl-ex]. [738] ALICE Collaboration, J. Adam et al. Eur. Phys. J. C77 (2017) no. 4, 245, arXiv:1605.06963 [nucl-ex]. [739] ALICE Collaboration, S. Acharya et al. arXiv:1805.04367 [nucl-ex]. [740] ALICE Collaboration, J. Adam et al. JHEP 07 (2015) 051, arXiv:1504.07151 [nucl-ex]. [741] ALICE Collaboration, J. Adam et al. JHEP 11 (2015) 127, arXiv:1506.08808 [nucl-ex]. [742] ALICE Collaboration, J. Adam et al. JHEP 06 (2016) 050, arXiv:1603.02816 [nucl-ex]. [743] ALICE Collaboration, J. Adam et al. Eur. Phys. J. C76 (2016) no. 4, 184, arXiv:1509.08258 [hep-ex]. [744] ALICE Collaboration, J. Adam et al. JHEP 05 (2016) 179, arXiv:1506.08804 [nucl-ex].

- [745] ALICE Collaboration, D. Adamová et al. Phys. Lett. B776 (2018) 91-104, arXiv:1704.00274 [nucl-ex].
- [746] ALICE Collaboration, S. Acharya et al. Eur. Phys. J. C77 (2017) no. 6, 392, arXiv:1702.00557 [hep-ex].
- [747] CMS Collaboration, A. M. Sirunyan et al. Eur. Phys. J. C77 (2017) no. 4, 269, arXiv:1702.01462 [nucl-ex].
- [748] CMS Collaboration, A. M. Sirunyan et al. Phys. Rev. Lett. 118 (2017) no. 16, 162301, arXiv:1611.01438 [nucl-ex].
- [749] LHCb Collaboration, R. Aaij et al. Phys. Lett. B774 (2017) 159–178, arXiv:1706.07122 [hep-ex].
- [750] J. Jia and S. Radhakrishnan Phys. Rev. C92 (2015) no. 2, 024911, arXiv:1412.4759 [nucl-ex].
- [751] M. Gyulassy, P. Levai, I. Vitev, and T. S. Biro Phys. Rev. D90 (2014) no. 5, 054025, arXiv:1405.7825 [hep-ph].
- [752] L. McLerran and V. V. Skokov Acta Phys. Polon. B46 (2015) no. 8, 1513, arXiv:1407.2651 [hep-ph].
- [753] X. Zhu, Y. Zhou, H. Xu, and H. Song Phys. Rev. C95 (2017) no. 4, 044902, arXiv:1608.05305 [nucl-th].
- [754] ALICE Collaboration, J. Adam et al. Eur. Phys. J. C77 (2017) no. 1, 33, arXiv:1509.07541 [nucl-ex].
- [755] ATLAS Collaboration, G. Aad et al. New J. Phys. 13 (2011) 053033, arXiv:1012.5104 [hep-ex].
- [756] ATLAS Collaboration, G. Aad et al. Eur. Phys. J. C76 (2016) no. 7, 403, arXiv:1603.02439 [hep-ex].
- [757] CMS Collaboration, V. Khachatryan et al. JHEP 01 (2011) 079, arXiv:1011.5531 [hep-ex].
- [758] J. F. Grosse-Oetringhaus and K. Reygers J. Phys. G37 (2010) 083001, arXiv:0912.0023 [hep-ex].
- [759] ALICE Collaboration, S. Acharya et al. Eur. Phys. J. C77 (2017) no. 12, 852, arXiv:1708.01435 [hep-ex].
- [760] A. Bzdak, B. Schenke, P. Tribedy, and R. Venugopalan Phys. Rev. C87 (2013) no. 6, 064906, arXiv:1304.3403 [nucl-th].
- [761] ALICE Collaboration, B. B. Abelev et al. Phys. Lett. B727 (2013) 371-380, arXiv:1307.1094 [nucl-ex].
- [762] C. Loizides, J. Kamin, and D. d'Enterria Phys. Rev. C97 (2018) no. 5, 054910, arXiv:1710.07098 [nucl-ex].
- [763] CMS Collaboration, V. Khachatryan et al. Eur. Phys. J. C76 (2016) no. 3, 155, arXiv:1512.00815 [hep-ex].
- [764] ALICE Collaboration, B. Abelev et al. JHEP 09 (2013) 049, arXiv:1307.1249 [nucl-ex].
- [765] J. L. Albacete, H. Petersen, and A. Soto-Ontoso Phys. Lett. B778 (2018) 128–136, arXiv:1707.05592 [hep-ph].
- [766] J. Jia J. Phys. G41 (2014) no. 12, 124003, arXiv:1407.6057 [nucl-ex].
- [767] CMS Collaboration CMS-PAS-FTR-18-025, CERN, Geneva, 2018. https://cds.cern.ch/record/2651892.
- [768] ATLAS CollaborationATL-PHYS-PUB-2018-039 (Nov, 2018). https://cds.cern.ch/record/2649445.
- [769] ALICE Collaboration, J. Adam et al. Phys. Rev. C91 (2015) no. 6, 064905, arXiv:1412.6828 [nucl-ex].
- [770] M. L. Miller, K. Reygers, S. J. Sanders, and P. Steinberg Ann. Rev. Nucl. Part. Sci. 57 (2007) 205–243, arXiv:nucl-ex/0701025 [nucl-ex].
- [771] J. S. Moreland, J. E. Bernhard, and S. A. Bass Phys. Rev. C92 (2015) no. 1, 011901, arXiv:1412.4708 [nucl-th].
- [772] ALICE Collaboration, S. Acharya et al. arXiv:1812.01312 [nucl-ex].
- [773] H.-J. Drescher, A. Dumitru, C. Gombeaud, and J.-Y. Ollitrault Phys. Rev. C76 (2007) 024905, arXiv:0704.3553 [nucl-th].
- [774] ALICE Collaboration, S. Acharya et al. Phys. Lett. B784 (2018) 82-95, arXiv:1805.01832 [nucl-ex].
- [775] Y. V. Kovchegov and E. Levin, *Quantum chromodynamics at high energy*, vol. 33. Cambridge University Press, 2012. http://www.cambridge.org/de/knowledge/isbn/item6803159.
- [776] L. V. Gribov, E. M. Levin, and M. G. Ryskin Phys. Rept. 100 (1983) 1–150.
- [777] A. H. Mueller and J.-w. Qiu Nucl. Phys. B268 (1986) 427-452.
- [778] F. Gelis, E. Iancu, J. Jalilian-Marian, and R. Venugopalan Ann. Rev. Nucl. Part. Sci. 60 (2010) 463–489, arXiv:1002.0333 [hep-ph].
- [779] J. Liu, C. Shen, and U. Heinz Phys. Rev. C91 (2015) no. 6, 064906, arXiv:1504.02160 [nucl-th]. [Erratum: Phys. Rev.C92,no.4,049904(2015)].
- [780] H. Paukkunen Nucl. Phys. A967 (2017) 241-248, arXiv:1704.04036 [hep-ph].
- [781] N. Armesto Proceedings of **DIS2018** (2018). https://indico.cern.ch/event/656250/contributions/2889208/.
- [782] K. J. Eskola, P. Paakkinen, H. Paukkunen, and C. A. Salgado Eur. Phys. J. C77 (2017) no. 3, 163, arXiv:1612.05741 [hep-ph].
- [783] K. Kovarik et al. Phys. Rev. D93 (2016) no. 8, 085037, arXiv:1509.00792 [hep-ph].
- [784] D. de Florian, R. Sassot, P. Zurita, and M. Stratmann Phys. Rev. D85 (2012) 074028, arXiv:1112.6324 [hep-ph].
- [785] H. Khanpour and S. Atashbar Tehrani Phys. Rev. D93 (2016) no. 1, 014026, arXiv:1601.00939 [hep-ph].
- [786] L. Frankfurt, V. Guzey, and M. Strikman Phys. Rept. 512 (2012) 255–393, arXiv:1106.2091 [hep-ph].
- [787] N. Armesto, A. Capella, A. B. Kaidalov, J. Lopez-Albacete, and C. A. Salgado Eur. Phys. J. C29 (2003) 531–540, arXiv:hep-ph/0304119 [hep-ph].
- [788] R. D. Ball, V. Bertone, M. Bonvini, et al. Eur. Phys. J. C78 (2018) no. 4, 321, arXiv:1710.05935 [hep-ph].

- [789] xFitter Developers' Team Collaboration, H. Abdolmaleki et al. Eur. Phys. J. C78 (2018) no. 8, 621, arXiv:1802.00064 [hep-ph].
- [790] ZEUS Collaboration, M. Przybycień Proceedings of ICHEP2018 (2018). https://indico.cern.ch/event/ 686555/contributions/2975347/attachments/1681772/2741853/ichep_2018_zeus.pdf.
- [791] J. L. Albacete, A. Dumitru, and C. Marquet Int. J. Mod. Phys. A28 (2013) 1340010, arXiv:1302.6433 [hep-ph].
- [792] T. Lappi in Proceedings, 7th International Conference on Hard and Electromagnetic Probes of High-Energy Nuclear Collisions (Hard Probes 2015): Montréal, Québec, Canada, June 29-July 3, 2015. 2016. arXiv:1509.04503 [hep-ph].
- [793] LHeC Study Group Collaboration, J. L. Abelleira Fernandez et al. J. Phys. G39 (2012) 075001, arXiv:1206.2913 [physics.acc-ph].
- [794] A. Accardi et al. Eur. Phys. J. A52 (2016) no. 9, 268, arXiv:1212.1701 [nucl-ex].
- [795] LHCb Collaboration, R. Aaij et al. J. Phys. G40 (2013) 045001, arXiv:1301.7084 [hep-ex].
- [796] LHCb Collaboration, R. Aaij et al. J. Phys. G41 (2014) 055002, arXiv:1401.3288 [hep-ex].
- [797] LHCb Collaboration, R. Aaij et al. JHEP **09** (2015) 084, arXiv:1505.08139 [hep-ex].
- [798] LHCb Collaboration, R. Aaij et al. JHEP 10 (2018) 167, arXiv:1806.04079 [hep-ex].
- [799] ALICE MFT Working Group Collaboration, A. Uras and f. t. A. M. W. Group J. Phys. Conf. Ser. 446 (2013) 012054, arXiv:1212.6517 [hep-ex].
- [800] A. Collaboration. https://cds.cern.ch/record/2285584.
- [801] CMS Collaboration CERN-LHCC-2017-027, 2017. https://cds.cern.ch/record/2296612.
- [802] CMS Collaboration CMS-NOTE-2018-006, CERN, Geneva, Dec, 2018. https://cds.cern.ch/record/2650976.
- [803] LHCb Collaboration LHCb-CONF-2018-003, CERN, Geneva, May, 2018. http://cds.cern.ch/record/2320135.
- [804] LHCb Collaboration CERN-LHCC-2014-016, LHCB-TDR-016, May, 2014. https://cds.cern.ch/record/1701361.
- [805] ATLAS Collaboration, M. Aaboud et al. Nature Phys. 13 (2017) no. 9, 852–858, arXiv:1702.01625 [hep-ex].
- [806] CMS Collaboration CMS-PAS-FSQ-16-012 (2018) no. CMS-PAS-FSQ-16-012, . https://cds.cern.ch/record/2319158.
- [807] ALICE Collaboration, J. Adam et al. JHEP 09 (2015) 095, arXiv:1503.09177 [nucl-ex].
- [808] STAR Collaboration, B. I. Abelev et al. Phys. Rev. C77 (2008) 034910, arXiv:0712.3320 [nucl-ex].
- [809] STAR Collaboration, B. I. Abelev et al. Phys. Rev. C81 (2010) 044901, arXiv:0912.0604 [nucl-ex].
- [810] V. Guzey, E. Kryshen, and M. Zhalov Phys. Rev. C93 (2016) no. 5, 055206, arXiv:1602.01456 [nucl-th].
- [811] S. Klein and J. Nystrand Phys. Rev. C60 (1999) 014903, arXiv:hep-ph/9902259 [hep-ph].
- [812] V. Guzey, M. Strikman, and M. Zhalov Eur. Phys. J. C74 (2014) no. 7, 2942, arXiv:1312.6486 [hep-ph].
- [813] V. Guzey, E. Kryshen, M. Strikman, and M. Zhalov Phys. Lett. B726 (2013) 290-295, arXiv:1305.1724 [hep-ph].
- [814] ALICE Collaboration, B. Abelev et al. Phys. Lett. B718 (2013) 1273-1283, arXiv:1209.3715 [nucl-ex].
- [815] ALICE Collaboration, E. Abbas et al. Eur. Phys. J. C73 (2013) no. 11, 2617, arXiv:1305.1467 [nucl-ex].
- [816] ALICE Collaboration, J. Adam et al. Phys. Lett. B751 (2015) 358-370, arXiv:1508.05076 [nucl-ex].
- [817] CMS Collaboration, V. Khachatryan et al. Phys. Lett. B772 (2017) 489-511, arXiv:1605.06966 [nucl-ex].
- [818] V. Rebyakova, M. Strikman, and M. Zhalov Phys. Lett. B710 (2012) 647-653, arXiv:1109.0737 [hep-ph].
- [819] A. Adeluyi and C. A. Bertulani Phys. Rev. C85 (2012) 044904, arXiv:1201.0146 [nucl-th].
- [820] V. Guzey and M. Zhalov JHEP 10 (2013) 207, arXiv:1307.4526 [hep-ph].
- [821] V. P. Goncalves and M. V. T. Machado Phys. Rev. C84 (2011) 011902, arXiv:1106.3036 [hep-ph].
- [822] T. Lappi and H. Mantysaari Phys. Rev. C87 (2013) no. 3, 032201, arXiv:1301.4095 [hep-ph].
- [823] A. Cisek, W. Schafer, and A. Szczurek Phys. Rev. C86 (2012) 014905, arXiv:1204.5381 [hep-ph].
- [824] L. Frankfurt, V. Guzey, M. Strikman, and M. Zhalov Phys. Lett. B752 (2016) 51-58, arXiv:1506.07150 [hep-ph].
- [825] ALICE Collaboration, M. Krivda et al. PoS TWEPP-17 (2017) 149, arXiv:1711.01200 [physics.ins-det].
- [826] A. J. Baltz, S. R. Klein, and J. Nystrand Phys. Rev. Lett. 89 (2002) 012301, arXiv:nucl-th/0205031 [nucl-th].
- [827] G. Baur, K. Hencken, A. Aste, D. Trautmann, and S. R. Klein Nucl. Phys. A729 (2003) 787-808,
- arXiv:nucl-th/0307031 [nucl-th].
- [828] STAR Collaboration, S. R. Klein PoS **DIS2016** (2016) 188, arXiv:1606.02754 [nucl-ex].
- [829] M. Diehl in Presented at the INT workshop on Gluons and the quark sea at high energies: distributions, polarization, tomography, 2010. 1st Nov, 2010.
 - http://www.int.washington.edu/talks/WorkShops/int_10_3/People/Diehl_M/Diehl1.pdf.
- [830] T. Toll and T. Ullrich Phys. Rev. C87 (2013) no. 2, 024913, arXiv:1211.3048 [hep-ph].

- [831] STAR Collaboration, L. Adamczyk et al. Phys. Rev. C96 (2017) no. 5, 054904, arXiv:1702.07705 [nucl-ex].
- [832] STAR Collaboration, S. R. Klein in 26th International Workshop on Deep Inelastic Scattering and Related Subjects (DIS 2018) Port Island, Kobe, Japan, April 16-20, 2018. 2018. arXiv:1807.00455 [nucl-ex].
- [833] L. Frankfurt, M. Strikman, D. Treleani, and C. Weiss Phys. Rev. Lett. 101 (2008) 202003, arXiv:0808.0182 [hep-ph].
- [834] H. Mäntysaari and B. Schenke Phys. Rev. Lett. 117 (2016) no. 5, 052301, arXiv:1603.04349 [hep-ph].
- [835] H. Mäntysaari and B. Schenke Phys. Rev. D94 (2016) no. 3, 034042, arXiv:1607.01711 [hep-ph].
- [836] H. Mäntysaari and B. Schenke Phys. Lett. B772 (2017) 832-838, arXiv:1703.09256 [hep-ph].
- [837] J. Cepila, J. G. Contreras, and J. D. Tapia Takaki Phys. Lett. B766 (2017) 186–191, arXiv:1608.07559 [hep-ph].
- [838] J. Cepila, J. G. Contreras, and M. Krelina Phys. Rev. C97 (2018) no. 2, 024901, arXiv:1711.01855 [hep-ph].
- [839] V. Guzey, M. Strikman, and M. Zhalov arXiv:1808.00740 [hep-ph].
- [840] L. Frankfurt, M. Strikman, and M. Zhalov Phys. Rev. Lett. 102 (2009) 232001, arXiv:0811.0368 [hep-ph].
- [841] L. Frankfurt, M. Strikman, and M. Zhalov Phys. Lett. B670 (2008) 32–36, arXiv:0807.2208 [hep-ph].
- [842] M. Strikman, R. Vogt, and S. N. White Phys. Rev. Lett. **96** (2006) 082001, arXiv:hep-ph/0508296 [hep-ph].
- [843] I. Helenius in 26th International Workshop on Deep Inelastic Scattering and Related Subjects (DIS 2018) Port Island, Kobe, Japan, April 16-20, 2018. 2018. arXiv:1806.07326 [hep-ph].
- [844] ATLAS Collaboration ATLAS-CONF-2017-011, CERN, Geneva, Feb, 2017. http://cds.cern.ch/record/2244822.
- [845] ZEUS Collaboration, M. Derrick et al. Phys. Lett. B384 (1996) 401-413, arXiv:hep-ex/9605009 [hep-ex].
- [846] H1 Collaboration, C. Adloff et al. Phys. Lett. B483 (2000) 36-48, arXiv:hep-ex/0003011 [hep-ex].
- [847] R. D. Ball et al. Nucl. Phys. B867 (2013) 244–289, arXiv:1207.1303 [hep-ph].
- [848] I. Helenius in 9th International Conference on Hard and Electromagnetic Probes of High-Energy Nuclear Collisions: Hard Probes 2018 (HP2018) Aix-Les-Bains, Savoie, France, October 1-5, 2018. 2018. arXiv:1811.10931 [hep-ph].
- [849] F. Cornet, P. Jankowski, M. Krawczyk, and A. Lorca Phys. Rev. D68 (2003) 014010, arXiv:hep-ph/0212160 [hep-ph].
- [850] M. Gluck, E. Reya, and A. Vogt Phys. Rev. D46 (1992) 1973–1979.
- [851] G. A. Schuler and T. Sjöstrand Z. Phys. C68 (1995) 607–624, arXiv:hep-ph/9503384 [hep-ph].
- [852] V. Guzey and M. Klasen arXiv:1811.10236 [hep-ph].
- [853] M. Klasen Rev. Mod. Phys. 74 (2002) 1221-1282, arXiv:hep-ph/0206169 [hep-ph].
- [854] V. Guzey and M. Klasen JHEP 04 (2016) 158, arXiv:1603.06055 [hep-ph].
- [855] S. R. Klein, J. Nystrand, and R. Vogt Phys. Rev. C66 (2002) 044906, arXiv:hep-ph/0206220 [hep-ph].
- [856] V. P. Gonçalves, G. Sampaio dos Santos, and C. R. Sena Nucl. Phys. A976 (2018) 33–45, arXiv: 1711.04497 [hep-ph].
- [857] K. J. Eskola, V. J. Kolhinen, and P. V. Ruuskanen Nucl. Phys. B535 (1998) 351-371, arXiv:hep-ph/9802350 [hep-ph].
- [858] R. Bala J. Phys. Conf. Ser. 381 (2012) 012034, arXiv:1201.0729 [nucl-ex].
- [859] S. R. Klein, J. Nystrand, and R. Vogt Eur. Phys. J. C21 (2001) 563–566, arXiv:hep-ph/0005157 [hep-ph].
- [860] V. P. Gonçalves Phys. Rev. D88 (2013) no. 5, 054025.
- [861] ATLAS Collaboration, G. Aad et al. Phys. Rev. C92 (2015) no. 4, 044915, arXiv:1507.06232 [hep-ex].
- [862] ATLAS Collaboration ATLAS-CONF-2015-056, CERN, Geneva, Sep, 2015. https://cds.cern.ch/record/2055677.
- [863] CMS Collaboration, V. Khachatryan et al. Phys. Lett. B750 (2015) 565-586, arXiv:1503.05825 [nucl-ex].
- [864] CMS Collaboration, V. Khachatryan et al. Phys. Lett. B759 (2016) 36-57, arXiv:1512.06461 [hep-ex].
- [865] CMS Collaboration, A. M. Sirunyan et al. Phys. Rev. Lett. 121 (2018) no. 6, 062002, arXiv:1805.04736 [hep-ex].
- [866] CMS Collaboration, A. M. Sirunyan et al. Phys. Rev. Lett. 119 (2017) no. 24, 242001, arXiv: 1709.07411 [nucl-ex].
- [867] CMS Collaboration CMS-PAS-HIN-17-007, CERN, Geneva, 2018. https://cds.cern.ch/record/2318138.
- [868] CMS Collaboration CMS-PAS-FTR-18-027 (2018).
- [869] S. Dulat, T.-J. Hou, J. Gao, et al. Phys. Rev. D93 (2016) no. 3, 033006, arXiv:1506.07443 [hep-ph].
- [870] E. C. Aschenauer, S. Fazio, M. A. C. Lamont, H. Paukkunen, and P. Zurita Phys. Rev. D96 (2017) no. 11, 114005, arXiv:1708.05654 [nucl-ex].
- [871] LHCb Collaboration, R. Aaij et al. JHEP 10 (2017) 090, arXiv:1707.02750 [hep-ex].

- [872] ALICE Collaboration, S. Acharya et al. Phys. Lett. B770 (2017) 459-472, arXiv:1702.01479 [nucl-ex].
- [873] I. Helenius and H. Paukkunen JHEP 05 (2018) 196, arXiv: 1804.03557 [hep-ph].
- [874] I. Helenius, K. J. Eskola, and H. Paukkunen JHEP 09 (2014) 138, arXiv:1406.1689 [hep-ph].
- [875] D. Kharzeev, E. Levin, and L. McLerran Nucl. Phys. A748 (2005) 627-640, arXiv:hep-ph/0403271 [hep-ph].
- [876] A. van Hameren, P. Kotko, K. Kutak, C. Marquet, and S. Sapeta Phys. Rev. D89 (2014) no. 9, 094014, arXiv:1402.5065 [hep-ph].
- [877] F. Dominguez, C. Marquet, B.-W. Xiao, and F. Yuan Phys. Rev. D83 (2011) 105005, arXiv:1101.0715 [hep-ph].
- [878] C. Marquet, E. Petreska, and C. Roiesnel JHEP 10 (2016) 065, arXiv:1608.02577 [hep-ph].
- [879] J. L. Albacete, G. Giacalone, C. Marquet, and M. Matas arXiv:1805.05711 [hep-ph].
- [880] E. Akcakaya, A. Schäfer, and J. Zhou Phys. Rev. D87 (2013) no. 5, 054010, arXiv:1208.4965 [hep-ph].
- [881] C. Marquet, C. Roiesnel, and P. Taels Phys. Rev. D97 (2018) no. 1, 014004, arXiv:1710.05698 [hep-ph].
- [882] C. Marquet Nucl. Phys. A796 (2007) 41-60, arXiv:0708.0231 [hep-ph].
- [883] P. J. Mulders and J. Rodrigues Phys. Rev. D63 (2001) 094021, arXiv:hep-ph/0009343 [hep-ph].
- [884] D. Boer, P. J. Mulders, and C. Pisano Phys. Rev. D80 (2009) 094017, arXiv:0909.4652 [hep-ph].
- [885] D. Boer, S. J. Brodsky, P. J. Mulders, and C. Pisano Phys. Rev. Lett. 106 (2011) 132001, arXiv:1011.4225 [hep-ph].
- [886] A. Metz and J. Zhou Phys. Rev. D84 (2011) 051503, arXiv:1105.1991 [hep-ph].
- [887] F. Dominguez, J.-W. Qiu, B.-W. Xiao, and F. Yuan Phys. Rev. D85 (2012) 045003, arXiv:1109.6293 [hep-ph].
- [888] LHCb Collaboration, R. Aaij et al. JHEP 06 (2012) 141, arXiv:1205.0975 [hep-ex]. [Addendum: JHEP03,108(2014)].
- [889] S. Peigné and R. Kolevatov JHEP 01 (2015) 141, arXiv:1405.4241 [hep-ph].
- [890] F. Arleo, R. Kolevatov, and S. Peigné Phys. Rev. D93 (2016) 014006, arXiv:1402.1671 [hep-ph].
- [891] F. Arleo and S. Peigné Phys. Rev. Lett. 109 (2012) 122301, arXiv:1204.4609 [hep-ph].
- [892] F. Arleo and S. Peigné JHEP 03 (2013) 122, arXiv:1212.0434 [hep-ph].
- [893] F. Arleo, R. Kolevatov, S. Peigné, and M. Rustamova JHEP 05 (2013) 155, arXiv:1304.0901 [hep-ph].
- [894] T. Stavreva, I. Schienbein, F. Arleo, et al. JHEP 01 (2011) 152, arXiv:1012.1178 [hep-ph].
- [895] LHCb Collaboration, R. Aaij et al. Eur. Phys. J. C74 (2014) no. 5, 2839, arXiv:1401.6932 [hep-ex].
- [896] NNPDF Collaboration, R. D. Ball et al. Eur. Phys. J. C77 (2017) no. 10, 663, arXiv:1706.00428 [hep-ph].
- [897] S. Alekhin, J. Blümlein, S. Moch, and R. Placakyte Phys. Rev. D96 (2017) no. 1, 014011, arXiv:1701.05838 [hep-ph].
- [898] CHORUS Collaboration, G. Onengut et al. Phys. Lett. B632 (2006) 65–75.
- [899] NuTeV Collaboration, M. Goncharov et al. Phys. Rev. D64 (2001) 112006, arXiv:hep-ex/0102049 [hep-ex].
- [900] D. A. Mason, Measurement of the strange antistrange asymmetry at NLO in QCD from NuTeV dimuon data. PhD thesis, Oregon U., 2006. http://lss.fnal.gov/archive/thesis/2000/fermilab-thesis-2006-01.pdf.
- [901] NOMAD Collaboration, O. Samoylov et al. Nucl. Phys. B876 (2013) 339–375, arXiv:1308.4750 [hep-ex].
- [902] A. Kayis-Topaksu et al. New J. Phys. 13 (2011) 093002, arXiv:1107.0613 [hep-ex].
- [903] M. Hirai, S. Kumano, and T. H. Nagai Phys. Rev. C70 (2004) 044905, arXiv:hep-ph/0404093 [hep-ph].
- [904] ATLAS Collaboration, M. Aaboud et al. Eur. Phys. J. C77 (2017) no. 6, 367, arXiv:1612.03016 [hep-ex].
- [905] A. Kusina, F. Lyonnet, D. B. Clark, et al. Eur. Phys. J. C77 (2017) no. 7, 488, arXiv:1610.02925 [nucl-th].
- [906] nCTEQ Collaboration, E. Godat et al. in 26th International Workshop on Deep Inelastic Scattering and Related Subjects (DIS 2018) Port Island, Kobe, Japan, April 16-20, 2018. arXiv:1808.07514 [hep-ph].
- [907] LHCb Collaboration, R. Aaij et al. JHEP 09 (2014) 030, arXiv:1406.2885 [hep-ex].
- [908] ALICE Collaboration, J. Adam et al. JHEP 02 (2017) 077, arXiv:1611.03002 [nucl-ex].
- [909] ATLAS Collaboration, G. Aad et al. Phys. Rev. Lett. 109 (2012) 012001, arXiv:1203.4051 [hep-ex].
- [910] A. M. Cooper-Sarkar and K. Wichmann Phys. Rev. D98 (2018) no. 1, 014027, arXiv:1803.00968 [hep-ex].
- [911] H. Paukkunen and C. A. Salgado JHEP 03 (2011) 071, arXiv:1010.5392 [hep-ph].
- [912] H. Paukkunen and P. Zurita JHEP 12 (2014) 100, arXiv:1402.6623 [hep-ph].
- [913] LHeC study Group Collaboration, H. Paukkunen PoS DIS2017 (2018) 109, arXiv:1709.08342 [hep-ph].
- [914] A. Kusina, J. P. Lansberg, I. Schienbein, and H. S. Shao Acta Phys. Polon. B49 (2018) 1185.
- [915] ALICE Collaboration, B. B. Abelev et al. Phys. Rev. Lett. 113 (2014) no. 23, 232301, arXiv:1405.3452 [nucl-ex].
- [916] ALICE Collaboration, J. Adam et al. Phys. Rev. C94 (2016) no. 5, 054908, arXiv:1605.07569 [nucl-ex].
- [917] ALICE Collaboration, J. Adam et al. JHEP 06 (2015) 055, arXiv:1503.07179 [nucl-ex].

- [918] ALICE Collaboration, B. B. Abelev et al. JHEP 02 (2014) 073, arXiv:1308.6726 [nucl-ex].
- [919] LHCb Collaboration, R. Aaij et al. JHEP **02** (2014) 072, arXiv:1308.6729 [nucl-ex].
- [920] LHCb Collaboration LHCb-CONF-2018-004, CERN, Geneva, Oct, 2018. https://cds.cern.ch/record/2641671.
- [921] ALICE Collaboration, B. B. Abelev et al. Phys. Lett. **B740** (2015) 105–117, arXiv:1410.2234 [nucl-ex].
- [922] LHCb Collaboration, R. Aaij et al. JHEP 07 (2014) 094, arXiv:1405.5152 [nucl-ex].
- [923] ATLAS Collaboration ATLAS-CONF-2017-073, CERN, Geneva, Sep, 2017. https://cds.cern.ch/record/2285811.
- [924] CMS Collaboration, A. M. Sirunyan et al. Submitted to: Eur. Phys. J. (2018), arXiv:1809.11080 [hep-ex].
- [925] CMS Collaboration, A. M. Sirunyan et al. arXiv:1805.02248 [hep-ex].
- [926] LHCb Collaboration, R. Aaij et al. Submitted to: JHEP (2018), arXiv:1809.01404 [hep-ex].
- [927] B. A. Kniehl, G. Kramer, I. Schienbein, and H. Spiesberger Phys. Rev. D79 (2009) 094009, arXiv:0901.4130 [hep-ph].
- [928] S. J. Brodsky, A. Kusina, F. Lyonnet, et al. Adv. High Energy Phys. 2015 (2015) 231547, arXiv:1504.06287 [hep-ph].
- [929] F. Lyonnet, A. Kusina, T. Ježo, et al. JHEP 07 (2015) 141, arXiv:1504.05156 [hep-ph].
- [930] ATLAS Collaboration ATLAS-CONF-2017-072, CERN, Geneva, Sep, 2017. https://cds.cern.ch/record/2285810.
- [931] S. Catani, L. Cieri, D. de Florian, G. Ferrera, and M. Grazzini JHEP 04 (2018) 142, arXiv:1802.02095 [hep-ph].
- [932] B. Boussaha, F. Iddir, and L. Semlala Adv. High Energy Phys. 2018 (2018) 4174602, arXiv:1803.09176 [hep-ph].
- [933] ATLAS, CDF, CMS, D0 Collaboration, M. Dunford EPJ Web Conf. 49 (2013) 06004.
- [934] ATLAS Collaboration, P. H. Beauchemin PoS ICHEP2012 (2013) 263.
- [935] ATLAS, CDF, CMS, D0 Collaboration, L. Di Ciaccio EPJ Web Conf. 60 (2013) 05003.
- [936] CMS Collaboration, V. Candelise Nuovo Cim. C38 (2015) no. 1, 6.
- [937] CMS Collaboration, S. Chatrchyan et al. JHEP 02 (2014) 013, arXiv:1310.1138 [hep-ex].
- [938] ATLAS Collaboration, G. Aad et al. JHEP 05 (2014) 068, arXiv:1402.6263 [hep-ex].
- [939] D. d'Enterria, K. Krajczár, and H. Paukkunen Phys. Lett. B746 (2015) 64–72, arXiv:1501.05879 [hep-ph].
- [940] J. M. Campbell, R. K. Ellis, and W. T. Giele Eur. Phys. J. C75 (2015) no. 6, 246, arXiv: 1503.06182 [physics.comp-ph].
- [941] L. Apolinário, J. G. Milhano, G. P. Salam, and C. A. Salgado Phys. Rev. Lett. 120 (2018) no. 23, 232301, arXiv:1711.03105 [hep-ph].
- [942] H. Paukkunen in 9th International Conference on Hard and Electromagnetic Probes of High-Energy Nuclear Collisions: Hard Probes 2018 (HP2018) Aix-Les-Bains, Savoie, France, October 1-5, 2018. 2018. arXiv:1811.01976 [hep-ph].
- [943] V. Emel'yanov, A. Khodinov, S. R. Klein, and R. Vogt Phys. Rev. C59 (1999) 1860–1863, arXiv:hep-ph/9809222 [hep-ph].
- [944] E. G. Ferreiro, F. Fleuret, J. P. Lansberg, and A. Rakotozafindrabe Phys. Lett. B680 (2009) 50-55, arXiv:0809.4684 [hep-ph].
- [945] I. Helenius, K. J. Eskola, H. Honkanen, and C. A. Salgado JHEP 07 (2012) 073, arXiv:1205.5359 [hep-ph].
- [946] A. Kovner, T. Altinoluk, N. Armesto, G. Beuf, and M. Lublinsky PoS QCDEV2016 (2017) 010.
- [947] CMS Collaboration, A. M. Sirunyan et al. JHEP 10 (2018) 138, arXiv:1809.00201 [hep-ex].
- [948] ATLAS Collaboration ATLAS-CONF-2018-007, CERN, Geneva, May, 2018. https://cds.cern.ch/record/2318588.
- [949] K. Zapp, J. Stachel, and U. A. Wiedemann PoS High-pT physics09 (2009) 022, arXiv:0904.4885 [hep-ph].
- [950] R. C. Barrett and D. F. Jackson United Kingdom: Clarendon Press (1977).
- [951] Belle Collaboration, C.-C. Kuo et al. Phys. Lett. B621 (2005) 41-55, arXiv:hep-ex/0503006 [hep-ex].
- [952] M. Kłusek-Gawenda, P. Lebiedowicz, O. Nachtmann, and A. Szczurek Phys. Rev. D96 (2017) no. 9, 094029, arXiv:1708.09836 [hep-ph].
- [953] M. Diehl, P. Kroll, and C. Vogt Eur. Phys. J. C26 (2003) 567-577, arXiv:hep-ph/0206288 [hep-ph].
- [954] ALICE Collaboration, E. L. Kryshen Nucl. Phys. A967 (2017) 273-276, arXiv:1705.06872 [nucl-ex].
- [955] CMS Collaboration, A. M. Sirunyan et al. arXiv:1810.04602 [hep-ex].
- [956] ATLAS Collaboration, G. Aad et al. CERN-LHCC-2013-018, ATLAS-TDR-023, 2013. https://cds.cern.ch/record/1602235.

- [957] CMS Collaboration, A. Tapper and D. Acosta CERN-LHCC-2013-011. CMS-TDR-12, Jun, 2013. https://cds.cern.ch/record/1556311.
- [958] L. A. Harland-Lang, V. A. Khoze, and M. G. Ryskin arXiv:1810.06567 [hep-ph].
- [959] ALICE Collaboration, B. B. Abelev et al. Eur. Phys. J. C74 (2014) no. 10, 3108, arXiv:1405.3794 [nucl-ex].
- [960] M. Kłusek-Gawenda arXiv:1809.03823 [hep-ph].
- [961] D. d'Enterria and G. G. da Silveira Phys. Rev. Lett. 111 (2013) 080405, arXiv:1305.7142 [hep-ph]. [Erratum: Phys. Rev. Lett.116,no.12,129901(2016)].
- [962] M. Kłusek-Gawenda, P. Lebiedowicz, and A. Szczurek Phys. Rev. C93 (2016) no. 4, 044907, arXiv:1601.07001 [nucl-th].
- [963] M. Kłusek-Gawenda, W. Schäfer, and A. Szczurek Phys. Lett. B761 (2016) 399-407, arXiv:1606.01058 [hep-ph].
- [964] ALICE Collaboration, S. Acharya et al. Phys. Rev. (2018), arXiv:1803.05490 [nucl-ex]. [Phys. Rev.C98,044901(2018)].
- [965] E. Govorkova Phys. Atom. Nucl. 79 (2016) no. 11-12, 1474–1476, arXiv:1505.02960 [hep-ex].
- [966] G. Jikia and A. Tkabladze Phys. Lett. B323 (1994) 453-458, arXiv:hep-ph/9312228 [hep-ph].
- [967] Z. Bern, A. De Freitas, L. J. Dixon, A. Ghinculov, and H. L. Wong JHEP 11 (2001) 031, arXiv:hep-ph/0109079 [hep-ph].
- [968] D. Bardin, L. Kalinovskaya, and E. Uglov Phys. Atom. Nucl. 73 (2010) 1878–1888, arXiv:0911.5634 [hep-ph].
- [969] Particle Data Group Collaboration, C. Patrignani et al. Chin. Phys. C40 (2016) no. 10, 100001.
- [970] S. R. Klein Phys. Rev. D98 (2018) 118501, arXiv:1808.08253 [hep-ph].
- [971] M. Kłusek-Gawenda and A. Szczurek Phys. Rev. C87 (2013) no. 5, 054908, arXiv:1302.4204 [nucl-th].
- [972] R. D. Peccei and H. R. Quinn Phys. Rev. Lett. 38 (1977) 1440-1443.
- [973] R. D. Peccei and H. R. QuinnPhys. Rev. Lett. 38 (Jun, 1977) 1440–1443. https://link.aps.org/doi/10.1103/PhysRevLett.38.1440.
- [974] M. Bauer, M. Neubert, and A. Thamm JHEP 12 (2017) 044, arXiv:1708.00443 [hep-ph].
- [975] K. Mimasu and V. Sanz JHEP 06 (2015) 173, arXiv:1409.4792 [hep-ph].
- [976] J. Jaeckel and M. Spannowsky Phys. Lett. B753 (2016) 482-487, arXiv:1509.00476 [hep-ph].
- [977] S. Knapen, T. Lin, H. K. Lou, and T. Melia Phys. Rev. Lett. 118 (2017) no. 17, 171801, arXiv:1607.06083 [hep-ph].
- [978] I. Brivio, M. B. Gavela, L. Merlo, et al. Eur. Phys. J. C77 (2017) no. 8, 572, arXiv:1701.05379 [hep-ph].
- [979] S. Knapen, T. Lin, H. K. Lou, and T. Melia in Photon 2017: International Conference on the Structure and the Interactions of the Photon and 22th International Workshop on Photon-Photon Collisions and the International Workshop on High Energy Photon Colliders CERN, Geneva, Switzerland, May 22-26, 2017. 2017. arXiv:1709.07110 [hep-ph].
- [980] A. L. Read J. Phys. G28 (2002) 2693–2704.
- [981] C. Baldenegro, S. Fichet, G. Von Gersdorff, and C. Royon JHEP 06 (2018) 131, arXiv:1803.10835 [hep-ph].
- [982] K.-H. Kampert and M. Unger Astropart. Phys. 35 (2012) 660-678, arXiv:1201.0018 [astro-ph.HE].
- [983] Pierre Auger Collaboration, A. Aab et al. Phys. Rev. D91 (2015) no. 3, 032003, arXiv:1408.1421 [astro-ph.HE]. [Erratum: Phys. Rev.D91,no.5,059901(2015)].
- [984] Liverpool Telescope, MAGIC, H.E.S.S., AGILE, Kiso, VLA/17B-403, INTEGRAL, Kapteyn, Subaru, HAWC, Fermi-LAT, ASAS-SN, VERITAS, Kanata, IceCube, Swift NuSTAR Collaboration, Liverpool Telescope, MAGIC, H.E.S.S., et al. Science 361 (2018) no. 6398, eaat1378, arXiv:1807.08816 [astro-ph.HE].
- [985] K. Werner, F.-M. Liu, and T. Pierog Phys. Rev. C74 (2006) 044902, arXiv:hep-ph/0506232 [hep-ph].
- [986] S. Ostapchenko Phys. Rev. D83 (2011) 014018, arXiv:1010.1869 [hep-ph].
- [987] F. Riehn, H. P. Dembinski, R. Engel, et al. PoS ICRC2017 (2017) 301, arXiv:1709.07227 [hep-ph].
- [988] LHCb Collaboration, R. Aaij et al. JHEP 06 (2018) 100, arXiv:1803.10974 [hep-ex].
- [989] H. Dembinski, J. Arteaga, L. Cazon, et al., *Report on Tests and Measurements of Hadronic Interaction Properties with Air Showers*, UHECR 2018, Paris, France, 2018.
- [990] IceCube Collaboration, H. Dembinski in *Proceedings, 19th International Symposium on Very High Energy Cosmic Ray Interactions (ISVHECRI 2016): Moscow, Russia, August 22-27, 2016, vol. 145, p. 01003. 2017.*
- [991] R. P. Kokoulin, A. G. Bogdanov, G. Mannocchi, et al. Nucl. Phys. Proc. Suppl. 196 (2009) 106-109.
- [992] Telescope Array Collaboration, R. U. Abbasi et al. Phys. Rev. D98 (2018) no. 2, 022002, arXiv:1804.03877 [astro-ph.HE].
- [993] R. Ulrich, R. Engel, and M. Unger Phys. Rev. D83 (2011) 054026, arXiv:1010.4310 [hep-ph].
- [994] T. Pierog and K. Werner Phys. Rev. Lett. 101 (2008) 171101, arXiv:astro-ph/0611311 [astro-ph].

- [995] ALICE Collaboration, B. Kim in 27th International Conference on Ultrarelativistic Nucleus-Nucleus Collisions (Quark Matter 2018) Venice, Italy, May 14-19, 2018. 2018. arXiv:1807.09061 [hep-ex].
- [996] D. d'Enterria, T. Pierog, and G. Sun arXiv:1809.06406 [astro-ph.HE].
- [997] LHCf Collaboration, O. Adriani et al. Phys. Rev. D94 (2016) no. 3, 032007, arXiv:1507.08764 [hep-ex].
- [998] NA61/SHINE Collaboration, A. Aduszkiewicz CERN-SPSC-2017-035. SPSC-P-330-ADD-9, CERN, Geneva, Oct, 2017. https://cds.cern.ch/record/2287004.
- [999] NA61/SHINE Collaboration, A. Aduszkiewicz CERN-SPSC-2018-008. SPSC-P-330-ADD-10, CERN, Geneva, Mar, 2018. https://cds.cern.ch/record/2309890.
- [1000] Y. Genolini, D. Maurin, I. V. Moskalenko, and M. Unger Phys. Rev. C98 (2018) no. 3, 034611, arXiv:1803.04686 [astro-ph.HE].
- [1001] J. P. Lansberg, G. Cavoto, C. Hadjidakis, et al. Adv. High. Energy Phys., Article ID 319654, 2015. https://doi.org/10.1155/2015/319654.
- [1002] C. Barschel, Precision luminosity measurement at LHCb with beam-gas imaging. PhD thesis, RWTH Aachen U., 2014. CERN-THESIS-2013-301.
- [1003] LHCb Collaboration, R. Aaij et al. Submitted to: Phys. Rev. Lett. (2018), arXiv:1808.06127 [hep-ex].
- [1004] Eur. Phys. J. A52 (2016) no. 6, https://epja.epj.org/component/toc/?task=topic&id=591.
- [1005] S. J. Brodsky, P. Hoyer, C. Peterson, and N. Sakai Phys. Lett. 93B (1980) 451-455.
- [1006] D. F. Geesaman, K. Saito, and A. W. Thomas Ann. Rev. Nucl. Part. Sci. 45 (1995) 337–390.
- [1007] J. L. Albacete and C. Marquet Prog. Part. Nucl. Phys. 76 (2014) 1–42, arXiv:1401.4866 [hep-ph].
- [1008] R. Vogt Nucl. Phys. A700 (2002) 539-554, arXiv:hep-ph/0107045 [hep-ph].
- [1009] H. Song and U. W. Heinz Phys. Rev. C77 (2008) 064901, arXiv:0712.3715 [nucl-th].
- [1010] B. Schenke, S. Jeon, and C. Gale Phys. Rev. Lett. 106 (2011) 042301, arXiv:1009.3244 [hep-ph].
- [1011] I. A. Karpenko, P. Huovinen, H. Petersen, and M. Bleicher Phys. Rev. C91 (2015) no. 6, 064901, arXiv:1502.01978 [nucl-th].
- [1012] C. Shen and B. Schenke Phys. Rev. C97 (2018) no. 2, 024907, arXiv:1710.00881 [nucl-th].
- [1013] M. A. Stephanov Prog. Theor. Phys. Suppl. 153 (2004) 139–156, arXiv:hep-ph/0402115 [hep-ph]. [Int. J. Mod. Phys.A20,4387(2005)].
- [1014] PHENIX Collaboration, A. Adare et al. arXiv:1807.11928 [nucl-ex].
- [1015] V. P. Goncalves and W. K. Sauter Phys. Rev. D91 (2015) no. 9, 094014, arXiv:1503.05112 [hep-ph].
- [1016] V. P. Gonçalves and M. M. Jaime Eur. Phys. J. C78 (2018) no. 9, 693, arXiv:1802.04713 [hep-ph].
- [1017] PAMELA collaboration, V. Bonvicini et al. Nucl. Instrum. Meth. A461 (2001) 262-268.
- [1018] A. Kounine Int. J. Mod. Phys. E21 (2012) no. 08, 1230005.
- [1019] A. Reinert and M. W. Winkler JCAP 1801 (2018) no. 01, 055, arXiv:1712.00002 [astro-ph.HE].
- [1020] M. Korsmeier, F. Donato, and M. Di MauroPhys. Rev. D 97 (May, 2018) 103019, arXiv:1802.03030 [astro-ph.HE]. https://link.aps.org/doi/10.1103/PhysRevD.97.103019.
- [1021] IceCube collaboration, M. G. Aartsen et al. Science 342 (2013) 1242856, arXiv:1311.5238 [astro-ph.HE].
- [1022] R. Laha and S. J. Brodsky Phys. Rev. D96 (2017) no. 12, 123002, arXiv:1607.08240 [hep-ph].
- [1023] HERMES collaboration, A. Airapetian et al. Nucl. Instrum. Meth. A540 (2005) 68-101, arXiv:physics/0408137 [physics].
- [1024] D. Kikoła, M. G. Echevarria, C. Hadjidakis, et al. Few Body Syst. 58 (2017) no. 4, 139, arXiv:1702.01546 [hep-ex].
- [1025] G. Denicol, A. Monnai, and B. Schenke Phys. Rev. Lett. 116 (2016) no. 21, 212301, arXiv:1512.01538 [nucl-th].
- [1026] F. Becattini and J. Cleymans J. Phys. G34 (2007) S959–964, arXiv:hep-ph/0701029 [hep-ph].
- [1027] I. Karpenko arXiv:1805.11998 [nucl-th].
- [1028] V. Begun, D. Kikola, V. Vovchenko, and D. Wielanek arXiv:1806.01303 [nucl-th].
- [1029] I. V. Moskalenko and A. W. Strong Astrophys. J. 493 (1998) 694-707, arXiv:astro-ph/9710124 [astro-ph].
- [1030] A. E. Vladimirov, S. W. Digel, G. Johannesson, et al. Comput. Phys. Commun. 182 (2011) 1156–1161, arXiv:1008.3642 [astro-ph.HE].
- [1031] E. Orlando, G. Johannesson, I. V. Moskalenko, T. A. Porter, and A. Strong 2017. arXiv:1712.09755 [astro-ph.HE]. http://inspirehep.net/record/1645188/files/arXiv:1712.09755.pdf.
- [1032] STAR Collaboration, B. I. Abelev et al. Phys. Rev. D80 (2009) 111102, arXiv:0910.1428.
- [1033] COMPASS Collaboration, M. Alekseev et al. Eur. Phys. J. C64 (2009) 171-179, arXiv:0907.0388 [hep-ex].
- [1034] D. Yu. Ivanov, A. Schafer, L. Szymanowski, and G. Krasnikov Eur. Phys. J. C34 (2004) no. 3, 297–316,

arXiv:hep-ph/0401131 [hep-ph]. [Erratum: Eur. Phys. J.C75,no.2,75(2015)].

[1035] L. Massacrier, J. P. Lansberg, L. Szymanowski, and J. Wagner in Photon 2017: International Conference on the Structure and the Interactions of the Photon and 22th International Workshop on Photon-Photon Collisions and the International Workshop on High Energy Photon Colliders CERN, Geneva, Switzerland, May 22-26, 2017. 2017. arXiv:1709.09044 [nucl-ex].

http://inspirehep.net/record/1625748/files/arXiv:1709.09044.pdf.

- [1036] J. Koempel, P. Kroll, A. Metz, and J. Zhou Phys. Rev. D85 (2012) 051502, arXiv:1112.1334 [hep-ph].
- [1037] Q. Wang, X.-H. Liu, and Q. Zhao Phys. Rev. D92 (2015) 034022, arXiv:1508.00339 [hep-ph].
- [1038] ALICE Collaboration, S. Acharya et al. arXiv:1809.03235 [nucl-ex].
- [1039] M. Mangano, P. Azzi, M. Benedikt, et al. CERN-ACC-2018-0056, CERN, Geneva, Dec, 2018. https://cds.cern.ch/record/2651294.
- [1040] D. d'Enterria Nucl. Part. Phys. Proc. 289-290 (2017) 237-240, arXiv:1701.08047 [hep-ex].
- [1041] ALICE Collaboration, K. Aamodt et al. Phys. Rev. Lett. 105 (2010) 252301, arXiv:1011.3916 [nucl-ex].
- [1042] ALICE Collaboration, S. Acharya et al. arXiv:1805.04432 [nucl-ex].
- [1043] J. M. Campbell and R. K. Ellis Nucl. Phys. Proc. Suppl. 205-206 (2010) 10-15, arXiv:1007.3492 [hep-ph].
- [1044] A. V. Baskakov, E. E. Boos, L. V. Dudko, I. P. Lokhtin, and A. M. Snigirev Phys. Rev. C92 (2015) no. 4, 044901, arXiv:1502.04875 [hep-ph].
- [1045] D. d'Enterria and C. Loizides arXiv:1809.06832 [hep-ph].
- [1046] E. L. Berger, J. Gao, A. Jueid, and H. Zhang arXiv:1804.06858 [hep-ph].
- [1047] J. Casalderrey-Solana, Y. Mehtar-Tani, C. A. Salgado, and K. Tywoniuk Phys. Lett. B725 (2013) 357–360, arXiv:1210.7765 [hep-ph].
- [1048] Y. Liu and C.-M. Ko J. Phys. G43 (2016) no. 12, 125108, arXiv:1604.01207 [nucl-th].
- [1049] CMS Collaboration, S. Chatrchyan et al. JHEP 05 (2012) 063, arXiv:1201.5069 [nucl-ex].
- [1050] Y.-p. Liu, Z. Qu, N. Xu, and P.-f. Zhuang Phys. Lett. B678 (2009) 72-76, arXiv:0901.2757 [nucl-th].
- [1051] A. Andronic, P. Braun-Munzinger, K. Redlich, and J. Stachel J. Phys. G38 (2011) 124081, arXiv:1106.6321 [nucl-th].
- [1052] M. Arneodo Phys. Rept. 240 (1994) 301-393.
- [1053] D. d'Enterria, M. Klasen, and K. Piotrzkowski Nucl. Phys. Proc. Suppl. 179B (2008) 1.
- [1054] CMS Collaboration, S. Chatrchyan et al. JHEP 11 (2012) 080, arXiv:1209.1666 [hep-ex].
- [1055] CMS Collaboration, S. Chatrchyan et al. JHEP 07 (2013) 116, arXiv:1305.5596 [hep-ex].
- [1056] ATLAS Collaboration, G. Aad et al. Phys. Lett. B749 (2015) 242-261, arXiv:1506.07098 [hep-ex].
- [1057] G. J. Gounaris, P. I. Porfyriadis, and F. M. Renard Eur. Phys. J. C9 (1999) 673-686, arXiv:hep-ph/9902230 [hep-ph].
- [1058] I. F. Ginzburg and A. Schiller Phys. Rev. D57 (1998) 6599-6603, arXiv:hep-ph/9802310 [hep-ph].
- [1059] D. Bernard Nuovo Cim. A110 (1997) 1339–1346.
- [1060] T. Kikuchi, N. Okada, and M. Takeuchi Phys. Rev. D77 (2008) 094012, arXiv:0801.0018 [hep-ph].
- [1061] K.-m. Cheung Phys. Rev. D61 (2000) 015005, arXiv:hep-ph/9904266 [hep-ph].
- [1062] J. L. Hewett, F. J. Petriello, and T. G. Rizzo Phys. Rev. D64 (2001) 075012, arXiv:hep-ph/0010354 [hep-ph].